UNIVERSITY OF CALGARY

Effects Of A Three-Phase Circular Inclusion With Two Imperfect Interfaces On A Radial Crack

by

Pan Gi Park

A THESIS

SUBMITTED TO THE FACULTY OF GRADUATE STUDIES IN PARTIAL FULFILMENT OF THE REQUIREMENTS FOR THE DEGREE OF MASTER OF SCIENCE

DEPARTMENT OF MECHANICAL AND MANUFACTURING ENGINEERING

CALGARY, ALBERTA

August, 2007

© Pan Gi Park 2007

.1

UNIVERSITY OF CALGARY FACULTY OF GRADUATE STUDIES

The undersigned certify that they have read, and recommend to the Faculty of Graduate Studies for acceptance, a thesis entitled "Effects Of A Three-Phase Circular Inclusion With Two Imperfect Interfaces On A Radial Crack" submitted by Pan Gi Park in partial fulfilment of the requirements for the degree of Master of Science.

Supervisor, Dr. Leszek J. Sudak Department of Mechanical and Manufacturing Engineering

Mr. Costain

Dr. Marcelo Epstein

Department of Mechanical and Manufacturing Engineering.

und -

Dr.Jedrzej Sniatycki

Department of Mathematics and Statistics

uly 30 2007

Date

ABSTRACT

Studies of material failure using fracture mechanics have been considered of significant practical importance to engineering applications for their ability to predict and prevent catastrophic failures. A popular model of fracture mechanics, the two-phase single inclusion model, has provided insight into the interaction between cracking and the inclusion. However, it ignores the profound effects of degraded bonding quality and other surrounding medium on the composite materials.

Thus, the author considers the three-phase inclusion model consisting of two circular regions, an infinitely extended region and the imperfect interfaces between the regions, which incorporates the profound effects of micromechanics, and thus, is of fundamental importance in understanding failure mechanisms and enhancing the performance of composite materials. In addition, its application can be extended to deal with functionally graded interphases and cemented implants embedded in bone. Although the three-phase model has recently been adapted to address imperfect adhesion and to provide fundamental solutions to a generalized self-consistent scheme, all of the previous works deal with a single imperfect interface. A comprehensive literature search found no research addressing the situation where both interfaces are imperfect.

In this study, a semi-analytic solution to the interaction between a pre-existing crack and a three-phase inclusion with spring-layer imperfect interfaces on both boundaries is presented. For the analytic approach, theories of linear elastic fracture mechanics (LEFM) and linear elasticity with complex variable techniques are applied to address how interfacial damage on both interfaces simultaneously affects the tendency for the crack to propagate during mode I loading.

ACKNOWLEDGEMENTS

Thank you, God, for your generosity throughout my life.

.*

The last 4 years have been the most challenging, but most valuable and productive time of my life. It was difficult to return to the academic setting eleven years after graduation and to manage both my job and my studies in Canada, my new homeland. However, the lessons gained during this time have been significant and have provided me with invaluable pleasure.

I would like to express my sincere gratitude to my supervisors, Dr. Les Sudak and Dr. X. Wang for their enthusiastic mentoring and encouragement throughout my graduate years. I would like to thank my committee members, Dr. Epstein and Dr. Sniatycki for their invaluable comments and advice. I am also grateful for the advice and encouragment of Dr. Simon S. Park.

I owe a debt of gratitude to Mr. Garth Corsar, a chief officer of Caltech Design, and Mr. Robert Nardei, a principal of Nardei Fabricator, for their kind help and encouragement. I appreciate the support I received from SNC Lavalin. Without their help, I could not have accomplished my study.

The love of my sweet daughter, Sung-In Lauren, beautiful son, Jun-Yong Andrew and lovely wife Hyo-Jung Theresa, was always there to encourage me in overcoming difficulties encountered during these last four years. I extend my deepest gratitude to my mother, Iy-Soon Agatha for her unwavering confidence. Without my family's love, patience and sacrifices, I could never have started my study, much less completed it. This dissertation is dedicated to my Lord Jesus and Holy Mary.

v

.

.

٠

.-

TABLE OF CONTENTS

	Approval Pageii
	Abstractiii
	Acknowledgementsiv
	Dedicationv
	Table of Contents
	List of Tablesix
	List of Figuresx
	CHAPTER 1: INTRODUCTION1
	1.1 Composite Materials1
	1.2 Applied Mechanics of Composite Materials
	1.2.1 Introduction
	1.2.2 Interface Conditions5
	1.2.3 Two-Phase Micromechanics Scheme
	1.2.4 Three-Phase Micromechanics Scheme10
	1.3 Applications of the Three-Phase Model13
	1.3.1 Interphase Layer of Composite Materials
	1.3.2 A Model of A Cemented Implant
	1.4 Overview of the Current Study15
	CHAPTER 2: ANALYTICAL FORMULATION OF A BOUNDARY VALUE PROB-
	LEM
	2.1 Introduction
•	2.2 Formulation
	2.3 Boundary Value Problem23
. .	2.4 Series Representation for Stress Potentials

.

.

.

.

CHAPTER 3: ALGEBRAIC EQUATIONS	
3.1 Introduction	
3.2 General Algebraic Equations on Inner Interface Γ_0	
3.3 General Algebraic Equations on Outer Interface Γ_1	····· 45
3.4 Homogeneous Imperfect Interfaces	
CHAPTER 4: NUMERICAL ANALYSIS RESULTS / DISCUSSION	
4.1 Introduction	62
4.2 Stress Fields around Crack Tips	
4.3 Stress Intensity Factor	
4.4 Numerical Results	67
4.4.1 Verification of Formulae	67
4.4.2 Influence of Imperfect Interface $\Gamma_0 \& \Gamma_1 \dots \dots$	
4.4.3 Influence of Inclusion Stiffness μ_0 on Cracking	
4.4.4 Influence of Interphase Stiffness μ_1 on Cracking	
4.4.5 Influence of Crack Length on Cracking	
4.4.6 The Stable Equilibrium Position	
4.4.7 The Stable Zone	73
CHAPTER 5: AN EXAMPLE OF APPLICATION - BONE / IMPLAN	Γ MODEL-
ING	
5.1 Introduction	85
5.2 Formulation	
5.3 Discussion	90
CHAPTER 6: CONCLUSIONS AND FUTURE WORKS	
6.1 Conclusions	
6.2 Future Works	101

.

÷

REF	'ERENCES	103
APP	ENDIX 1: DERIVATION OF BOUNDARY CONDITION $\partial \Gamma_0 \dots$	
APP	ENDIX 2: DERIVATION OF BOUNDARY CONDITION $\partial \Gamma_1 \dots$	120
APP	ENDIX 3: STRESS POTENTIALS	124
APP	ENDIX 4: GENERAL ALGEBRAIC EQUATIONS	
APP	ENDIX 5: 38 ALGEBRAIC EQUATIONS	
APP	ENDIX 6: STRESS FIELDS AROUND CRACK TIPS	

ş

LIST OF TABLES

Table 5.1 Properties	of materials for	orthopaedic a	rthroplasty	94
		• •		
				·
۶				
		•		
				· .
			· · ·	
		ix		

.

LIST OF FIGURES

Figure 1.1 Phases of a composite materials19
Figure 1.2 Classification of Composite Materials
Figure 2.1 Three-phase circular inclusion model with a radial crack
Figure 4.1 Comparison with Luo results with perfect interfaces along Γ_0 and $\Gamma_1 \dots 75$
Figure 4.2 Influence of imperfect Γ_0 with perfect Γ_1 $(M_1 = 100)$
Figure 4.3 Influence of imperfect Γ_0 with imperfect Γ_1 $(M_1 = 5)$
Figure 4.4 Debonding $(M_1 = 0.01)$
Figure 4.5 Influence of stiffness of an inclusion
Figure 4.6 Influence of stiffness of the matrix phase
Figure 4.7 Influence of the crack length
Figure 4.8 Influence of imperfect Γ_0 with perfect Γ_1
Figure 4.9 Influence of stiffer composite phase on the stable zone $(M_1 = 100)83$
Figure 4.10 Influence of stiffer composite phase on the stable zone $(M_1 = 5)$ 84
Figure 5.1 Three-phase stem-cement-bone system
Figure 5.2 Normalized SIFs of crack tip " a " and " b " vs. Crack locations. No-slip
imperfect stem-cement/ cement-bone bonding $(m_1 = 10)$
Figure 5.3 Normalized SIFs of crack tip " a " and " b " vs. Crack locations. No-slip
imperfect cement-bone bonding $(m_1 = 10)$
Figure 5.4 Normalized SIFs of crack tip " a " and " b " vs. Crack locations. Weakened
no-slip imperfect stem-cement; No-slip imperfect cement-bone bonding

CHAPTER 1 Introduction

1.1 Composite Materials

A composite material is a material system comprised of two or more materials that form a new material with new individual properties or characteristics. This can be called a multiphase material. Within the multiphase, is a region that is very stiff and is called the *reinforcement*. The continuous and compliant phase surrounding the reinforcement is called the *matrix*. Applied loads are usually transmitted into the reinforcement material through the matrix phase, with a small amount of the load being sustained by the matrix. By restraining brittle cracks from propagating between reinforcements, the matrix provides a barrier to cracking in the composite materials [1]. In addition, an independent phase between the matrix and the reinforcement is occasionally considered and is called the *interphase* (see Figure 1.1). The interphase is the result of a chemical interaction or other processing effect.

Composite materials are prevalent in nature. Rocks are an aggregate of many different types of crystals and grains, and often include salt water or oil [2]. Bone and wood are the most common natural composites. From an engineering application standpoint, the composite is a multiphase material, which is artificially made to meet the specific performance requirements of everything from machinery, such as bicycles, automobiles and aircrafts, to biomechanical applications, such as carbon/epoxy artificial limbs and carbon/polymer stems. One simple scheme for the classification of engineering composite materials is shown in Figure 1.2. This includes particulate composites, fibre-reinforced composites and laminate composites.

Particulate composite materials are comprised of a continuous matrix phase and

particles randomly embedded within the matrix as a reinforcement phase. In particulate composites, the matrix bears a majority of the applied load and the particles strengthen the composite by preventing motion of dislocation. Laminate composites consist of two or more layers bonded together, called *lamina*, which are thin, planar and unidirectional, and have a preferred high-strength direction. The orientation and combination of stacked lamina vary with the desired orientation of the high-strength reinforcement. Fibre-reinforced composites are the most dominant and important composite materials. With stiff and brittle fibres embedded in a soft and ductile matrix, fibre reinforcement composites dramatically increase strength, fatigue limits, resistance and stiffness. Stiffness of a material along the fibre direction is usually governed by the mechanical properties of the fibres, while enhancement in the overall strength of the composite is achieved through the properties of the matrix [3]. Since stiffness transverse to the fibres direction is much lower than stiffness normal to the fibres, combinations of both laminate and fibre-reinforced composites, which stack multi-layers of fibres in the different directions, have been developed.

Many new applications have been developed through the engineering of composite materials. Composite materials were first introduced in the form of steel-reinforced concrete in the nineteenth century and have been followed by applications such as the fibreglass boat in 1942, high-strength carbon fibre in the early 1960's and today's nano-composite materials [3]. The demand for such high performance materials which can overcome extreme and severe conditions—conditions which single-phase conventional materials cannot sustain—fuels the study of composite materials that have advantages and dominant features of each constituent combined. Consequently, the advent of many advanced composites and the significant progress made in materials science and analytical approaches, such as micromechanics and macromechanics which save time and costs associated with the expensive experiments necessary to test composites, considering the large number of permutations represented by composites, have attracted the attention of many scientists. Micromechanics is the analysis of the interactions between constituent materials, on the level of the individual phases and at the macromechanical level, to evaluate the equivalent overall response of a composite.

A composite is a multiphase material that consists of many inhomogeneities on a length scale much greater than the atomic scale, but which contain regions large enough to be considered locally continuous and essentially homogeneous at the macroscopic length scale [2, 4]. Also, the inhomogeneities bond with each other at interfaces. Thus, classical mechanics, such as linear elasticity and fracture mechanics can be applied to the material system without loosing generality, given certain assumptions. Namely, each phase, including the reinforcement and the matrix, shows linear elastic behaviour, and they bond together. The singular surface between the phases can be expressed in terms of special parameters such as the spring-type interface parameter, which connects the two elastic constituent materials together smoothly. Thus, this thesis is developed based on the premise that three different imperfectly bonded phases show linear elastic behaviour and spring-type interface parameters represent degradation of bonding. Linear elasticity and fracture mechanics are applied to the composite.

1.2 Applied Mechanics of Composite Materials

1.2.1 Introduction

The underlying studies of the theory and analysis of composite materials were introduced by Maxwell, Rayleigh and Einstein following Poisson's presentation of the theory of induced magnetism, in which a composite material consists of conducting spherical inclusions and a nonconducting matrix, in 1826 [2]. With the advent of fibres as reinforcement constituents in the 1960s, came the dramatic development of micro/macromechanics with respect to composite materials. A macromechanical approach determines the effective, or overall, parameters of composites and predicts their average response. A micromechanical analysis depicts the interaction and relationship between fibres and the matrix; thus, the analysis illustrates the behaviour of deformation and stress within each constituent and local failure, such as matrix/fibre cracking and bonding failure at the interface or the interphase layer [3]. For the study of strength and fracture toughness, which cannot be averaged, the micromechanical approach should be adapted. A well-known model of micromechanics, which considers three-phase system is the generalized self-consistent model (GSC) [5, 6].

Micromechanical analysis involves a representative volume element (RVE), which is defined as a sample volume having the same properties as the homogeneous composite. Making the assumption that an inclusion like a fibre within the RVE is far enough apart from the others so that the interaction between inclusions can be ignored, the RVE taken from composites is considered a single inclusion embedded within the matrix [7]. Hence, since it provides a much simpler approach to micromechanics, the single inclusion problem has attracted a great deal of attention, and has been considered a fundamental solution to the micromechanics analysis of composites.

The interaction between a single inclusion within a matrix and material flaws (such as a crack [8, 9, 10, 11, 12] or a dislocation [13, 14, 15, 16]) has received a considerable amount of attention in the area of fracture mechanics. Understanding crack behaviour within an inclusion (fibre cracking) or within the matrix (matrix cracking) leads to the ability to predict strengthening, hardening and sudden brittle

4

failures of advanced composite materials. Thus, the solution to a single inclusion such as a fibre interacting with a crack or a dislocation in the vicinity of the inclusion is of fundamental importance in understanding the failure analysis of composite materials.

The widely accepted interface assumptions associated with micromechanics can be divided into the following categories: the interface is assumed perfect [5, 13, 16, 17] or imperfect [18, 19, 20, 21]; the imperfect interface is homogeneous [22, 23] or inhomeogeneous along its boundary [24, 25]; the imperfect interface is modeled as a spring layer [26, 27] or thin independent layer [28, 29].

1.2.2 Interface Conditions

The most simple and commonly made assumption regarding interface bonding conditions between constituent materials in composites is that the interface is perfectly bonded. It is an idealization of a complicated adhesion that assumes there is no bonding degradation or damage between the fibre and the surrounding matrix, such as weak adhesion and voids, or crack propagation along the interface. From an analytical point of view, the perfect interface means that tractions and displacements are continuous across the interface. Eshelby contributed to the development of this mechanical model [17]. Although Eshelby's model is inadequate for complicated modern applications, its simple mathematical advantage, contributing to the early development of micromechanics of composite materials, has been widely accepted [17, 29, 30, 31, 32].

Since the actual degradation of interface bonding is not taken into account with the perfect interface, imperfect interface is introduced to simulate intermediate states of bonding, from initial perfect bonding to complete debonding. The imperfect interface condition is also simulated as an interphase layer, which is a thin annular layer between

a fibre and a matrix. In a micromechanic analysis, the intermediate zone is usually represented in two different ways: a finitely thin independent layer [29, 33, 34, 35] and a spring layer [18, 20, 21, 25, 26, 27].

As an independent phase, the mechanical properties of the interphase layer vary between the fibre and the surrounding matrix. The properties match those of the fibre on one side and the matrix on the other [33]. However, the interphase layer is considered a non-uniform, anisotropic region, with mechanical parameters distinct from both the fibre and the matrix [36]. Qaissaunee and Santare [35] investigated an edge dislocation interacting with a three-phase elliptical inclusion in which all the interphases were assumed perfectly bonded. They addressed the presence of an interfacial zone, which had significant influence on the stress field in the matrix and the inclusion. Three parameters, namely, two elastic constants and the thickness of the layer are required to elucidate the behaviour of the interphase layer. However, it is very difficult to determine the constants, and the experiments required are very expensive. To overcome this difficulty, a spring-layer model is often used.

The spring-layer model does not have an independent finite thickness layer. Thus, the three unknown parameters are reduced to two spring constant parameters: in other words, the two elastic constants and the thickness are incorporated into two spring constants defined along the whole length of the interphase [37, 38]. The mathematical representation of this model is based on the premise that tractions are continuous, and displacements are discontinuous across the interface. Displacement jumps across the interface are proportional, in terms of "spring-factor type" interface parameters, to their respective traction components at the boundary. Sudak [22] considered the interaction between a dislocation and a circular inhomogeneity subjected to antiplane shear, with the assumption of homogeneous spring-type inter-

face parameters. It was found that a soft inclusion always attracts the dislocation, regardless of the state of adhesion at the interface. Thus, the equilibrium position of the dislocation cannot be obtained. Alternatively, a stiff inclusion causes an unstable equilibrium position that is affected by imperfect interface conditions and the shear modulus ratio. Amenyah et al. [26] investigated cracking within an fibre embedded in an infinite matrix subjected to thermal loadings. They incorporated the imperfect interface with spring-type interface parameters into the single inclusion (fibre)system and found that the imperfection of the interface had significant effects on the stress intensity factor (SIF). They also addressed the fact that the perfect interface model results in overestimation of the SIF by up to 100% in a hard inclusion, and 200% in a soft inclusion. Consequently, the imperfect interface condition is of significant importance in analysis. A relationship between an imperfect interface and crack propagation in the matrix is proposed by Liu et al. [27]. In contrast to the perfect bonding condition, the imperfect interface causes greater SIFs at the nearby crack tip than at the distant tip, even when the fibre is stiffer than the matrix. Also, the imperfectness of adhesion leads to crack propagation in the interface, which eventually results in debonding.

If the interface parameters are constant along the entire interface, it is called homogeneously imperfect, while it is non-homogeneously imperfect if it is not constant. The effects of an inhomogeneously imperfect interface on the stress fields induced within a fibre are addressed in [25]. The two spring-type interface parameters vary along normal and tangential directions, resulting in inhomogeneously imperfect interface conditions, which lead to average stresses different from those of a homogeneously imperfect interface.

Combining the two different methods mentioned above, authors have recently tried

to derive a more descriptive and detailed interpretation of crack propagation interactions with respect to imperfect interfaces [39, 40, 41]. In this model, the interphase layer is considered an independent region with finite thickness. Inner interface bonding, between the region and a fibre, is considered imperfect and is depicted by two spring-type interface parameters. The outer interface, which is in contact with the surrounding matrix, is assumed perfectly bonded. Adapting a three-phase circular inclusion model with the imperfect interface between a fibre and interphase layer, Kim and Sudak indicate that the degradation of interface bonding quality and mechanical properties of the interphase layer alter SIFs of a radial matrix crack and their influence on micromechanical analysis of composite constituents should not be ignored [39]. The interaction between an inhomogeneity and a screw dislocation in the interphase layer with imperfect interface has been studied [41]. It has been demonstrated that the interphase thickness, imperfection of the inner interface, and ratio of material properties compete with each other to alter the equilibrium position and stability of the dislocation. Henceforth, both mechanical properties of the independent region and imperfect bonding conditions should be taken into account.

In this research, the interphase layer is assumed to be an independent region having imperfect bonding interfaces along both sides and the imperfection of the boundary is assumed to be homogeneously imperfect.

1.2.3 Two-Phase Micromechanics Scheme

In this section, a simplified two-phase composite materials model is briefly introduced. From the analytical point of view, a inclusion represents a void or reinforcement such as a fibre. One of the most significant contributors to the establishment and shaping of the conceptual framework of the two-phase micromechanics scheme is Eshelby. In

his 1957 paper [17], Eshelby's developed a tensor to solve for an elastic field perturbation within an inclusion and the matrix, caused by the inclusion being perfectly embedded in the matrix. His tensor has provided a foundation for the study and application of linear elastostatic theories, such as the theory of dislocations and interactions between a crack and the matrix. In 1964 [13], Dundurs and Mura depicted the interaction between a circular inclusion embedded in an infinite elastic region and an edge dislocation in the neighbourhood of the inclusion by determining Airy's stress functions as a function of the Burger's vector. They addressed the fact that a stable equilibrium position exists in the matrix, even in the vicinity of a perfect interface. In addition, Poisson's ratios of the inclusion and the matrix and the orientation of the Burger's vector, with respect to the inclusion, are shown to have strong influences on matrix cracking. In 1965, Dundurs and Sendeckyj [14] turned their attention to fibre cracking, by solving a similar problem with a perfect interface, but which had a dislocation inside the inclusion. They found that the behaviours of the stable equilibrium position within the inclusion were similar. In 1967, Dundurs [42] evaluated screw dislocation interaction with an inclusion. The influence of inclusion stiffness on the behaviour of the screw dislocation were addressed. A stiff inclusion repels the screw dislocation, whereas a compliant inclusion attracts the dislocation. This result has been proven by many authors in different ways [8, 27, 39, 43].

The interaction between a circular inclusion and an external radial crack was defined in terms of SIFs by Tamate in 1968 [8]. He stated that a relatively stiff inclusion causes lower SIFs, while a softer inclusion results in higher SIFs. His findings were in agreement with the results of Dundurs' 1967 work. In addition, he showed that the near crack tip is more vulnerable to the effects of an inclusion's parameters than the distant crack tip. To overcome the limitations encountered in the form of the series expansion for complex stress potentials used by Tamate, Atkinson [9], in 1972, introduced a dislocation density method in which singular integral equations for the distribution of dislocations which form a crack, are set up and solved numerically. Integrating over the dislocation density yields the stress fields. The solution for a single dislocation interacting with an inclusion was addressed by Dundurs and Mura [13] in 1964. Their solution was generalized to an arbitrarily oriented line crack by Erdogan et al. in 1974 [16]. Over the last two decades, the condition in which an imperfect cohesion exists along an interface including an interphase layer has been incorporated into the two-phase model as described in the previous section [18, 21, 44, 45].

Research into the behaviours of a crack or dislocations interacting with an inclusion in the two-phase composite system has had a significant impact on the development of the fundamental theories of micromechanics. However, in considering the effects of an independent interphase layer; the influence of other inclusions surrounding an inhomogeneity, as in the GSC scheme; or other general three-phase cases such as cemented implant systems (implant/bone/PMMA); the three-phase micromechanics scheme is of great practical importance and interest.

1.2.4 Three-Phase Micromechanics Scheme

.*

The three-phase model, known as the GSC model, was first introduced by Christensen and Lo in 1979 [5]. They addressed the fact that perturbation caused by neighbouring inclusions should be taken into consideration when determining the effective shear modulus of the composite materials. The three-phase model consists of two circular regions and an infinitely extended region. The advantage of this model is that it provides a mathematical simplification of the single inclusion problem and takes into account the effects of neighbouring inclusions and the matrix. Thus, it is a simplified and generalized composite model, which offers more precise determination of the effective parameters, and effects the local characteristics and fracture mechanism of many physical applications. From a classical micromechanics perspective, the internal circular region is considered part of the inclusion, the annulus a part of the matrix phase, and the infinitely extending outer region a part of the composite phase, which is affected by both the neighbouring inclusions and the matrix [5]. Thus, this model presents the fundamental solution for the GSC model analysis.

From an engineering application standpoint, a common physical example of the model is coated fibres embedded in a matrix. The internal component is the fibre, the intermediate region is the coating layer and the outer region is the surrounding matrix. The three-phase model is valuable for simulating the interphase layer associated with bonding in composite materials. Furthermore, the methodology addressed in the three-phase model can easily be expanded into a multi-layer composite material without losing generality [46] and can also be adapted to simulate particulate composite materials in two-dimensional micromechanical analysis [6].

Since Christensen and Lo [5] introduced the three-phase scheme in 1979, it has attracted the attention of many researchers interested in simulating the influence of both neighbouring inclusions and the surrounding matrix in the microstructure analysis of composite materials, and in representing an interphase layer. In 1989, Luo and Weng [47] addressed how stress and strain fields are uniform in a fibre when Eshelby's S-tensor is applied to the three-phase scheme with perfect bonding interfaces. Stress fields in a composite with a coated inclusion were subjected to thermoelastic loading and evaluated by Benveniste et al. in 1989 [28]. In 1991, Luo and Chen compared two-phase models and three-phase models by considering

the effects of volume fibre fraction, and the thickness and material properties of an intermediate phase [48]. They demonstrated different results for the two models. By solving a problem of two concentric circles surrounded by unbounded medium, they addressed how the volume fibre fraction has a profound influence upon cracking in the matrix and how the stiffer surrounding medium, which is the infinitely extending outer region, suppresses crack propagation in the matrix. Also, they illustrated how the three-phase model shows a stable equilibrium position of the dislocation under much less stringent conditions, when compared with the two-phase model used by Dundurs and Mura [13], and how the trapping mechanism of the dislocation is more likely to take place in the three-phase model [49]. Of particular interest is how the stability of the dislocation is observed to be fundamentally independent of the orientation of its Burger's vector. These results are significantly different than the results seen using the two-phase model. In 1998, Ru provided the exact closedform solution for stress fields within an inclusion surrounded by an interphase layer under thermal loads, showing that the interphase layer has strong effects on the local stress, but moderate influences on the mean stress values [50]. In 2002, Wang and Shen, further developed the model by considering imperfect interface such as the inner circumferentially homogeneous sliding interface between an inclusion and the surrounding matrix [51]. By investigating how crack behaviours in the matrix interact with the imperfect interface, they depicted how the sliding interface influences the SIFs—the degradation of the initial bonding condition plays a predominant role in altering SIFs.

1.3 Applications of the Three-Phase Model

1.3.1 Interphase Layer of Composite Materials

It is well known that the quality of adhesion along an interface between a fibre and the matrix has a profound influence on the performance and failure of composite materials, since the effective load transfer between the fibre and the matrix are compromised due to bonding defects, such as voids, impurities and cracking. The intermediate zone representing a state of adhesion between a fibre and matrix is known as the interphase layer. The interphase layer can be illustrated by a two-phase model having an imperfect interface, with the assumption that its thickness is very thin and there are three parameters, the thickness and two elastic constants, which are incorporated into two spring-type interface parameters. However, as addressed in Section 1.2.2, the three-phase model is preferred, since it provides a more precise analysis of the interphase layer's influence on the failure mechanism of composite materials.

The interphase layer is a chemical interaction by-product of the bonding between the constituents. However, it may also be introduced in the design stage to improve performance of the composite constituents. For example, fibres may be coated to improve the bonding between the fibres and the matrix. Xiao and Chen [52, 53] investigated the effect of coated fibre composites on bonding strength. They demonstrated that the thickness and mechanical properties of the coating materials compete with each other to alter the stress field around the circular inclusion. With a thick coating layer, the elastic properties of an inclusion have limited effects on the force of a dislocation, while the mechanical properties of both an inclusion and coating have substantial influence on the equilibrium position and the stability of the dislocation in the case of a thin coating layer. The interphase layer can also include a thermal barrier coating in order to avoid thermal mismatch induced stresses in passivated interconnect lines in integrated circuits. Ru addressed how the interphase layer, with an intermediate thermal expansion coefficient between that of the matrix and the inclusion, should be implemented in order to reduce thermal stresses within both the inclusion and interphase layer [50]. In addition, in order to study stress and electric field concentrations in piezoelectric composites and devices, Sudak investigated the relationship between residual electroelastic stresses induced by electromechanical loadings, and lattice mismatch between buried active components and surrounding materials [54]. Considering the interphase layer as a continuum with finite thickness, the works mentioned above have adapted the three-phase model to investigate its effects on the behaviour of composite materials.

1.3.2 A Model of A Cemented Implant

According to the National Hospital Discharge Survey of 2003, 36,000 of the 217,000 total hip replacement (THR) surgeries and 33,000 of the 402,000 total knee replacement (TKR) operations performed in the United States underwent revision surgery. Cemented implants, usually PMMA, are among the most popular types of prostheses. The predominant failure of cemented implants is aseptic loosening of the stem from a host bone. Of the many factors causing aseptic loosening, damage to the mechanical interlocking along the interfaces is the primary reason for loosening of a stem from the bone cement.

It is important to understand the behaviours of each interface, namely, bone/cement and cement/implant, to predict failure of the cemented prosthesis system. First, the interface between an implant and the bone cement can have either a smooth polished

...

surface, to facilitate implant insertion into the cement, or a porous surface, to increase interdigitation between the constituents.

The interface between bone cement and the host bone has strong interdigitation. The interdigitation in a well-fixed cementing keeps its initially strong mechanical interlocking and prohibits any relative shear slip. Compared to the adhesion between an implant and the bone cement, the mechanical interlocking and bonding along this interface is strong and less vulnerable to loosening. The three-phase model with two independently imperfect interfaces is of fundamental importance in predicting failure in an implant/cement/bone system.

1.4 Overview of the Current Study

Studying the behaviour of cracks or dislocations interacting with inclusions and the surrounding matrix is of fundamental importance in order to understand failure mechanisms and to enhance the performance of composite materials. A simplified scheme consisting of a single inclusion and a crack or a dislocation has been adopted for theoretical analyses by many researchers and has provided reasonable and reliable results. The application of the three-phase model includes bonding layers, any special coatings and cemented implants embedded into bone.

The majority of the research dealing with the three-phase model has assumed that both inner and outer interfaces are perfectly bonded or that only the inner boundary has imperfect adhesion. Although the imperfect interface has been considered a predominant parameter in micromechanical analysis and the degree of bonding degeneration at the interfaces has been known to dramatically alter stress fields in the composite constituents [26, 27], applying the imperfect interface condition to the three phase model is rare in the literature due to mathematical difficulties involved. To illustrate this point, we will examine the dislocation density method.

The dislocation density method, also called the singular integral equation method, was introduced and developed by Atkinson [9] and Erdogan et al. [16] and has been the most popular and widely accepted for simulating crack-inclusion or dislocationinclusion interactions. Although this procedure has contributed to the development of micromechanics of composite materials, the computational difficulties associated with solving the singular integral equations for imperfect interfaces are still extremely challenging. To overcome the difficulties, a relatively simple series method has been incorporated into a two-phase imperfect interface model by Amenyah et al. [26] and Liu et al. [27], and into a three-phase model with an imperfect inner interface by Sudak [41] and Kim [39]. Sudak studied interactions between a screw dislocation and an inclusion in anti-plane shear employing a three-phase model with an imperfect interface by determining unknown complex coefficients using the series method. By establishing coefficients of complex potentials, Kim and Sudak also solved the threephase inclusion-crack interaction with general imperfect bonding conditions. Wang and Shen proposed the decoupled strategy, in which the dislocation density method and the series method were combined to show crack-inclusion interaction with sliding interface condition [51]. In this view, the three-phase model with imperfect bonding along both inner and outer interfaces is considered to be the preferred and more precise model. However, although the three-phase model with inner/outer imperfect interfaces is identified, to the best of my knowledge, there is no research in the literature dealing with the three-phase scheme having imperfect bonding along both inner and outer interfaces.

In the present work, both internal and external annular interfaces are considered imperfectly bonded and the series method is used to resolve the mathematical difficul-

ties. Imperfectness of interfaces is modeled with dimensionless spring-type interface parameters and the effect of the parameters on the SIFs at crack tips is shown along crack locations. Thus, the aim of this research is to address how imperfections of both interfaces simultaneously affects propagation of a pre-existing radial crack in the annular region in plane elastostatics during the mechanical cracking processes.

This thesis is organized into six chapters. Following this introduction which reviewed the basis of the three-phase model with imperfect interfaces, Chapter 2 provides a derivation of the mathematical formulations of the boundary value problem. Specifically, a semi-analytical solution to plane elastic deformation of a single inclusion interacting with a pre-existing radial crack is presented. The boundary conditions of an imperfect bonding along two interfaces are expressed in terms of stress potentials, which are represented solely in series forms. In Chapter 3, a rigorous solution to the boundary value problem is derived from the two imperfect bonding interfaces and a set of general algebraic equations is obtained. Chapter 4 outlines the numerical analysis procedures used to solve the algebraic equations and lays a foundation for SIFs to describe the behaviour of a radial crack under mode I loading. For the verification of formulations, numerical results from the derived formulations with perfect interfaces are compared to those published. In particular, a new phenomenon, the socalled stable zone associated with debonding, is discussed in this chapter. Chapter 5 deals with another application example of this rigorous model, namely implant/bone cement/bone system. The semi-analytical solution to the three-phase model is applied to the cemented implant design to show how the non-slip surface conditions of the implant affect performance and failure of the prosthesis. Chapter 6 summarizes the new findings of this research and concludes that the three-phase model having imperfect bonding along both inner and outer interfaces is general and preferred from

.

17

application perspective and the imperfectness of both interfaces should be taken into account for a more precise analysis. A brief insight into future work is presented at the end of the chapter.

.



Figure 1.1 Phases of a composite materials (from Daniel et al. [3])

÷



Figure 1.2 Classification of Composite Materials

.

4

.

20

CHAPTER 2

Analytical Formulation of A Boundary Value Problem

2.1 Introduction

We consider a domain in \mathbb{R}^2 , infinite in extent, containing an inclusion, a matrix zone with a pre-existing radial crack and a composite phase (Figure 2.1). The inclusion, with center at the origin of the coordinate system and radius R_0 occupies a region denoted by S_0 and the matrix zone is modeled as an annulus with outer radius R_1 and represented by S_1 . The surrounding composite phase is represented by S_2 . All the materials occupying S_0 , S_1 and S_2 are assumed to be homogeneous and isotropic. The inclusion/matrix interface and the matrix/composite phase interface are denoted by curves $\partial \Gamma_0$ and $\partial \Gamma_1$, respectively and the interfaces $\partial \Gamma_0$ and $\partial \Gamma_1$ are assumed to be homogeneously imperfect. The crack has length of 2l and unless otherwise stated, the subscripts 0, 1 and 2 will denote quantities in the domain S_0 , S_1 and S_2 , respectively.

2.2 Formulation

For plane deformation, the elastic stresses and their respective displacements in the polar coordinate system can be given in terms of two complex potentials $\varphi(z)$ and $\psi(z)$ as follows [55]:

$$2\mu (u_r + iu_\theta) = e^{-i\theta} \left[\kappa \varphi(z) - z \overline{\varphi'(z)} - \overline{\psi(z)} \right], \qquad (2.1)$$

$$\sigma_{rr} + \sigma_{\theta\theta} = 2 \left[\varphi'(z) + \overline{\varphi'(z)} \right], \qquad (2.1)$$

$$\sigma_{rr} - i\sigma_{r\theta} = \varphi'(z) + \overline{\varphi'(z)} - e^{2i\theta} \left[\overline{z} \varphi''(z) + \psi'(z) \right],$$

where

 $z = x + iy = re^{i\theta}$ is the complex coordinate,

 μ = the shear modulus,

 $\nu =$ Poisson's ratio,

 $\kappa = 3 - 4\nu$ for plane strain, $(3 - \nu) / (1 + \nu)$ for plane stress.

Also, the resultant force acting on an arbitrary arc AB in an elastic body is given by [55]

$$F_x + iF_y = -i\left[\varphi(z) + z\overline{\varphi'(z)} + \overline{\psi(z)}\right]_A^B \quad , \tag{2.2}$$

where $[f(*)]_{A}^{B} = f(B) - f(A)$.

Among the various mechanical descriptions of an imperfect interface, one of the most widely accepted models is based on the premise that tractions are continuous but displacements are discontinuous. Specifically, displacement jumps are proportional, in terms of 'spring-factor type' interface parameters, to their respective traction components. In view of this, let us assume that the inclusion is imperfectly bonded to the matrix along the circular curve $\partial \Gamma_0$ and the matrix zone is also imperfectly bonded to the composite phase along $\partial \Gamma_1$. Then, the boundary value equations describing the problem above can be formulated. The traction continuity and displacement jump condition along Γ_0 are given by [40]

$$\|\sigma_{rr} - i\sigma_{r\theta}\| = 0,$$

$$\sigma_{rr} = m_0 \|u_r\| - m_0 u_r^0 \quad , \quad \sigma_{r\theta} = n_0 \|u_\theta\| - n_0 u_\theta^0 \quad \partial \Gamma_0.$$
 (2.3)

In the same way, the other boundary conditions along Γ_1 are given by

 $\|\sigma_{rr} - i\sigma_{r\theta}\| = 0,$

$$\sigma_{rr} = m_1 \|u_r\| - m_1 u_r^0 \quad , \quad \sigma_{r\theta} = n_1 \|u_\theta\| - n_1 u_\theta^0 \quad \partial \Gamma_1, \tag{2.4}$$

where m and n are non-negative spring-factor type interface parameter representing the degree of interface damage. Physically, these parameters represent the mechanical properties of the interface such as strength, stiffness and the overall degree of adhesion along the interface [37]. The bracket expression $\|*\|$ denotes jump across the interface, namely $\|*\| = \|*\|_1 - \|*\|_0$ along Γ_0 and $\|*\| = \|*\|_2 - \|*\|_1$ along Γ_1 , and u_r^0 and u^0_{θ} are additional displacements induced by the uniform stress-free eigenstrains $\left(\varepsilon_x^0,\varepsilon_y^0,\varepsilon_z^0\right)$ prescribed within the inclusion and the bonding medium which might be induced by thermal mismatch between adjoining materials. The advantage of defining the interface parameters m in the normal direction and n in the tangential direction in (2.3) and (2.4) is that they allow representation of degree of the interface imperfectness from complete debonding to perfect bonding through intermediate states. It is shown that the parameters, $m = n = \infty$ in (2.3) and (2.4) lead the displacement jumps across the interfaces to zero, which corresponds to a perfect bonding condition. Interface tractions vanish as the value of the parameters approaches zero, which is associated with complete debonding. Any positive value of these parameters represents a degree of bonding inadequacy along the interfaces.

2.3 Boundary Value Problem

The boundary conditions along Γ_0 and Γ_1 in (2.3) and (2.4) need to be expressed in terms of analytic functions $\varphi(z)$ and $\psi(z)$. Each region such as S_0 , S_1 and S_2 requires two stress potentials to describe its stress fields so that it is required that six analytic functions $\varphi_0(z)$, $\psi_0(z), \varphi_1(z), \psi_1(z), \varphi_2(z)$ and $\psi_2(z)$ be defined. Attention should be given to the boundary conditions along Γ_0 and Γ_1 in region S_1 since a radial crack causes multi-valuedness. As a result, the analytic functions $\varphi_1(z)$ and $\psi_1(z)$ are divided into two pairs such as $\varphi_{11}(z)$, $\psi_{11}(z)$ and $\varphi_{12}(z)$, $\psi_{12}(z)$: thus, $\varphi_{11}(z)$ and $\psi_{11}(z)$ are defined on and outward from Γ_0 while $\varphi_{12}(z)$ and $\psi_{12}(z)$ are defined on and inward from Γ_1 (See section 2.4 for details).

The traction continuation along Γ_0 in $(2.3)_1$ leads to the following expression

$$(\sigma_{rr} - i\sigma_{r\theta})_1 = (\sigma_{rr} - i\sigma_{r\theta})_0.$$
(2.5)

In view of $(2.1)_3$, the above equation is re-written in terms of analytic functions as

$$\varphi_1'(z) + \overline{\varphi_1'(z)} - e^{2i\theta} \left[\overline{z} \varphi_1''(z) + \psi_1'(z) \right] = \varphi_0'(z) + \overline{\varphi_0'(z)} - e^{2i\theta} \left[\overline{z} \varphi_0''(z) + \psi_0'(z) \right].$$
(2.6)

Since $z\overline{z} = R_0^2$ and $z = R_0 e^{i\theta}$ at $\partial\Gamma_0$ and $\overline{\varphi'(\overline{z})} = \overline{\varphi}'(z)$ according to analytic continuation $(\overline{\varphi'(z)} = \overline{\varphi}'(\overline{z})), \ \overline{z} = \frac{R_0^2}{z}$ and $e^{i\theta} = \frac{z}{R_0}$ are substituted into the equation (2.6) and it can be recast into the following form

$$\varphi_{11}'(z) + \overline{\varphi_{11}'}\left(\frac{R_0^2}{z}\right) - z\varphi_{11}''(z) - \frac{z^2}{R_0^2}\psi_{11}'(z) = \varphi_0'(z) + \overline{\varphi_0'}\left(\frac{R_0^2}{z}\right) - z\varphi_0''(z) - \frac{z^2}{R_0^2}\psi_0'(z) \ \partial\Gamma_0.$$
(2.7)

It is noted that the stress potentials φ_1 and ψ_1 are replaced with φ_{11} and ψ_{11} .

The normal and tangential displacement jump condition along Γ_0 in (2.3)₂ is rewritten based on (2.3)₁ as

$$\sigma_{rr} - i\sigma_{r\theta} = \left(m_0 \|u_r\| - m_0 u_r^0\right) - i\left(n_0 \|u_\theta\| - n_0 u_\theta^0\right), \quad \partial\Gamma_0.$$
(2.8)

According to the assumption of traction continuation condition in $(2.3)_1$, the left-

٠.

hand-side of (2.8) can be either $(\sigma_{rr} - i\sigma_{r\theta})_0 = (\sigma_{rr} - i\sigma_{r\theta})_1$ along Γ_0 . Letting $\sigma_{rr} - i\sigma_{r\theta} = (\sigma_{rr} - i\sigma_{r\theta})_0$, we write the displacement jump condition in the following form (see Appendix 1 for detailed derivation)

$$(\sigma_{rr} - i\sigma_{r\theta})_0 = \left(\frac{m_0 - n_0}{2}\right) \|u_r + iu_\theta\| + \left(\frac{m_0 + n_0}{2}\right) \|u_r - iu_\theta\| - \left(m_0 u_r^0 - in_0 u_\theta^0\right).$$
(2.9)

The additional displacement induced by the eigenstrains prescribed in the inclusion can be written as [10]

$$u_r^0 = R\left(\varepsilon_x^0 \cos^2\theta + \varepsilon_y^0 \sin^2\theta + \varepsilon_{xy}^0 \sin 2\theta\right), \quad u_\theta^0 = R\left(\frac{\varepsilon_y^0 - \varepsilon_x^0}{2} \sin 2\theta + \varepsilon_{xy}^0 \cos 2\theta\right).$$
(2.10)

Sudak et.al.[25] have shown that the last term in (2.9), the displacement induced by the uniform eigenstrains can be written in the following form

$$\left(m_0 u_r^0 - i n_0 u_\theta^0\right) = m R_0 \varepsilon_1 + \left(\frac{m+n}{2R_0}\right) \left(\varepsilon_2 - i\varepsilon_3\right) z^2 + \left(\frac{m-n}{2z^2}\right) R_0^3 \left(\varepsilon_2 + i\varepsilon_3\right), \quad \partial \Gamma_{0,1}$$

$$(2.11)$$

where $\varepsilon_1 = \frac{\varepsilon_x^0 + \varepsilon_y^0}{2}$, $\varepsilon_2 = \frac{\varepsilon_x^0 - \varepsilon_y^0}{2}$ and $\varepsilon_3 = \varepsilon_{xy}^0$. By substituting (2.11) into (2.9) and then using (2.1) and (2.2), we can express the displacement jump condition (2.3)₂ in the following form (see Appendix 1 for detailed derivation)

$$(m_0 - n_0)(\kappa_1 + 1)\frac{R_0}{z}\varphi_{11}(z) + (m_0 + n_0)(\kappa_1 + 1)\frac{z}{R_0}\overline{\varphi_{11}}\left(\frac{R_0^2}{z}\right)$$
(2.12)
= $4\mu_1 \left[\varphi_0'(z) + \overline{\varphi_0'}\left(\frac{R_0^2}{z}\right) - z\varphi_0''(z) - \frac{z^2}{R_0^2}\psi_0'(z)\right] + (m_0 - n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})\frac{R_0}{z}\varphi_0(z)$

$$+ (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0}) \left[R_0 \overline{\varphi'_0} \left(\frac{R_0^2}{z} \right) + \frac{R_0}{z} \overline{\psi_0} \left(\frac{R_0^2}{z} \right) \right]$$

$$+ (m_0 + n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0}) \frac{z}{R_0} \overline{\varphi_0} (\frac{R_0^2}{z}) + (m_0 + n_0)(1 - \frac{\mu_1}{\mu_0}) \left[R_0 \varphi'_0(z) + \frac{z}{R_0} \psi_0(z) \right]$$

$$+ 4m_0 R_0 \mu_1 \varepsilon_1 + \frac{2\mu_1 (m_0 + n_0)(\varepsilon_2 - i\varepsilon_3)}{R_0} z^2 + \frac{2\mu_1 (m_0 - n_0)(\varepsilon_2 + i\varepsilon_3) R_0^3}{z^2}$$

 $,z\in\Gamma_{0}.$

In a similar fashion of (2.7), the traction continuation along Γ_1 in (2.4)₁ can be written as

$$\varphi_2'(z) + \overline{\varphi_2'} \left(\frac{R_1^2}{z}\right) - z\varphi_2''(z) - \frac{z^2}{R_1^2} \psi_2'(z) = \varphi_{12}'(z) + \overline{\varphi_{12}'} \left(\frac{R_1^2}{z}\right) - z\varphi_{12}''(z) - \frac{z^2}{R_1^2} \psi_{12}'(z) \quad (2.13)$$

 $z \in \Gamma_1$. It is noted that the stress potentials φ_{12} and ψ_{12} are used. Also, the displacement jump condition along Γ_1 in $(2.4)_2$ can be expressed in terms of stress potentials as follows (see Appendix 2 for detailed derivation)

$$(m_{1} - n_{1}) (\kappa_{1} + 1) \frac{R_{1}}{z} \varphi_{12} (z) + (m_{1} + n_{1}) (\kappa_{1} + 1) \frac{z}{R_{1}} \overline{\varphi_{12}} \left(\frac{R_{1}^{2}}{z}\right)$$
(2.14)

$$= -4\mu_{1} \left[\varphi_{2}'(z) + \overline{\varphi_{2}'} \left(\frac{R_{1}^{2}}{z}\right) - z\varphi_{2}''(z) - \frac{z^{2}}{R_{1}^{2}} \psi_{2}'(z) \right]$$

$$+ (m_{1} - n_{1}) \left(1 - \frac{\mu_{1}}{\mu_{2}}\right) \left[R_{1} \overline{\varphi_{2}'} \left(\frac{R_{1}^{2}}{z}\right) + \frac{R_{1}}{z} \overline{\psi_{2}} \left(\frac{R_{1}^{2}}{z}\right) \right]$$

$$+ (m_{1} - n_{1}) \left(1 + \frac{\mu_{1}}{\mu_{2}} \kappa_{2}\right) \frac{R_{1}}{z} \varphi_{2} (z) + (m_{1} + n_{1}) \frac{z}{R_{1}} \left(1 + \frac{\mu_{1}}{\mu_{2}} \kappa_{2}\right) \overline{\varphi_{2}} \left(\frac{R_{1}^{2}}{z}\right)$$

$$+ (m_{1} + n_{1}) \frac{z}{R_{1}} \left(1 - \frac{\mu_{1}}{\mu_{2}}\right) \left[\psi_{2} (z) + \overline{z} \varphi_{2}' (z)\right]$$

 $, z \in \Gamma_1.$

In (2.14) there are no terms associated with the displacements induced by the uniform eigenstrains. Also, two different pairs of stress potentials in region S_1 are defined. The $\varphi_{11}(z)$ and $\psi_{11}(z)$ defined along $\partial \Gamma_0$ are incorporated into (2.7) and

•
(2.12): the other pair of analytic functions $\varphi_{12}(z)$ and $\psi_{12}(z)$ are used in (2.13) and (2.14) along $\partial \Gamma_1$. Thus, the boundary value problem describing the imperfect bonding condition along both inclusion/matrix and matrix/composite is expressed in the four equations such as (2.7), (2.12), (2.13) and (2.14) which are given solely in terms of stress potentials.

2.4 Series Representation for Stress Potentials

The stress potentials can be expressed solely in series representation. As addressed in Chapter 1, to overcome mathematical difficulties related to solving singular integral equations, this relatively simple series method is often adapted in the three-phase model. Now, the eight stress potentials shown in (2.7), (2.12), (2.13) and (2.14) are represented as series expansions in the variable of z within each domain.

Let's consider region S_0 first. The stress potentials $\varphi_0(z)$ and $\psi_0(z)$ are analytic within a inclusion and can be expanded into a Taylor series in S_0 as follows

$$\varphi_0(z) = \sum_{k=0}^{\infty} a_k z^k, \qquad \psi_0(z) = \sum_{k=0}^{\infty} b_k z^k,$$
(2.15)

where a_k and b_k are undetermined complex coefficients.

Secondly, the infinitely expanded region S_2 , namely the composite phase is considered. The stresses X_x , Y_y and X_y are assumed to exist and to be bounded and finite at infinity. In this view, the equations of $(2.1)_{2,3}$ lead to a conclusion that $\varphi_2(z)$ and $\psi_2(z)$ must have O(1) as $|z| \to \infty$. Thus, the remote loading at intinite can be characterized by the uniform stress field

$$\varphi_2(z) = Az + O(1), \quad \psi_2(z) = Bz + O(1), \quad \text{as } |z| \to \infty,$$
(2.16)

where A is a given real number and B is a given complex number. Stresses in a Cartesian system [55] are given as

$$X_{x} + Y_{y} = 2 \left[\varphi_{2}'(z) + \overline{\varphi_{2}'(z)} \right], \qquad (2.17)$$

$$Y_{y} - X_{x} + 2iX_{y} = 2 \left[\overline{z} \varphi_{2}''(z) + \psi_{2}'(z) \right].$$

By substituting (2.16) into (2.17) and solving the equations, we get a relationship between stresses and the constants in the following form

$$X_x + Y_y = 4A, \ Y_y + iX_y = 2A + B, \ X_x - iX_y = 2A - B.$$
 (2.18)

Now, let X_x, Y_y and X_y denote principal stresses $(\sigma_x^{\infty}, \sigma_y^{\infty}, \sigma_{xy}^{\infty})$ at $z = \infty$, respectively. We re-write (2.18) as

$$\sigma_x^{\infty} + \sigma_y^{\infty} = 4A, \ \sigma_y^{\infty} + i\sigma_{xy}^{\infty} = 2A + B, \ \sigma_x^{\infty} - i\sigma_{xy}^{\infty} = 2A - B.$$
(2.19)

In this dissertation, we consider the case of a uniaxial load normal to the radial crack, namely, $\sigma_x^{\infty} = 0$, $\sigma_y^{\infty} = \sigma^{\infty}$ and $\sigma_{xy}^{\infty} = 0$, and then the constants A and B are determined as

$$A = \frac{\sigma^{\infty}}{4}, B = \bar{B} = 2A = \frac{\sigma^{\infty}}{2}.$$
(2.20)

Thus, the stress potentials $\varphi_2(z)$ and $\psi_2(z)$ are expanded into a standard Laurent series in region S_2 in the following form

$$\varphi_2(z) = Az + \sum_{k=1}^{\infty} c_k z^{-k}, \qquad \psi_2(z) = Bz + \sum_{k=1}^{\infty} d_k z^{-k},$$
(2.21)

÷

ન

where $A = \frac{\sigma^{\infty}}{4}$, $B = \overline{B} = \frac{\sigma^{\infty}}{2}$.

Finally, the stress potentials $\varphi_1(z)$ and $\psi_1(z)$ in region S_1 containing a pre-existing radial crack have been expanded as series forms. In the present problem, since the intermediate matrix region S_1 contains a crack, $\varphi_1(z)$ and $\psi_1(z)$ are not analytic in the region. To solve the difficulties, let us employ analytic continuation [55] to express $\varphi_1(z)$ and $\psi_1(z)$ in terms of two new functions that are analytic in a new domain Drepresenting the intermediate region without the radial crack. Namely, $D = S_1 - 2l$. Thus, the stress potentials $\varphi_1(z)$ and $\psi_1(z)$ can be expanded into standard Laurent series in D. In view of (2.2) and traction free condition along the crack-face 2l, we can get the following condition

$$F_x + iF_y = -i\left[\varphi(z) + z\overline{\varphi'(z)} + \overline{\psi(z)}\right]_A^B = 0.$$
(2.22)

The above condition is expressed along the crack face in the upper and lower half planes as follows

$$\varphi_{1}(z)^{+} + z\overline{\varphi_{1}'(z)^{+}} + \overline{\psi_{1}(z)^{+}} = 0, \quad z \in 2l^{+},$$

$$\varphi_{1}(z)^{-} + z\overline{\varphi_{1}'(z)^{-}} + \overline{\psi_{1}(z)^{-}} = 0, \quad z \in 2l^{-}.$$
(2.23)

Let's consider the analytic continuation across the boundary which is the real axis. In particular, if the line of symmetry is the real axis, Schwarz's Reflection Principle can be employed into any analytic function f_1 in the upper plane D^+ . f_2 defined in the lower plane D^- by $f_2(z) = \overline{f_1(\overline{z})}$ is the analytic continuation of f_1 to D^- [56]. Muskhelishvili [55] defines the $f_2(z)$ as a function to have the conjugate complex value of f_1 at the point \overline{z} , namely $\overline{f}(z) = \overline{f(\overline{z})}$. By analytic continuation,

$$\begin{bmatrix} \overline{\varphi_{1}'(z)} \end{bmatrix}^{+} = \overline{\varphi}_{1}'(\overline{z}) = [\overline{\varphi}_{1}'(z)]^{-} \text{ and } \begin{bmatrix} \overline{\psi}_{1}(z) \end{bmatrix}^{+} = \overline{\psi}_{1}(\overline{z}) = [\overline{\psi}_{1}(z)]^{-}. \text{ Thus,}$$

$$\varphi_{1}(z)^{+} + \begin{bmatrix} z\overline{\varphi}_{1}'(z) + \overline{\psi}_{1}(z) \end{bmatrix}^{-} = 0, \quad z \in 2l, \qquad (2.24)$$

$$\varphi_{1}(z)^{-} + \begin{bmatrix} z\overline{\varphi}_{1}'(z) + \overline{\psi}_{1}(z) \end{bmatrix}^{+} = 0, \quad z \in 2l.$$

By adding and subtracting the above equations, we get

$$\varphi_{1}(z)^{+} - \left[z\overline{\varphi}_{1}'(z) + \overline{\psi}_{1}(z)\right]^{+} = \varphi_{1}(z)^{-} - \left[z\overline{\varphi}_{1}'(z) + \overline{\psi}_{1}(z)\right]^{-}, \quad z \in 2l, (2.25)$$

$$\varphi_{1}(z)^{+} + \left[z\overline{\varphi}_{1}'(z) + \overline{\psi}_{1}(z)\right]^{+} = -\left\{\varphi_{1}(z)^{-} + \left[z\overline{\varphi}_{1}'(z) + \overline{\psi}_{1}(z)\right]^{-}\right\}, \quad z \in 2l.$$

First, let's consider $(2.25)_1$. From the equation $(2.25)_1$, a new analytic function is defined as follows

$$X(z) \equiv \varphi_1(z) - \left[z \overline{\varphi}_1'(z) + \overline{\psi}_1(z) \right].$$
(2.26)

In view of $(2.25)_1$, the following relation holds

$$X(z)^{+} = G(z)X(z)^{-} + f(z), \qquad z \in 2l,$$
(2.27)

where G(z) = 1 and f(z) = 0. The above problem is called the problem of linear relationship or the Hilbert Problem [55]. Equation $(2.25)_1$ is expressed in terms of the new function such as $X(z)^+ = X(z)^-$, which means that X(z) is continuous across the crack 2*l*. Thus, X(z) is analytic in *D* representing a domain S_1 minus 2*l* and across the crack face 2*l* and can be expanded into a Laurent series in *D* as follows

$$X(z) = \sum_{k=-\infty}^{\infty} e_k z^k, \quad z \in D,$$
(2.28)

•

where e_k are unknown complex coefficients.

Secondly, $(2.25)_2$ shows that the value of $\varphi_1(z) + [z\overline{\varphi}'_1(z) + \overline{\psi}_1(z)]$ has opposite sign across the crack face and this is an another *Hilbert Problem* with G(z) = -1 and f(z) = 0. By Plemelj's function and a solution given by Muskhelishvili [55], we get

$$\varphi_1(z)^+ + \left[z \overline{\varphi}_1'(z) + \overline{\psi}_1(z) \right] = \frac{P(z)}{\sqrt{(z-a)(z-b)}} , \qquad (2.29)$$

where P(z) is an arbitrary polynomial and

.

$$\left[\sqrt{(z-a)(z-b)}\right]^{+} = -\left[\sqrt{(z-a)(z-b)}\right]^{-}, \ z \in 2l = (a,b).$$

For example, if the outer radius R_1 is extended to the infinity and the value of the above equation (2.29) is finite and bounded at infinity, the polynomial P(z) would be taken as z. In the case where (2.29) holds in intermediate zone S_1 , P(z) is taken to be 1 for easy formulation. Now, let's define another new analytic function such as

$$Y(z) \equiv \sqrt{(z-a)(z-b)} \left[\varphi_1(z) + z\overline{\varphi}_1'(z) + \overline{\psi}_1(z)\right]. \qquad (2.30)$$

Similarly to X(z), Y(z) is continuous across the crack $2l (Y(z)^+ = Y(z)^-)$ and is analytic in D. Hence, the new analytic function can be expanded into a standard Laurent series as follows

$$Y(z) = \sum_{k=-\infty}^{\infty} f_k z^k, \qquad z \in D,$$
(2.31)

where f_k are unknown complex coefficients. Consequently, from (2.26) and (2.30),

the stress potentials $\varphi_1(z)$ and $\psi_1(z)$ defined in S_1 take the following forms

$$\varphi_1(z) = \frac{1}{2\sqrt{(z-a)(z-b)}}Y(z) + \frac{1}{2}X(z), \qquad (2.32)$$

and

$$\psi_1(z) = -\overline{X}(z) + \overline{\varphi_1}(z) - z\varphi_1'(z).$$
(2.33)

It is noted that the aim of derivations of the stress potentials $\varphi_1(z)$ and $\psi_1(z)$ in S_1 is to investigate crack behaviours around crack tips, in other words, the inward crack propagation from crack tip a to the inner interface Γ_0 and the outward propagation from the other tip b to the outer interface Γ_1 . In (2.32), the term of $\frac{1}{2\sqrt{(z-a)(z-b)}}$ is a multi-valued function across the crack face but is analytic within a circular domain |z| < a including $\Gamma_0 \cup S_0$ and within an outward circular domain |z| > b including $\Gamma_1 \cup S_2$. Let F(z) denote $\frac{1}{2\sqrt{(z-a)(z-b)}}$ and expand it into standard series forms. Since two different domains are considered, two notations, $F(z)_{11}$ and $F(z)_{12}$ represent the value of $\frac{1}{2\sqrt{(z-a)(z-b)}}$ within a domain |z| < a including $\Gamma_0 \cup S_0$ and |z| > b including $\Gamma_1 \cup S_2$, respectively.

First, $F(z)_{11}$ in a domain |z| < a is expanded into a Taylor series as follows

$$F_{11}(z) = \frac{1}{2\sqrt{(z-a)(z-b)}} = \frac{1}{2}(z-a)^{-\frac{1}{2}}(z-b)^{-\frac{1}{2}}$$
(2.34)
= $\sum_{k=0}^{\infty} g_k z^k = g_0 + g_1 z + g_2 z^2 + g_3 z^3 + g_4 z^4 + g_5 z^5 + g_6 z^6 + g_7 z^7 + \dots,$

where g_k are coefficients which are determined in terms of the crack tip position a and b. For example, the first eight coefficients are the following (see Appendix 3 for

detailed derivation)

÷

$$\begin{split} g_{0} &= \frac{1}{2\sqrt{ab}}, \\ g_{1} &= \left(\frac{1}{4}\frac{1}{\sqrt{a^{1}b^{3}}} + \frac{1}{4}\frac{1}{\sqrt{a^{3}b^{1}}}\right), \\ g_{2} &= \left(\frac{3}{16}\frac{1}{\sqrt{a^{1}b^{5}}} + \frac{1}{8}\frac{1}{\sqrt{a^{3}b^{3}}} + \frac{3}{16}\frac{1}{\sqrt{a^{5}b^{1}}}\right), \\ g_{3} &= \left(\frac{5}{32}\frac{1}{\sqrt{a^{1}b^{7}}} + \frac{3}{32}\frac{1}{\sqrt{a^{3}b^{5}}} + \frac{3}{32}\frac{1}{\sqrt{a^{5}b^{3}}} + \frac{5}{32}\frac{1}{\sqrt{a^{7}b^{1}}}\right), \\ g_{4} &= \left(\frac{35}{256}\frac{1}{\sqrt{a^{1}b^{9}}} + \frac{5}{64}\frac{1}{\sqrt{a^{3}b^{7}}} + \frac{9}{128}\frac{1}{\sqrt{a^{5}b^{5}}} + \frac{5}{64}\frac{1}{\sqrt{a^{7}b^{3}}} + \frac{35}{256}\frac{1}{\sqrt{a^{9}b^{1}}}\right), \\ g_{5} &= \left(\frac{\frac{63}{512}\frac{1}{\sqrt{a^{1}b^{11}}} + \frac{35}{512}\frac{1}{\sqrt{a^{3}b^{9}}} + \frac{15}{256}\frac{1}{\sqrt{a^{5}b^{7}}} + \frac{15}{256}\frac{1}{\sqrt{a^{7}b^{5}}} + \frac{35}{512}\frac{1}{\sqrt{a^{9}b^{3}}}\right), \\ g_{6} &= \left(\frac{\frac{231}{2048}\frac{1}{\sqrt{a^{1}b^{13}}} + \frac{63}{1024}\frac{1}{\sqrt{a^{3}b^{11}}} + \frac{105}{2048}\frac{1}{\sqrt{a^{5}b^{9}}} + \frac{25}{512}\frac{1}{\sqrt{a^{7}b^{7}}} + \frac{105}{2048}\frac{1}{\sqrt{a^{9}b^{3}}}}\right), \\ g_{7} &= \left(\frac{\frac{3003}{28672}\frac{1}{\sqrt{a^{1}b^{16}}} + \frac{231}{4096}\frac{1}{\sqrt{a^{3}b^{13}}} + \frac{189}{4096}\frac{1}{\sqrt{a^{13}b^{3}}} + \frac{3003}{28672}\frac{1}{\sqrt{a^{1}b^{16}}}}\right). \end{split}$$

Now, let's expand $F(z)_{12}$ in a domain |z| > b into Laurent series as follows

$$F_{12}(z) = \frac{1}{2\sqrt{(z-a)(z-b)}} = \frac{1}{\sqrt{2z}\sqrt{(1-\frac{a}{z})(1-\frac{b}{z})}}$$

$$= \frac{1}{2z}(1-\frac{a}{z})^{-\frac{1}{2}}(1-\frac{b}{z})^{-\frac{1}{2}} = \sum_{k=1}^{\infty} h_{-k}z^{-k}$$

$$= \begin{pmatrix} h_{-1}z^{-1} + h_{-2}z^{-2} + h_{-3}z^{-3} + h_{-4}z^{-4} + h_{-5}z^{-5} \\ +h_{-6}z^{-6} + h_{-7}z^{-7} + h_{-8}z^{-8} + \dots \end{pmatrix},$$
(2.35)

where h_{-k} are also coefficients which are determined in terms of the crack tip position

a and b in the following form (see Appendix 3 for detailed derivation)

$$\begin{split} h_{-1} &= \frac{1}{2}, \\ h_{-2} &= \frac{1}{2} \left(\frac{a+b}{2} \right), \\ h_{-3} &= \frac{1}{2} \left(\frac{3}{8}a^2 + \frac{1}{4}ab + \frac{3}{8}b^2 \right), \\ h_{-4} &= \frac{1}{2} \left(\frac{5}{16}a^3 + \frac{3}{16}a^2b + \frac{3}{16}ab^2 + \frac{5}{16}b^3 \right), \\ h_{-5} &= \frac{1}{2} \left(\frac{35}{128}a^4 + \frac{5}{32}a^3b + \frac{9}{64}a^2b^2 + \frac{5}{32}ab^3 + \frac{35}{128}b^4 \right), \\ h_{-6} &= \frac{1}{2} \left(\frac{63}{256}a^5 + \frac{35}{256}a^4b + \frac{15}{128}a^3b^2 + \frac{15}{128}a^2b^3 + \frac{35}{256}ab^4 + \frac{63}{256}b^5 \right), \\ h_{-7} &= \frac{1}{2} \left(\frac{\frac{231}{1024}a^6 + \frac{63}{512}a^5b + \frac{105}{1024}a^4b^2 + \frac{25}{256}a^3b^3 + \frac{105}{1024}a^2b^4 + \frac{63}{512}ab^5 + \frac{231}{1024}b^6 \right), \\ h_{-8} &= \frac{1}{2} \left(\frac{\frac{429}{2048}a^7 + \frac{231}{2048}a^6b + \frac{189}{2048}a^5b^2 + \frac{175}{2048}a^4b^3 + \frac{175}{2048}a^3b^4 + \frac{189}{2048}a^2b^5 + \frac{231}{2048}ab^6 + \frac{429}{2048}b^7 \right). \end{split}$$

For approximation to F(x), the first 8 coefficients of $F_{11}(z)$ in (2.34) are taken. Let $\varphi_{11}(z)$ and $\psi_{11}(z)$ denote stress potentials $\varphi_1(z)$ and $\psi_1(z)$ in a domain |z| < aaround the crack tip *a*. Substituting (2.28), (2.31) and (2.34)₂ into (2.32) for $\varphi_{11}(z)$ and (2.33) for $\psi_{11}(z)$ yields (see Appendix 3 for detailed derivation)

$$\varphi_{11}(z) = p_7 z^7 + p_6 z^6 + p_5 z^5 + p_4 z^4 + p_3 z^3 + p_2 z^2 + p_1 z^1 + p_0 \qquad (2.36)$$
$$+ p_{-1} z^{-1} + p_{-2} z^{-2} + p_{-3} z^{-3} + p_{-4} z^{-4},$$

and

$$\psi_{11}(z) = -6p_7 z^7 - 5p_6 z^6 - 4p_5 z^5 - (3p_4 + \overline{e}_4) z^4 - (2p_3 + \overline{e}_3) z^3$$
(2.37)

$$-(p_{2}+\overline{e}_{2})z^{2}-\overline{e}_{1}z^{1}+(p_{0}-\overline{e}_{0})+(2p_{-1}-\overline{e}_{-1})z^{-1}$$

+(3p_{-2}-\overline{e}_{-2})z^{-2}+(4p_{-3}-\overline{e}_{-3})z^{-3}+(5p_{-4}-\overline{e}_{-4})z^{-4},

where

..

$$\begin{split} p_{6} &= \left(g_{7}f_{-1} + g_{6}f_{0} + g_{5}f_{1} + g_{4}f_{2} + g_{3}f_{3} + g_{2}f_{4}\right), \\ p_{5} &= \left(g_{7}f_{-2} + g_{6}f_{-1} + g_{5}f_{0} + g_{4}f_{1} + g_{3}f_{2} + g_{2}f_{3} + g_{1}f_{4}\right), \\ p_{4} &= \left(g_{7}f_{-3} + g_{6}f_{-2} + g_{5}f_{-1} + g_{4}f_{0} + g_{3}f_{1} + g_{2}f_{2} + g_{1}f_{3} + g_{0}f_{4} + \frac{1}{2}e_{4}\right), \\ p_{3} &= \left(g_{6}f_{-3} + g_{5}f_{-2} + g_{4}f_{-1} + g_{3}f_{0} + g_{2}f_{1} + g_{1}f_{2} + g_{0}f_{3} + \frac{1}{2}e_{3}\right), \\ p_{2} &= \left(g_{5}f_{-3} + g_{4}f_{-2} + g_{3}f_{-1} + g_{2}f_{0} + g_{1}f_{1} + g_{0}f_{2} + \frac{1}{2}e_{2}\right), \\ p_{1} &= \left(g_{4}f_{-3} + g_{3}f_{-2} + g_{2}f_{-1} + g_{1}f_{0} + g_{0}f_{1} + \frac{1}{2}e_{1}\right), \\ p_{0} &= \left(g_{3}f_{-3} + g_{2}f_{-2} + g_{1}f_{-1} + g_{0}f_{0} + \frac{1}{2}e_{0}\right), \\ p_{-1} &= \left(g_{2}f_{-3} + g_{1}f_{-2} + g_{0}f_{-1} + \frac{1}{2}e_{-1}\right), \\ p_{-2} &= \left(g_{1}f_{-3} + g_{0}f_{-2} + \frac{1}{2}e_{-2}\right), \\ p_{-3} &= \left(g_{0}f_{-3} + \frac{1}{2}e_{-3}\right), \\ p_{-4} &= \left(\frac{1}{2}e_{-4}\right). \end{split}$$

In the similar way, the first 8 coefficients of $F_{12}(z)$ in (2.35) are taken for approximation to F(x) in a domain |z| > b. Now $\varphi_{12}(z)$ and $\psi_{12}(z)$ denoting stress potentials $\varphi_1(z)$ and $\psi_1(z)$ in a domain |z| > b around a crack tip b are considered. Substituting (2.28), (2.31) and (2.35)₂ into (2.32) for $\varphi_{12}(z)$ and (2.33) for $\psi_{12}(z)$ leads to the following form (see Appendix 3 for detailed derivation)

$$\varphi_{12}(z) = q_3 z^3 + q_2 z^2 + q_1 z^1 + q_0 + q_{-1} z^{-1} + q_{-2} z^{-2} + q_{-3} z^{-3} + q_{-4} z^{-4} + q_{-5} z^{-5} + q_{-6} z^{-6},$$
(2.38)

٠

and

$$\begin{aligned} & \vdots \\ \psi_{12}(z) &= 7q_{-6}z^{-6} + 6q_{-5}z^{-5} + (5q_{-4} - \overline{e}_{-4}) z^{-4} + (4q_{-3} - \overline{e}_{-3}) z^{-3} \\ & + (3q_{-2} - \overline{e}_{-2}) z^{-2} + (2q_{-1} - \overline{e}_{-1}) z^{-1} + (q_0 - \overline{e}_0) - \overline{e}_1 z^1 \\ & + (-\overline{e}_2 - q_2) z^2 + (-\overline{e}_3 - 2q_3) z^3 - \overline{e}_4 z^4, \end{aligned}$$

$$\end{aligned}$$

where

$$\begin{split} q_{3} &= \left(h_{-1}f_{4} + \frac{1}{2}e_{3}\right), \\ q_{2} &= \left(h_{-2}f_{4} + h_{-1}f_{3} + \frac{1}{2}e_{2}\right), \\ q_{1} &= \left(h_{-3}f_{4} + h_{-2}f_{3} + h_{-1}f_{2} + \frac{1}{2}e_{1}\right), \\ q_{0} &= \left(h_{-4}f_{4} + h_{-3}f_{3} + h_{-2}f_{2} + h_{-1}f_{1} + \frac{1}{2}e_{0}\right), \\ q_{-1} &= \left(h_{-5}f_{4} + h_{-4}f_{3} + h_{-3}f_{2} + h_{-2}f_{1} + h_{-1}f_{0} + \frac{1}{2}e_{-1}\right), \\ q_{-2} &= \left(h_{-6}f_{4} + h_{-5}f_{3} + h_{-4}f_{2} + h_{-3}f_{1} + h_{-2}f_{0} + h_{-1}f_{-1} + \frac{1}{2}e_{-2}\right), \\ q_{-3} &= \left(h_{-7}f_{4} + h_{-6}f_{3} + h_{-5}f_{2} + h_{-4}f_{1} + h_{-3}f_{0} + h_{-2}f_{-1} + h_{-1}f_{-2} + \frac{1}{2}e_{-3}\right), \\ q_{-4} &= \left(\begin{array}{c}h_{-8}f_{4} + h_{-7}f_{3} + h_{-6}f_{2} + h_{-5}f_{1} + h_{-4}f_{0} + h_{-3}f_{-1} + h_{-2}f_{-2} \\ &+h_{-1}f_{-3} + \frac{1}{2}e_{-4}\end{array}\right), \end{split}$$

$$q_{-5} = \begin{pmatrix} h_{-8}f_3 + h_{-7}f_2 + h_{-6}f_1 + h_{-5}f_0 + h_{-4}f_{-1} + h_{-3}f_{-2} + h_{-2}f_{-3} \\ + h_{-1}f_{-4} + \frac{1}{2}e_{-5} \end{pmatrix},$$

$$q_{-6} = \begin{pmatrix} h_{-8}f_2 + h_{-7}f_1 + h_{-6}f_0 + h_{-5}f_{-1} + h_{-4}f_{-2} + h_{-3}f_{-3} + h_{-2}f_{-4} + \frac{1}{2}e_{-6} \end{pmatrix}.$$

Finally, all the stress potentials $\varphi_0(z)$ and $\psi_0(z)$ in S_0 , $\varphi_{11}(z)$ and $\psi_{11}(z)$ in a region $(R_0 \leq |z| < a) \subset S_1$, $\varphi_{12}(z)$ and $\psi_{12}(z)$ in a region $(b < |z| \leq R_1) \subset S_1$ and $\varphi_2(z)$ and $\psi_2(z)$ in S_2 are expressed solely in terms of finite series coefficients. Of the coefficients, a_k , b_k , c_k , d_k , e_k and f_k are unknown coefficients while g_k and h_k are known coefficients determined in terms of crack position a and b. In addition, p_k are functions of e_k , f_k and g_k , and q_k are determined by e_k , f_k and h_k . Consequently, this boundary problem is reduced to determining the unknown complex coefficients such that the boundary condition (2.7), (2.12), (2.13) and (2.14) along $\partial \Gamma_0$ and $\partial \Gamma_1$ hold.

:



Figure 2.1 Three-phase circular inclusion model with a radial crack

CHAPTER 3 Algebraic Equations

3.1 Introduction

In this chapter, the eight stress potentials expressed in terms of finite series coefficients in Chapter 2 are substituted into the four boundary conditions (2.7), (2.12), (2.13) and (2.14). By comparing coefficients of powers of z, a set of 38 coupled linear algebraic equations are derived to determine 38 unknown coefficients of the stress potentials. In particular, uniaxial tensile loads are considered as the prescribed remote load normal to crack surface since Mode I stress intensity factor is the most critical factor leading composite material failure. Since a pre-existing crack is located and moving on the real axis, namely, the x-axis all the coefficients that we consider are assumed on the real axis as well. Henceforth, all the complex coefficients of stress potentials and their respective conjugates are the same. The imperfect interfaces are assumed homogeneous, which means the spring-factor type interface parameters do not vary along the interfaces.

3.2 General Algebraic Equations on Inner Interface Γ_0

There are two boundary conditions along the inner interface $\partial \Gamma_0$: one is the traction continuity and the other is the displacement jump across the interface. First, let's consider the traction continuity condition. Substituting the stress potentials $\varphi_0(z)$, $\psi_0(z)$, $\varphi_{11}(z)$ and $\psi_{11}(z)$ into (2.7) gives the following expression (See Appendix 4 for the details)

$$\frac{42}{R_0^2} p_7 z^8 + \frac{30}{R_0^2} p_6 z^7 + \left(-35 p_7 + \frac{20}{R_0^2} p_5\right) z^6 \tag{3.1}$$

$$\begin{split} &+ \left(\frac{1}{R_0^2} \left(12p_4 + 4e_4\right) - 24p_6 - \frac{4p_{-4}}{R_0^{10}}\right) z^5 + \left(\frac{1}{R_0^2} \left(6p_3 + 3e_3\right) - 15p_5 - \frac{3p_{-3}}{R_0^8}\right) z^4 \\ &+ \left(\frac{1}{R_0^2} \left(2p_2 + 2e_2\right) - 8p_4 - \frac{2p_{-2}}{R_0^6}\right) z^3 + \left(\frac{1}{R_0^2} e_1 - 3p_3 - \frac{p_{-1}}{R_0^4}\right) z^2 + (0) z \\ &+ \left(\frac{1}{R_0^2} \left(2p_{-1} - e_{-1}\right) + 2p_1\right) \\ &+ \left(\frac{1}{R_0^2} \left(12p_{-3} - 3e_{-3}\right) - 3p_{-1} + 3R_0^4 p_3\right) z^{-2} \\ &+ \left(\frac{1}{R_0^2} \left(20p_{-4} - 4e_{-4}\right) - 8p_{-2} + 4R_0^6 p_4\right) z^{-3} + \left(5R_0^8 p_5 - 15p_{-3}\right) z^{-4} \\ &+ \left(6R_0^{10} p_6 - 24p_{-4}\right) z^{-5} + 7p_7 R_0^{12} z^{-6} \\ &= -\frac{7}{R_0^2} b_7 z^8 - \frac{6}{R_0^2} b_6 z^7 + \left(-35a_7 - \frac{5}{R_0^2} b_5\right) z^6 + \left(-24a_6 - \frac{4}{R_0^2} b_4\right) z^5 \\ &+ \left(-15a_5 - \frac{3}{R_0^2} b_3\right) z^4 + \left(-8a_4 - \frac{2}{R_0^2} b_2\right) z^3 + \left(-3a_3 - \frac{1}{R_0^2} b_1\right) z^2 + (0) z \\ &+ 2a_1 + 2R_0^2 a_2 z^{-1} + 3R_0^4 a_3 z^{-2} + 4R_0^6 a_4 z^{-3} + 5R_0^8 a_5 z^{-4} + 6R_0^{10} a_6 z^{-5} + 7R_0^{12} a_7 z^{-6}. \end{split}$$

Also, deploying the stress potentials into (2.12) leads to the expression (See Appendix 4 for the details)

$$(m_{0} - n_{0})(\kappa_{1} + 1)R_{0}p_{7}z^{6} + \left((m_{0} - n_{0})(\kappa_{1} + 1)R_{0}p_{6} + (m_{0} + n_{0})(\kappa_{1} + 1)\frac{1}{R_{0}^{9}}p_{-4}\right)z^{5} + \left((m_{0} - n_{0})(\kappa_{1} + 1)R_{0}p_{5} + (m_{0} + n_{0})(\kappa_{1} + 1)\frac{1}{R_{0}^{7}}p_{-3}\right)z^{4}$$

$$+ \left((m_{0} - n_{0})(\kappa_{1} + 1)R_{0}p_{4} + (m_{0} + n_{0})(\kappa_{1} + 1)\frac{1}{R_{0}^{5}}p_{-2}\right)z^{3} + \left((m_{0} - n_{0})(\kappa_{1} + 1)R_{0}p_{3} + (m_{0} + n_{0})(\kappa_{1} + 1)\frac{1}{R_{0}^{9}}p_{-1}\right)z^{2} + \left((m_{0} - n_{0})(\kappa_{1} + 1)R_{0}p_{2} + (m_{0} + n_{0})(\kappa_{1} + 1)\frac{1}{R_{0}}p_{0}\right)z + 2m_{0}(\kappa_{1} + 1)R_{0}p_{1} + \left[(m_{0} - n_{0})(\kappa_{1} + 1)R_{0}p_{-1} + (m_{0} + n_{0})(\kappa_{1} + 1)R_{0}p_{3}\right]z^{-2} + \left[(m_{0} + n_{0})(\kappa_{1} + 1)R_{0}p_{4} + (m_{0} - n_{0})(\kappa_{1} + 1)R_{0}p_{-2}\right]z^{-3}$$

.

••

+
$$[(m_0 + n_0)(\kappa_1 + 1)R_0^9 p_5 + (m_0 - n_0)(\kappa_1 + 1)R_0 p_{-3}]z^{-4}$$

+ $[(m_0 - n_0)(\kappa_1 + 1)R_0 p_{-4} + (m_0 + n_0)(\kappa_1 + 1)R_0^{11} p_6]z^{-5}$
+ $(m_0 + n_0)(\kappa_1 + 1)R_0^{13} p_7 z^{-6}$

·

$$\begin{split} &= \left(-4\mu_{1}\frac{7}{R_{0}^{2}}b_{7}+(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})\frac{1}{R_{0}}b_{7}\right)z^{8} \\ &+ \left(-4\mu_{1}\frac{6}{R_{0}^{2}}b_{6}+(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})\frac{1}{R_{0}}b_{6}\right)z^{7} \\ &+ \left(\begin{array}{c} 4\mu_{1}\left(-35a_{7}-\frac{5}{R_{0}^{2}}b_{5}\right)+(m_{0}-n_{0})(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}a_{7} \\ +7(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})R_{0}a_{7}+(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})\frac{1}{R_{0}}b_{5} \end{array}\right)z^{6} \\ &+ \left(\begin{array}{c} 4\mu_{1}\left(-24a_{6}-\frac{4}{R_{0}^{2}}b_{4}\right)+(m_{0}-n_{0})(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}a_{6} \\ +6(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})R_{0}a_{6}+(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})\frac{1}{R_{0}}b_{4} \end{array}\right)z^{5} \\ &+ \left(\begin{array}{c} 4\mu_{1}\left(-15a_{5}-\frac{3}{R_{0}^{2}}b_{3}\right)+(m_{0}-n_{0})(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}a_{5} \\ +5(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})R_{0}a_{5}+(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})\frac{1}{R_{0}}b_{3} \end{array}\right)z^{4} \\ &+ \left(\begin{array}{c} 4\mu_{1}\left(-8a_{4}-\frac{2}{R_{0}^{2}}b_{2}\right)+(m_{0}-n_{0})(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}a_{4} \\ +4(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})R_{0}a_{4}+(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})\frac{1}{R_{0}}b_{2} \end{array}\right)z^{3} \\ &+ \left(\begin{array}{c} 4\mu_{1}\left(-3a_{3}-\frac{1}{R_{0}^{2}}b_{1}\right)+(m_{0}-n_{0})(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}a_{3}+\frac{2\mu_{1}(m_{0}+n_{0})(e_{2}-ie_{3})}{R_{0}}}{1-R_{0}}\right)z^{4} \\ &+ \left(\begin{array}{c} (m_{0}-n_{0})(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}a_{2}+(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})\frac{1}{R_{0}}}a_{0} \\ +3(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})R_{0}a_{3}+(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})\frac{1}{R_{0}}}a_{2} \end{array}\right)z^{4} \\ &+ \left(\begin{array}{c} 8\mu_{1}R_{0}^{2}a_{2}+(m_{0}-n_{0})(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}a_{1}+R_{0}(m_{0}-n_{0})(1-\frac{\mu_{1}}{\mu_{0}})R_{0}a_{1}} \\ +(m_{0}+n_{0})(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}a_{1}+4m_{0}R_{0}\mu_{1}\varepsilon_{1}+(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}}})R_{0}a_{2} \\ +(m_{0}-n_{0})(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}}})R_{0}a_{0}+2(m_{0}-n_{0})(1-\frac{\mu_{1}}{\mu_{0}}})R_{0}^{3}a_{2} \end{array}\right)z^{-1} \end{split}$$

.

÷

.

$$+ \begin{pmatrix} 12\mu_{1}R_{0}^{4}a_{3} + 3(m_{0} - n_{0})(1 - \frac{\mu_{1}}{\mu_{0}})R_{0}^{5}a_{3} + (m_{0} - n_{0})(1 - \frac{\mu_{1}}{\mu_{0}})R_{0}^{3}b_{1} \\ + (m_{0} + n_{0})(1 + \kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}^{5}a_{3} + 2\mu_{1}(m_{0} - n_{0})(\varepsilon_{2} + i\varepsilon_{3})R_{0}^{3} \end{pmatrix} z^{-2} \\ + \begin{pmatrix} 16\mu_{1}R_{0}^{6}a_{4} + 4(m_{0} - n_{0})(1 - \frac{\mu_{1}}{\mu_{0}})R_{0}^{7}a_{4} + (m_{0} - n_{0})(1 - \frac{\mu_{1}}{\mu_{0}})R_{0}^{5}b_{2} \\ + (m_{0} + n_{0})(1 + \kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}^{7}a_{4} \end{pmatrix} z^{-3} \\ + \begin{pmatrix} 20\mu_{1}R_{0}^{8}a_{5} + 5(m_{0} - n_{0})(1 - \frac{\mu_{1}}{\mu_{0}})R_{0}^{9}a_{5} + (m_{0} - n_{0})(1 - \frac{\mu_{1}}{\mu_{0}})R_{0}^{7}b_{3} \\ + (m_{0} + n_{0})(1 + \kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}^{9}a_{5} \end{pmatrix} z^{-4} \\ + \begin{pmatrix} 24\mu_{1}R_{0}^{10}a_{6} + 6(m_{0} - n_{0})(1 - \frac{\mu_{1}}{\mu_{0}})R_{0}^{11}a_{6} + (m_{0} - n_{0})(1 - \frac{\mu_{1}}{\mu_{0}})R_{0}^{9}b_{4} \\ + (m_{0} + n_{0})(1 + \kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}^{11}a_{6} \end{pmatrix} z^{-5} \\ + \begin{pmatrix} 28\mu_{1}R_{0}^{12}a_{7} + 7(m_{0} - n_{0})(1 - \frac{\mu_{1}}{\mu_{0}})R_{0}^{13}a_{7} + (m_{0} - n_{0})(1 - \frac{\mu_{1}}{\mu_{0}})R_{0}^{11}b_{5} \\ + (m_{0} + n_{0})(1 + \kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}^{13}a_{7} + (m_{0} - n_{0})(1 - \frac{\mu_{1}}{\mu_{0}})R_{0}^{15}b_{7}z^{-8}. \end{pmatrix} z^{-6} \\ \end{pmatrix} z^{-6}$$

By comparing coefficients of the same power of z, from z^5 to z^{-4} in (3.1), we get the following set of coupled linear algebraic equations

$$\left(\frac{12}{R_0^2}p_4 + \frac{4}{R_0^2}e_4 - 24p_6 - \frac{4p_{-4}}{R_0^{10}}\right)z^5 = \left(-24a_6 - \frac{4}{R_0^2}b_4\right)z^5,\tag{3.3}$$

$$\left(\frac{6}{R_0^2}p_3 + \frac{3}{R_0^2}e_3 - 15p_5 - \frac{3p_{-3}}{R_0^8}\right)z^4 = \left(-15a_5 - \frac{3}{R_0^2}b_3\right)z^4,\tag{3.4}$$

$$\left(\frac{2}{R_0^2}p_2 + \frac{2}{R_0^2}e_2 - 8p_4 - \frac{2p_{-2}}{R_0^6}\right)z^3 = \left(-8a_4 - \frac{2}{R_0^2}b_2\right)z^3,\tag{3.5}$$

$$\left(\frac{1}{R_0^2}e_1 - 3p_3 - \frac{p_{-1}}{R_0^4}\right)z^2 = \left(-3a_3 - \frac{1}{R_0^2}b_1\right)z^2,\tag{3.6}$$

$$\left(\frac{2}{R_0^2}p_{-1} - \frac{1}{R_0^2}e_{-1} + 2p_1\right) = 2a_1,$$
(3.7)

$$\left(2R_0^2 p_2 + \frac{6}{R_0^2} p_{-2} - \frac{2}{R_0^2} e_{-2}\right) z^{-1} = 2R_0^2 a_2 z^{-1},$$
(3.8)

$$\left(\frac{12}{R_0^2}p_{-3} - \frac{3}{R_0^2}e_{-3} - 3p_{-1} + 3R_0^4p_3\right)z^{-2} = 3R_0^4a_3z^{-2},\tag{3.9}$$

$$\left(\frac{20}{R_0^2}p_{-4} - \frac{4}{R_0^2}e_{-4} - 8p_{-2} + 4R_0^6p_4\right)z^{-3} = 4R_0^6a_4z^{-3},$$
(3.10)

$$\left(5R_0^8 p_5 - 15p_{-3}\right) z^{-4} = 5R_0^8 a_5 z^{-4}. \tag{3.11}$$

It is noted that the coefficients of z^1 are 0 in both sides $(0 * z^1 = 0 * z^1)$.

Similarly, comparing coefficients of the same power of z, from z^5 to z^{-4} (3.2) gives another set of coupled linear algebraic equations as following

$$\begin{pmatrix} (m_0 - n_0)(\kappa_1 + 1)R_0p_6 \\ +(m_0 + n_0)(\kappa_1 + 1)\frac{1}{R_0^5}p_{-4} \end{pmatrix} z^5 = \begin{pmatrix} 4\mu_1 \left(-24a_6 - \frac{4}{R_0^2}b_4\right) \\ +(m_0 - n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})R_0a_6 \\ +6(m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})R_0a_6 \\ +(m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})\frac{1}{R_0}b_4 \end{pmatrix} z^5, (3.12)$$

$$\begin{pmatrix} (m_0 - n_0)(\kappa_1 + 1)R_0p_5 \\ +(m_0 + n_0)(\kappa_1 + 1)\frac{1}{R_0'}p_{-3} \end{pmatrix} z^4 = \begin{pmatrix} 4\mu_1 \left(-15a_5 - \frac{3}{R_0^2}b_3\right) \\ +(m_0 - n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})R_0a_5 \\ +5(m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})R_0a_5 \\ +(m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})\frac{1}{R_0}b_3 \end{pmatrix} z^4, \quad (3.13)$$

$$\begin{pmatrix} (m_0 - n_0)(\kappa_1 + 1)R_0p_4 \\ +(m_0 + n_0)(\kappa_1 + 1)\frac{1}{R_0^5}p_{-2} \end{pmatrix} z^3 = \begin{pmatrix} 4\mu_1 \left(-3u_4 - \frac{1}{R_0^2}b_2\right) \\ +(m_0 - n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})R_0a_4 \\ +4(m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})R_0a_4 \\ +(m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})\frac{1}{R_0}b_2 \end{pmatrix} z^3, \quad (3.14)$$

$$\begin{pmatrix} (m_0 - n_0)(\kappa_1 + 1)R_0p_3 \\ +(m_0 + n_0)(\kappa_1 + 1)\frac{1}{R_0^3}p_{-1} \end{pmatrix} z^2 = \begin{pmatrix} 4\mu_1 \left(-3a_3 - \frac{1}{R_0^2}b_1\right) + \frac{2\mu_1(m_0 + n_0)(\varepsilon_2 - i\varepsilon_3)}{R_0} \\ +(m_0 - n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})R_0a_3 \\ +3(m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})R_0a_3 \\ +(m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})\frac{1}{R_0}b_1 \end{pmatrix} z^2,$$

$$(3.15)$$

.

.

•

$$\begin{pmatrix} (m_0 - n_0)(\kappa_1 + 1)R_0p_2 \\ +(m_0 + n_0)(\kappa_1 + 1)\frac{1}{R_0}p_0 \end{pmatrix} z = \begin{pmatrix} (m_0 - n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})R_0a_2 \\ +(m_0 + n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})\frac{1}{R_0}a_0 \\ +(m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})\frac{1}{R_0}b_0 \\ +2(m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})R_0a_2 \end{pmatrix} z, \quad (3.16)$$

$$2m_{0}(\kappa_{1}+1)R_{0}p_{1} = \begin{pmatrix} 8\mu_{1}a_{1} + (m_{0}-n_{0})(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}a_{1} \\ +R_{0}(m_{0}-n_{0})(1-\frac{\mu_{1}}{\mu_{0}})a_{1} \\ +(m_{0}+n_{0})(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}a_{1} \\ +4m_{0}R_{0}\mu_{1}\varepsilon_{1} + (m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})R_{0}a_{1} \end{pmatrix}, \quad (3.17)$$

*

$$\begin{pmatrix} (m_0 - n_0)(\kappa_1 + 1)R_0p_0 \\ +(m_0 + n_0)(\kappa_1 + 1)R_0^3p_2 \end{pmatrix} z^{-1} = \begin{pmatrix} 8\mu_1 R_0^2 a_2 + (m_0 - n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})R_0 a_0 \\ +2(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^3 a_2 \\ +(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0 b_0 \\ +(m_0 + n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})R_0^3 a_2 \end{pmatrix} z^{-1},$$
(3.18)

÷

.

44

÷

4

÷

$$\begin{pmatrix} (m_0 - n_0)(\kappa_1 + 1)R_0p_{-1} \\ +(m_0 + n_0)(\kappa_1 + 1)R_0^5p_3 \end{pmatrix} z^{-2} = \begin{pmatrix} 12\mu_1 R_0^4 a_3 + 3(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^5 a_3 \\ +(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^3 b_1 \\ +(m_0 + n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})R_0^5 a_3 \\ +2\mu_1(m_0 - n_0)(\varepsilon_2 + i\varepsilon_3)R_0^3 \end{pmatrix} z^{-2},$$
(3.19)

$$\begin{pmatrix} (m_0 + n_0)(\kappa_1 + 1)R_0^7 p_4 \\ +(m_0 - n_0)(\kappa_1 + 1)R_0 p_{-2} \end{pmatrix} z^{-3} = \begin{pmatrix} 16\mu_1 R_0^6 a_4 + 4(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^7 a_4 \\ +(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^5 b_2 \\ +(m_0 + n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})R_0^7 a_4 \end{pmatrix} z^{-3},$$
(3.20)

$$\begin{pmatrix} (m_0 + n_0)(\kappa_1 + 1)R_0^9 p_5 \\ +(m_0 - n_0)(\kappa_1 + 1)R_0 p_{-3} \end{pmatrix} z^{-4} = \begin{pmatrix} 20\mu_1 R_0^8 a_5 + 5(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^9 a_5 \\ +(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^7 b_3 \\ +(m_0 + n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})R_0^9 a_5 \end{pmatrix} z^{-4}.$$
(3.21)

3.3 General Algebraic Equations on Outer Interface Γ_1

Along the outer interface Γ_1 , there are also two boundary conditions as addressed in the previous section. The traction continuity condition is considered first. Substituting the stress potentials $\varphi_2(z)$, $\psi_2(z)$, $\varphi_{12}(z)$ and $\psi_{12}(z)$ into (2.13) we have the following expression (See Appendix 4 for the details)

•

$$-\frac{8}{R_1^{18}}c_8z^9 - \frac{7}{R_1^{16}}c_7z^8 - \frac{6}{R_1^{14}}c_6z^7 - \frac{5}{R_1^{12}}c_5z^6 - \frac{4}{R_1^{10}}c_4z^5 - \frac{3}{R_1^8}c_3z^4 \qquad (3.22)$$

45

,

$$\begin{split} &-\frac{2}{R_1^6}c_2z^3+\left(-\frac{1}{R_1^4}c_1-\frac{1}{R_1^2}B\right)z^2+\left(2A+\frac{1}{R_1^2}d_1\right)+\frac{2}{R_1^2}d_2z^{-1}\\ &+\left(-3c_1+\frac{3}{R_1^2}d_3\right)z^{-2}+\left(-8c_2+\frac{4}{R_1^2}d_4\right)z^{-3}+\left(-15c_3+\frac{5}{R_1^2}d_5\right)z^{-4}\\ &+\left(-24c_4+\frac{6}{R_1^2}d_6\right)z^{-5}+\left(-35c_5+\frac{7}{R_1^2}d_7\right)z^{-6}+\left(-48c_6+\frac{8}{R_1^2}d_8\right)z^{-7}\\ &-63c_7z^{-8}-80c_8z^{-9}\\ =&\left(-6q_{-6}\frac{1}{R_1^{14}}z^7\right)+\left(-5q_{-5}\frac{1}{R_1^{12}}z^6\right)+\left(4\overline{e}_4\frac{1}{R_1^2}-4q_{-4}\frac{1}{R_1^{10}}\right)z^5\\ &+\left(-3\left(-\overline{e}_3-2q_3\right)\frac{1}{R_1^2}-3q_{-3}\frac{1}{R_1^8}\right)z^4+\left(-2\left(-\overline{e}_2-q_2\right)\frac{1}{R_1^2}-2q_{-2}\frac{1}{R_1^6}\right)z^3\\ &+\left(\overline{e}_1\frac{1}{R_1^2}-q_{-1}\frac{1}{R_1^4}+3q_3-6q_3\right)z^2+\left(2q_2-2q_2\right)z\\ &+\left(q_1+q_1+\left(2q_{-1}-\overline{e}_{-1}\right)\frac{1}{R_1^2}\right)+\left(2q_2R_1^2+2\left(3q_{-2}-\overline{e}_{-2}\right)\frac{1}{R_1^2}\right)z^{-1}\\ &+\left(3q_3R_1^4-q_{-1}-2q_{-1}+3\left(4q_{-3}-\overline{e}_{-3}\right)\frac{1}{R_1^2}\right)z^{-2}\\ &+\left(-3q_{-3}+30q_{-5}\frac{1}{R_1^2}-12q_{-3}\right)z^{-4}+\left(-4q_{-4}+42q_{-6}\frac{1}{R_1^2}-20q_{-4}\right)z^{-5}\\ &+\left(-5q_{-5}-30q_{-5}\right)z^{-6}+\left(-6q_{-6}-42q_{-6}\right)z^{-7}. \end{split}$$

,

.

.

Finally, with $\varphi_2(z)$, $\psi_2(z)$, $\varphi_{12}(z)$ and $\psi_{12}(z)$ substituted into (2.14), the last boundary condition (2.14) yields the following expression (See Appendix 4 for the details)

$$(m_{1} + n_{1}) (k_{1} + 1) q_{-6} \frac{1}{R_{1}^{13}} z^{7} + (m_{1} + n_{1}) (k_{1} + 1) q_{-5} \frac{1}{R_{1}^{11}} z^{6}$$

$$+ (m_{1} + n_{1}) (k_{1} + 1) q_{-4} \frac{1}{R_{1}^{9}} z^{5} + (m_{1} + n_{1}) (k_{1} + 1) q_{-3} \frac{1}{R_{1}^{7}} z^{4}$$

$$+ (m_{1} + n_{1}) (k_{1} + 1) q_{-2} \frac{1}{R_{1}^{5}} z^{3}$$

$$+ \left((m_{1} - n_{1}) (k_{1} + 1) R_{1} q_{3} + (m_{1} + n_{1}) (k_{1} + 1) q_{-1} \frac{1}{R_{1}^{3}} \right) z^{2}$$

$$(3.23)$$

.

••

$$\begin{split} &+ \left((m_1 - n_1) \left(k_1 + 1\right) R_1 q_2 + (m_1 + n_1) \left(k_1 + 1\right) \frac{1}{R_1} q_0 \right) z + 2m_1 \left(k_1 + 1\right) R_1 q_1 \\ &+ \left((m_1 - n_1) \left(k_1 + 1\right) R_1 q_0 + (m_1 + n_1) \left(k_1 + 1\right) R_1^3 q_2 \right) z^{-1} \\ &+ \left((m_1 - n_1) \left(k_1 + 1\right) R_1 q_{-1} + (m_1 + n_1) \left(k_1 + 1\right) R_1^5 q_3 \right) z^{-2} \\ &+ (m_1 - n_1) \left(k_1 + 1\right) R_1 q_{-2} z^{-3} + (m_1 - n_1) \left(k_1 + 1\right) R_1 q_{-3} z^{-4} \\ &+ (m_1 - n_1) \left(k_1 + 1\right) R_1 q_{-4} z^{-5} + (m_1 - n_1) \left(k_1 + 1\right) R_1 q_{-5} z^{-6} \\ &+ (m_1 - n_1) \left(k_1 + 1\right) R_1 q_{-6} z^{-7} \\ &= \left(4\mu_1 \frac{4}{R_1^{10}} c_4 - (m_1 - n_1) \left(1 - \frac{\mu_1}{\mu_2}\right) \left(4\frac{1}{R_1^{12}} - \frac{3}{R_1^{1}} c_3 \right) \\ &+ \left(4\mu_1 \frac{3}{R_1^{10}} c_3 + (m_1 - n_1) \left(1 - \frac{\mu_1}{\mu_2}\right) \left(4\frac{1}{R_1^{12}} - \frac{3}{R_1^{1}} c_3 \right) \\ &+ \left(4\mu_1 \frac{3}{R_1^{12}} c_2 + (m_1 - n_1) \left(1 - \frac{\mu_1}{\mu_2}\right) \left(4\frac{1}{R_1^{12}} - \frac{2}{R_1^{12}} c_2 \right) \\ &+ \left(4\mu_1 \left(\frac{1}{R_1^{12}} c_1 + \frac{1}{R_1^{2}} B \right) + (m_1 - n_1) \left(1 - \frac{\mu_1}{\mu_2} \right) \left(4\frac{1}{R_1^{12}} - \frac{1}{R_1^{12}} c_1 \right) \\ &+ \left(m_1 - n_1 \right) \left(1 - \frac{\mu_1}{\mu_2} \right) d_2 \frac{1}{R_1^{3}} z \\ &- 4\mu_1 \left(2A + \frac{1}{R_1^{2}} d_1 \right) + (m_1 + n_1) \left(1 + \frac{\mu_1}{\mu_2} k_2 \right) AR_1 \\ &+ \left(m_1 - n_1 \right) \left(1 - \frac{\mu_1}{\mu_2} \right) \left(AR_1 + d_1 \frac{1}{R_1} \right) + (m_1 - n_1) \left(1 + \frac{\mu_1}{\mu_2} k_2 \right) AR_1 \\ &+ \left(-4\mu_1 \frac{2}{R_1^{2}} d_2 + (m_1 + n_1) \left(1 - \frac{\mu_1}{\mu_2} \right) d_2 \frac{1}{R_1} \right) z^{-1} \\ &+ \left(-4\mu_1 \left(-3c_1 + \frac{3}{R_1^{2}} d_3 \right) + (m_1 - n_1) \left(1 - \frac{\mu_1}{\mu_2} \right) BR_1^{3} \\ &+ \left(m_1 + n_1 \right) \left(1 - \frac{\mu_1}{\mu_2} \right) \left(d_3 \frac{1}{R_1} - R_1c_1 \right) \right) z^{-2} \end{split}$$

.

.

••

,

$$+ \left(\begin{array}{c} -4\mu_{1} \left(-8c_{2} + \frac{4}{R_{1}^{2}}d_{4} \right) + (m_{1} - n_{1}) \left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2} \right) R_{1}c_{2} \\ + (m_{1} + n_{1}) \left(1 - \frac{\mu_{1}}{\mu_{2}} \right) \left(d_{4}\frac{1}{R_{1}} - 2R_{1}c_{2} \right) \end{array} \right) z^{-3} \\ + \left(\begin{array}{c} -4\mu_{1} \left(-15c_{3} + \frac{5}{R_{1}^{2}}d_{5} \right) + (m_{1} - n_{1}) \left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2} \right) R_{1}c_{3} \\ + (m_{1} + n_{1}) \left(1 - \frac{\mu_{1}}{\mu_{2}} \right) \left(d_{5}\frac{1}{R_{1}} - 3R_{1}c_{3} \right) \end{array} \right) z^{-4} \\ + \left(\begin{array}{c} -4\mu_{1} \left(-24c_{4} + \frac{6}{R_{1}^{2}}d_{6} \right) + (m_{1} - n_{1}) \left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2} \right) R_{1}c_{4} \\ - (m_{1} + n_{1}) \left(1 - \frac{\mu_{1}}{\mu_{2}} \right) 4R_{1}c_{4} \end{array} \right) z^{-5}.$$

By comparing coefficients of the same power of z, from z^5 to z^{-4} in (3.22), we get the following set of coupled linear algebraic equations

$$-\frac{4}{R_1^{10}}c_4 z^5 = \left(4e_4 \frac{1}{R_1^2} - 4q_{-4} \frac{1}{R_1^{10}}\right) z^5,\tag{3.24}$$

$$-\frac{3}{R_1^8}c_3z^4 = \left(+3e_3\frac{1}{R_1^2} + 6q_3\frac{1}{R_1^2} - 3q_{-3}\frac{1}{R_1^8}\right)z^4,\tag{3.25}$$

$$-\frac{2}{R_1^6}c_2z^3 = \left(+2e_2\frac{1}{R_1^2} + 2q_2\frac{1}{R_1^2} - 2q_{-2}\frac{1}{R_1^6}\right)z^3,\tag{3.26}$$

$$\left(-\frac{1}{R_1^4}c_1 - \frac{1}{R_1^2}B\right)z^2 = \left(e_1\frac{1}{R_1^2} - q_{-1}\frac{1}{R_1^4} - 3q_3\right)z^2,$$
(3.27)

$$\left(2A + \frac{1}{R_1^2}d_1\right) = \left(+2q_1 + 2q_{-1}\frac{1}{R_1^2} - e_{-1}\frac{1}{R_1^2}\right),\tag{3.28}$$

$$\frac{2}{R_1^2} d_2 z^{-1} = \left(2q_2 R_1^2 + 6q_{-2} \frac{1}{R_1^2} - 2e_{-2} \frac{1}{R_1^2} \right) z^{-1}, \tag{3.29}$$

$$\left(-3c_1 + \frac{3}{R_1^2}d_3\right)z^{-2} = \left(3q_3R_1^4 - 3q_{-1} + 12q_{-3}\frac{1}{R_1^2} - 3e_{-3}\frac{1}{R_1^2}\right),\tag{3.30}$$

$$\left(-8c_2 + \frac{4}{R_1^2}d_4\right)z^{-3} = \left(4\left(5q_{-4} - e_{-4}\right)\frac{1}{R_1^2} - 8q_{-2}\right)z^{-3},\tag{3.31}$$

$$\left(-15c_3 + \frac{5}{R_1^2}d_5\right)z^{-4} = \left(+30q_{-5}\frac{1}{R_1^2} - 15q_{-3}\right)z^{-4}.$$
(3.32)

•*

Similar to the case of traction continuation along Γ_0 , the coefficients of z^1 are 0 in both sides $(0 * z^1 = 0 * z^1)$.

Finally, the last set of coupled linear algebraic equations is obtained by comparing coefficients of the same power of z, from z^5 to z^{-4} in (3.23)

$$(m_{1}+n_{1})(k_{1}+1)q_{-4}\frac{1}{R_{1}^{9}}z^{5} = \begin{pmatrix} 4\mu_{1}\frac{4}{R_{1}^{10}}c_{4} - (m_{1}-n_{1})\left(1-\frac{\mu_{1}}{\mu_{2}}\right)\frac{4}{R_{1}^{9}}c_{4} \\ + (m_{1}+n_{1})\left(1+\frac{\mu_{1}}{\mu_{2}}k_{2}\right)c_{4}\frac{1}{R_{1}^{9}} \end{pmatrix} z^{5}, (3.33)$$

$$(m_{1}+n_{1})(k_{1}+1)q_{-3}\frac{1}{R_{1}^{7}}z^{4} = \begin{pmatrix} 4\mu_{1}\frac{3}{R_{1}^{8}}c_{3} + (m_{1}-n_{1})\left(1-\frac{\mu_{1}}{\mu_{2}}\right)\left(d_{5}\frac{1}{R_{1}^{9}}-\frac{3}{R_{1}^{7}}c_{3}\right) \\ + (m_{1}+n_{1})\left(1+\frac{\mu_{1}}{\mu_{2}}k_{2}\right)c_{3}\frac{1}{R_{1}^{7}} \end{pmatrix} z^{4};$$
(3.34)

$$(m_{1}+n_{1})(k_{1}+1)q_{-2}\frac{1}{R_{1}^{5}}z^{3} = \begin{pmatrix} 4\mu_{1}\frac{2}{R_{1}^{6}}c_{2} + (m_{1}-n_{1})\left(1-\frac{\mu_{1}}{\mu_{2}}\right)\left(d_{4}\frac{1}{R_{1}^{7}}-\frac{2}{R_{1}^{5}}c_{2}\right)\\ + (m_{1}+n_{1})\left(1+\frac{\mu_{1}}{\mu_{2}}k_{2}\right)c_{2}\frac{1}{R_{1}^{5}} \end{pmatrix} z^{3},$$
(3.35)

$$\begin{pmatrix} (m_{1}-n_{1})(k_{1}+1)R_{1}q_{3} \\ +(m_{1}+n_{1})(k_{1}+1)q_{-1}\frac{1}{R_{1}^{3}} \end{pmatrix} z^{2} = \begin{pmatrix} 4\mu_{1}\left(\frac{1}{R_{1}^{4}}c_{1}+\frac{1}{R_{1}^{2}}B\right) \\ +(m_{1}-n_{1})\left(1-\frac{\mu_{1}}{\mu_{2}}\right)\left(d_{3}\frac{1}{R_{1}^{5}}-\frac{1}{R_{1}^{3}}c_{1}\right) \\ +(m_{1}+n_{1})\left(1+\frac{\mu_{1}}{\mu_{2}}k_{2}\right)c_{1}\frac{1}{R_{1}^{3}} \\ +(m_{1}+n_{1})\left(1-\frac{\mu_{1}}{\mu_{2}}\right)B\frac{1}{R_{1}} \end{pmatrix} z^{2},$$

$$(3.36)$$

$$\left(\begin{array}{c} \left(b \overline{b} \frac{k}{r \overline{A}} + c^{3} 8 - \right) t u \overline{b} - u \overline{b}$$

۰.

$$\begin{pmatrix} & (3.40) \\ & (3.40) \\ & (1.40$$

(65.8)

۰

•

$$\left(\begin{array}{c} \left(-\frac{1}{2} u^{2} - \frac{1}{2} u^{2} \right) \left(-\frac{1}{2} u^{2} - \frac{1}{2} u^{2} \right) \right) = \left(-\frac{1}{2} u^{2} - \frac{1}{2} u^{2} + \frac{1}{2} u^{2} + \frac{1}{2} u^{2} + \frac{1}{2} u^{2} \right) \left(-\frac{1}{2} u^{2} - \frac{1}{2} u^{2} + \frac{1}{2} u^{2} \right) \left(-\frac{1}{2} u^{2} - \frac{1}{2} u^{2} + \frac{1}{2} u^{2} \right) \left(-\frac{1}{2} u^{2} - \frac{1}{2} u^{2} + \frac{1}{2} u^{2} \right) \left(-\frac{1}{2} u^{2} - \frac{1}{2} u^{2} + \frac{1}{2} u^{2} \right) \left(-\frac{1}{2} u^{2} - \frac{1}{2} u^{2} + \frac{1}{2} u^{2} \right) \left(-\frac{1}{2} u^{2} - \frac{1}{2} u^{2} + \frac{1}{2} u^{2} \right) \left(-\frac{1}{2} u^{2} - \frac{1}{2} u^{2} + \frac{1}{2} u^{2} \right) \left(-\frac{1}{2} u^{2} - \frac{1}{2} u^{2} + \frac{1}{2} u^{2} \right) \left(-\frac{1}{2} u^{2} - \frac{1}{2} u^{2} - \frac{1}{2} u^{2} \right) \left(-\frac{1}{2} u^{2} - \frac{1}{$$

$$(85.6), \left(\begin{array}{c}_{I}\mathcal{A}_{z^{4}}\left(\underline{\imath}\mathcal{A}_{z^{4}}\right)\left(\underline{\imath}n+\underline{\imath}m\right)+\left(\underline{\imath}b\frac{1}{\overline{\imath}\overline{n}}+\underline{\Lambda}\mathcal{L}\right)\underline{\imath}\mu\underline{\imath}\\\left(\frac{1}{\overline{\imath}\overline{n}}\underline{\imath}b+\underline{\imath}\mathcal{A}\mathcal{N}\right)\left(\underline{\imath}n+\underline{\imath}m\right)+\left(\underline{\imath}n-\underline{\imath}n\right)\left(\underline{\imath}n-\underline{\imath}m\right)+\\\left(\frac{1}{\overline{\imath}\overline{n}}\underline{\imath}b+\underline{\imath}\mathcal{A}\mathcal{N}\right)\left(\underline{\imath}^{4\mu}-\underline{\imath}\right)\left(\underline{\imath}n-\underline{\imath}m\right)+\\\left(\frac{1}{\overline{\imath}\overline{n}}\underline{\imath}h+\underline{\imath}\mathcal{A}\mathcal{N}\right)\left(\underline{\imath}^{4\mu}-\underline{\imath}\right)\left(\underline{\imath}n-\underline{\imath}m\right)+\\\left(\frac{1}{\overline{\imath}\overline{n}}\mathcal{A}\mathcal{N}\left(\underline{\imath}\mathcal{A}_{z^{4}}\right)+\underline{1}\right)\left(\underline{\imath}n-\underline{\imath}m\right)+\end{array}\right)$$

$$z_{\overline{s}}^{T} z_{\overline{s}} \int (z^{\mu} - 1) (z^{\mu} - 1) (z^{\mu} - z^{\mu}) = z \left(\frac{1}{2} \int (z^{\mu} - z^{\mu}) (z^{\mu} - z^{\mu}) (z^{\mu} - z^{\mu}) (z^{\mu} - z^{\mu}) \right)$$

$$(75.8)$$

$$(m_{1} - n_{1}) (k_{1} + 1) R_{1} q_{-3} z^{-4} = \begin{pmatrix} -4\mu_{1} \left(-15c_{3} + \frac{5}{R_{1}^{2}}d_{5} \right) \\ + (m_{1} - n_{1}) \left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2} \right) R_{1}c_{3} \\ + (m_{1} + n_{1}) \left(1 - \frac{\mu_{1}}{\mu_{2}} \right) \left(d_{5}\frac{1}{R_{1}} - 3R_{1}c_{3} \right) \end{pmatrix} z^{-4}.$$

$$(3.42)$$

Now, 38 coupled linear algebraic equations have been derived from the four boundary conditions along imperfect interfaces Γ_0 and Γ_1 . With e_k and f_k encapsulated in p_k and q_k , the 38 equations contain 38 unknown coefficients of a_k , b_k , c_k , d_k , e_k and f_k for 8 stress potentials.

3.4 Homogeneous Imperfect Interfaces

In this present research, we consider the homogeneous imperfect interface: namely, the normal (m) and tangential (n) spring-factor type interface parameters are constant along the interface Γ_0 and Γ_1 . For convenience, let us introduce non-dimensional parameter $M_0 = \frac{m_0+n_0}{2\mu_1}R_0$, $M_1 = \frac{m_1+n_1}{2\mu_1}R_1$ characterizing the degree of damage along the interface. In this view, a very small value of M (say M = 0.01) represents complete debonding and a large value of M (say M = 100) corresponds to the case of perfect bonding condition. The other values between 0.01 and 100 are assumed to characterize the state of imperfect adhesion.

Also, the uniform eigenstrains $(\varepsilon_x^0, \varepsilon_y^0, \varepsilon_z^0)$ prescribed in the inclusion and the displacements induced by them are assumed zero since their effects on crack propagations under the Model I load are negligible.

Substituting p_k and q_k into (3.3) thru (3.21) and (3.24) thru (3.42) and deploying M_0 , and M_1 into the equations we can write the following 38 coupled linear algebraic

equations (see Appendix 5 for detail calculations):

.

.

.

$$0 = 24a_{6} + \frac{4}{R_{0}^{2}}b_{4} - \frac{4}{R_{0}^{10}}\frac{1}{2}e_{-4} + \frac{10}{R_{0}^{2}}e_{4} + \frac{12}{R_{0}^{2}}g_{7}f_{-3} + \frac{12}{R_{0}^{2}}g_{6}f_{-2}$$

$$+ \left(\frac{12}{R_{0}^{2}}g_{5} - 24g_{7}\right)f_{-1} + \left(\frac{12}{R_{0}^{2}}g_{4} - 24g_{6}\right)f_{0} + \left(\frac{12}{R_{0}^{2}}g_{3} - 24g_{5}\right)f_{1}$$

$$+ \left(\frac{12}{R_{0}^{2}}g_{2} - 24g_{4}\right)f_{2} + \left(\frac{12}{R_{0}^{2}}g_{1} - 24g_{3}\right)f_{3} + \left(\frac{12}{R_{0}^{2}}g_{0} - 24g_{2}\right)f_{4},$$

$$\vdots \qquad (3.43)$$

$$0 = 15a_{5} + \frac{3}{R_{0}^{2}}b_{3} - \frac{3}{R_{0}^{8}}\frac{1}{2}e_{-3} + \frac{6}{R_{0}^{2}}e_{3} + \left(\frac{6}{R_{0}^{2}}g_{6} - \frac{3}{R_{0}^{8}}g_{0}\right)f_{-3}$$

$$+ \left(\frac{6}{R_{0}^{2}}g_{5} - 15g_{7}\right)f_{-2} + \left(\frac{6}{R_{0}^{2}}g_{4} - 15g_{6}\right)f_{-1} + \left(\frac{6}{R_{0}^{2}}g_{3} - 15g_{5}\right)f_{0}$$

$$+ \left(\frac{6}{R_{0}^{2}}g_{2} - 15g_{4}\right)f_{1} + \left(\frac{6}{R_{0}^{2}}g_{1} - 15g_{3}\right)f_{2} + \left(\frac{6}{R_{0}^{2}}g_{0} - 15g_{2}\right)f_{3} - 15g_{1}f_{4},$$

$$(3.44)$$

$$0 = 8a_4 + \frac{2}{R_0^2}b_2 - \frac{1}{R_0^6}e_{-2} + \frac{3}{R_0^2}e_2 - 4e_4 + \left(\frac{2}{R_0^2}g_5 - 8g_7 - \frac{2}{R_0^6}g_1\right)f_{-3} \quad (3.45)$$

+ $\left(\frac{2}{R_0^2}g_4 - 8g_6 - \frac{2}{R_0^6}g_0\right)f_{-2} + \left(\frac{2}{R_0^2}g_3 - 8g_5\right)f_{-1} + \left(\frac{2}{R_0^2}g_2 - 8g_4\right)f_0$
+ $\left(\frac{2}{R_0^2}g_1 - 8g_3\right)f_1 + \left(\frac{2}{R_0^2}g_0 - 8g_2\right)f_2 - 8g_1f_3 - 8g_0f_4,$

$$0 = 3a_{3} + \frac{1}{R_{0}^{2}}b_{1} - \frac{1}{2}\frac{1}{R_{0}^{4}}e_{-1} + \frac{1}{R_{0}^{2}}e_{1} - 3\frac{1}{2}e_{3} + \left(-3g_{6} - \frac{1}{R_{0}^{4}}g_{2}\right)f_{-3} \qquad (3.46)$$

+ $\left(-3g_{5} - \frac{1}{R_{0}^{4}}g_{1}\right)f_{-2} + \left(-3g_{4} - \frac{1}{R_{0}^{4}}g_{0}\right)f_{-1}$
 $-3g_{3}f_{0} - 3g_{2}f_{1} - 3g_{1}f_{2} - 3g_{0}f_{3},$

$$0 = -2a_{1} + e_{1} + \left(2g_{4} + \frac{2}{R_{0}^{2}}g_{2}\right)f_{-3} + \left(2g_{3} + \frac{2}{R_{0}^{2}}g_{1}\right)f_{-2} + \left(2g_{2} + \frac{2}{R_{0}^{2}}g_{0}\right)f_{-1} + 2g_{1}f_{0} + 2g_{0}f_{1},$$
(3.47)

.

.

$$0 = -2R_0^2 a_2 + \frac{1}{R_0^2} e_{-2} + R_0^2 e_2 + \left(2R_0^2 g_5 + \frac{6}{R_0^2} g_1\right) f_{-3}$$

$$+ \left(2R_0^2 g_4 + \frac{6}{R_0^2} g_0\right) f_{-2} + 2R_0^2 g_3 f_{-1} + 2R_0^2 g_2 f_0 + 2R_0^2 g_1 f_1 + 2R_0^2 g_0 f_2,$$
(3.48)

$$0 = -3R_0^4 a_3 + \frac{3}{R_0^2} e_{-3} - \frac{3}{2} e_{-1} + 3R_0^4 \frac{1}{2} e_3 \qquad (3.49)$$

+ $\left(3R_0^4 g_6 - 3g_2 + \frac{12}{R_0^2} g_0\right) f_{-3} + \left(3R_0^4 g_5 - 3g_1\right) f_{-2}$
+ $\left(3R_0^4 g_4 - 3g_0\right) f_{-1} + 3R_0^4 g_3 f_0 + 3R_0^4 g_2 f_1 + 3R_0^4 g_1 f_2 + 3R_0^4 g_0 f_3,$

•

ų

$$0 = -4R_0^6 a_4 + \frac{6}{R_0^2} e_{-4} - 4e_{-2} + 2R_0^6 e_4$$

$$+ (4R_0^6 g_7 - 8g_1) f_{-3} + (4R_0^6 g_6 - 8g_0) f_{-2} + 4R_0^6 g_5 f_{-1}$$

$$+ 4R_0^6 g_4 f_0 + 4R_0^6 g_3 f_1 + 4R_0^6 g_2 f_2 + 4R_0^6 g_1 f_3 + 4R_0^6 g_0 f_4,$$
(3.50)

$$0 = -5R_0^8 a_5 - 15\frac{1}{2}e_{-3} - 15g_0 f_{-3} + 5R_0^8 g_7 f_{-2} + 5R_0^8 g_6 f_{-1}$$

$$+5R_0^8 g_5 f_0 + 5R_0^8 g_4 f_1 + 5R_0^8 g_3 f_2 + 5R_0^8 g_2 f_3 + 5R_0^8 g_1 f_4,$$
(3.51)

$$0 = \left(48\frac{1}{M_0} - \frac{(m_0 - n_0)}{(m_0 + n_0)}(1 + \kappa_0\frac{\mu_1}{\mu_0}) - 6(1 - \frac{\mu_1}{\mu_0})\right)a_6 \qquad (3.52)$$

$$+ \left(-(1 - \frac{\mu_1}{\mu_0})\frac{1}{R_0^2} + \frac{8}{M_0}\frac{1}{R_0^2}\right)b_4 + (\kappa_1 + 1)\frac{1}{2R_0^{10}}e_{-4}$$

$$+ \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_7f_{-1} + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_6f_0$$

$$+ \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_5f_1 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_4f_2$$

$$+ \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_3f_3 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_2f_4,$$

$$0 = \left(30\frac{1}{M_0} - \frac{(m_0 - n_0)}{(m_0 + n_0)}(1 + \kappa_0\frac{\mu_1}{\mu_0}) - 5(1 - \frac{\mu_1}{\mu_0})\right)a_5 \qquad (3.53)$$

$$+ \left(-(1 - \frac{\mu_1}{\mu_0})\frac{1}{R_0^2} + \frac{1}{M_0}\frac{6}{R_0^2}\right)b_3 + (\kappa_1 + 1)\frac{1}{2R_0^8}e_{-3}$$

$$+ (\kappa_1 + 1)\frac{1}{R_0^8}g_0f_{-3} + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_7f_{-2} + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_6f_{-1}$$

$$+ \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_5f_0 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_4f_1 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_3f_2$$

$$+ \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_2f_3 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_1f_4,$$

.

•

• ...

4

,

$$0 = \left(16\frac{1}{M_0} - \frac{(m_0 - n_0)}{(m_0 + n_0)}(1 + \kappa_0\frac{\mu_1}{\mu_0}) - 4(1 - \frac{\mu_1}{\mu_0})\right)a_4$$
(3.54)
+ $\left(-(1 - \frac{\mu_1}{\mu_0})\frac{1}{R_0^2} + \frac{1}{M_0}\frac{4}{R_0^2}\right)b_2 + (\kappa_1 + 1)\frac{1}{R_0^6}\frac{1}{2}e_{-2} + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)\frac{1}{2}e_4$
+ $\left(\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_7 + (\kappa_1 + 1)\frac{1}{R_0^6}g_1\right)f_{-3}$
+ $\left(\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_6 + (\kappa_1 + 1)\frac{1}{R_0^6}g_0\right)f_{-2} + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_5f_{-1}$
+ $\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_4f_0 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_3f_1$
+ $\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_2f_2 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_1f_3 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_0f_4,$

$$0 = \left(6\frac{1}{M_{0}} - \frac{(m_{0} - n_{0})}{(m_{0} + n_{0})}(1 + \kappa_{0}\frac{\mu_{1}}{\mu_{0}}) - 3(1 - \frac{\mu_{1}}{\mu_{0}})\right)a_{3} \qquad (3.55)$$

$$+ \left(-(1 - \frac{\mu_{1}}{\mu_{0}})\frac{1}{R_{0}^{2}} + \frac{1}{M_{0}}\frac{2}{R_{0}^{2}}\right)b_{1} + (\kappa_{1} + 1)\frac{1}{R_{0}^{4}}\frac{1}{2}e_{-1}$$

$$+ \frac{(m_{0} - n_{0})}{(m_{0} + n_{0})}(\kappa_{1} + 1)\frac{1}{2}e_{3} + \left(\frac{(m_{0} - n_{0})}{(m_{0} + n_{0})}(\kappa_{1} + 1)g_{6} + (\kappa_{1} + 1)\frac{1}{R_{0}^{4}}g_{2}\right)f_{-3}$$

$$+ \left(\frac{(m_{0} - n_{0})}{(m_{0} + n_{0})}(\kappa_{1} + 1)g_{5} + (\kappa_{1} + 1)\frac{1}{R_{0}^{4}}g_{1}\right)f_{-2}$$

$$+ \left(\frac{(m_{0} - n_{0})}{(m_{0} + n_{0})}(\kappa_{1} + 1)g_{4} + (\kappa_{1} + 1)\frac{1}{R_{0}^{4}}g_{0}\right)f_{-1}$$

,

$$+\frac{(m_0-n_0)}{(m_0+n_0)}(\kappa_1+1)g_3f_0+\frac{(m_0-n_0)}{(m_0+n_0)}(\kappa_1+1)g_2f_1\\+\frac{(m_0-n_0)}{(m_0+n_0)}(\kappa_1+1)g_1f_2+\frac{(m_0-n_0)}{(m_0+n_0)}(\kappa_1+1)g_0f_3,$$

$$0 = \left(-\frac{(m_0 - n_0)}{(m_0 + n_0)}(1 + \kappa_0\frac{\mu_1}{\mu_0})R_0 - 2(1 - \frac{\mu_1}{\mu_0})R_0\right)a_2 - (1 + \kappa_0\frac{\mu_1}{\mu_0})\frac{1}{R_0}a_0 (3.56)$$

$$-(1 - \frac{\mu_1}{\mu_0})\frac{1}{R_0}b_0 + (\kappa_1 + 1)\frac{1}{R_0}\frac{1}{2}e_0 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)R_0\frac{1}{2}e_2$$

$$+ \left(\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)R_0g_5 + (\kappa_1 + 1)\frac{1}{R_0}g_3\right)f_{-3}$$

$$+ \left(\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)R_0g_4 + (\kappa_1 + 1)\frac{1}{R_0}g_2\right)f_{-2}$$

$$+ \left(\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)R_0g_3 + (\kappa_1 + 1)\frac{1}{R_0}g_1\right)f_{-1}$$

$$+ \left(\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)R_0g_2 + (\kappa_1 + 1)\frac{1}{R_0}g_0\right)f_0$$

$$+ \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)R_0g_1f_1 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)R_0g_0f_2,$$

$$0 = \begin{pmatrix} -4\frac{1}{M_0} - \frac{(m_0 - n_0)}{(m_0 + n_0)} (1 + \kappa_0 \frac{\mu_1}{\mu_0}) - \frac{(m_0 - n_0)}{(m_0 + n_0)} (1 - \frac{\mu_1}{\mu_0}) \\ -(1 + \kappa_0 \frac{\mu_1}{\mu_0}) - (1 - \frac{\mu_1}{\mu_0}) \end{pmatrix} a_1 \qquad (3.57)$$
$$+ \frac{m_0(\kappa_1 + 1)}{(m_0 + n_0)} e_1 + \frac{2m_0(\kappa_1 + 1)}{(m_0 + n_0)} g_4 f_{-3} + \frac{2m_0(\kappa_1 + 1)}{(m_0 + n_0)} g_3 f_{-2} \\ + \frac{2m_0(\kappa_1 + 1)}{(m_0 + n_0)} g_2 f_{-1} + \frac{2m_0(\kappa_1 + 1)}{(m_0 + n_0)} g_1 f_0 + \frac{2m_0(\kappa_1 + 1)}{(m_0 + n_0)} g_0 f_1,$$

$$0 = -\frac{(m_0 - n_0)}{(m_0 + n_0)} (1 + \kappa_0 \frac{\mu_1}{\mu_0}) a_0 + \begin{pmatrix} -4\frac{1}{M_0} R_0^2 - \frac{(m_0 - n_0)}{(m_0 + n_0)} 2(1 - \frac{\mu_1}{\mu_0}) R_0^2 \\ -(1 + \kappa_0 \frac{\mu_1}{\mu_0}) R_0^2 \end{pmatrix} a_2 \quad (3.58)$$
$$-\frac{(m_0 - n_0)}{(m_0 + n_0)} (1 - \frac{\mu_1}{\mu_0}) b_0 + \frac{(m_0 - n_0)}{(m_0 + n_0)} (\kappa_1 + 1) \frac{1}{2} e_0 + (\kappa_1 + 1) R_0^2 \frac{1}{2} e_2$$

$$+ \left(\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_3 + (\kappa_1 + 1)R_0^2g_5\right)f_{-3} + \left(\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_2 + (\kappa_1 + 1)R_0^2g_4\right)f_{-2} + \left(\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_1 + (\kappa_1 + 1)R_0^2g_3\right)f_{-1} + \left(\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_0 + (\kappa_1 + 1)R_0^2g_2\right)f_0 + (\kappa_1 + 1)R_0^2g_1f_1 + (\kappa_1 + 1)R_0^2g_0f_2,$$

•

$$0 = \left(-\frac{6}{M_0}R_0^4 - \frac{(m_0 - n_0)}{(m_0 + n_0)}3(1 - \frac{\mu_1}{\mu_0})R_0^4 - (1 + \kappa_0\frac{\mu_1}{\mu_0})R_0^4\right)a_3 \qquad (3.59)$$

$$-\frac{(m_0 - n_0)}{(m_0 + n_0)}(1 - \frac{\mu_1}{\mu_0})R_0^2b_1 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)\frac{1}{2}e_{-1}$$

$$+(\kappa_1 + 1)R_0^4\frac{1}{2}e_3 + \left(\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_2 + (\kappa_1 + 1)R_0^4g_6\right)f_{-3}$$

$$+ \left(\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_1 + (\kappa_1 + 1)R_0^4g_5\right)f_{-2}$$

$$+ \left(\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_0 + (\kappa_1 + 1)R_0^4g_4\right)f_{-1} + (\kappa_1 + 1)R_0^4g_3f_0$$

$$+(\kappa_1 + 1)R_0^4g_2f_1 + (\kappa_1 + 1)R_0^4g_1f_2 + (\kappa_1 + 1)R_0^4g_0f_3,$$

$$0 = \left(-8\frac{1}{M_0}R_0^6 - \frac{(m_0 - n_0)}{(m_0 + n_0)}4(1 - \frac{\mu_1}{\mu_0})R_0^6 - (1 + \kappa_0\frac{\mu_1}{\mu_0})R_0^6\right)a_4 \qquad (3.60)$$

$$-\frac{(m_0 - n_0)}{(m_0 + n_0)}(1 - \frac{\mu_1}{\mu_0})R_0^4b_2 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)\frac{1}{2}e_{-2}$$

$$+(\kappa_1 + 1)R_0^6\frac{1}{2}e_4 + \left((\kappa_1 + 1)R_0^6g_7 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_1\right)f_{-3}$$

$$+ \left((\kappa_1 + 1)R_0^6g_6 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_0\right)f_{-2} + (\kappa_1 + 1)R_0^6g_5f_{-1}$$

$$+(\kappa_1 + 1)R_0^6g_4f_0 + (\kappa_1 + 1)R_0^6g_3f_1 + (\kappa_1 + 1)R_0^6g_2f_2$$

$$+(\kappa_1 + 1)R_0^6g_1f_3 + (\kappa_1 + 1)R_0^6g_0f_4,$$

÷

$$0 = \left(-10\frac{1}{M_0}R_0^8 - \frac{(m_0 - n_0)}{(m_0 + n_0)}5(1 - \frac{\mu_1}{\mu_0})R_0^8 - (1 + \kappa_0\frac{\mu_1}{\mu_0})R_0^8\right)a_5 \qquad (3.61)$$

$$-\frac{(m_0 - n_0)}{(m_0 + n_0)}(1 - \frac{\mu_1}{\mu_0})R_0^6b_3 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)\frac{1}{2}e_{-3}$$

$$+\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_0f_{-3} + (\kappa_1 + 1)R_0^8g_7f_{-2} + (\kappa_1 + 1)R_0^8g_6f_{-1}$$

$$+(\kappa_1 + 1)R_0^8g_5f_0 + (\kappa_1 + 1)R_0^8g_4f_1 + (\kappa_1 + 1)R_0^8g_3f_2$$

$$+(\kappa_1 + 1)R_0^8g_2f_3 + (\kappa_1 + 1)R_0^8g_1f_4,$$

$$\vec{0} = + \frac{4}{R_1^{10}} c_4 + 4 \frac{1}{R_1^2} e_4 - \frac{2}{R_1^{10}} e_{-4} - \frac{4}{R_1^{10}} h_{-1} f_{-3} - \frac{4}{R_1^{10}} h_{-2} f_{-2}$$

$$- \frac{4}{R_1^{10}} h_{-3} f_{-1} - \frac{4}{R_1^{10}} h_{-4} f_0 - \frac{4}{R_1^{10}} h_{-5} f_1 - \frac{4}{R_1^{10}} h_{-6} f_2$$

$$- \frac{4}{R_1^{10}} h_{-7} f_3 - \frac{4}{R_1^{10}} h_{-8} f_4,$$

$$(3.62)$$

$$0 = +\frac{3}{R_{1}^{8}}c_{3} + \frac{6}{R_{1}^{2}}e_{3} + \left(\frac{6}{R_{1}^{2}}h_{-1} - \frac{3}{R_{1}^{8}}h_{-7}\right)f_{4}$$

$$-\frac{3}{R_{1}^{8}}h_{-6}f_{3} - \frac{3}{R_{1}^{8}}h_{-5}f_{2} - \frac{3}{R_{1}^{8}}h_{-4}f_{1} - \frac{3}{R_{1}^{8}}h_{-3}f_{0}$$

$$-\frac{3}{R_{1}^{8}}h_{-2}f_{-1} - \frac{3}{R_{1}^{8}}h_{-1}f_{-2} - \frac{1}{R_{1}^{8}}\frac{3}{2}e_{-3},$$
(3.63)

$$0 = \frac{2}{R_1^6} c_2 + \frac{3}{R_1^2} e_2 - \frac{1}{R_1^6} e_{-2} + \left(\frac{2}{R_1^2} h_{-2} - \frac{2}{R_1^6} h_{-6}\right) f_4 \qquad (3.64) \\ + \left(\frac{2}{R_1^2} h_{-1} - \frac{2}{R_1^6} h_{-5}\right) f_3 - \frac{2}{R_1^6} h_{-4} f_2 - \frac{2}{R_1^6} h_{-3} f_1 - \frac{2}{R_1^6} h_{-2} f_0 - \frac{2}{R_1^6} h_{-1} f_{-1},$$

$$-\frac{1}{R_1^2}B = +\frac{1}{R_1^4}c_1 - \frac{3}{2}e_3 + \frac{1}{R_1^2}e_1 - \frac{1}{R_1^4}\frac{1}{2}e_{-1} + \left(-3h_{-1} - \frac{1}{R_1^4}h_{-5}\right)f_4 \quad (3.65)$$
$$-\frac{1}{R_1^4}h_{-4}f_3 - \frac{1}{R_1^4}h_{-3}f_2 - \frac{1}{R_1^4}h_{-2}f_1 - \frac{1}{R_1^4}h_{-1}f_0,$$

1

.

57

...

$$2A = -\frac{1}{R_1^2} d_1 + e_1 + \left(2h_{-3} + \frac{2}{R_1^2}h_{-5}\right) f_4 + \left(2h_{-2} + \frac{2}{R_1^2}h_{-4}\right) f_3 \qquad (3.66)$$
$$+ \left(2h_{-1} + \frac{2}{R_1^2}h_{-3}\right) f_2 + \frac{2}{R_1^2}h_{-2}f_1 + \frac{2}{R_1^2}h_{-1}f_0,$$

$$0 = -\frac{2}{R_1^2} d_2 + \dot{R_1^2} e_2 + \frac{1}{R_1^2} e_{-2} + \left(2R_1^2 h_{-2} + \frac{6}{R_1^2} h_{-6}\right) f_4 \qquad (3.67)$$

+ $\left(2R_1^2 h_{-1} + \frac{6}{R_1^2} h_{-5}\right) f_3 + \frac{6}{R_1^2} h_{-4} f_2 + \frac{6}{R_1^2} h_{-3} f_1 + \frac{6}{R_1^2} h_{-2} f_0 + \frac{6}{R_1^2} h_{-1} f_{-1},$

$$0 = +3c_{1} - \frac{3}{R_{1}^{2}}d_{3} + R_{1}^{4}\frac{3}{2}e_{3} - \frac{3}{2}e_{-1} + \frac{3}{R_{1}^{2}}e_{-3}$$

$$+ \left(\frac{12}{R_{1}^{2}}h_{-7} + 3R_{1}^{4}h_{-1} - 3h_{-5}\right)f_{4} + \left(\frac{12}{R_{1}^{2}}h_{-6} - 3h_{-4}\right)f_{3} + \left(\frac{12}{R_{1}^{2}}h_{-5} - 3h_{-3}\right)f_{2}$$

$$+ \left(\frac{12}{R_{1}^{2}}h_{-4} - 3h_{-2}\right)f_{1} + \left(\frac{12}{R_{1}^{2}}h_{-3} - 3h_{-1}\right)f_{0} + \frac{12}{R_{1}^{2}}h_{-2}f_{-1} + \frac{12}{R_{1}^{2}}h_{-1}f_{-2},$$

$$(3.68)$$

$$0 = +8c_2 - \frac{4}{R_1^2}d_4 - 4e_{-2} + \frac{6}{R_1^2}e_{-4} + \left(\frac{20}{R_1^2}h_{-8} - 8h_{-6}\right)f_4 \qquad (3.69)$$

+ $\left(\frac{20}{R_1^2}h_{-7} - 8h_{-5}\right)f_3 + \left(\frac{20}{R_1^2}h_{-6} - 8h_{-4}\right)f_2 + \left(\frac{20}{R_1^2}h_{-5} - 8h_{-3}\right)f_1$
+ $\left(\frac{20}{R_1^2}h_{-4} - 8h_{-2}\right)f_0 + \left(\frac{20}{R_1^2}h_{-3} - 8h_{-1}\right)f_{-1} + \frac{20}{R_1^2}h_{-2}f_{-2} + \frac{20}{R_1^2}h_{-1}f_{-3},$

$$0 = 15c_3 - \frac{5}{R_1^2}d_5 - \frac{15}{2}e_{-3} + \frac{15}{R_1^2}e_{-5} - 15h_{-7}f_4 \qquad (3.70)$$

+ $\left(\frac{30}{R_1^2}h_{-8} - 15h_{-6}\right)f_3 + \left(\frac{30}{R_1^2}h_{-7} - 15h_{-5}\right)f_2$
+ $\left(\frac{30}{R_1^2}h_{-6} - 15h_{-4}\right)f_1 + \left(\frac{30}{R_1^2}h_{-5} - 15h_{-3}\right)f_0$
+ $\left(\frac{30}{R_1^2}h_{-4} - 15h_{-2}\right)f_{-1} + \left(\frac{30}{R_1^2}h_{-3} - 15h_{-1}\right)f_{-2} + \frac{30}{R_1^2}h_{-2}f_{-3} + \frac{30}{R_1^2}h_{-1}f_{-4},$

÷

$$0 = \left(\frac{1}{M_{1}}\frac{8}{R_{1}^{9}} - \frac{(m_{1} - n_{1})}{(m_{1} + n_{1})}\left(1 - \frac{\mu_{1}}{\mu_{2}}\right)\frac{4}{R_{1}^{9}} + \left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2}\right)\frac{1}{R_{1}^{9}}\right)c_{4}$$
(3.71)
$$- (k_{1} + 1)\frac{1}{R_{1}^{9}}h_{-8}f_{4} - (k_{1} + 1)\frac{1}{R_{1}^{9}}h_{-7}f_{3} - (k_{1} + 1)\frac{1}{R_{1}^{9}}h_{-6}f_{2}$$
$$- (k_{1} + 1)\frac{1}{R_{1}^{9}}h_{-5}f_{1} - (k_{1} + 1)\frac{1}{R_{1}^{9}}h_{-4}f_{0} - (k_{1} + 1)\frac{1}{R_{1}^{9}}h_{-3}f_{-1}$$
$$- (k_{1} + 1)\frac{1}{R_{1}^{9}}h_{-2}f_{-2} - (k_{1} + 1)\frac{1}{R_{1}^{9}}h_{-1}f_{-3} - (k_{1} + 1)\frac{1}{R_{1}^{9}}\frac{1}{2}e_{-4},$$

.

$$0 = \left(\frac{1}{M_{1}}\frac{6}{R_{1}^{7}} - \frac{(m_{1} - n_{1})}{(m_{1} + n_{1})} \left(1 - \frac{\mu_{1}}{\mu_{2}}\right) \frac{3}{R_{1}^{7}} + \left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2}\right) \frac{1}{R_{1}^{7}}\right) c_{3}$$
(3.72)
$$+ \frac{(m_{1} - n_{1})}{(m_{1} + n_{1})} \left(1 - \frac{\mu_{1}}{\mu_{2}}\right) \frac{1}{R_{1}^{9}} d_{5} - \frac{(k_{1} + 1)}{R_{1}^{7}} \frac{1}{2}e_{-3} - \frac{(k_{1} + 1)}{R_{1}^{7}}h_{-7}f_{4}$$
$$- \frac{(k_{1} + 1)}{R_{1}^{7}}h_{-6}f_{3} - \frac{(k_{1} + 1)}{R_{1}^{7}}h_{-5}f_{2} - \frac{(k_{1} + 1)}{R_{1}^{7}}h_{-4}f_{1} - \frac{(k_{1} + 1)}{R_{1}^{7}}h_{-3}f_{0}$$
$$- \frac{(k_{1} + 1)}{R_{1}^{7}}h_{-2}f_{-1} - \frac{(k_{1} + 1)}{R_{1}^{7}}h_{-1}f_{-2},$$

.

۰

$$0 = \left(\frac{1}{M_{1}}\frac{4}{R_{1}^{5}} - \frac{(m_{1} - n_{1})}{(m_{1} + n_{1})} \left(1 - \frac{\mu_{1}}{\mu_{2}}\right) \frac{2}{R_{1}^{5}} + \left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2}\right) \frac{1}{R_{1}^{5}}\right) c_{2}$$
(3.73)
+ $\frac{(m_{1} - n_{1})}{(m_{1} + n_{1})} \left(1 - \frac{\mu_{1}}{\mu_{2}}\right) \frac{1}{R_{1}^{7}} d_{4} - \frac{(k_{1} + 1)}{R_{1}^{5}} \frac{1}{2}e_{-2} - \frac{(k_{1} + 1)}{R_{1}^{5}} h_{-6}f_{4} - \frac{(k_{1} + 1)}{R_{1}^{5}} h_{-5}f_{3}$
- $\frac{(k_{1} + 1)}{R_{1}^{5}} h_{-4}f_{2} - \frac{(k_{1} + 1)}{R_{1}^{5}} h_{-3}f_{1} - \frac{(k_{1} + 1)}{R_{1}^{5}} h_{-2}f_{0} - \frac{(k_{1} + 1)}{R_{1}^{5}} h_{-1}f_{-1},$

$$\begin{pmatrix} -\left(1-\frac{\mu_{1}}{\mu_{2}}\right)\frac{1}{R_{1}}-\frac{1}{M_{1}}\frac{2}{R_{1}}\right)B$$

$$= \left(\frac{1}{M_{1}}\frac{2}{R_{1}^{3}}-\frac{(m_{1}-n_{1})}{(m_{1}+n_{1})}\left(1-\frac{\mu_{1}}{\mu_{2}}\right)\frac{1}{R_{1}^{3}}+\left(1+\frac{\mu_{1}}{\mu_{2}}k_{2}\right)\frac{1}{R_{1}^{3}}\right)c_{1}$$

$$+\frac{(m_{1}-n_{1})}{(m_{1}+n_{1})}\left(1-\frac{\mu_{1}}{\mu_{2}}\right)\frac{1}{R_{1}^{5}}d_{3}-\frac{(m_{1}-n_{1})}{(m_{1}+n_{1})}\left(k_{1}+1\right)R_{1}\frac{1}{2}e_{3}$$

$$-\frac{(k_{1}+1)}{R_{1}^{3}}\frac{1}{2}e_{-1}+\left(-\frac{(m_{1}-n_{1})}{(m_{1}+n_{1})}\left(k_{1}+1\right)R_{1}h_{-1}-\frac{(k_{1}+1)}{R_{1}^{3}}h_{-5}\right)f_{4}$$

$$-\frac{(k_{1}+1)}{R_{1}^{3}}h_{-4}f_{3}-\frac{(k_{1}+1)}{R_{1}^{3}}h_{-3}f_{2}-\frac{(k_{1}+1)}{R_{1}^{3}}h_{-2}f_{1}-\frac{(k_{1}+1)}{R_{1}^{3}}h_{-1}f_{0},$$

$$(3.74)$$

÷

59

۲

:

$$0 = \frac{(m_1 - n_1)}{(m_1 + n_1)} \left(1 - \frac{\mu_1}{\mu_2} \right) \frac{1}{R^3} d_2 - \frac{(m_1 - n_1)}{2(m_1 + n_1)} (k_1 + 1) R_1 e_2$$
(3.75)
- $(k_1 + 1) \frac{1}{R_1} \frac{1}{2} e_0 + \left(-\frac{(m_1 - n_1)}{(m_1 + n_1)} (k_1 + 1) R_1 h_{-2} - (k_1 + 1) \frac{1}{R_1} h_{-4} \right) f_4$
+ $\left(-\frac{(m_1 - n_1)}{(m_1 + n_1)} (k_1 + 1) R_1 h_{-1} - (k_1 + 1) \frac{1}{R_1} h_{-3} \right) f_3$
- $(k_1 + 1) \frac{1}{R_1} h_{-2} f_2 - (k_1 + 1) \frac{1}{R_1} h_{-1} f_1,$

.

$$\frac{4A}{M_{1}} - \frac{2m_{1}A}{(m_{1}+n_{1})} \left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2}\right) - \frac{2m_{1}A}{(m_{1}+n_{1})} \left(1 - \frac{\mu_{1}}{\mu_{2}}\right) \qquad (3.76)$$

$$= \left(-\frac{2}{R_{1}^{2}}\frac{1}{M_{1}} + \left(1 - \frac{\mu_{1}}{\mu_{2}}\right)\frac{1}{R_{1}^{2}} + \frac{(m_{1}-n_{1})}{(m_{1}+n_{1})} \left(1 - \frac{\mu_{1}}{\mu_{2}}\right)\frac{1}{R_{1}^{2}}\right) d_{1} - \frac{m_{1}(k_{1}+1)}{(m_{1}+n_{1})}e_{1} - \frac{2m_{1}(k_{1}+1)}{(m_{1}+n_{1})}h_{-3}f_{4} - \frac{2m_{1}(k_{1}+1)}{(m_{1}+n_{1})}h_{-2}f_{3} - \frac{2m_{1}(k_{1}+1)}{(m_{1}+n_{1})}h_{-1}f_{2},$$

$$0 = \left(-\frac{4}{R_{1}}\frac{1}{M_{1}} + \left(1 - \frac{\mu_{1}}{\mu_{2}}\right)\frac{1}{R_{1}}\right)d_{2} - (k_{1}+1)R_{1}^{3}\frac{1}{2}e_{2}$$

$$-\frac{(m_{1}-n_{1})}{(m_{1}+n_{1})}(k_{1}+1)R_{1}\frac{1}{2}e_{0} + \left(-(k_{1}+1)R_{1}^{3}h_{-2} - \frac{(m_{1}-n_{1})}{(m_{1}+n_{1})}(k_{1}+1)R_{1}h_{-4}\right)f_{4}$$

$$+ \left(-(k_{1}+1)R_{1}^{3}h_{-1} - \frac{(m_{1}-n_{1})}{(m_{1}+n_{1})}(k_{1}+1)R_{1}h_{-3}\right)f_{3}$$

$$-\frac{(m_{1}-n_{1})}{(m_{1}+n_{1})}(k_{1}+1)R_{1}h_{-2}f_{2} - \frac{(m_{1}-n_{1})}{(m_{1}+n_{1})}(k_{1}+1)R_{1}h_{-1}f_{1},$$

$$(3.77)$$

,

$$-\frac{(m_{1}-n_{1})}{(m_{1}+n_{1})}\left(1-\frac{\mu_{1}}{\mu_{2}}\right)BR_{1}^{2}$$

$$= \left(\frac{6}{M_{1}}+\frac{(m_{1}-n_{1})}{(m_{1}+n_{1})}\left(1+\frac{\mu_{1}}{\mu_{2}}k_{2}\right)-\left(1-\frac{\mu_{1}}{\mu_{2}}\right)\right)c_{1}$$

$$+\left(-\frac{6}{R_{1}^{2}}\frac{1}{M_{1}}+\left(1-\frac{\mu_{1}}{\mu_{2}}\right)\frac{1}{R_{1}^{2}}\right)d_{3}-(k_{1}+1)R_{1}^{4}\frac{1}{2}e_{3}-\frac{(m_{1}-n_{1})}{(m_{1}+n_{1})}(k_{1}+1)\frac{1}{2}e_{-1}$$

$$+\left(-(k_{1}+1)R_{1}^{4}h_{-1}-\frac{(m_{1}-n_{1})}{(m_{1}+n_{1})}(k_{1}+1)h_{-5}\right)f_{4}-\frac{(m_{1}-n_{1})}{(m_{1}+n_{1})}(k_{1}+1)h_{-4}f_{3}$$

$$-\frac{(m_{1}-n_{1})}{(m_{1}+n_{1})}(k_{1}+1)h_{-3}f_{2}-\frac{(m_{1}-n_{1})}{(m_{1}+n_{1})}(k_{1}+1)h_{-2}f_{1}-\frac{(m_{1}-n_{1})}{(m_{1}+n_{1})}(k_{1}+1)h_{-1}f_{0},$$
(3.78)

•

60

÷

$$0 = \left(\frac{16}{M_1} + \frac{(m_1 - n_1)}{(m_1 + n_1)} \left(1 + \frac{\mu_1}{\mu_2} k_2\right) - 2 \left(1 - \frac{\mu_1}{\mu_2}\right)\right) c_2 \qquad (3.79)$$

$$+ \left(-\frac{1}{R_1^2} \frac{8}{M_1} + \left(1 - \frac{\mu_1}{\mu_2}\right) \frac{1}{R_1^2}\right) d_4 - \frac{(m_1 - n_1) (k_1 + 1)}{2 (m_1 + n_1)} e_{-2}$$

$$- \frac{(m_1 - n_1)}{(m_1 + n_1)} (k_1 + 1) h_{-6} f_4 - \frac{(m_1 - n_1)}{(m_1 + n_1)} (k_1 + 1) h_{-5} f_3$$

$$- \frac{(m_1 - n_1)}{(m_1 + n_1)} (k_1 + 1) h_{-4} f_2 - \frac{(m_1 - n_1)}{(m_1 + n_1)} (k_1 + 1) h_{-3} f_1$$

$$- \frac{(m_1 - n_1)}{(m_1 + n_1)} (k_1 + 1) h_{-2} f_0 - \frac{(m_1 - n_1)}{(m_1 + n_1)} (k_1 + 1) h_{-1} f_{-1},$$

$$0 = \left(\frac{30}{M_1} + \frac{(m_1 - n_1)}{(m_1 + n_1)} \left(1 + \frac{\mu_1}{\mu_2} k_2\right) - 3 \left(1 - \frac{\mu_1}{\mu_2}\right)\right) c_3$$

$$+ \left(-\frac{1}{R_1^2} \frac{10}{M_1} + \left(1 - \frac{\mu_1}{\mu_2}\right) \frac{1}{R_1^2}\right) d_5 - \frac{(m_1 - n_1) (k_1 + 1)}{(m_1 + n_1)} h_{-7} f_4$$

$$- \frac{(m_1 - n_1) (k_1 + 1)}{(m_1 + n_1)} h_{-6} f_3 - \frac{(m_1 - n_1) (k_1 + 1)}{(m_1 + n_1)} h_{-5} f_2 - \frac{(m_1 - n_1) (k_1 + 1)}{(m_1 + n_1)} h_{-4} f_1$$

$$- \frac{(m_1 - n_1) (k_1 + 1)}{(m_1 + n_1)} h_{-3} f_0 - \frac{(m_1 - n_1) (k_1 + 1)}{(m_1 + n_1)} h_{-2} f_{-1}$$

$$- \frac{(m_1 - n_1) (k_1 + 1)}{(m_1 + n_1)} h_{-1} f_{-2} - \frac{(m_1 - n_1) (k_1 + 1)}{2 (m_1 + n_1)} e_{-3}.$$
(3.80)

÷

.

,

.

.

4

CHAPTER 4 Numerical Analysis Results / Discussion

4.1 Introduction

In this chapter, the numerical analysis of 38 coupled linear algebraic equations derived in Chapter 3 is presented to determine the 38 unknown coefficients. The unknown coefficients vary according to mechanical properties of the composites such as Poisson's ratio and crack locations. Based on the resulting coefficients, stress intensity factors at the crack tips under Mode I are derived to investigate crack behaviour around the tips. Thus, this chapter first presents derivations of stress intensity factors at crack tips a and b in terms of the resulting coefficients of stress potentials.

The equations of (3.43) thru (3.80) can be expressed in a matrix form such as $\{b\} = [A] \{x\}$ where [A] is a 38 x 38 coefficient matrix, $\{b\}$ corresponds to a load vector and $\{x\}$ is the solution vector representing the resulting coefficients. Using MATLAB, the inverse matrix of [A] is calculated and the solution vector $\{x\}$ is readily calculated by $\{x\}=[A]^{-1}\{b\}$. For the verification of the formulation, a comparison of the results corresponding to a three-phase model with the perfect interfaces reported in the literature is made. Finally, some numerical results such as effects of inclusion stiffness, interphase stiffness and crack length on cracking are addressed.

4.2 Stress Fields around Crack Tips

In many engineering applications, a study of the behaviour of materials under certain loads in the presence of material flaws such as internal voids, minor cracks and different materials embedded inside is quite important to prevent their failures. In particular, it is well known that failures associated with sudden crack propagation in
solid materials take place even at the far lower than the failure limits of the stress. Eventually, to evaluate the effects of flaw size, fracture toughness and applied stress on material failures, the early frame works of fracture mechanics has been intensively addressed and developed by Inglis, Griffith and Westergaard. One of the most dominant researchers in 1950s is G.R. Irwin leading the Naval Research Laboratory. He further developed the early works and introduced a new idea that stresses and displacements in the neighbourhood of crack tips could be characterized by a single constant, called the stress intensity factor (SIF). The stress intensity factor represents the driving force to cause fracture in the vicinity of the crack tips.

There are three types of loadings that a crack can experience: Mode I represents a crack opening deformation, Mode II an in-plane shear and Mode III an out-of-plane deformation. A system with a crack can undergo any one of these modes, or arbitrary combinations of two or three modes. However, Mode I is known as the most dominant failure scenario [38] compared to the other modes. Furthermore, under uniform remote tensile stress, a singularity-dominated zone is defined as a region where the singularity $1/\sqrt{r}$ (r: distance from a crack tip) dominates the stress fields and the stress intensity factor in the zone defines the amplitude of the crack-tip singularity [57]. Thus, in this study, stress intensity factor under Mode I loading near crack tip a and b, namely within singularity dominated zone, is investigated. Also, it is assumed that there is no crack growth at the tips so that the Model I stress intensity factor K_I is smaller than the critical stress intensity factor K_{IC} .

Using the Cartesian form of (2.1), we can re-write it as

$$\sigma_{xx} + \sigma_{yy} = 2\left[\varphi'(z) + \overline{\varphi'(z)}\right], \qquad (4.1)$$

$$\sigma_{xx} - i\sigma_{xy} = \varphi'(z) + \overline{\varphi'(z)} - \left[\overline{z}\varphi''(z) + \psi'(z)\right].$$
(4.2)

Subtracting the conjugate of (4.2) from (4.1) leads

$$\sigma_{yy} - i\sigma_{xy} = \varphi'(z) + \overline{\varphi'(z)} + z\overline{\varphi''(z)} + \overline{\psi'(z)}.$$
(4.3)

It is noted that $\operatorname{Re}(z) = \operatorname{Re}(\overline{z})$ and $\operatorname{Im}(z) = -\operatorname{Im}(\overline{z})$. Thus, we get the stress field from (4.2) and (4.3) as follows

$$\sigma_{xx} = \operatorname{Re}\left[\varphi'(z) + \overline{\varphi'(z)} - z\overline{\varphi''(z)} - \overline{\psi'(z)}\right], \qquad (4.4)$$

$$\sigma_{yy} = \operatorname{Re}\left[\varphi'(z) + \overline{\varphi'(z)} + z\overline{\varphi''(z)} + \overline{\psi'(z)}\right], \qquad (4.4)$$

$$\sigma_{xy} = \operatorname{Im}\left[\varphi'(z) + \overline{\varphi'(z)} - z\overline{\varphi''(z)} - \overline{\psi'(z)}\right].$$

Let's express $\psi(z)$ in terms of $\varphi(z)$ and X(z) by using (2.33).

.

$$\psi_1'(z) = \frac{d}{dz} \left[-\overline{X}(z) + \overline{\varphi_1}(z) - z\varphi_1'(z) \right]$$

$$= \overline{\varphi_1'}(z) - \varphi_1'(z) - z\varphi_1''(z) - \overline{X'}(z).$$
(4.5)

Taking conjugate of (4.5) gives

..

$$\overline{\psi_1'(z)} = \varphi_1'(\overline{z}) - \overline{\varphi_1'(z)} - \overline{z}\overline{\varphi_1''(z)} - X'(\overline{z}).$$
(4.6)

Substituting (4.5) and (4.6) into (4.4), we get the stress fields in region 1 as follows

à

$$\sigma_{xx} = \operatorname{Re}\left[\varphi_{1}'(z) - \varphi_{1}'(\bar{z}) + 2\overline{\varphi_{1}'(z)} + (\bar{z} - z)\overline{\varphi_{1}''(z)} + X'(\bar{z})\right], \qquad (4.7)$$

$$\sigma_{yy} = \operatorname{Re}\left[\varphi_{1}'(z) + \varphi_{1}'(\bar{z}) + [z - \bar{z}]\overline{\varphi_{1}''(z)} - X'(\bar{z})\right], \qquad (4.7)$$

$$\sigma_{xy} = \operatorname{Im}\left[\varphi_{1}'(z) - \varphi_{1}'(\bar{z}) + 2\overline{\varphi_{1}'(z)} + (\bar{z} - z)\overline{\varphi_{1}''(z)} + X'(\bar{z})\right].$$

64

Considering leading order terms of each stress potential shown on (4.7), the explicit expressions of the stress fields in the neighbourhood of the crack tips a are given as (See Appendix 6 for detail derivation)

$$\sigma_{yy} = \frac{1}{\sqrt{r_1}\sqrt{2l}} \left(-\frac{5}{8} \sin\frac{\theta_1}{2} + \frac{1}{8} \sin\frac{5\theta_1}{2} \right) \left[\sum_{k=-\infty}^{\infty} kf_k a^{k-1} \right] + O(r_1), \quad (4.8)$$

$$\sigma_{xx} = \frac{1}{\sqrt{r_1}\sqrt{2l}} \left(-\frac{3}{8} \sin\frac{\theta_1}{2} - \frac{1}{8} \sin\frac{5\theta_1}{2} \right) \left[\sum_{k=-\infty}^{\infty} kf_k a^{k-1} \right] + O(r_1), \quad (4.8)$$

$$\sigma_{xy} = \frac{1}{\sqrt{r_1}\sqrt{2l}} \left(-\frac{1}{8} \cos\frac{\theta_1}{2} + \frac{1}{8} \cos\frac{5\theta_1}{2} \right) \left[\sum_{k=-\infty}^{\infty} kf_k a^{k-1} \right] + O(r_1).$$

where 2l = b - a and $(z - a) = r_1 e^{i\theta_1} \ (0 \le \theta_1 \le 2\pi)$.

Also, the stresses near crack tip b are determined in the following forms

$$\sigma_{yy} = \frac{1}{\sqrt{r_2}\sqrt{2l}} \left(\frac{5}{8} \cos \frac{\theta_2}{2} - \frac{1}{8} \cos \frac{5\theta_2}{2} \right) \left[\sum_{k=-\infty}^{\infty} kf_k b^{k-1} \right] + O(r_2), \quad (4.9)$$

$$\sigma_{xx} = \frac{1}{\sqrt{r_2}\sqrt{2l}} \left(\frac{3}{8} \cos \frac{\theta_2}{2} + \frac{1}{8} \cos \frac{5\theta_2}{2} \right) \left[\sum_{k=-\infty}^{\infty} kf_k b^{k-1} \right] + O(r_2), \quad (4.9)$$

$$\sigma_{xy} = \frac{1}{\sqrt{r_2}\sqrt{2l}} \left(-\frac{1}{8} \sin \frac{\theta_2}{2} + \frac{1}{8} \sin \frac{5\theta_2}{2} \right) \left[\sum_{k=-\infty}^{\infty} kf_k b^{k-1} \right] + O(r_2).$$

where b - a = 2l and $(z - b) = r_2 e^{i\theta_2}$ $(-\pi \le \theta_2 \le \pi)$.

..

It should be noted that the remote loading term is not shown in (4.8) and (4.9) since the load is not directly applied to the crack in the matrix but transferred to the domain through imperfect interface Γ_1 which alters the effect of the load on the crack. The undetermined coefficients f_k play a dominant role in transferring the load from a infinite point to the crack tips through the interface and the square bracketed terms represent the influence of the imperfect bonding condition.

4.3 Stress Intensity Factor

The normalized Mode I stress intensity factor in the vicinity of tips is defined as

$$\frac{K_I}{K_{I \ w/o \ inclusion}},\tag{4.10}$$

where $K_I = \sigma_{yy} \sqrt{2\pi r}$ and $K_I w/o inclusion = \sigma_{\infty} \sqrt{\pi l}$. K_I represents Mode I stress intensity factor in the intermediate matrix zone under uni-axial loading while $K_I w/o$ inclusion denotes the Mode I SIF for the same crack in the homogeneous matrix without an inclusion. It shows the pertubation in stress fields caused by mechanical properties of an inclusion, matrix, composite phase and imperfect interfaces.

In view of (4.8), the normalized SIF at tip a where θ_1 is π is

$$\begin{bmatrix} \frac{K_I}{K_I \ w/o \ inclusion} \end{bmatrix}_{at \ a}$$
(4.11)
= $\frac{\sqrt{2\pi r}}{\sigma_{\infty}\sqrt{\pi l}} \left\{ \frac{1}{\sqrt{r_1}\sqrt{2l}} \left(-\frac{5}{8} \sin \frac{\theta_1}{2} + \frac{1}{8} \sin \frac{5\theta_1}{2} \right) \left(\sum_{k=-\infty}^{\infty} k f_k z^{k-1} \right) + O\left(r_1^0\right) \right\}_{\theta_1 = \pi}$
= $\frac{1}{\sqrt{l}} \left\{ \frac{1}{\sqrt{l}} \left(-\frac{5}{8} + \frac{1}{8} \right) \left(\sum_{k=-\infty}^{\infty} k f_k z^{k-1} \right) + O\left(r_1\right) \right\}$
= $-\frac{1}{2l} \left(-3f_{-3}a^{-4} - 2f_{-2}a^{-3} - f_{-1}a^{-2} + f_1 + 2f_2a + 3f_3a^2 + 4f_4a^3 + ... \right) + O\left(r_1\right).$

In the similar way, the normalized SIF at tip b where θ_2 is 0 is

$$\left[\frac{K_{I}}{K_{I \ w/o \ inclusion}}\right]_{at \ b} \qquad (4.12)$$

$$= \frac{\sqrt{2\pi r}}{\sigma_{\infty}\sqrt{\pi l}} \left\{ \frac{1}{\sqrt{r_{2}}\sqrt{2l}} \left(\frac{5}{8} \cos \frac{\theta_{2}}{2} - \frac{1}{8} \cos \frac{\theta_{2}}{2} \right) \left(\sum_{k=-\infty}^{\infty} k f_{k} z^{k-1} \right) + O\left(r_{2}^{0}\right) \right\}_{\theta_{2}=0}$$

$$= \frac{1}{\sqrt{l}} \left\{ \frac{1}{\sqrt{l}} \left(\frac{5}{8} - \frac{1}{8} \right) \left(\sum_{k=-\infty}^{\infty} k f_{k} z^{k-1} \right) + O\left(r_{2}^{0}\right) \right\}$$

$$= \frac{1}{2l} \left(-3f_{-3}b^{-4} - 2f_{-2}b^{-3} - f_{-1}b^{-2} + f_1 + 2f_2b + 3f_3b^2 + 4f_4b^3 + \ldots \right) + O(r_2).$$

4.4 Numerical Results

4.4.1 Verification of Formulae

For convenience, the imperfect interface parameters m and n introduced in Chapter 2 are expressed in the terms of new dimensionless interface parameter M, defined by $M_0 = \frac{m_0+n_0}{2\mu_1}R_0$ on Γ_0 and $M_1 = \frac{m_1+n_1}{2\mu_1}R_1$ on Γ_0 . As mentioned before, these parameters represent the effectiveness of the bonding at the interface in transferring load thru interface. A small value of M or m (say, 0.01) represents debonding between adjoining materials while a large value of M or m (say, 100) simulates the perfect bonding condition. The varying values of M, n and m from 0.01 to 100 correspond to intermediate states between perfect bonding and complete debonding. These figures just represent the relative degree of bonding deterioration at the interfaces, not absolute.

The comparison method is to simulate the three phase perfect interface condition with a radial crack between the two interfaces as evaluated by Luo and Chen [48]. Taking M_0 and M_1 equal to 100, it is shown that the current formulation is in good agreement with those of Luo (see Figure 4.1). There are small differences in the actual values around interfaces and the differences seem to be from the number of unknown coefficients taken in the series expansions. Therefore, the good agreement with published results of crack behaviour confirms the validity of the equations. Expansion of the terms from z^{-4} to z^5 represents the minimum of terms in this analysis. Addition of more terms makes the numerical results converge to the exact solution addressed by Luo and Chen.

4.4.2 Influence of Imperfect Interface $\Gamma_0 \& \Gamma_1$

The effect of an imperfect bonding condition at the inclusion-matrix and matrixcomposite interfaces on SIFs at crack tips is evaluated for stiffer inclusion and softer than the matrix phase composites (Figure 4.2 and Figure 4.3). With respect to the GSC scheme, the softer composite phase results in mass damage in the surrounding matrix and inclusions outside the RVE. This case also represents the bonding medium that is stiffer than the matrix. In the case of a perfect bonding at the compositematrix interface $(M_1=100)$, the remote load is assumed to be fully transferred into the matrix and the inclusion-matrix interface. In Figure 4.2, it is clearly shown that the greater the degradation of the inclusion-matrix bonding Γ_0 , the higher the stress intensity factor at the crack tips in the vicinity of the interface; hence, the degeneration of the inclusion-matrix interface fixation accelerates propagation from both crack tips adjacent to the interface in both directions. It shall be noted that the influence of the deterioration plays a dominant role in increasing SIFs, not only near tip a but also distant tip b. Cracks located around the interface Γ_0 easily propagate in both directions, deteriorating the integrity of the interface fixation, resulting in increased degradation, which accelerates the cracking process further. Another interesting phenomenon is the existence of a stable zone associated with debonding in which the SIFs corresponding to debonding are lower than those of the initial perfect bonding condition. As the distance between a crack and the inclusion-matrix interface increases, the SIFs corresponding to the imperfect interfaces decrease up to a certain crack location and increase beyond the minimum point. The SIFs associated with the degraded bonding are lower within the zone and higher beyond this zone than those with perfect bonding. In addition, within the zone, the debonding of the matrix-inclusion interface suppresses matrix cracking. In the vicinity of the

matrix-composite interface, the influence of an imperfect matrix-inclusion interface increases SIFs at crack tip a and suppresses crack propagation at crack tip b. At the initial adhesion, when both interfaces are perfect, the stiff inclusion repels crack propagation from tip a and b around the inclusion-matrix interface and from tip aaround the matrix-composite. It increases SIFs at crack tip b as the crack approaches the matrix-composite phase interface.

With respect to the influence of the composite-matrix interface, as the fixation at the composite-matrix interface Γ_1 deteriorates, the stable zone associated with debonding becomes wider and deeper, and the influence of both interfaces diminishes, since the load transferred into the matrix thru the matrix-composite interface decreases, Figure 4.3. The outer interface does not directly affect the crack propagation at tip a and b, but indirectly shifts the SIFs downward in conjunction with the influence of the inner matrix-inclusion interface. SIFs at both crack tips go to zero, regardless of crack locations, as the matrix becomes totally debonded $(M_1=0.01)$ from the composite phase, indicating no load is being transferred thru this interface (Figure 4.4). The matrix-inclusion bonding condition is more critical for matrix cracking than matrix-composite adhesion quality. Liu and Kim addressed how imperfect bonding plays a dominant role in cracking outside an inclusion [27, 39]. Based on this work, it can be concluded that loosened matrix-composite phase fixation affects crack propagation outside the matrix, resulting in cracking or damage in the composite phase, namely altering the influence of neighbouring inclusions and the matrix. This type of cracking in the composite phase will be investigated in a future study.

4.4.3 Influence of Inclusion Stiffness μ_0 on Cracking

Figure 4.5 shows the influence of inclusion stiffness and bonding at the matrixinclusion interface on SIFs of a crack fixed at $d/R_0=0.4$ for a soft composite phase. SIFs at both crack tips dramatically increase when the matrix-inclusion interface fixation becomes imperfect, M_0 close to zero. Stiffer inclusions lead to lower SIFs than compliant inclusions, although SIFs corresponding to stiff inclusions are still greater than 1. This phenomenon agrees with reported results that stiff inclusions repel cracks from the interface [27, 39]. Increasing inclusion stiffness beyond 25 times stiffer than the matrix, does not have a significant impact on suppressing crack propagation. Thus, determining proper stiffness of inclusions, up to the critical ratio of inclusion to matrix stiffness, is recommended to discourage crack propagation. In cases involving a stiffer composite phase -2 times harder than the matrix-the overall behaviour of the SIFs are similar, however, their values are less than 1. Thus, a stiffer composite phase repels crack propagation in the matrix; keeping the composite phase harder than the matrix is the best way to suppress crack propagation. When the outer interface degenerates, all of the SIFs shift downward. This phenomenon is illustrated in Figure 4.3.

4.4.4 Influence of Interphase Stiffness μ_1 on Cracking

Figure 4.6 presents the effect of matrix stiffness and degeneration of the matrixinclusion fixation on cracking in the matrix. Similar to the results realized for inclusion stiffness, the deterioration of matrix-inclusion bonding accelerates matrix cracking in the vicinity of the interface. Contrary to the case of inclusion stiffness, the normalized SIF values increase as the stiffness of the matrix increases. Stiff matrix accelerates the cracking process in the matrix, while soft matrix suppresses crack propagation. Comparing the magnitude of the changes in SIFs in Figure 4.6 with those in Figure 4.5, it can be seen that the ratio of matrix stiffness to the composite phase is more critical than that of the ratio between inclusions and the matrix. However, it should be noted that the soft matrix suppresses matrix cracking on itself, but may cause damage to the neighbouring matrix and inclusions in the vicinity of the matrix-composite phase interface. Thus, the advantages and disadvantages of the soft matrix properties should be weighed with regard to the total damage experienced by the inclusion-matrixcomposite phase system. SIFs corresponding to matrix softer than the composite phase shift downward much more readily than SIFs corresponding to hard matrix as the matrix-composite interface starts to loosen. The effect of the loosening of the outer interface is similar to the effects mentioned above; however, its influence on a soft matrix is stronger than on a hard matrix.

4.4.5 Influence of Crack Length on Cracking

Interactions between imperfect matrix-inclusion bonding, SIFs and the length of a crack fixed in the vicinity of the matrix-inclusion interface $(d/R_0=0.4)$ are presented in Figure 4.7. SIFs at crack tip *a* dramatically increase when fixation is highly degenerated: imperfection parameter M_0 goes to zero. Under poor bonding conditions, namely for $M_0 < 5$, the influence of the crack length on the propagation at tip *a* is not significant. However, with $M_0 > 10$, the longer crack results in higher SIFs at tip *a* than the shorter crack; therefore, under certain conditions involving adhesion damage, longer cracks are more vulnerable to cracking than shorter cracks at tip *a*. On the contrary, the behaviour of crack tip *b* is quite different from cracks at tip *a*. A Shorter crack's SIFs at tip *b* corresponding to highly imperfect bonding, $M_0 < 5$, are dramatically higher than those of a longer crack. The location of a longer

crack's tip b is further away from the interface Γ_0 than that of a shorter crack and the degeneration at the interface has less influence on a longer crack's tip b. Microcracks are more vulnerable to cracking at tip b. With $M_0 > 10$, the effect of crack length on the propagation at tip b diminishes. Consequently, as the initial perfect bonding is damaged, long cracks are more vulnerable to propagating at tip a than short cracks, the propagation would cause further damage to the interface, and beyond certain damage, namely $M_0 < 5$, microcracks become more susceptible to proceeding from tip b than long cracks.

4.4.6 The Stable Equilibrium Position

The stable equilibrium position, called the trapping mechanism by Dundurs and Mura [13], is confirmed in this present work. As Luo and Chen [49] mentioned, the stable equilibrium position is more likely to occur in a three-phase model than a two-phase model and is dependent on material properties such as those addressed by Poisson's ratio. Luo and Chen explain that when an inclusion and a composite phase are stiffer than the matrix ($\mu_0 = \mu_2 = 1.25 * \mu_1$) with $v_0 = v_2 = 0.2 \& v_1 = 0.4$, the dislocation does not have a stable equilibrium position. A simulation of this case was conducted as part of this study and the findings agree with their results. Given the values of Poisson's ratio and stiffness, all the SIFs at tip *a* are greater than 1, which means that there is no stable equilibrium position. Around tip *b*, all of the SIFs are between 0.94 and 1.06. With the assumption that interfaces are perfectly bonded, Luo and Chen addressed how stable equilibrium positions exist for any combination of Poisson's ratio as in the previous case, SIFs at tip *a* corresponding to relatively good bonding quality ($M_0 = 100 \& 10$) and SIFs at tip *b* are less than 1, Figure 4.8.

These results show that there is a stable equilibrium position, which agrees with Luo and Chen's analysis. It is noted that the loosening of the inner interface is likely to decrease the range of the stable equilibrium position ($M_0 = 0.01$ to 1). When $\mu_0 = \mu_2 = 3\mu_1$, all the SIFs at tip *a* and *b* are less than 1. Thus, the interface damage and Poisson's ratio affect the trapping mechanism, but the stiffness of the composite phase and inclusion play a significant role in crack propagation. Luo and Chen did not identify the location of the stable equilibrium position within the intermediate region. However, in this study, the actual locations of the position are shown to be along the real axis.

4.4.7 The Stable Zone

Figures 4.9 and 4.10 show the effect of a stiffer composite phase on the stable zone associated with debonding. This new phenomenon called The Stable Zone is introduced in section 4.4.2. All of the conditions in Figure 4.9 and Figure 4.10 are the same as those in Figure 4.2 and 4.3, except for the stiffness of the composite phase. Compared to Figure 4.2 and 4.3, the hard composite phase definitely suppresses the SIFs at tip *a* for both M_1 =100 and M_1 =5. The stable zone near tip *a* with a compliant composite phase, as shown in Figure 4.2, does not take place for the stiffer composite phase shown in Figure 4.9. Regarding crack tip *b*, the stable zone does not take place with the soft composite phase (Figure 4.2). In Figure 4.9, shifting all the SIFs at tip *b* downward, the stiffer composite phase causes a stable zone to occur again. However, once the adhesion on the outer interface deteriorates (M_1 : from 100 to 5), the stable zone goes away (Figure 4.10). Figure 4.10 shows that the stiffer composite phase changes the stable zone, resulting in a more narrow zone for M_1 =5. It is important to note that the stiffer composite phase decreases or removes the effect of the zone and the debonding of the matrix-composite diminishes the effect of a stiff composite and causes the stable zone associated with debonding. Both the stiffness and the outer bonding condition M_1 affect the existence and shape of the stable zone. Consequently, the outer interface bonding and the stiffness of the outer composite phase are leading factors in the development of a stable zone associated with debonding.

••

:



Figure 4.1 Comparison with Luo's results with perfect interfaces along Γ_0 and Γ_1 .



Figure 4.2 Influence of imperfect Γ_0 with perfect Γ_1 ($M_1 = 100$). Normalized SIFs of crack tip "a" & "b" vs. crack locations. $\mu_0 = 2\mu_1, \mu_2 = 0.5\mu_1$.

÷



Figure 4.3 Influence of imperfect Γ_0 with imperfect Γ_1 ($M_1 = 5$). Normalized SIFs of crack tip "a" & "b" vs. crack locations. $\mu_0 = 2\mu_1, \mu_2 = 0.5\mu_1$.

÷



Figure 4.4 Debonding $(M_1 = 0.01)$





••







Figure 4.7 Influence of the crack length. Normalized SIFs of crack tip "a" vs. Γ_0 imperfect parameter M_0 at the fixed crack location. $\mu_0 = 2\mu_1$, $\mu_2 = 0.5\mu_1$ and

 $d/R_0 = 0.4.$



Figure 4.8 Influence of imperfect Γ_0 with perfect Γ_1 ($M_1 = 100$). Normalized SIFs of crack tip "a" & "b" vs. Crack locations. $\mu_0 = \mu_2 = \sqrt{3}\mu_1$.



Figure 4.9 Influence of stiffer composite phase on the stable zone $(M_1 = 100)$.

 $\mu_0 = 2\mu_1, \, \mu_2 = 2\mu_1.$



:

4

Figure 4.10 Influence of stiffer composite phase on the stable zone (M1 = 5).

 $\mu_0 = 2\mu_1, \, \mu_2 = 2\mu_1.$

CHAPTER 5

An Example of Application - Bone/Implant Modeling

5.1 Introduction

According to the National Hospital Discharge Survey of 2003, approximately 217,000 and 402,000 patients in the U.S. underwent Total Hip Replacement (THR) and Total Knee Replacement (TKR) operations, respectively, and \$24.7 billion dollars were spent in hospitalization related to these replacement surgeries. In addition, there were 36,000 revision hip replacements and 33,000 revision knee replacements. However, the revision rate of joint replacement is greater for those who are heavier or more active, such as young patients. Many surgeons and biomechanical scientists have asked the question whether femoral/tibial component fixation of total hip/knee arthroplasty shows better performance with or without bone cement and the answer is still quite controversial. With respect to fixation, performance of cemented prostheses was reported to be better than that of cementless protheses [58]. However, some researchers have demonstrated no differences between these two techniques [59]. Currently, the cemented surgery is more widely performed and several mechanisms of debonding of cemented implants from bone have been suggested.

Biologically, wear particles and debris from bone cement and implants spread into tissue within the vicinity of the interface and provoke osteolysis. Another important . factor leading to debonding is the mechanical properties of cement and cementing techniques used during surgery. Although the dramatic improvements made in cementing techniques from the earlier hand-packed cementing to centrifugation and vacuum mixing have decreased the revision surgery rate [60, 61], poor mechanical properties of the bone cement and inadequate interdigitation adjacent to the bone cement interface are still being addressed as primary failure factors initiating the aseptic loosening of cemented prosthetics [62, 63, 64]. In addition, variables such as a surgeon's experience, patient weight and implant type are also implicated as factors leading to poor mechanical properties in the mantle [65]. Through such a multifunctional failure mechanism, the majority of popular cemented implants continue to migrate over time even though the fixation is perfect at the initial stage [58] and the loosening in the cement mantle is believed inevitable.

Interdigitation between bone and cement in well-fixed cemented implants keeps its mechanical interlocking long after implantation [66]. Poor interdigitation or permeation at the cement-bone interface [67], and deteriorated resistance at the cementimplant boundary, result in imperfect interface bonding. However, from the standpoint of sliding motion, the strong interdigitation between the cement and cancellous bone prevents relative slip at the interface and provides strong mechanical interlocking, which is referred to as a no-slip interface. The no-slip bonding enables shear loads to be sustained along the interface even when tractions fail in the normal direction. This no-slip condition is of considerable practical interest and is suggested as an adequate model of cement-bone interface [68, 69]. In contrast, implant-cement interface is either smooth for polished implants or mechanically interlocked for rough surface implants. The rough surface implants were introduced to provide mechanical interlocking over the whole interface and prevent the prosthesis from subsiding. It was reported that rough implants increased friction between cement and an implant and reduced its subsidence and the effects of debonding at the interface [70]. It is noted that this mechanical interlocking is not as strong as interlocking on a cement-bone interface. In contrast to the longevity of cement-bone interdigitation, mechanical interlocking between a rough stem and cement vanishes once loosening of the stem

progresses.

Of the many factors causing aseptic loosening, mechanical loading is the most commonly accepted as the leading failure factor [71, 72]. Repeated mechanical loads, in conjunction with other factors mentioned earlier, have resulted in the continuing migration in the cement mantle and finally loosening of prosthetic components. This multifunctional mechanism has attracted the attention of many researchers. The aseptic loosening of cemented implants is most frequently a long-term complication [73, 74]. To evaluate the influence of the loads, researchers have investigated the stress field in the region adjacent to the implant and cracking at the cement-bone and cement-implant interfaces under cyclic loads.

With finite element analysis, interactions and relationships between stress magnitude in the cement mantle and failure have been reported [75]. However, experimental investigations have addressed how the predicted magnitude of the peak stress is not a dominant factor leading to clinical failure [76]. Damage accumulation theory was suggested as a failure scenario of joint replacement [77]. The propagation process of cracks in the cement mantle was investigated experimentally, and pre-load and loadinitiated cracks have been reported to play a determinant role in progress of failure [78]. Furthermore, it has been demonstrated that failure of the cement mantle caused by radial cracks in the mantle initiated loosening of cemented implants [79] and many radial cracks generated adjacent to the cement-implant interface were found by SEM micrographs after stem failure [70]. Also, the bonding conditions of cement-bone and cement-implant interfaces are believed to affect aseptic loosening. Deterioration of the interface condition between cancellous bone and implants is considered to be a significant failure mechanism, which results in mechanical fracture and debonding of the prosthesis from the bone [80, 81, 82].

The interface between an implant and the cement has also been attracting the attention of researchers. Different surface finishing techniques, designed to improve the strength and fixation at the interface, have been suggested [83]. Repeated cyclic stress imposed on artificial implants results in loosening between the prosthesis and the bone, and the development of load-initiated cracks in the vicinity of the cementbone interface [78]. Debonding and resistance of the cement-implant interface was investigated with respect to residual stress developed during cement curing [84]. A theoretical mechanical failure process of metal-cement interface, with respect to the cohesive zone, was proposed [85]. Whether the cracks and debonding begin first from the cement-implant interface or from the cement mantle has not yet been clearly determined [86, 87, 88]. Gharpuray developed a hypothesis that the initial cracks were emanating from the perfectly bonded interfaces, and consequently, they accelerated degradation of the prosthesis fixation, micro-motions between bone and implant, and bone resorption around the implant [71]. Gharpuray proposed a theoretical model to show cracking around a void and a circular elastic inclusion in the vicinity of the cement-implant interface and demonstrated how the cracking caused loosening of the fixation. A majority of this research has assumed perfect bonding in which displacement compatibility at the cement-implant and cement bone interface holds. However, with respect to the unavoidable degradation of the fixation, this study demonstrates that the imperfectness of the interface resistance should be considered and the inevitable interaction between deterioration in the interface bonding quality and crack behaviours within the cement mantle, or local region adjacent to the interfaces, causes catastrophic failure of joint replacement arthroplasty.

As addressed in Chapter 1, imperfect bonding between two materials with different mechanical properties has been known to be a dominant factor affecting crack propagation located outside the inclusion in the vicinity of the interface [27]. Furthermore, this thesis provides a general three-phase model, which considers imperfect bonding conditions along both interfaces. Thus, the proposed model is well suited to the bone cement cracking interacting with an implant and the bone matrix under damaged adhesion conditions along both boundaries. This research provides, for the first time, analytical rationale for the possible failure mechanism of a cemented implant system in the full range of bone cement. This study addresses how the imperfectness of both the cement-bone and cement-implant interface simultaneously affects radial crack propagation in bone cement during the mechanical cracking process. Specifically, a no-slip condition on cement-bone and cement-stem interfaces are taken into account. The procedures addressed in the previous chapters are applied to this specific application. Thus, the imperfectness of the interfaces is modeled with interface parameters and the effect of the parameters on SIFs at crack tips is evaluated along a radial crack location from implant-cement interface to cement-bone interface.

5.2 Formulation

Figure 5.1 shows a simplified model of cemented joint arthroplasty. We consider a domain of \mathbb{R}^2 , infinite in extent, containing an implant, bone cement PMMA and bone matrix with a radial crack in cement. The circular implant, with center at the origin of the coordinate system and radius of R_0 , occupies a region denoted by the domain S_0 and the cement is also modeled as a circle with radius of R_1 represented by the domain S_1 . The surrounding bone matrix is represented by S_2 . All the materials are assumed to be homogeneous and isotropic with shear moduli μ_0 , μ_1 and μ_2 , respectively. Interfaces Γ_0 and Γ_1 are imperfect and represent implant-cement and cement-bone interface, respectively and the adhesion imperfectness in the radial and

the tangential direction is represented by the non-negative interface parameter m and n, respectively. All the notations, symbols and boundary conditions are same as ones specified in Chapter 2.

5.3 Discussion

The crack length is fixed as $l = R_0$ and the other mechanical properties used during numerical calculation are shown in Table 5.1. Linear elastic fracture mechanics is applicable to brittle materials which do not have a considerable plastic zone near the crack tips. PMMA, the most common bone cement material, is an amorphous brittle material with very high mechanical strength. Thus, our approach to applying linear elastic fracture mechanics to this brittle PMMA is reasonable. Although the bone cement contains some inclusions introduced during its formation, such as beads and voids which cause less isotropic and homogeneous properties locally, the influences of the inclusions on cement properties are small and can be ignored. Therefore, the bone cement is considered isotropic and homogeneous. While the cement-implant interface is clearly shown physically, interface between cement and bone can not be simply defined, due to interdigitation and permeation. However, the simple model of the interfaces as a single layer is enough to give us insight into the cracking process initiated by imperfect bonding. Uniaxial compression is the most likely loading in bone cement; however, uniaxial tension is occasionally applied to bone cement. As addressed by Gharpuray, the uniaxial compression case can also be obtained as the negative of the tension value [71].

The result of the no-slip conditions on both interfaces is shown on Figure 5.2. These conditions can be achieved by using rough surface implants. No-slip interfaces dramatically reduce the effects of loosening in the normal direction on SIFs at the crack tips. Tractions in the tangential direction transfer to the cement and then to an implant through interfaces, and compensates for reduction of tractions in the normal direction. Thus, imperfect bonding in the normal direction has little influence on the SIFs at the crack tips. SIFs at tip a continuously increase as the crack approaches the outer interface. Since the outer bone matrix is much more compliant than cement, it attracts crack propagation and increases stress intensity in the vicinity of its interface. Similarly, SIFs at tip b have a maximum value near the outer bone cement interface. It is noticed that the stiff implant - 26 times the cement shear modulus - does not have a noticeable effect on the SIFs at both tips in the region adjacent to the cement-implant interface, while the change in shear modulus across the cement-bone interface dramatically alters the magnitude of SIFs near the outer interface. The cement-bone interface is the first barrier which transforms the tractions initiated from remote loading conditions and has greater influence on crack behaviour than the cement-implant interface. Preventing slip on the cement-implant interface results in very stable behaviour of tip b except at the region adjacent to the cement-bone interface.

Figure 5.3 shows crack behaviour with no-slip and imperfect bonding in the normal direction on the cement-bone interface. This is a case in which a polished implant has been embedded into cement and the cement is interdigitated with cancellous bone. A rough implant can be modeled with this type of debonding, when bonding at the cement-implant interface deteriorates and mechanical interlocking in tangential direction is no longer sustained. The imperfect cement-implant interface attracts propagation at tip a and accelerates opening of crack tip b in the vicinity of the cement-implant interface. As for tip a behaviour, debonding at the cement-implant interface accelerates opening of the cement-implant interface effect of the debonding decreases, since the crack is far from the interface. Soft bone

matrix attracts crack propagation at tip a as long as the cement-implant bonding is strong enough, and attracts propagation of cracks at tip b near the cement-bone interface at the initial perfect bonding. SIFs increase up to a critical point around the cement-bone interface as the degeneration of cement-implant bonding worsens. Debonding at the cement-implant interface results in failure of load transfer from the remote load to the implant. Thus, the SIFs at the tip close to the cement-bone interface dramatically decrease beyond the critical point. The interesting behaviours at tip b are the existence of a suppression region where the normalized SIF is below 1. At the initial stage, when integrity of the bonding is perfect, there is a stable region for crack tip b which disappears as the implant-cement interface starts to loosen. Thus, debonding at the cement-implant interface accelerates crack propagation at tip b and promotes crack opening at both tips in the region adjacent to the cement-implant interface.

While the cement-cancellous bone interface has strong interdigitation and no-slip bonding, mechanical interlocking of the cement-implant interface in the tangential direction is weaker than interlocking of the cement-cancellous interface. Figure 5.4 shows the effect of the weaker, no-slip cement-implant interface on crack behaviour. Parameter n in the tangential direction on the cement-implant interface is reduced to 10 to simulate the weakened no-slip bonding condition. The result shows that this is an intermediate stage between Figure 5.2 and 5.3. While cement-implant bonding in the normal direction degrades from 100 to 0.01, the constant bonding parameter n in the tangential direction continues to carry loads and reduces the effect of the degeneration of bonding. Overall SIFs are lower than those in Figure 5.3.

The behaviours of crack tips near both interfaces are of considerable practical interest, since crack propagation in the vicinity of the interfaces accelerates degradation of interface bonding. The existence of mechanical interlocking in the tangential direction on the cement-implant interface reduces SIFs within the region adjacent to the interfaces. Specifically, it repels crack propagation near the cement-implant interface by decreasing the effect of imperfect bonding. One outstanding advantage of a no-slip cement-stem interface is that it prevents further progress of cement cracking from a good initial condition by maintaining the initial pattern of crack behaviour. Furthermore, from the standpoint of damage accumulation theory, a no-slip cement-stem interface suppresses overall SIFs in the cement mantle; hence, it decreases damage accumulation throughout the mantle. However, longevity and durability of a no-slip cement-implant interface is still questionable, since the rough implants do not provide strong mechanical interlocking and can result in breaking of the interdigitation, inappropriate permeation and breaking debris. Further design features to prevent slip on the interface are required.

In this present study, a titanium alloy implant is used during numerical calculations. Crack behaviours with stiffer alloys such as Cr/Co and stainless steel alloys produce similar results as those with titanium alloy. It is shown that changes in implant properties do not have a substantial influence on crack propagation. A similar result was reached by a study that noted how the elastic modulus of an implant does not alter the load transfer mechanism between the cement and the prosthesis, since the elastic modulus of the implant material is much higher than that of the cement and bone [79].

Material	Poisson's Ratio, ν	Shear Modulus(GPa), μ
Titanium	0.3	41
PMMA	0.3	1.54
Bone	0.3	0.13

Table 5.1 Properties of materials for orthopaedic arthroplasty



Figure 5.1 Three-phase stem-cement-bone system.



Figure 5.2 Normalized SIFs of crack tip "a" and "b" vs. Crack locations. No-slip imperfect stem-cement/ cement-bone bonding $(m_1 = 10)$



Figure 5.3 Normalized SIFs of crack tip "a" and "b" vs. Crack locations. No-slip imperfect cement-bone bonding $(m_1 = 10)$.



Figure 5.4 Normalized SIFs of crack tip "a" and "b" vs. Crack locations. Weakened no-slip imperfect stem-cement; No-slip imperfect cement-bone bonding.

CHAPTER 6 Conclusions and Future Works

6.1 Conclusions

The most commonly used model of classical micromechanics is a two-phase single inclusion surrounded by an infinite elastic medium, called matrix phase, with perfect interface. This simple model has been providing insight into the interaction between cracking, or fracture mechanism, and an inclusion. However, since the presence of the interphase layer has been identified as a major parameter in altering stress fields around the inclusion, the study of the three-phase model, in which the interphase layer is considered, has become of practical and theoretical interest. Also, the threephase model is quite general and is applicable to the generalized self-consistent and implant/cement/bone system. Furthermore, many researchers have shown that perturbation in the stress fields around the inclusion is caused by interface bonding conditions, which draws attention to the importance of the imperfect bonding. They have also addressed how there are significant differences between previous perfect bonding and imperfect bonding in micromechanics. There are a few studies which consider the three-phase model with imperfect bonding along one interface. However, due to the cumbersome mathematical problems involved, there has been no research in the literature to address the imperfectness of both interfaces using the three-phase model.

In this study, a semi-analytic solution to the interaction between a pre-exiting crack and a three-phase inclusion with imperfect bonding along both boundaries is presented. For the analytic approach, linear elasticity with complex variable techniques and linear elastic fracture mechanics are applied to the system. Also, to
overcome the mathematical difficulties regarding imperfect interfaces, analytical continuations and series methods are employed. With the fundamental theories applied to the three-phase single inclusion system, the problem reduces to a boundary value problem. By solving 38 coupled linear algebraic equations numerically, unknown coefficients of stress potentials representing stress fields near crack tips have been determined. The coefficients reflect the effects of damaged interfaces, Poisson's ratio and material stiffness on SIFs at each of the radial crack tips. An interpretation of the numerical results is provided in Chapter 4 and an analytical solution to the implant/bone/cement system is presented as an application example in Chapter 5. For verification of the formulae derived in this dissertation, comparisons are drawn with corresponding cases published in the literature.

New findings established through the numerical analysis results presented in Chapter 4 are as follows:

-. The proper stiffness of fibres as an inclusion must be determined. Benefit is seen up to a critical ratio of fibre to the matrix; increasing fibre stiffness beyond this critical ratio has little effect on crack propagation. Also, keeping the composite phase harder than the matrix is the best way to suppress crack propagation.

-. Outer interface degeneration shifts all of the SIFs at both crack tips downward.

^{*}-. Ratio of composite phase stiffness to matrix phase stiffness is more critical than the ratio for stiffness between inclusions and the matrix. SIFs corresponding to a softer than the matrix composite phase, shift downward more readily than those of a harder matrix, as the matrix-composite interface starts to loosen.

-. Regarding crack lengths, longer cracks under certain damaged adhesion conditions are more vulnerable to cracking than short cracks at near tip a, while microcracks are more vulnerable to cracking at tip b. Consequently, as the initial perfect bonding becomes damaged, long cracks are more vulnerable to propagating at tip a, and the propagation can result in further damage on the inner interface M_0 . Beyond certain damages, microcracks are more susceptible to proceeding from tip b.

-. The stable equilibrium position, called the trapping mechanism by Dundurs and Mura, is known to be affected by other factors, such as the interface condition, Poisson's ratios and stiffness. In their analysis, they demonstrated that the stable equilibrium position does not take place under certain conditions that lead to inner interface loosening. Thus, the damage of interface, Poisson's ratio and stiffness play a pronounced role in the trapping mechanism.

-. A new phenomenon, called the *stable zone*, which is associated with debonding, is introduced. In this zone, SIFs corresponding to debonding are lower than those of an initial perfect adhesion. This zone is affected by Poisson's ratio, material stiffness and crack location. The degradation of adhesion accelerates cracking overall; however, in a certain region, it depresses crack propagation. Stiffer composite phase decreases or removes the effect of the zone and the debonding of the matrix-composite diminishes the effect on the stable zone.

Thus, cracking at the matrix is affected by various factors such as degree of imperfect interface bonding, crack length, mechanical properties of the matrix and inclusion, and crack location. Of the parameters, the bonding condition along the inclusion and the matrix has a profound effect on the failure mechanism in the vicinity of the interface, regardless of other factors.

In addition, Chapter 5 addresses how the mechanical interlocking in the tangential direction on the cement-stem interface substantially suppresses crack development throughout the cement mantle and debonding on the interface accelerates cement cracking in the vicinity of the implant. The design of special stems with strong interdigitation, namely non-slip interface, is of practical importance. It is noted that changing implant materials does not have a substantial impact on crack propagation within the bone cement mantle.

This research contributes to the understanding of the interaction between matrix cracking and homogeneously imperfect interfaces, and introduces a new phenomenon, called the stable zone, associated with debonding. In addition, the interaction between cement crack propagation and the influence of non-slip implant-cement interface, namely rough implant interface, is explored and guidance for the design of implants is presented.

6.2 Future Works

This research deals with single inclusion embedded in an infinite homogeneous isotropic elastic medium that interacts with an external pre-existing crack and imperfect adhesion interfaces. To further the knowledge gained through this dissertation, the author anticipates the following future works:

-. The schematic study of the effect of inhomogeneous, imperfect bonding along both interfaces in the three-phase model. Homogeneous imperfection along the interfaces is an idealization of inhomogeneous bonding. Incorporating the inhomogeneous imperfect bonding case into the three-phase model will provide a more realistic solution. -. The schematic study of cracking within the outer infinite phase. Kim and Sudak [39] studied this matrix cracking in the three-phase model by employing inner imperfect interface only, which might underestimate the perturbation effect caused by the outer bonding condition. This future study, I believe, will provide helpful information regarding the bone damage process which is caused by damage to the implant/cement and cement/bone bonding.

.

-. The schematic study of interaction between two cracks within the intermediate matrix phase. In real life situations, there are many microcracks and voids in the intermediate layer, such as bone cements, and it is well known that the aggregation of the near cracks/voids causes catastrophic failures in implant arthroplasty and composite materials. The study of the behaviour of cracks interacting with each other under imperfect interface conditions will undoubtedly provide a foundation for exploring the relationships between crack aggregation and fracture mechanisms.

Hence, deploying parameters governing fracture mechanisms, such as mechanical properties (stiffness and Poisson's ratio) and physical conditions (bonding condition, crack length and multiple cracks) in a micromechanics analysis, and working to understand the fracture mechanism analytically, will lead to the development of a foundation through which failures in many engineering applications can be predicted and prevented.

..

References

- Callister, W.D., 2007, Materials Science and Engineering: An Introduction, 7th Edition, John Wiley & Sons.
- [2] Milton, Graeme Walter, 2001, Theory Of Composites, Cambridge University Press.
- [3] Daniel, I. M., Ishai, O., 2006, Engineering Mechanics Of Composite Materials, Oxford University Press.
- [4] Hashin, Z., 1983, "Analysis Of Composite Materials A Survey", ASME Journal of Applied Mechanics, 50, pp. 481-505.
- [5] Christensen, R. M., Lo, K. H., 1979, "Solution For Effective Shear Properties In Three Phase Sphere And Cylinder Models", Journal of Mechanics and Physics of Solids, 27, pp. 315-330.
- [6] Christensen, R. M., 1990 "A Critical Evaluation For A Class Of Micromechanics Models", Journal of Mechanics and Physics of Solids, 38, pp. 379-404.
- [7] Sudak, L.J., 2000, PhD Thesis: A Circular Inclusion With Circumferentially Inhomogeneous Imperfect Interface In Plane Elasticity, University of Alberta.
- [8] Tamate, 0., 1968, "The Effect Of A Circular Inclusion On The Stresses Around A Line Crack In A Sheet Under Tension", International Journal of Fracture Mechanics, 4, pp. 257-265.
- [9] Atkinson, C., 1972, "The Interaction Between A Crack And An Inclusion", International Journal of Engineering Science", 10, pp. 127-136.

- Anlas G., Santare M. H.,1993, "Arbitrarily Oriented Crack Inside An Elliptical Inclusion," ASME Journal of Applied Mechanics, 60, pp. 589-594.
- [11] Wu, C.H., Chen, C.H., 1991, "A Crack In A Confocal Elliptic Inhomogeneity Embedded In An Infinite Medium", ASME Journal of Applied Mechanics, 57, pp. 91-96.
- [12] Wu, L.Z., 2000, "A Crack In A Confocal Elliptic Piezoelectric Inhomogeneity Embedded In An Infinite Piezoelectric Medium", International Journal of Fracture, 104, pp. 1-14.
- [13] Dundurs, J., Mura, T., 1964, "Interaction Between an Edge Dislocation and a Circular Inclusion", Journal of the Mechanics and Physics of Solids, 12, pp. 177-189.
- [14] Dundurs, J., Sendeckyj, G. P., 1965, "Edge Dislocation Inside A Circular Inclusion", Journal of the Mechanics and Physics of Solids, 13, pp. 141-147.
- [15] Stagni, L., Lizzio, R., 1983, "Shape Effects In The Interaction Between An Edge Dislocation And An Elliptical Inhomogeneity", Applied Physics A: Materials Science & Processing, 30, pp. 217-221.
- [16] Erdogan, F., Gupta, G. D., Ratwani, M., 1974, "Interaction Between A Circular Inclusion And An Arbitrarily Oriented Crack", ASME Journal of Applied Mechanics, 41, pp. 1007-1013.
- [17] Eshelby, J. D., 1957, "The Determination Of The Elastic Field Of An Ellipsoidal Inclusion And Related Problems", Preceedings of the Royal Society of London, A241, pp. 376-396

- [18] Hashin, Z., 1990, "Thermoelastic Properties Of Fiber Composites With Imperfect Interface", Mechanics of Materials, 8, pp. 333-348.
- [19] Hashin, Z., 1991, "The Spherical Inclusion With Imperfect Interface", ASME Journal of Applied Mechanics, 58, pp. 444-449.
- [20] Achenbach, J. D., Zhu, H., 1989, "Effect Of Interfacial Zone On Mechanical Behavior And Failure Of Fiber-reinforced Composites", Journal of The Mechanics and Physics of Solids, 37, pp. 381-393.
- [21] Aboudi, J., 1987, "Damage In Composites Modeling Of Imperfect Bonding", Composite Science and Technology, 28, pp. 103-128.
- [22] Sudak, L.J., 2003, "On The Interaction Between A Dislocation And A Circular Inhomogeneity With Imperfect Interface In Antiplane Shear", Mechanics Research Communications, 30, pp. 53-59.
- [23] Stagni, L., 1999, "The Effect Of The Interface On The Interaction Of An Interior Edge Dislocation With An Elliptical Inhomogeneity", Zeitschrift fur Angewandte Mathematik und Physik, 50, pp. 327-337.
- [24] Ru, C. Q., Schiavone, P.,1997, "A Circular Inclusion With Circumferentially Inhomogeneous Interface In Antiplane Shear", Preceedings of the Royal Society of London, A453, pp. 2551-2572.
- [25] Sudak, L. J., Ru, C. Q., Schiavone, P., Mioduchowski, A., 1999, "A Circular Inclusion With Inhomogeneously Imperfect Interface In Plane Elasticity", Journal of Elasticity, 55, pp. 19-41.

- [26] Amenyah, W., and Schiavone, P., et al., 2001, "Interior Cracking Of A Circular Inclusion With Imperfect Interface Under Thermal Loading", Mathematics and Mechanics of Solids, 6, pp. 525-540.
- [27] Liu, Y., and Ru, C. Q., et al., 2001, "New Phenomena Concerning The Effect Of Imperfect Bonding On Radial Matrix Cracking In Fiber Composites", International Journal of Engineering Science, 39, pp. 2033-2050.
- [28] Benveniste, Y., Dvorak, G. J., Chen, T., 1989, "Stress Fields In Composites With Coated Inclusions", Mechanics of Materials, 7, pp. 305-317.
- [29] Cheeseman, B. A., Santare, M. H., 2001, "The Effect Of The Interphase On Crack Inclusion Interactions", International Journal of Fracture, 109, pp. 303-323.
- [30] Eshelby, J. D., 1959, "The Elastic Field Outside An Ellipsoidal Inclusion", Preceedings of the Royal Society of London, A252, pp. 561-569.
- [31] Aboudi, J., 1983, "The Effective Moduli Of Short-Fiber Composites", International Journal of Solids and Structures, 19, pp. 693-707.
- [32] Sendeckyj, G. P., 1970, "Elastic Inclusion Problems In Plane Elastostatics," International Journal of Solids and Structures, 6, pp. 1535-1543.
- [33] Theocaris, P.S., 1985, "The Unfolding Model For The Representation Of The Mesophase Layer In Composites", Journal of Polymer Science., 30, pp. 621-645.
- [34] Theocaris, P. S., Sideridis, E. P., Papanicolaou, G. C., 1985, "The Elastic Longitudinal Modulus And Poisson's Ratio Of Fiber Composites", Journal of Reinforced Plastics and Composites, 4, pp. 396-418.

- [35] Qaissaunee, M.T., Santare, M.H., 1995, "Edge Dislocation Interacting With An Elliptic Inclusion Surrounded By An Interfacial Zone", Quarterly Journal of Mechanics and Applied Mathematics, 48, pp. 465-482.
- [36] Piggot, M. R., Sandai, A., Chua, P. S., Andison, D., 1986, Mechanical Interactions In The Interphasial Region Of Fibre Reinforced Thermosets, Composite Interfaces, H.Ishida and J.L. Koeing Eds.
- [37] Bigoni, D., Serkov, S. K., Valentini, M., Movchan, A. B., 1998, "Asymptotic Models Of Dilute Composites With Imperfectly Bonded Inclusions", International Journal of Solids and Structures, 35, pp. 3239-3258.
- [38] Zhu, H., Achenbach, J. D., 1991, "Radial Matrix Cracking And Interphase Failure In Transversely Loaded Fiber Composites", Mechanics of Materials, 11, pp. 347-356.
- [39] Kim, K, Sudak, L.J., 2005, "Interaction Between A Radial Matrix Crack And A Three-Phase Circular Inclusion With Imperfect Interface In Plane Elasticity", International Journal of Fracture, 131, pp. 155-172.
- [40] Sudak, L.J., Mioduchowski, A., 2002, "A Three-Phase Circular Inhomogeneity With Imperfect Interface Under Therrnomechanical Loadings In Plane Elasticity", Acta Mechanica, 158, pp. 43-56.
- [41] Sudak, L.J., 2003, "Interaction Between A Screw Dislocation And A Three-Phase Circular Inhomogeneity With Imperfect Interface", Mathematics and Mechanics of Solids, 8, pp. 171-188.
- [42] Dundurs, J., 1967, "On The Interaction Of A Screw Dislocation With Inhomogeneities", Recent Advances in Engineering Science, 2, pp. 223-233.

- [43] Hu, K. X., Chandra, A., Huang, Y., 1993, "Fundamental Solutions For Dilute Distributions Of Inclusions Embedded In Microcracked Solids", Mechanics of Materials, 16, pp.281-294.
- [44] Lene, F., Leguillon, D., 1982, "Homogenized Constitutive Law For A Partially Cohesive Composite Materials", International Journal of Solids and Structures, 18, pp. 443-458.
- [45] Benveniste, Y., 1985, "The Effective Mechanical Behaviour Of Composite Materials With Imperfect Contact Between The Constituents", Mechanics of Materials, 4, pp. 197-208.
- [46] Ru, C. Q., 1999, "A New Method for an Inhomogeneity With Stepwise Graded Interphase Under Thermomechanical Loadings", Journal of Elasticity, 56, pp. 107-127.
- [47] Luo, H. A., Weng, G. J., 1989, "On Eshelby's S-tensor In A Three-Phase Cylindrically Concentric Solid, And The Elastic Moduli Of Fiber-Reinforced Composites", Mechanics of Materials, 8, pp. 77-88.
- [48] Luo, H.A., Chen, Y., 1991, "Matrix Cracking In Fiber-Reinforced Composite Materials", ASME Journal of Applied Mechanics, 58, pp. 846-848.
- [49] Luo, H.A., Chen, Y., 1991, "An Edge Dislocation In Three-Phase Composite Cylinder model", ASME Journal of Applied Mechanics, 58, pp.75-86.
- [50] Ru, C. Q., 1998, "Effect Of Interphase Layers On Thermal Stresses Within An Elliptical Inclusion", Journal of Applied Physics, 84, pp. 4872-4879.

- [51] Wang, X., Shen, Y.P., 2002, "An Edge Dislocation In A Three-Phase Composite Cylinder Model With A Sliding Interface", ASME Journal of Applied Mechanics, 69, pp. 527-538.
- [52] Xiao, Z.M., Chen, B.J., 2000, "A Screw Dislocation Interacting With A Coated Fiber", Mechanics of Materials, 32, pp. 485-495.
- [53] Xiao, Z.M., Chen, B.J., 2001, "On The Interaction Between An Edge Dislocation And A Coated Inclusion", International Journal of Solids and Structures, 38, pp. 2533-2548.
- [54] Sudak, L. J., 2003, "Effect Of An Interphase Layer On The Electroelastic Stresses Within A Three-Phase Elliptic Inclusion", International Journal of Engineering Science, 41, pp. 1019-1039.
- [55] Muskhelishvili, N. I., 1963, Some Basic Problems of the Mathematical Theory of Elasticity, P. Noordhoff Ltd.
- [56] Osborne, A. D., 1999, Complex Variables and their Applications, Addison Wesley Longman.
- [57] Anderson, T. L., 2005, Fracture Mechanics Fundamentals and Applications, 3rd
 Edition, CRC Press.
- [58] Carlsson, A., Bjorkman, A., Besjakov, J., Onsten, I., 2005, "Cemented Tibial Component Fixation Performs Better Than Cementless Fixation - A Randomized Radiostereometric Study Comparing Porous-Coated, Hydroxyapatite-Coated And Cemented Tibial Components Over 5 Years", Acta Orthopaedica, 76, pp. 362-369.

- [59] Nilsson, K. G., Karrholm, J., Carlsson, L., Dalen, T., 1999, "Hydroxyapatite Coating Versus Cemented Fixation Of The Tibial Component In Total Knee Arthroplasty: Prospective Randomized Comparison Of Hydroxyapatite-Coated And Cemented Tibial Components With 5-year Follow-Up Using Radiostereometry", Journal of Arthroplasty, 14, pp. 9-20.
- [60] Macaulay, W., DiGiovanni, C. W., Restrepo, A., et al., 2002, "Differences In Bone-Cement Porosity By Vacuum Mixing, Centrifugation, And Hand Mixing", Journal of Arthroplasty, 17, pp. 569-575.
- [61] Davies, J. P., O'Connor, D. O., Burke, D. W., et al., 1988, "The Effect Of Centrifugation On The Fatigue Life Of Bone-Cement In The Presence Of Surface Irregularities", Clinical Orthopaedics and Related Research, 229, pp. 156-161.
- [62] Mann, K. A., Ayers, D. C., Werner, F. W., Nicoletta, R. J., Fortino, M. D., 1997, "Tensile Strength Of The Cement-Bone Interface Depends On The Amount Of Bone Interdigitated With PMMA Cement", Journal of Biomechanics, 30, pp. 339-346.
- [63] Linden, U., 1991, "Mechanical-Properties Of Bone-Cement Importance Of The Mixing Technique", Clinical Orthopaedics and Related Research, 272, pp. 274-278.
- [64] Maltry, J. A., Noble, P. C., Kamaric E., et al., 1995, "Factors Influencing Pressurization Of The Femoral Canal During Cemented Total Hip Arthroplasty", Journal of Arthroplasty, 10, pp. 492-497.

- [65] Hierton, C., Blomgren, G., Lindgren, U., 1983, "Factors Associated With Early Loosening Of Cemented Total Hip Prostheses", Acta Orthopaedica Scandinavica, 54, 168-173.
- [66] Jasty, M., Maloney, W.J., Bragdon, C.R., Haire, T., Harris, W.H, 1990, "Histomorphological Studies Of The Long-Term Skeletal Responses To Well Fixed Cemented Femoral Components", Journal of Bone and Joint Surgery, 72, pp. 1220-1229.
- [67] Dohmae, Y., Bechtold, J. E., Sherman, R. E., et al., 1988, "Reduction In Cement-Bone Interface Shear Strength Between Primary And Revision Arthroplasty", Clinical Orthopaedics and Related Research, 236, pp. 214-220.
- [68] Mak, A.F., Keer, L.M., Chen, S.H., Lewis, J.L., 1980, "A No-Slip Interface Crack", ASME Journal of Applied Mechanics, 47, pp. 347-350.
- [69] Mak, A.F., Keer, L.M., 1980, "A No-Slip Edge Crack On A Bimaterial Interface", ASME Journal of Applied Mechanics, 47, pp. 816-820.
- [70] Verdonschot, N., Huiskes, R., 1998, "Surface Roughness Of Debonded Straight-Tapered Stems In Cemented THA Reduces Subsidence But Not Cement Damage", Biomaterials, 19, pp. 1773-1779.
- [71] Gharpuray, V. M., and Keer, L. M., et al., 1990, "Cracks Emanating From Circular Voids Or Elastic Inclusions In PMMA Near A Bone-Implant Interface", ASME Journal of Biomechanical Engineering, 112, pp. 22-28.
- [72] Steege, J. W., Polizos, T., Lewis, J. L., and Wixon, R. L., 1986, "Failure Mechanism In PMMA Around Loaded Tibial Components", Transactions, Orthopedic Research Society, 11, p. 355.

- [73] Sutherland, C. J., Wilde, A. H., Borden, L. S., et al., 1982, "A 10-year Follow-Up Of 100 Consecutive Muller Curved-Stem Total Hip-Replacement Arthroplasties", Journal of Bone and Joint Surgery, 64, pp. 970-982.
- [74] Beckenhaugh, R. D., Ilstrup, D. M., 1978, "Total Hip Arthroplasty Review Of 333 Cases With Long Follow-Up", Journal of Bone and Joint Surgery, 60A, pp. 306-313.
- [75] Bartel, D. L., 1977, "Stress Analysis: Effects Of Geometry In Bone-Implant Systems", Bulletin of the Hospital for Joint Diseases, 40, pp. 90-103.
- [76] Lu, Z., Ebramzadeh, E., McKellop, H., Sarmiento, A., 1996, "Stable Partial Debonding Of The Cement Interface Indicated By A Finite Element Model Of A Total Hip Prosthesis", Journal of Orthopaedic Research, 14, pp. 238-244.
- [77] Verdonschot, N., Huiskes, R., 1996, "A Combination Of Continuum Damage Mechanics And The Finite Element Method To Analyze Acrylic Cement Cracking Around Implants", In: Middleton, J. Jones, M.N., Pande, G.N. (Eds.), 2nd International symposium on Computer methods in Biomechanics and Biomedical Engineering, Gordon and Breach, Amsterdam, pp. 25-33.
- [78] McCormack, B. A. O., Prendergast, P. J., 1999, "Microdamage Accumulation In The Cement Layer Of Hip Replacements Under Flexural Loading", Journal of Biomechanics, 32, pp. 467-475.
- [79] Mann, K.A., Bartel, D.L., Wright T.M., Burstein, A.H., 1995, "Coulomb Frictional Interfaces In Modeling Cemented Total Hip Replacement: A more realistic model", Journal of Biomechanics, 28, pp. 1067-1078.

- [80] Hori, R. Y., Lewis, J. L., Zimmerman, J. R., Compere, C. L., 1978, "The Number Of Total Joint Replacements In The U.S.", Clinical Orthopaedics and Related Research, 132, pp. 46-52.
- [81] Mak, A. F., Lewis, J. L., 1980, "Fracture Mechanics Of The PMMA/ Cancellous Bone Interface", ASME Advances in Bioengineering, 1980, pp. 209-212.
- [82] Robinson, R. P., Lovell, T. P., Green, T. M., Bailey, G. A., 1989, "Early Femoral Component Loosening In DF-80 Total Hip Arthroplasty", J. Arthroplasty, 4, pp. 55-64.
- [83] Karrholm, J., Nivbrant, B., Thanner, J., et al., 2000, "Radiostereometric Evaluation Of Hip Implant Design And Surface Finish. Micromotion Of Cemented Femoral Stem", In: 67th Annual Meeting of the American Academy of Orthopaedic Surgeons (Orlando, USA), Scientific exhibition, pp. 1-16.
- [84] Nuno, N., Amabili, M., 2002, "Modelling Debonded Stem-Cement Interface For Hip Implants: Effects Of Residual Stresses", Clinical Biomechanics, 17, pp. 41-48.
- [85] Clech, J. P., Keer, L. M., Lewis, J. L., 1985, "A Model Of Tension And Compression Cracks With Cohesive Zone At A Bone-Cement Interface", ASME Journal of Biomechanical Engineering, 107, pp. 175-182.
- [86] McCormack, B. A. O., Walsh, C. D., Wilson, S. P., Prendergast, P. J., 1998, "A Statistical Analysis Of Microcrack Accumulation In PMMA Under Fatigue Loading: Aapplications To Orthopaedic Implant Fixation", International Journal Fatigue, 20, pp.581-593.

- [87] Maloney, W. J., Jasty, M., Burke, D. W., et al., 1989, "Biomechanical And Histologic Investigation Of Cemented Total Hip Arthroplasties - A Study Of Autopsy-Retrieved Femurs After Invivo Cycling", Clinical Orthopaedics and Related Research, 249, pp. 129-140.
- [88] James, S. P., Schmalzried, T. P., McGarry, F. J., Harris, W. H., 1993, "Extensive Porosity At The Cement Femoral Prosthesis Interface - A Preliminary Study", Journal of Biomedical Materials Research, 27, pp. 71-78.

$\begin{array}{c} \mbox{APPENDIX 1} \\ \mbox{Derivation of Boundary Condition } \partial \Gamma_0 \end{array}$

A1.1 Derivation of Equation (2.9)

$$\begin{aligned} (\sigma_{rr} - i\sigma_{r\theta})_{0} &= \left(m_{0} \|u_{r}\| - m_{0}u_{r}^{0}\right) - i\left(n_{0} \|u_{\theta}\| - n_{0}u_{\theta}^{0}\right) \qquad (A1.1) \\ &= \left(\frac{m_{0}}{2} \|u_{r}\| + \frac{m_{0}}{2} \|u_{r}\|\right) - \left(i\frac{n_{0}}{2} \|u_{\theta}\| + i\frac{n_{0}}{2} \|u_{\theta}\|\right) - \left(m_{0}u_{r}^{0} - in_{0}u_{\theta}^{0}\right) \\ &= \left(\frac{m_{0}}{2} \|u_{r}\| + \frac{m_{0}}{2} \|u_{r}\|\right) - \left(i\frac{n_{0}}{2} \|u_{\theta}\| + i\frac{n_{0}}{2} \|u_{\theta}\|\right) - \left(m_{0}u_{r}^{0} - in_{0}u_{\theta}^{0}\right) \\ &+ \left(\frac{n_{0}}{2} \|u_{r}\| - \frac{n_{0}}{2} \|u_{r}\|\right) + \left(i\frac{m_{0}}{2} \|u_{\theta}\| - i\frac{m_{0}}{2} \|u_{\theta}\|\right) \\ &= \left(\frac{m_{0}}{2} \|u_{r}\| - \frac{n_{0}}{2} \|u_{r}\| + \frac{m_{0}}{2} \|u_{\theta}\| + i\frac{n_{0}}{2} \|u_{\theta}\|\right) \\ &- \left(i\frac{n_{0}}{2} \|u_{\theta}\| - i\frac{m_{0}}{2} \|u_{\theta}\| + i\frac{n_{0}}{2} \|u_{\theta}\| + i\frac{m_{0}}{2} \|u_{\theta}\|\right) \\ &= \left(\frac{m_{0} - n_{0}}{2}\right) \|u_{r}\| + \left(\frac{m_{0} + n_{0}}{2}\right) \|u_{r}\| - \left(m_{0}u_{r}^{0} - in_{0}u_{\theta}^{0}\right) \\ &+ \left(\frac{m_{0} - n_{0}}{2}\right) \|u_{r} + iu_{\theta}\| + \left(\frac{m_{0} + n_{0}}{2}\right) \|u_{r} - iu_{\theta}\| - \left(m_{0}u_{r}^{0} - in_{0}u_{\theta}^{0}\right). \end{aligned}$$

A1.2 Derivation of Equation (2.12)

•

From equation (2.9)

· ·

.

$$(\sigma_{rr} - i\sigma_{r\theta})_0 = \left(\frac{m_0 - n_0}{2}\right) \|u_r + iu_\theta\| + \left(\frac{m_0 + n_0}{2}\right) \|u_r - iu_\theta\| - \left(m_0 u_r^0 - in_0 u_\theta^0\right).$$
(A1.2)

By taking the right-hand-side in (2.7), the left-hand-side of (A1.2) is

$$(\sigma_{rr} - i\sigma_{r\theta})_0 = \varphi_0'(z) + \ \overline{\varphi_0'}\left(\frac{R_0^2}{z}\right) - z\varphi_0''(z) - \frac{z^2}{R_0^2}\psi_0'(z).$$
(A1.3)

,

By (2.1)₁,

$$u_{r} + iu_{\theta} = \frac{e^{-i\theta}}{2\mu} \left[\kappa \varphi(z) - z\overline{\varphi'(z)} - \overline{\psi(z)} \right]$$

$$= \frac{R_{0}}{z2\mu} \left[\kappa \varphi(z) - z\overline{\varphi'(z)} - \overline{\psi(z)} \right].$$
(A1.4)

Thus,

:

4

$$\|u_{r} + iu_{\theta}\| = \frac{R_{0}}{z} \frac{1}{2\mu_{1}} \left[\kappa_{1}\varphi_{11}(z) - z\overline{\varphi_{11}'(z)} - \overline{\psi_{11}(z)} \right]$$

$$-\frac{R_{0}}{z} \frac{1}{2\mu_{0}} \left[\kappa_{0}\varphi_{0}(z) - z\overline{\varphi_{0}'(z)} - \overline{\psi_{0}(z)} \right].$$
(A1.5)

Taking conjugate of (A1.4) leads to

.

$$u_{r} - iu_{\theta} = \frac{e^{i\theta}}{2\mu} \left[\kappa \overline{\varphi(z)} - \overline{z} \varphi'(z) - \psi(z) \right]$$

$$= \frac{z}{R_{0} 2\mu} \left[\kappa \overline{\varphi(z)} - \overline{z} \varphi'(z) - \psi(z) \right].$$
(A1.6)

Thus,

$$\|u_{r} - iu_{\theta}\| = \frac{z}{2R_{0}} \frac{1}{\mu_{1}} \left[\kappa_{1} \overline{\varphi_{11}(z)} - \overline{z} \varphi_{11}'(z) - \psi_{11}(z) \right]$$

$$- \frac{z}{2R_{0}} \frac{1}{\mu_{0}} \left[\kappa_{0} \overline{\varphi_{0}(z)} - \overline{z} \varphi_{0}'(z) - \psi_{0}(z) \right].$$
(A1.7)

•

.

••

•

Substituting (A1.5), (A1.7) and (2.11) into (2.9), we can express the right-hand-side in the following form

,

$$\frac{m_{0} - n_{0}}{2} \|u_{r} + iu_{\theta}\| + \frac{m_{0} + n_{0}}{2} \|u_{r} - iu_{\theta}\| - (m_{0}u_{r}^{0} - in_{0}u_{\theta}^{0}) \quad (A1.8)$$

$$= \frac{m_{0} - n_{0}}{2} \frac{R_{0}}{2z} \begin{cases} \frac{1}{\mu_{1}} \left[\kappa_{1}\varphi_{11}(z) - z\overline{\varphi_{11}'(z)} - \overline{\psi_{11}(z)}\right] \\ -\frac{1}{\mu_{0}} \left[\kappa_{0}\varphi_{0}(z) - z\overline{\varphi_{0}'(z)} - \overline{\psi_{0}(z)}\right] \end{cases} \\
+ \frac{m_{0} + n_{0}}{2} \frac{z}{2R_{0}} \begin{cases} \frac{1}{\mu_{1}} \left[\kappa_{1}\overline{\varphi_{11}(z)} - \overline{z}\varphi_{11}'(z) - \psi_{11}(z)\right] \\ -\frac{1}{\mu_{0}} \left[\kappa_{0}\overline{\varphi_{0}(z)} - \overline{z}\varphi_{0}'(z) - \psi_{0}(z)\right] \end{cases} \\
- m_{0}R_{0}\varepsilon_{1} - \frac{(m_{0} + n_{0})(\varepsilon_{2} - i\varepsilon_{3})}{2R_{0}} z^{2} - \frac{(m_{0} - n_{0})(\varepsilon_{2} + i\varepsilon_{3})R_{0}^{3}}{2z^{2}} \\
= \frac{m_{0} - n_{0}}{2} \frac{R_{0}}{2z} \begin{cases} \frac{1}{\mu_{1}} \left[k_{1}\varphi_{11}(z) - z\overline{\varphi_{11}'}\left(\frac{R_{0}^{2}}{z}\right) - \overline{\psi_{0}}\left(\frac{R_{0}^{2}}{z}\right)\right] \\ -\frac{1}{\mu_{0}} \left[k_{0}\varphi_{0}(z) - z\overline{\varphi_{0}'}\left(\frac{R_{0}^{2}}{z}\right) - \overline{\psi_{0}}\left(\frac{R_{0}^{2}}{z}\right)\right] \end{cases} \\
+ \frac{m_{0} + n_{0}}{2} \frac{z}{2R_{0}} \begin{cases} \frac{1}{\mu_{1}} \left[k_{1}\overline{\varphi_{11}}\left(\frac{R_{0}^{2}}{z}\right) - \overline{z}\varphi_{11}'(z) - \psi_{11}(z)\right] \\ -\frac{1}{\mu_{0}} \left[k_{0}\overline{\varphi_{0}}\left(\frac{R_{0}^{2}}{z}\right) - \overline{z}\varphi_{0}'(z) - \psi_{0}(z)\right] \end{cases} \\
- m_{0}R_{0}\varepsilon_{1} - \frac{(m_{0} + n_{0})(\varepsilon_{2} - i\varepsilon_{3})}{2R_{0}} z^{2} - \frac{(m_{0} - n_{0})(\varepsilon_{2} + i\varepsilon_{3})R_{0}^{3}}{2z^{2}}.$$

Let's consider resultant forces expressed in (2.2) along a contour Γ_0 . The contour is a circle: namely, a point "B" on the arc coincides with the starting point "A", which means the resultant force between "A" and "B" is zero. Let "A" be a reference point which is 0 and "B" any point along the circle. Thus, (2.2) is

$$F_x + iF_y = -i\left[\varphi_0(z) + z\overline{\varphi_0'(z)} + \overline{\psi_0(z)}\right]_0^B = -i\left[\varphi_{11}(z) + z\overline{\varphi_{11}'(z)} + \overline{\psi_{11}(z)}\right]_0^B,$$
(A1.9)

and

,

$$z\overline{\varphi_{11}'(z)} + \overline{\psi_{11}(z)} = -\varphi_{11}(z) + \varphi_0(z) + z\overline{\varphi_0'(z)} + \overline{\psi_0(z)},$$
 (A1.10)

•-

$$z\overline{\varphi_{11}'}\left(\frac{R_0^2}{z}\right) + \overline{\psi_{11}}\left(\frac{R_0^2}{z}\right) = -\varphi_{11}\left(z\right) + \varphi_0\left(z\right) + z\overline{\varphi_0'}\left(\frac{R_0^2}{z}\right) + \overline{\psi_0}\left(\frac{R_0^2}{z}\right).$$

.

Taking conjugate of (A1.10) leads to

$$\overline{z}\varphi_{11}'(z) + \psi_{11}(z) = -\overline{\varphi_{11}(z)} + \overline{\varphi_0(z)} + \overline{z}\varphi_0'(z) + \psi_0(z), \qquad (A1.11)$$

$$\overline{z}\varphi_{11}'(z) + \psi_{11}(z) = -\overline{\varphi_{11}}\left(\frac{R_0^2}{z}\right) + \overline{\varphi_0}\left(\frac{R_0^2}{z}\right) + \overline{z}\varphi_0'(z) + \psi_0(z).$$

By substituting (A1.10) and (A1.11) into (A1.8), and then (A1.8) and (A1.3) into (2.9), we obtain an another notation of (2.9) in the following form

$$\begin{aligned}
\varphi_{0}'(z) + \overline{\varphi_{0}'}\left(\frac{R_{0}^{2}}{z}\right) - z\varphi_{0}''(z) - \frac{z^{2}}{R_{0}^{2}}\psi_{0}'(z) \quad (A1.12) \\
&= \frac{m-n}{2}\frac{R_{0}}{2z} \left\{ \begin{array}{c} \frac{1}{\mu_{1}}\left[k_{1}\varphi_{11}\left(z\right) - \left(-\varphi_{11}\left(z\right) + \varphi_{0}\left(z\right) + z\overline{\varphi_{0}'}\left(\frac{R_{0}^{2}}{z}\right) + \overline{\psi_{0}}\left(\frac{R_{0}^{2}}{z}\right)\right)\right] \\ &- \frac{1}{\mu_{0}}\left[k_{0}\varphi_{0}\left(z\right) - z\overline{\varphi_{0}'}\left(\frac{R_{0}^{2}}{z}\right) - \overline{\psi_{0}}\left(\frac{R_{0}^{2}}{z}\right)\right] \end{array} \right\} \\
&+ \frac{m+n}{2}\frac{z}{2R_{0}} \left\{ \begin{array}{c} \frac{1}{\mu_{1}}\left[k_{1}\overline{\varphi_{11}}\left(\frac{R_{0}^{2}}{z}\right) - \left(-\overline{\varphi_{11}}\left(\frac{R_{0}^{2}}{z}\right) + \overline{\varphi_{0}}\left(\frac{R_{0}^{2}}{z}\right) + \overline{z}\varphi_{0}'\left(z\right) + \psi_{0}\left(z\right)\right)\right] \\ &- \frac{1}{\mu_{0}}\left[k_{0}\overline{\varphi_{0}}\left(\frac{R_{0}^{2}}{z}\right) - \overline{z}\varphi_{0}'\left(z\right) - \psi_{0}\left(z\right)\right] \\ &- m_{0}R_{0}\varepsilon_{1} - \frac{(m_{0}+n_{0})(\varepsilon_{2}-i\varepsilon_{3})}{2R_{0}}z^{2} - \frac{(m_{0}-n_{0})(\varepsilon_{2}+i\varepsilon_{3})R_{0}^{3}}{2z^{2}}.
\end{aligned}$$

Multiplying $4\mu_1$ to both side yields

$$4\mu_{1}\left[\varphi_{0}'(z) + \overline{\varphi_{0}'}\left(\frac{R_{0}^{2}}{z}\right) - z\varphi_{0}''(z) - \frac{z^{2}}{R_{0}^{2}}\psi_{0}'(z)\right]$$
(A1.13)
= $(m-n)\frac{R_{0}}{z}\left\{ \begin{cases} \left[k_{1}\varphi_{11}\left(z\right) + \varphi_{11}\left(z\right) - \varphi_{0}\left(z\right) - z\overline{\varphi_{0}'}\left(\frac{R_{0}^{2}}{z}\right) - \overline{\psi_{0}}\left(\frac{R_{0}^{2}}{z}\right)\right] \\ -\frac{\mu_{1}}{\mu_{0}}\left[k_{0}\varphi_{0}\left(z\right) - z\overline{\varphi_{0}'}\left(\frac{R_{0}^{2}}{z}\right) - \overline{\psi_{0}}\left(\frac{R_{0}^{2}}{z}\right)\right] \end{cases} \right\}$
+ $(m+n)\frac{z}{R_{0}}\left\{ \begin{cases} \left[k_{1}\overline{\varphi_{11}}\left(\frac{R_{0}^{2}}{z}\right) + \overline{\varphi_{11}}\left(\frac{R_{0}^{2}}{z}\right) - \overline{\varphi_{0}}\left(\frac{R_{0}^{2}}{z}\right) - \overline{z}\varphi_{0}'\left(z\right) - \psi_{0}\left(z\right)\right] \\ -\frac{\mu_{1}}{\mu_{0}}\left[k_{0}\overline{\varphi_{0}}\left(\frac{R_{0}^{2}}{z}\right) - \overline{z}\varphi_{0}'\left(z\right) - \psi_{0}\left(z\right)\right] \\ -4\mu_{1}m_{0}R_{0}\varepsilon_{1} - 4\mu_{1}\frac{(m_{0}+n_{0})(\varepsilon_{2}-i\varepsilon_{3})}{2R_{0}}z^{2} - 4\mu_{1}\frac{(m_{0}-n_{0})(\varepsilon_{2}+i\varepsilon_{3})R_{0}^{3}}{2z^{2}}. \end{cases}$

By making arrangement of left-hand side for φ_{11} and right-hand-side for φ_0 and ψ_0 , we get (2.12)

.

$$(m_0 - n_0)(\kappa_1 + 1)\frac{R_0}{z}\varphi_{11}(z) + (m_0 + n_0)(\kappa_1 + 1)\frac{z}{R_0}\overline{\varphi_{11}}\left(\frac{R_0^2}{z}\right)$$

$$= 4\mu_1 \left[\varphi_0'(z) + \overline{\varphi_0'}\left(\frac{R_0^2}{z}\right) - z\varphi_0''(z) - \frac{z^2}{R_0^2}\psi_0'(z)\right] + (m_0 - n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})\frac{R_0}{z}\varphi_0(z)$$

$$+ (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})\left[R_0\overline{\varphi_0'}\left(\frac{R_0^2}{z}\right) + \frac{R_0}{z}\overline{\psi_0}\left(\frac{R_0^2}{z}\right)\right]$$

$$+ (m_0 + n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})\frac{z}{R_0}\overline{\varphi_0}(\frac{R_0^2}{z}) + (m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})\left[R_0\varphi_0'(z) + \frac{z}{R_0}\psi_0(z)\right]$$

$$+ 4m_0R_0\mu_1\varepsilon_1 + \frac{2\mu_1(m_0 + n_0)(\varepsilon_2 - i\varepsilon_3)}{R_0}z^2 + \frac{2\mu_1(m_0 - n_0)(\varepsilon_2 + i\varepsilon_3)R_0^3}{z^2}$$

.

.

٠.

 $,z\in \Gamma _{1}.$

$\begin{array}{c} \mbox{APPENDIX 2} \\ \mbox{Derivation of Boundary Condition } \partial \Gamma_1 \end{array}$

A2.1 Derivation of equation (2.14)

In the similar fashion of (2.9), the displacement jump condition $(2.4)_2$ is

$$(\sigma_{rr} - i\sigma_{r\theta})_{1 \text{ or } 2} = \frac{m_1 - n_1}{2} \|u_r + iu_\theta\| + \frac{m_1 + n_1}{2} \|u_r - iu_\theta\|.$$
(A2.1)

Let's consider stresses at region S_2 in (2.13) and the left-hand-side of (A2.1) is

$$(\sigma_{rr} - i\sigma_{r\theta})_2 = \varphi_2'(z) + \overline{\varphi_2'}\left(\frac{R_1^2}{z}\right) - z\varphi_2''(z) - \frac{z^2}{R_1^2}\psi_2'(z).$$
(A2.2)

The right-hand-side of (A2.1) can be expressed as follows

$$\frac{m_{1} - n_{1}}{2} \|u_{r} + iu_{\theta}\| + \frac{m_{1} + n_{1}}{2} \|u_{r} - iu_{\theta}\| \tag{A2.3}$$

$$= \frac{m_{1} - n_{1}}{2} \frac{R_{1}}{2z} \left\{ \begin{array}{c} \frac{1}{\mu_{2}} \left[k_{2}\varphi_{2}\left(z\right) - z\overline{\varphi_{2}'}\left(\frac{R_{1}^{2}}{z}\right) - \overline{\psi_{2}}\left(\frac{R_{1}^{2}}{z}\right) \right] \\ -\frac{1}{\mu_{1}} \left[k_{1}\varphi_{12}\left(z\right) - z\overline{\varphi_{12}'}\left(\frac{R_{1}^{2}}{z}\right) - \overline{\psi_{12}}\left(\frac{R_{1}^{2}}{z}\right) \right] \\ + \frac{m_{1} + n_{1}}{2} \frac{z}{2R_{1}} \left\{ \begin{array}{c} \frac{1}{\mu_{2}} \left[k_{2}\overline{\varphi_{2}}\left(\frac{R_{1}^{2}}{z}\right) - \overline{z}\varphi_{2}'\left(z\right) - \psi_{2}\left(z\right) \right] \\ -\frac{1}{\mu_{1}} \left[k_{1}\overline{\varphi_{12}}\left(\frac{R_{1}^{2}}{z}\right) - \overline{z}\varphi_{12}'\left(z\right) - \psi_{12}\left(z\right) \right] \end{array} \right\}.$$

Thus,

$$\varphi_{2}'(z) + \overline{\varphi_{2}'}\left(\frac{R_{1}^{2}}{z}\right) - z\varphi_{2}''(z) - \frac{z^{2}}{R_{1}^{2}}\psi_{2}'(z) \qquad (A2.4)$$

$$= \frac{m-n}{2}\frac{R_{1}}{2z} \left\{ \begin{array}{c} \frac{1}{\mu_{2}}\left[k_{2}\varphi_{2}\left(z\right) - z\overline{\varphi_{2}'}\left(\frac{R_{1}^{2}}{z}\right) - \overline{\psi_{2}}\left(\frac{R_{1}^{2}}{z}\right)\right] \\ -\frac{1}{\mu_{1}}\left[k_{1}\varphi_{12}\left(z\right) - z\overline{\varphi_{12}'}\left(\frac{R_{1}^{2}}{z}\right) - \overline{\psi_{12}}\left(\frac{R_{1}^{2}}{z}\right)\right] \end{array} \right\}$$

÷

$$+\frac{m+n}{2}\frac{z}{2R_{1}}\left\{\begin{array}{c}\frac{1}{\mu_{2}}\left[k_{2}\overline{\varphi_{2}}\left(\frac{R_{1}^{2}}{z}\right)-\overline{z}\varphi_{2}'\left(z\right)-\psi_{2}\left(z\right)\right]\\-\frac{1}{\mu_{1}}\left[k_{1}\overline{\varphi_{12}}\left(\frac{R_{1}^{2}}{z}\right)-\overline{z}\varphi_{12}'\left(z\right)-\psi_{12}\left(z\right)\right]\end{array}\right\}$$

In addition, consider forces along the boundary Γ_1 ,

$$F_{x} + iF_{y} = -i \left[\varphi_{2}\left(z\right) + z\overline{\varphi_{2}'}\left(\frac{R_{1}^{2}}{z}\right) + \overline{\psi_{2}}\left(\frac{R_{1}^{2}}{z}\right) \right]_{0}^{B}$$

$$= -i \left[\varphi_{12}\left(z\right) + z\overline{\varphi_{12}'}\left(\frac{R_{0}^{2}}{z}\right) + \overline{\psi_{12}}\left(\frac{R_{0}^{2}}{z}\right) \right]_{0}^{B},$$
(A2.5)

and

$$\varphi_2(z) + z\overline{\varphi_2'}\left(\frac{R_1^2}{z}\right) + \overline{\psi_2}\left(\frac{R_1^2}{z}\right) = \varphi_{12}(z) + z\overline{\varphi_{12}'}\left(\frac{R_1^2}{z}\right) + \overline{\psi_{12}}\left(\frac{R_1^2}{z}\right). \quad (A2.6)$$

Thus,

۰

$$\overline{z\varphi_{12}'}\left(\frac{R_1^2}{z}\right) + \overline{\psi_{12}}\left(\frac{R_1^2}{z}\right) = -\varphi_{12}\left(z\right) + \varphi_2\left(z\right) + \overline{z\varphi_2'}\left(\frac{R_1^2}{z}\right) + \overline{\psi_2}\left(\frac{R_1^2}{z}\right) (A2.7)$$
$$\overline{z}\varphi_{12}'\left(z\right) + \psi_{12}\left(z\right) = -\overline{\varphi_{12}}\left(\frac{R_0^2}{z}\right) + \overline{\varphi_2}\left(\frac{R_1^2}{z}\right) + \overline{z}\varphi_2'\left(z\right) + \psi_2\left(z\right).$$

Substituting these equations into (A2.4) gives

$$\begin{aligned} \varphi_{2}'(z) + \ \overline{\varphi_{2}'}\left(\frac{R_{1}^{2}}{z}\right) - z\varphi_{2}''(z) - \frac{z^{2}}{R_{1}^{2}}\psi_{2}'(z) \qquad (A2.8) \end{aligned}$$

$$= \ \frac{m-n}{2}\frac{R_{1}}{2z} \left\{ \begin{array}{c} \frac{1}{\mu_{2}}\left[k_{2}\varphi_{2}\left(z\right) - z\overline{\varphi_{2}'}\left(\frac{R_{1}^{2}}{z}\right) - \overline{\psi_{2}}\left(\frac{R_{1}^{2}}{z}\right)\right] \\ -\frac{1}{\mu_{1}}\left[k_{1}\varphi_{12}\left(z\right) + \varphi_{12}\left(z\right) - \varphi_{2}\left(z\right) - z\overline{\varphi_{2}'}\left(\frac{R_{1}^{2}}{z}\right) - \overline{\psi_{2}}\left(\frac{R_{1}^{2}}{z}\right)\right] \\ +\frac{m+n}{2}\frac{z}{2R_{1}}\left\{ \begin{array}{c} \frac{1}{\mu_{1}}\left[k_{2}\overline{\varphi_{2}}\left(\frac{R_{1}^{2}}{z}\right) - \overline{z}\varphi_{2}'\left(z\right) - \psi_{2}\left(z\right)\right] \\ -\frac{1}{\mu_{1}}\left[k_{1}\overline{\varphi_{12}}\left(\frac{R_{1}^{2}}{z}\right) + \overline{\varphi_{12}}\left(\frac{R_{0}^{2}}{z}\right) - \overline{\varphi_{2}}\left(\frac{R_{1}^{2}}{z}\right) - \overline{z}\varphi_{2}'\left(z\right) - \psi_{2}\left(z\right)\right] \\ \end{array} \right\}. \end{aligned}$$

..

121

.

,

Multiplying $4\mu_2$ to both sides leads to the following expression

.

$$4\mu_{2}\left[\varphi_{2}'(z) + \overline{\varphi_{2}'}\left(\frac{R_{1}^{2}}{z}\right) - z\varphi_{2}''(z) - \frac{z^{2}}{R_{1}^{2}}\psi_{2}'(z)\right]$$

$$= (m-n)\frac{R_{1}}{z} \begin{cases} k_{2}\varphi_{2}(z) - z\overline{\varphi_{2}'}\left(\frac{R_{1}^{2}}{z}\right) - \overline{\psi_{2}}\left(\frac{R_{1}^{2}}{z}\right) \\ -\frac{\mu_{2}}{\mu_{1}}\left[k_{1}\varphi_{12}(z) + \varphi_{12}(z) - \varphi_{2}(z) - z\overline{\varphi_{2}'}\left(\frac{R_{1}^{2}}{z}\right) - \overline{\psi_{2}}\left(\frac{R_{1}^{2}}{z}\right)\right] \end{cases}$$

$$+ (m+n)\frac{z}{R_{1}} \begin{cases} k_{2}\overline{\varphi_{2}}\left(\frac{R_{1}^{2}}{z}\right) - \overline{\varphi_{2}}\left(\frac{R_{1}^{2}}{z}\right) - \overline{\varphi_{2}}\left(\frac{R_{1}^{2}}{z}\right) - \overline{\varphi_{2}}\left(\frac{R_{1}^{2}}{z}\right)\right] \end{cases}$$

$$(A2.9)$$

By making arrangement of the left-hand-side for $\varphi_{12}(z)$ and the right-hand-side for $\varphi_2(z)$ and $\psi_2(z)$, we have

$$(m-n)\frac{\mu_{2}}{\mu_{1}}(k_{1}+1)\frac{R_{1}}{z}\varphi_{12}(z) + (m+n)\frac{z}{R_{1}}\frac{\mu_{2}}{\mu_{1}}(k_{1}+1)\overline{\varphi_{12}}\left(\frac{R_{1}^{2}}{z}\right)(A2.10)$$

$$= -4\mu_{2}\left[\varphi_{2}'(z) + \overline{\varphi_{2}'}\left(\frac{R_{1}^{2}}{z}\right) - z\varphi_{2}''(z) - \frac{z^{2}}{R_{1}^{2}}\psi_{2}'(z)\right]$$

$$+ (m-n)\left(\frac{\mu_{2}}{\mu_{1}} - 1\right)\left[R_{1}\overline{\varphi_{2}'}\left(\frac{R_{1}^{2}}{z}\right) + \frac{R_{1}}{z}\overline{\psi_{2}}\left(\frac{R_{1}^{2}}{z}\right)\right]$$

$$+ (m-n)\left(\frac{\mu_{2}}{\mu_{1}} + k_{2}\right)\frac{R_{1}}{z}\varphi_{2}(z) + (m+n)\frac{z}{R_{1}}\left(\frac{\mu_{2}}{\mu_{1}} + k_{2}\right)\overline{\varphi_{2}}\left(\frac{R_{1}^{2}}{z}\right)$$

$$+ (m+n)\frac{z}{R_{1}}\left(\frac{\mu_{2}}{\mu_{1}} - 1\right)\left[\psi_{2}(z) + \overline{z}\varphi_{2}'(z)\right].$$

Finally, by multiplying $\frac{\mu_1}{\mu_2}$ to both sides, we get (2.14)

$$(m_1 - n_1) (k_1 + 1) \frac{R_1}{z} \varphi_1 (z) + (m_1 + n_1) (k_1 + 1) \frac{z}{R_1} \overline{\varphi_1} \left(\frac{R_1^2}{z}\right)$$

$$= -4\mu_1 \left[\varphi_2'(z) + \overline{\varphi_2'} \left(\frac{R_1^2}{z}\right) - z\varphi_2''(z) - \frac{z^2}{R_1^2} \psi_2'(z) \right]$$

$$+ (m_1 - n_1) \left(1 - \frac{\mu_1}{\mu_2}\right) \left[R_1 \overline{\varphi_2'} \left(\frac{R_1^2}{z}\right) + \frac{R_1}{z} \overline{\psi_2} \left(\frac{R_1^2}{z}\right) \right]$$

$$+ (m_1 - n_1) \left(1 + \frac{\mu_1}{\mu_2} k_2\right) \frac{R_1}{z} \varphi_2 (z) + (m_1 + n_1) \left(1 + \frac{\mu_1}{\mu_2} k_2\right) \frac{z}{R_1} \overline{\varphi_2} \left(\frac{R_1^2}{z}\right)$$

÷

.

 $+\left(m_{1}+n_{1}\right)\left(1-\frac{\mu_{1}}{\mu_{2}}\right)\left[R_{1}\varphi_{2}'\left(z\right)+\frac{z}{R_{1}}\psi_{2}\left(z\right)\right].$

. .

.

4

APPENDIX 3 Stress Potentials

A3.1 Coefficients g_k of $F_{11}(z)$ (2.34)

$$F_{11}(z) = \frac{1}{2\sqrt{(z-a)(z-b)}} = \frac{1}{2}(z-a)^{-\frac{1}{2}}(z-b)^{-\frac{1}{2}}$$
(A3.1)
$$= \sum_{k=0}^{\infty} g_k z^k = g_0 + g_1 z + g_2 z^2 + g_3 z^3 + g_4 z^4 + g_5 z^5 + g_6 z^6 + g_7 z^7.$$

Each term is expanded into Taylor series as follows

.

$$(z-a)^{-\frac{1}{2}}$$

$$= (-a)^{-\frac{1}{2}} + \left(-\frac{1}{2}\right)(-a)^{-\frac{3}{2}}z + \frac{1}{2!}\left(-\frac{1}{2}\right)\left(-\frac{3}{2}\right)(-a)^{-\frac{5}{2}}z^{2} + \frac{1}{3!}\left(-\frac{1}{2}\right)\left(-\frac{3}{2}\right)\left(-\frac{5}{2}\right)(-a)^{-\frac{7}{2}}z^{3} + \dots + \frac{1}{n!}\left(-\frac{1}{2}\right)^{n}\left[1*3*5*\dots*(2n-1)\right](-a)^{-\frac{2n+1}{2}}z^{n} + \dots + \frac{1}{n!}\left(\frac{1}{2}\right)^{n}\left[1*3*5*\dots*(2n-1)\right]\frac{1}{(-a)^{\frac{2n+1}{2}}}z^{n} + \dots + \frac{1}{(-a)^{\frac{1}{2}}}z^{1} + \frac{1}{2(-a)^{\frac{3}{2}}}z + \frac{3}{8(-a)^{\frac{5}{2}}}z^{2} + \frac{5}{16(-a)^{\frac{7}{2}}}z^{3} + \frac{35}{128(-a)^{\frac{9}{2}}}z^{4} + \frac{63}{256(-a)^{\frac{11}{2}}}z^{5} \dots,$$
(A3.2)

and

••

$$(z-b)^{-\frac{1}{2}} = (-b)^{-\frac{1}{2}} + \left(-\frac{1}{2}\right)(-b)^{-\frac{3}{2}}z + \frac{1}{2!}\left(-\frac{1}{2}\right)\left(-\frac{3}{2}\right)(-b)^{-\frac{5}{2}}z^{2} + \frac{1}{3!}\left(-\frac{1}{2}\right)\left(-\frac{3}{2}\right)\left(-\frac{5}{2}\right)(-b)^{-\frac{7}{2}}z^{3} + \dots + \frac{1}{n!}\left(-\frac{1}{2}\right)^{n} [1*3*5*\dots*(2n-1)](-b)^{-\frac{2n+1}{2}}z^{n} + \dots$$
(A3.3)

۰.

.

$$= \sum_{n=0}^{\infty} \frac{1}{n!} \left(\frac{1}{2}\right)^n [1*3*5*...*(2n-1)] \frac{1}{(-b)^{\frac{2n+1}{2}}} z^n$$

$$= \frac{1}{(-b)^{\frac{1}{2}}} + \frac{1}{2(-b)^{\frac{3}{2}}} z^2 + \frac{3}{8(-b)^{\frac{5}{2}}} z^2 + \frac{5}{16(-b)^{\frac{7}{2}}} z^3 + \frac{35}{128(-b)^{\frac{9}{2}}} z^4 + \frac{63}{256(-b)^{\frac{11}{2}}} z^5...$$

Consequently, (A3.1) is re-written as

$$\begin{aligned} \frac{1}{2}(z-a)^{-\frac{1}{2}}(z-b)^{-\frac{1}{2}} & (A3.4) \\ &= \frac{1}{2} \left[\begin{array}{c} \frac{1}{(-a)^{\frac{1}{2}}} + \frac{1}{2(-a)^{\frac{3}{2}}}z + \frac{3}{8(-a)^{\frac{3}{2}}}z^{2} + \frac{5}{16(-a)^{\frac{1}{2}}}z^{3} \\ &+ \frac{35}{128(-a)^{\frac{3}{2}}}z^{4} + \frac{63}{256(-a)^{\frac{1}{2}}}z^{5} \dots \end{array} \right] \\ &* \left[\begin{array}{c} \frac{1}{(-b)^{\frac{1}{2}}} + \frac{1}{2(-b)^{\frac{3}{2}}}z + \frac{3}{8(-b)^{\frac{3}{2}}}z^{2} + \frac{5}{16(-b)^{\frac{1}{2}}}z^{3} \\ &+ \frac{35}{128(-b)^{\frac{3}{2}}}z^{4} + \frac{63}{256(-a)^{\frac{1}{2}}}z^{5} \dots \end{array} \right] \end{aligned} \\ &= \frac{1}{2\sqrt{ab}} + \left(\frac{1}{4}\frac{1}{\sqrt{a^{1}b^{3}}} + \frac{1}{4}\frac{1}{\sqrt{a^{3}b^{1}}} \right)z + \left(\frac{3}{16}\frac{1}{\sqrt{a^{1}b^{5}}} + \frac{1}{8}\frac{1}{\sqrt{a^{3}b^{3}}} + \frac{3}{16}\frac{1}{\sqrt{a^{5}b^{1}}} \right)z^{2} \\ &+ \left(\frac{5}{32}\frac{1}{\sqrt{a^{1}b^{7}}} + \frac{3}{32}\frac{1}{\sqrt{a^{3}b^{5}}} + \frac{3}{32}\frac{1}{\sqrt{a^{5}b^{5}}} + \frac{5}{32}\frac{1}{\sqrt{a^{7}b^{1}}} \right)z^{3} \\ &+ \left(\frac{35}{256}\frac{1}{\sqrt{a^{1}b^{9}}} + \frac{5}{64}\frac{1}{\sqrt{a^{3}b^{7}}} + \frac{9}{128}\frac{1}{\sqrt{a^{5}b^{5}}} + \frac{5}{64}\frac{1}{\sqrt{a^{7}b^{3}}} + \frac{35}{256}\frac{1}{\sqrt{a^{9}b^{1}}} \right)z^{4} \\ &+ \left(\begin{array}{c} \frac{63}{512}\frac{1}{\sqrt{a^{1}b^{11}}} + \frac{35}{512}\frac{1}{\sqrt{a^{3}b^{5}}} + \frac{15}{256}\frac{1}{\sqrt{a^{1}b^{1}}} + \frac{15}{256}\frac{1}{\sqrt{a^{7}b^{5}}} + \frac{15}{256}\frac{1}{\sqrt{a^{7}b^{7}}} \right)z^{5} \\ &+ \left(\begin{array}{c} \frac{231}{2048}\frac{1}{\sqrt{a^{1}b^{13}}} + \frac{63}{1024}\frac{1}{\sqrt{a^{3}b^{13}}} + \frac{106}{2048}\frac{1}{\sqrt{a^{1}b^{13}}} + \frac{231}{2048}\frac{1}{\sqrt{a^{1}b^{1}}} \right)z^{6} \\ &+ \left(\begin{array}{c} \frac{3003}{28672}\frac{1}{\sqrt{a^{1}b^{15}}} + \frac{231}{4096}\frac{1}{\sqrt{a^{1}b^{15}}} + \frac{231}{4096}\frac{1}{\sqrt{a^{1}b^{1}}} + \frac{2303}{2067}\frac{1}{\sqrt{a^{1}b^{1}}} \right)z^{7} \\ &= g_{0} + g_{1}z + g_{2}z^{2} + g_{3}z^{3} + g_{4}z^{4} + g_{5}z^{5} + g_{6}z^{6} + g_{7}z^{7}. \end{array} \right)$$

.

••

,

A3.2 Coefficients h_k of $F_{12}(z)$ (2.35)

$$F_{12}(z) = \frac{1}{2\sqrt{(z-a)(z-b)}} = \frac{1}{2z\sqrt{(1-\frac{a}{z})(1-\frac{b}{z})}}$$

$$= \frac{1}{2z}(1-\frac{a}{z})^{-\frac{1}{2}}(1-\frac{b}{z})^{-\frac{1}{2}} = \sum_{k=1}^{\infty} h_{-k}z^{-k}$$

$$= \begin{pmatrix} h_{-1}z^{-1} + h_{-2}z^{-2} + h_{-3}z^{-3} + h_{-4}z^{-4} + h_{-5}z^{-5} \\ + h_{-6}z^{-6} + h_{-7}z^{-7} + h_{-8}z^{-8} \end{pmatrix}.$$
(A3.5)

Each term is expanded into Taylor series

$$\begin{pmatrix} \frac{a}{z} - 1 \end{pmatrix}^{-\frac{1}{2}}$$
(A3.6)

$$= (-1)^{-\frac{1}{2}} + \left(-\frac{1}{2}\right)(-1)^{-\frac{3}{2}} \left(\frac{a}{z}\right)^{1} + \left(\frac{1}{2!}\right) \left(-\frac{1}{2}\right) \left(-\frac{3}{2}\right)(-1)^{-\frac{5}{2}} \left(\frac{a}{z}\right)^{2}$$

$$+ \left(\frac{1}{3!}\right) \left(-\frac{1}{2}\right) \left(-\frac{3}{2}\right) \left(-\frac{5}{2}\right) (-1)^{-\frac{7}{2}} \left(\frac{a}{z}\right)^{3}$$

$$\dots + \left(\frac{1}{n!}\right) \left(-\frac{1}{2}\right)^{n} [1 * 3 * 5 * \dots * (2n-1)] (-1)^{n} (-1)^{-\frac{1}{2}} \left(\frac{a}{z}\right)^{n} \dots$$

$$= \sum_{n=0}^{\infty} (-1)^{-\frac{1}{2}} \left(\frac{1}{n!}\right) \left(\frac{1}{2}\right)^{n} [1 * 3 * 5 * \dots * (2n-1)] \left(\frac{a}{z}\right)^{n}$$

$$= (-1)^{-\frac{1}{2}} + (-1)^{-\frac{1}{2}} \frac{1}{2} \left(\frac{a}{z}\right) + (-1)^{-\frac{1}{2}} \frac{3}{8} \left(\frac{a}{z}\right)^{2} + (-1)^{-\frac{1}{2}} \frac{5}{16} \left(\frac{a}{z}\right)^{3}$$

$$+ (-1)^{-\frac{1}{2}} \frac{35_{4}}{128} \left(\frac{a}{z}\right)^{4} + (-1)^{-\frac{1}{2}} \frac{63}{256} \left(\frac{a}{z}\right)^{5} + (-1)^{-\frac{1}{2}} \frac{231}{1024} \left(\frac{a}{z}\right)^{6}$$

$$+ (-1)^{-\frac{1}{2}} \frac{429}{2048} \left(\frac{a}{z}\right)^{7} + (-1)^{-\frac{1}{2}} \frac{6435}{32768} \left(\frac{a}{z}\right)^{8} \dots$$

and

•

÷

$$(\frac{b}{z} - 1)^{-\frac{1}{2}}$$

$$= (-1)^{-\frac{1}{2}} + \left(-\frac{1}{2}\right)(-1)^{-\frac{3}{2}} \left(\frac{b}{z}\right)^{1} + \left(\frac{1}{2!}\right) \left(-\frac{1}{2}\right) \left(-\frac{3}{2}\right) (-1)^{-\frac{5}{2}} \left(\frac{b}{z}\right)^{2}$$
(A3.7)

$$+ \left(\frac{1}{3!}\right) \left(-\frac{1}{2}\right) \left(-\frac{3}{2}\right) \left(-\frac{5}{2}\right) (-1)^{-\frac{7}{2}} \left(\frac{b}{z}\right)^{3} \\ \dots + \left(\frac{1}{n!}\right) \left(-\frac{1}{2}\right)^{n} \left[1 * 3 * 5 * \dots * (2n-1)\right] \left(\frac{1}{i}\right)^{2n+1} \left(\frac{b}{z}\right)^{n} \dots \\ = \sum_{n=0}^{\infty} (-1)^{-\frac{1}{2}} \left(\frac{1}{n!}\right) \left(\frac{1}{2}\right)^{n} \left[1 * 3 * 5 * \dots * (2n-1)\right] \left(\frac{b}{z}\right)^{n} \\ = (-1)^{-\frac{1}{2}} + (-1)^{-\frac{1}{2}} \frac{1}{2} \left(\frac{b}{z}\right) + (-1)^{-\frac{1}{2}} \frac{3}{8} \left(\frac{b}{z}\right)^{2} + (-1)^{-\frac{1}{2}} \frac{5}{16} \left(\frac{b}{z}\right)^{3} \\ + (-1)^{-\frac{1}{2}} \frac{35}{128} \left(\frac{b}{z}\right)^{4} + (-1)^{-\frac{1}{2}} \frac{63}{256} \left(\frac{b}{z}\right)^{5} + (-1)^{-\frac{1}{2}} \frac{231}{1024} \left(\frac{b}{z}\right)^{6} \\ + (-1)^{-\frac{1}{2}} \frac{429}{2048} \left(\frac{b}{z}\right)^{7} + (-1)^{-\frac{1}{2}} \frac{6435}{32768} \left(\frac{b}{z}\right)^{8} \dots$$

.

Consequently, (A3.5) is re-written as

.

٠.

$$\begin{aligned} \frac{1}{2z} (1 - \frac{a}{z})^{-\frac{1}{2}} (1 - \frac{b}{z})^{-\frac{1}{2}} & (A3.8) \\ &= \frac{1}{2z} \begin{bmatrix} (-1)^{-\frac{1}{2}} + (-1)^{-\frac{1}{2}} \frac{1}{2} \left(\frac{a}{z}\right) + (-1)^{-\frac{1}{2}} \frac{3}{8} \left(\frac{a}{z}\right)^{2} + (-1)^{-\frac{1}{2}} \frac{5}{16} \left(\frac{a}{z}\right)^{3} \\ &+ (-1)^{-\frac{1}{2}} \frac{35}{128} \left(\frac{a}{z}\right)^{4} + (-1)^{-\frac{1}{2}} \frac{63}{65} \left(\frac{a}{z}\right)^{5} + (-1)^{-\frac{1}{2}} \frac{231}{1024} \left(\frac{a}{z}\right)^{6} \\ &+ (-1)^{-\frac{1}{2}} \frac{429}{2048} \left(\frac{a}{z}\right)^{7} + (-1)^{-\frac{1}{2}} \frac{6435}{32768} \left(\frac{a}{z}\right)^{8} \dots \end{bmatrix} \\ &\times \begin{bmatrix} (-1)^{-\frac{1}{2}} + (-1)^{-\frac{1}{2}} \frac{1}{2} \left(\frac{b}{z}\right) + (-1)^{-\frac{1}{2}} \frac{63}{3266} \left(\frac{b}{z}\right)^{2} + (-1)^{-\frac{1}{2}} \frac{5}{16} \left(\frac{b}{z}\right)^{3} \\ &+ (-1)^{-\frac{1}{2}} \frac{35}{128} \left(\frac{b}{z}\right)^{4} + (-1)^{-\frac{1}{2}} \frac{63}{2566} \left(\frac{b}{z}\right)^{5} + (-1)^{-\frac{1}{2}} \frac{51}{1024} \left(\frac{b}{z}\right)^{6} \\ &+ (-1)^{-\frac{1}{2}} \frac{429}{2048} \left(\frac{b}{z}\right)^{7} + (-1)^{-\frac{1}{2}} \frac{6435}{32768} \left(\frac{b}{z}\right)^{8} \dots \end{bmatrix} \end{bmatrix} \\ &= \frac{1}{2} \frac{1}{2z} + \frac{1}{2} \left(\frac{a+b}{2}\right) \frac{1}{z^{2}} + \frac{1}{2} \left(\frac{3}{8}a^{2} + \frac{1}{4}ab + \frac{3}{8}b^{2}\right) \frac{1}{z^{3}} \\ &+ \frac{1}{2} \left(\frac{5}{16}a^{3} + \frac{3}{16}a^{2}b + \frac{3}{16}ab^{2} + \frac{5}{16}b^{3}\right) \frac{1}{z^{4}} \\ &+ \frac{1}{2} \left(\frac{35}{128}a^{4} + \frac{5}{32}a^{3}b + \frac{9}{64}a^{2}b^{2} + \frac{5}{32}ab^{3} + \frac{35}{128}b^{4}\right) \frac{1}{z^{5}} \\ &+ \frac{1}{2} \left(\frac{63}{256}a^{5} + \frac{35}{256}a^{4}b + \frac{15}{128}a^{3}b^{2} + \frac{15}{128}a^{2}b^{3} + \frac{35}{256}ab^{4} + \frac{63}{256}b^{5}\right) \frac{1}{z^{6}} \end{aligned}$$

$$+ \frac{1}{2} \left(\begin{array}{c} \frac{231}{1024}a^6 + \frac{63}{512}a^5b + \frac{105}{1024}a^4b^2 + \frac{25}{256}a^3b^3 + \frac{105}{1024}a^2b^4 \\ + \frac{63}{512}ab^5 + \frac{231}{1024}b^6 \end{array} \right) \frac{1}{z^7} \\ + \frac{1}{2} \left(\begin{array}{c} \frac{429}{2048}a^7 + \frac{231}{2048}a^6b + \frac{189}{2048}a^5b^2 + \frac{175}{2048}a^4b^3 + \frac{175}{2048}a^3b^4 \\ + \frac{189}{2048}a^2b^5 + \frac{231}{2048}ab^6 + \frac{429}{2048}b^7 \end{array} \right) \frac{1}{z^8} \\ = \left(\begin{array}{c} h_{-1}z^{-1} + h_{-2}z^{-2} + h_{-3}z^{-3} + h_{-4}z^{-4} + h_{-5}z^{-5} \\ + h_{-6}z^{-6} + h_{-7}z^{-7} + h_{-8}z^{-8} \end{array} \right).$$

A3.3 Coefficients p_k of $\varphi_{11}(z)$ and $\psi_{11}(z)$

•

,

$$\varphi_1(z) = \frac{1}{2\sqrt{(z-a)(z-b)}}Y(z) + \frac{1}{2}X(z), \qquad (2.32)$$

$$\psi_1(z) = -\overline{X}(z) + \overline{\varphi_1}(z) - z\varphi_1'(z).$$
(2.33)

From (2.32), $\varphi_{11}(z)$ corresponding to $\varphi_1(z)$ in a domain |z| < a near a crack tip a is derived as follows

$$\begin{aligned} \varphi_{11}(z) & (A3.9) \\ &= \left[g_0 + g_1 z + g_2 z^2 + g_3 z^3 + g_4 z^4 + g_5 z^5 + g_6 z^6 + g_7 z^7 \right] \\ &\times \left[\dots f_{-4} z^{-4} + f_{-3} z^{-3} + f_{-2} z^{-2} + f_{-1} z^{-1} + f_0 + f_1 z^1 + f_2 z^2 + f_3 z^3 + f_4 z^4 \dots \right] \\ &+ \frac{1}{2} \left[\dots e_{-4} z^{-4} + e_{-3} z^{-3} + e_{-2} z^{-2} + e_{-1} z^{-1} + e_0 + e_1 z^1 + e_2 z^2 + e_3 z^3 + e_4 z^4 \dots \right] \\ &= \left(f_{0} g_7 + f_{1} g_6 + f_{2} g_5 + f_3 g_4 + f_4 g_3 \right) z^7 \\ &+ \left(f_{-1} g_7 + f_{0} g_6 + f_{1} g_5 + f_{2} g_4 + f_3 g_3 + f_4 g_2 \right) z^6 \\ &+ \left(f_{-2} g_7 + f_{-1} g_6 + f_{0} g_5 + f_{1} g_4 + f_{2} g_3 + f_3 g_2 + f_4 g_1 \right) z^5 \\ &+ \left(f_{-3} g_7 + f_{-2} g_6 + f_{-1} g_5 + f_{0} g_4 + f_{1} g_3 + f_{2} g_2 + f_3 g_1 + f_4 g_0 + \frac{1}{2} e_4 \right) z^4 \\ &+ \left(f_{-4} g_7 + f_{-3} g_6 + f_{-2} g_5 + f_{-1} g_4 + f_{0} g_3 + f_{1} g_2 + f_2 g_1 + f_3 g_0 + \frac{1}{2} e_3 \right) z^3 \end{aligned}$$

.

$$+ \begin{pmatrix} f_{-4}g_{6} + f_{-3}g_{5} + f_{-2}g_{4} + f_{-1}g_{3} + f_{0}g_{2} + f_{1}g_{1} + f_{2}g_{0} + \frac{1}{2}e_{2} \end{pmatrix} z^{2} \\ + \begin{pmatrix} f_{-4}g_{5} + f_{-3}g_{4} + f_{-2}g_{3} + f_{-1}g_{2} + f_{0}g_{1} + f_{1}g_{0} + \frac{1}{2}e_{1} \end{pmatrix} z^{1} \\ + \begin{pmatrix} f_{-4}g_{4} + f_{-3}g_{3} + f_{-2}g_{2} + f_{-1}g_{1} + f_{0}g_{0} + \frac{1}{2}e_{0} \end{pmatrix} z^{0} \\ + \begin{pmatrix} f_{-4}g_{4} + f_{-3}g_{3} + f_{-2}g_{2} + f_{-1}g_{1} + f_{-1}g_{0} + \frac{1}{2}e_{-1} \end{pmatrix} z^{-1} \\ + \begin{pmatrix} f_{-4}g_{3} + f_{-3}g_{2} + f_{-2}g_{1} + f_{-1}g_{0} + \frac{1}{2}e_{-2} \end{pmatrix} z^{-2} \\ + \begin{pmatrix} f_{-4}g_{2} + f_{-3}g_{1} + f_{-2}g_{0} + \frac{1}{2}e_{-2} \end{pmatrix} z^{-2} \\ + \begin{pmatrix} f_{-4}g_{2} + f_{-3}g_{1} + f_{-2}g_{0} + \frac{1}{2}e_{-2} \end{pmatrix} z^{-3} \\ + \begin{pmatrix} f_{-4}g_{1} + f_{-3}g_{0} + \frac{1}{2}e_{-3} \end{pmatrix} z^{-3} \\ + \begin{pmatrix} f_{-4}g_{0} + \frac{1}{2}e_{-4} \end{pmatrix} z^{-4} \\ \end{bmatrix} z^{-4}$$

.

4

.

Also, from (2.33), $\psi_{11}(z)$ corresponding to $\psi_1(z)$ in a domain |z| < a around a crack tip a is expressed as follows

$$\begin{split} \psi_{11}(z) & (A3.10) \\ &= -\overline{X}(z) + \overline{\varphi_{1}}(z) - z\varphi_{1}'(z) \\ &= -\left(\dots\overline{e}_{-4}z^{-4} + \overline{e}_{-3}z^{-3} + \overline{e}_{-2}z^{-2} + \overline{e}_{-1}z^{-1} + \overline{e}_{0} + \overline{e}_{1}z^{1} + \overline{e}_{2}z^{2} + \overline{e}_{3}z^{3} + \overline{e}_{4}z^{4} \dots\right) \\ &+ \left(\begin{array}{c} \overline{p}_{7}z^{7} + \overline{p}_{6}z^{6} + \overline{p}_{5}z^{5} + \overline{p}_{4}z^{4} + \overline{p}_{3}z^{3} + \overline{p}_{2}z^{2} + \overline{p}_{1}z^{1} + \overline{p}_{0} + \overline{p}_{-1}z^{-1} \\ &+ \overline{p}_{-2}z^{-2} + \overline{p}_{-3}z^{-3} + \overline{p}_{-4}z^{-4} \end{array} \right) \\ &- \left(\begin{array}{c} 7p_{7}z^{7} + 6p_{6}z^{6} + 5p_{5}z^{5} + 4p_{4}z^{4} + 3p_{3}z^{3} + 2p_{2}z^{2} + p_{1}z^{1} - p_{-1}z^{-1} \\ &- 2p_{-2}z^{-2} - 3p_{-3}z^{-3} - 4p_{-4}z^{-4} \end{array} \right) \\ &= -6p_{7}z^{7} - 5p_{6}z^{6} - 4p_{5}z^{5} - (3p_{4} + \overline{e}_{4})z^{4} - (2p_{3} + \overline{e}_{3})z^{3} - (p_{2} + \overline{e}_{2})z^{2} \\ &- \overline{e}_{1}z^{1} + (p_{0} - \overline{e}_{0}) + (2p_{-1} - \overline{e}_{-1})z^{-1} + (3p_{-2} - \overline{e}_{-2})z^{-2} \end{split}$$

•

$$+ (4p_{-3} - \overline{e}_{-3}) z^{-3} + (5p_{-4} - \overline{e}_{-4}) z^{-4}.$$

A3.4 Coefficients q_k of $\varphi_{12}(z)$ and $\psi_{12}(z)$

:

4

From (2.32), $\varphi_{12}(z)$ corresponding to $\varphi_1(z)$ in a domain |z| > b near a crack tip b takes the following form

$$\begin{split} \varphi_{12}(z) & (A3.11) \\ = \begin{bmatrix} h_{-1}z^{-1} + h_{-2}z^{-2} + h_{-3}z^{-3} + h_{-4}z^{-4} + h_{-5}z^{-5} \\ + h_{-6}z^{-6} + h_{-7}z^{-7} + h_{-8}z^{-8} \end{bmatrix} \\ & \times \begin{bmatrix} \dots f_{-4}z^{-4} + f_{-3}z^{-3} + f_{-2}z^{-2} + f_{-1}z^{-1} + f_0 + f_1z^1 \\ + f_2z^2 + f_3z^3 + f_4z^4 \dots \end{bmatrix} \\ & + \frac{1}{2} \begin{bmatrix} \dots e_{-4}z^{-4} + e_{-3}z^{-3} + e_{-2}z^{-2} + e_{-1}z^{-1} + e_0 + e_1z^1 \\ + e_2z^2 + e_3z^3 + e_4z^4 \dots \end{bmatrix} \\ & = \begin{pmatrix} h_{-1}f_4 + \frac{1}{2}e_3 \end{pmatrix} z^3 + \begin{pmatrix} h_{-2}f_4 + h_{-1}f_3 + \frac{1}{2}e_2 \end{pmatrix} z^2 \\ & + \begin{pmatrix} h_{-3}f_4 + h_{-2}f_3 + h_{-1}f_2 + \frac{1}{2}e_1 \end{pmatrix} z^1 \\ & + \begin{pmatrix} h_{-4}f_4 + h_{-3}f_3 + h_{-2}f_2 + h_{-1}f_1 + \frac{1}{2}e_0 \end{pmatrix} z^0 \\ & + \begin{pmatrix} h_{-5}f_4 + h_{-4}f_3 + h_{-3}f_2 + h_{-2}f_1 + h_{-1}f_0 + \frac{1}{2}e_{-1} \end{pmatrix} z^{-1} \\ & + \begin{pmatrix} h_{-6}f_4 + h_{-5}f_3 + h_{-4}f_2 + h_{-3}f_1 + h_{-2}f_0 + h_{-1}f_{-1} + \frac{1}{2}e_{-2} \end{pmatrix} z^{-2} \\ & + \begin{pmatrix} h_{-7}f_4 + h_{-6}f_3 + h_{-5}f_2 + h_{-4}f_1 + h_{-3}f_0 + h_{-2}f_{-1} \\ & + h_{-1}f_{-2} + \frac{1}{2}e_{-3} \end{pmatrix} z^{-3} \\ & + \begin{pmatrix} h_{-8}f_4 + h_{-7}f_3 + h_{-6}f_2 + h_{-5}f_1 + h_{-4}f_0 + h_{-3}f_{-1} + h_{-2}f_{-2} \\ & + h_{-1}f_{-3} + \frac{1}{2}e_{-4} \end{pmatrix} z^{-4} \end{split}$$

$$+ \begin{pmatrix} h_{-8}f_3 + h_{-7}f_2 + h_{-6}f_1 + h_{-5}f_0 + h_{-4}f_{-1} + h_{-3}f_{-2} + h_{-2}f_{-3} \\ + h_{-1}f_{-4} + \frac{1}{2}e_{-5} \end{pmatrix} z^{-5} \\ + \begin{pmatrix} h_{-8}f_2 + h_{-7}f_1 + h_{-6}f_0 + h_{-5}f_{-1} + h_{-4}f_{-2} + h_{-3}f_{-3} \\ + h_{-2}f_{-4} + \frac{1}{2}e_{-6} \end{pmatrix} z^{-6} \\ = q_3z^3 + q_2z^2 + q_1z^1 + q_0 + q_{-1}z^{-1} + q_{-2}z^{-2} + q_{-3}z^{-3} + q_{-4}z^{-4} \\ + q_{-5}z^{-5} + q_{-6}z^{-6}.$$

From (2.33), $\psi_{12}(z)$ corresponding to $\psi_1(z)$ in a domain |z| > b near a crack tip b can be written in the following form

÷

4

.

•

$$\begin{split} \psi_{12}(z) & (A3.12) \\ &= -\overline{X}(z) + \overline{\varphi_{12}}(z) - z\varphi_{12}'(z) \\ &= -\left(\dots\overline{e}_{-4}z^{-4} + \overline{e}_{-3}z^{-3} + \overline{e}_{-2}z^{-2} + \overline{e}_{-1}z^{-1} + \overline{e}_{0} + \overline{e}_{1}z^{1} + \overline{e}_{2}z^{2} + \overline{e}_{3}z^{3} + \overline{e}_{4}z^{4}\dots\right) \\ &+ \left(\begin{array}{c} q_{3}z^{3} + q_{2}z^{2} + q_{1}z^{1} + q_{0} + q_{-1}z^{-1} + q_{-2}z^{-2} + q_{-3}z^{-3} + q_{-4}z^{-4} \\ + q_{-5}z^{-5} + q_{-6}z^{-6} \end{array}\right) \\ &- \left(\begin{array}{c} 3q_{3}z^{3} + 2q_{2}z^{2} + q_{1}z^{1} - q_{-1}z^{-1} - 2q_{-2}z^{-2} - 3q_{-3}z^{-3} - 4q_{-4}z^{-4} \\ -5q_{-5}z^{-5} - 6q_{-6}z^{-6} \end{array}\right) \\ &= 7q_{-6}z^{-6} + 6q_{-5}z^{-5} + (5q_{-4}z^{-4} - \overline{e}_{-4}z^{-4}) + (4q_{-3}z^{-3} - \overline{e}_{-3}z^{-3}) \\ &+ (3q_{-2}z^{-2} - \overline{e}_{-2}z^{-2}) + (2q_{-1}z^{-1} - \overline{e}_{-1}z^{-1}) + (q_{0} - \overline{e}_{0}) - \overline{e}_{1}z^{1} \\ &+ (-\overline{e}_{2}z^{2} - q_{2}z^{2}) + (-\overline{e}_{3}z^{3} - 2q_{3}z^{3}) - \overline{e}_{4}z^{4}. \end{split}$$

...

APPENDIX 4 General Algebraic Equations

A4.1 Derivation of Equation (3.1)

The traction continuation condition along the Γ_0 (2.7) is

$$\varphi_{11}'(z) + \overline{\varphi_{11}'}\left(\frac{R_0^2}{z}\right) - z\varphi_{11}''(z) - \frac{z^2}{R_0^2}\psi_{11}'(z) = \varphi_0'(z) + \overline{\varphi_0'}\left(\frac{R_0^2}{z}\right) - z\varphi_0''(z) - \frac{z^2}{R_0^2}\psi_0'(z), \quad (2.7)$$

and the respective stress components defined in Chapter 2 are

$$\varphi_{11}(z) = p_7 z^7 + p_6 z^6 + p_5 z^5 + p_4 z^4 + p_3 z^3 + p_2 z^2 + p_1 z^1 + p_0 + p_{-1} z^{-1} + p_{-2} z^{-2} + p_{-3} z^{-3} + p_{-4} z^{-4},$$

$$\psi_{11}(z) = -6p_7 z^7 - 5p_6 z^6 - 4p_5 z^5 - (3p_4 + e_4) z^4$$

- $(2p_3 + e_3) z^3 - (p_2 + e_2) z^2 - e_1 z^1 + (p_0 - e_0)$
+ $(2p_{-1} - e_{-1}) z^{-1} + (3p_{-2} - e_{-2}) z^{-2} + (4p_{-3} - e_{-3}) z^{-3}$
+ $(5p_{-4} - e_{-4}) z^{-4}$,

$$\varphi_0(z) = \sum_{k=0}^{\infty} a_k z^k = a_0 + a_1 z + a_2 z^2 + a_3 z^3 + a_4 z^4 + a_5 z^5 + a_6 z^6 + a_7 z^7 + \dots ,$$

$$\psi_0(z) = \sum_{k=0}^{\infty} b_k z^k = b_0 + b_1 z + b_2 z^2 + b_3 z^3 + b_4 z^4 + b_5 z^5 + b_6 z^6 + b_7 z^7 + \dots .$$

Let's derive each item required to be substituted:

$$\varphi_{11}'(z) = 7p_7 z^6 + 6p_6 z^5 + 5p_5 z^4 + 4p_4 z^3 + 3p_3 z^2 + 2p_2 z^1$$

$$+ p_1 - p_{-1} z^{-2} - 2p_{-2} z^{-3} - 3p_{-3} z^{-4} - 4p_{-4} z^{-5},$$
(A4.1)

••

$$\begin{aligned} \overline{\varphi'_{11}} \left(\frac{R_0^2}{z}\right) &= 7p_7 \left(\frac{R_0^2}{z}\right)^6 + 6p_6 \left(\frac{R_0^2}{z}\right)^5 + 5p_5 \left(\frac{R_0^2}{z}\right)^4 + 4p_4 \left(\frac{R_0^2}{z}\right)^3 \quad (A4.2) \\ &+ 3p_3 \left(\frac{R_0^2}{z}\right)^2 + 2p_2 \left(\frac{R_0^2}{z}\right)^1 + p_1 - p_{-1} \left(\frac{R_0^2}{z}\right)^{-2} - 2p_{-2} \left(\frac{R_0^2}{z}\right)^{-3} \\ &- 3p_{-3} \left(\frac{R_0^2}{z}\right)^{-4} - 4p_{-4} \left(\frac{R_0^2}{z}\right)^{-5} \\ &= 7p_7 R_0^{12} z^{-6} + 6p_6 R_0^{10} z^{-5} + 5p_5 R_0^8 z^{-4} + 4p_4 R_0^6 z^{-3} + 3p_3 R_0^4 z^{-2} \\ &+ 2p_2 R_0^2 z^{-1} + p_1 - \frac{p_{-1}}{R_0^4} z^2 - \frac{2p_{-2}}{R_0^6} z^3 - \frac{3p_{-3}}{R_0^8} z^4 - \frac{4p_{-4}}{R_0^{10}} z^5, \end{aligned}$$

.

•

$$z\varphi_{11}''(z) = 42p_7 z^6 + 30p_6 z^5 + 20p_5 z^4 + 12p_4 z^3 + 6p_3 z^2 + 2p_2 z$$

$$+ 2p_{-1} z^{-2} + 6p_{-2} z^{-3} + 12p_{-3} z^{-4} + 20p_{-4} z^{-5},$$
(A4.3)

$$\frac{z^2}{R_0^2}\psi_{11}'(z) \qquad (A4.4)$$

$$= \frac{z^2}{R_0^2} \begin{pmatrix} -42p_7 z^6 - 30p_6 z^5 - 20p_5 z^4 - 4(3p_4 + e_4) z^3 - 3(2p_3 + e_3) z^2 \\ -2(p_2 + e_2) z^1 - e_1 - (2p_{-1} - e_{-1}) z^{-2} - 2(3p_{-2} - e_{-2}) z^{-3} \\ -3(4p_{-3} - e_{-3}) z^{-4} - 4(5p_{-4} - e_{-4}) z^{-5} \end{pmatrix}$$

$$= \frac{-\frac{42}{R_0^2} p_7 z^8 - \frac{30}{R_0^2} p_6 z^7 - \frac{20}{R_0^2} p_5 z^6 - \frac{1}{R_0^2} (12p_4 + 4e_4) z^5 - \frac{1}{R_0^2} (6p_3 + 3e_3) z^4 \\ = \frac{-\frac{1}{R_0^2} (2p_2 + 2e_2) z^3 - e_1 \frac{1}{R_0^2} z^2 - \frac{1}{R_0^2} (2p_{-1} - e_{-1})}{-\frac{1}{R_0^2} (6p_{-2} - 2e_{-2}) z^{-1} - \frac{1}{R_0^2} (12p_{-3} - 3e_{-3}) z^{-2} - \frac{1}{R_0^2} (20p_{-4} - 4e_{-4}) z^{-3},$$

$$\varphi_0'(z)' = (a_0 + a_1 z + a_2 z^2 + a_3 z^3 + a_4 z^4 + a_5 z^5 + a_6 z^6 + a_7 z^7)'$$

$$= 7a_7 z^6 + 6a_6 z^5 + 5a_5 z^4 + 4a_4 z^3 + 3a_3 z^2 + 2a_2 z + a_1,$$
(A4.5)

$$\overline{\varphi_0'}\left(\frac{R_0^2}{z}\right) = 7R_0^{12}a_7z^{-6} + 6R_0^{10}a_6z^{-5} + 5R_0^8a_5z^{-4} + 4R_0^6a_4z^{-3}$$
(A4.6)

•

.

.

.

••

$$+3R_0^4a_3z^{-2}+2R_0^2a_2z^{-1}+a_1,$$

$$z\varphi_0''(z) = z \left(7a_7 z^6 + 6a_6 z^5 + 5a_5 z^4 + 4a_4 z^3 + 3a_3 z^2 + 2a_2 z + a_1\right)'$$
(A4.7)
= $42a_7 z^6 + 30a_6 z^5 + 20a_5 z^4 + 12a_4 z^3 + 6a_3 z^2 + 2a_2 z,$

•

$$\frac{z^2}{R_0^2}\psi_0'(z) = \frac{z^2}{R_0^2} \left(b_0 + b_1z + b_2z^2 + b_3z^3 + b_4z^4 + b_5z^5 + b_6z^6 + b_7z^7\right)'$$
(A4.8)
$$= \frac{1}{R_0^2} b_1z^2 + \frac{2}{R_0^2} b_2z^3 + \frac{3}{R_0^2} b_3z^4 + \frac{4}{R_0^2} b_4z^5 + \frac{5}{R_0^2} b_5z^6 + \frac{6}{R_0^2} b_6z^7 + \frac{7}{R_0^2} b_7z^8.$$

Thus, by substituting (A4.1) thru (A4.4) into (2.7), the L.H.S. of (2.7) becomes

$$\begin{split} \varphi_{11}'(z) + \overline{\varphi_{11}'} \left(\frac{R_0^2}{z}\right) &- z\varphi_{11}''(z) - \frac{z^2}{R_0^2} \psi_{11}'(z) \qquad (A4.9) \\ &= \begin{pmatrix} 7p_7 z^6 + 6p_6 z^5 + 5p_5 z^4 + 4p_4 z^3 + 3p_3 z^2 + 2p_2 z^1 + p_1 - p_{-1} z^{-2} - 2p_{-2} z^{-3} \\ &- 3p_{-3} z^{-4} - 4p_{-4} z^{-5} \end{pmatrix} \\ &+ \begin{pmatrix} 7p_7 R_0^{12} z^{-6} + 6p_6 R_0^{10} z^{-5} + 5p_5 R_0^8 z^{-4} + 4p_4 R_0^6 z^{-3} + 3p_3 R_0^4 z^{-2} + 2p_2 R_0^2 z^{-1} \\ &+ p_1 - \frac{p_{-1}}{R_0^4} z^2 - \frac{2p_{-2}}{R_0^6} z^3 - \frac{3p_{-3}}{R_0^8} z^4 - \frac{4p_{-4}}{R_0^{10}} z^5 \end{pmatrix} \\ &- \begin{pmatrix} 42p_7 z^6 + 30p_6 z^5 + 20p_5 z^4 + 12p_4 z^3 + 6p_3 z^2 + 2p_2 z + 2p_{-1} z^{-2} + 6p_{-2} z^{-3} \\ &+ 12p_{-3} z^{-4} + 20p_{-4} z^{-5} \end{pmatrix} \\ &- \begin{pmatrix} -\frac{42}{R_0^2} p_7 z^8 - \frac{30}{R_0^2} p_6 z^7 - \frac{20}{R_0^2} p_5 z^6 - \frac{1}{R_0^2} (12p_4 + 4e_4) z^5 - \frac{1}{R_0^2} (6p_3 + 3e_3) z^4 \\ &- \frac{1}{R_0^2} (6p_{-2} - 2e_{-2}) z^{-1} - \frac{1}{R_0^2} (12p_{-3} - 3e_{-3}) z^{-2} - \frac{1}{R_0^2} (20p_{-4} - 4e_{-4}) z^{-3} \end{pmatrix} \\ &= \frac{42}{R_0^2} p_7 z^8 + \frac{30}{R_0^2} p_6 z^7 + \begin{pmatrix} -35p_7 + \frac{20}{R_0^2} p_5 \end{pmatrix} z^6 + \begin{pmatrix} \frac{1}{R_0^2} (12p_4 + 4e_4) - 24p_6 - \frac{4p_{-4}}{R_0^{10}} \end{pmatrix} z^5 \\ &+ \begin{pmatrix} \frac{1}{R_0^2} (6p_3 + 3e_3) - 15p_5 - \frac{3p_{-3}}{R_0^8} \end{pmatrix} z^4 + \begin{pmatrix} \frac{1}{R_0^2} (2p_2 + 2e_2) - 8p_4 - \frac{2p_{-2}}{R_0^6} \end{pmatrix} z^3 \end{split}$$

•

٠

.

÷
$$+ \left(\frac{1}{R_0^2}e_1 - 3p_3 - \frac{p_{-1}}{R_0^4}\right)z^2 + (0)z + \left(\frac{1}{R_0^2}\left(2p_{-1} - e_{-1}\right) + 2p_1\right) \\ + \left(2R_0^2p_2 + \frac{1}{R_0^2}\left(6p_{-2} - 2e_{-2}\right)\right)z^{-1} + \left(\frac{1}{R_0^2}\left(12p_{-3} - 3e_{-3}\right) - 3p_{-1} + 3R_0^4p_3\right)z^{-2} \\ + \left(\frac{1}{R_0^2}\left(20p_{-4} - 4e_{-4}\right) - 8p_{-2} + 4R_0^6p_4\right)z^{-3} + \left(5R_0^8p_5 - 15p_{-3}\right)z^{-4} \\ + \left(6R_0^{10}p_6 - 24p_{-4}\right)z^{-5} + 7p_7R_0^{12}z^{-6}.$$

Similarly, substituting (A4.5) thru (A5.8) into R.H.S. of (2.7) yields

$$\begin{split} \varphi_0'(z) + \overline{\varphi_0'} \left(\frac{R_0^2}{z}\right) &- z\varphi_0''(z) - \frac{z^2}{R_0^2} \psi_0'(z) \quad (A4.10) \\ = & \left(7a_7 z^6 + 6a_6 z^5 + 5a_5 z^4 + 4a_4 z^3 + 3a_3 z^2 + 2a_2 z + a_1\right) \\ &+ \left(7R_0^{12} a_7 z^{-6} + 6R_0^{10} a_6 z^{-5} + 5R_0^8 a_5 z^{-4} + 4R_0^6 a_4 z^{-3} + 3R_0^4 a_3 z^{-2} + 2R_0^2 a_2 z^{-1} + a_1\right) \\ &- \left(42a_7 z^6 + 30a_6 z^5 + 20a_5 z^4 + 12a_4 z^3 + 6a_3 z^2 + 2a_2 z\right) \\ &- \left(\frac{1}{R_0^2} b_1 z^2 + \frac{2}{R_0^2} b_2 z^3 + \frac{3}{R_0^2} b_3 z^4 + \frac{4}{R_0^2} b_4 z^5 + \frac{5}{R_0^2} b_5 z^6 + \frac{6}{R_0^2} b_6 z^7 + \frac{7}{R_0^2} b_7 z^8\right) \\ &= & -\frac{7}{R_0^2} b_7 z^8 - \frac{6}{R_0^2} b_6 z^7 + \left(-35a_7 - \frac{5}{R_0^2} b_5\right) z^6 + \left(-24a_6 - \frac{4}{R_0^2} b_4\right) z^5 \\ &+ \left(-15a_5 - \frac{3}{R_0^2} b_3\right) z^4 + \left(-8a_4 - \frac{2}{R_0^2} b_2\right) z^3 + \left(-3a_3 - \frac{1}{R_0^2} b_1\right) z^2 \\ &+ (0) z + 2a_1 + 7R_0^{12} a_7 z^{-6} + 6R_0^{10} a_6 z^{-5} + 5R_0^8 a_5 z^{-4} + 4R_0^6 a_4 z^{-3} \\ &+ 3R_0^4 a_3 z^{-2} + 2R_0^2 a_2 z^{-1}. \end{split}$$

Thus, (2.7) is re-written in the following form

$$\frac{42}{R_0^2} p_7 z^8 + \frac{30}{R_0^2} p_6 z^7 + \left(-35p_7 + \frac{20}{R_0^2} p_5\right) z^6$$

$$+ \left(\frac{1}{R_0^2} \left(12p_4 + 4e_4\right) - 24p_6 - \frac{4p_{-4}}{R_0^{10}}\right) z^5 + \left(\frac{1}{R_0^2} \left(6p_3 + 3e_3\right) - 15p_5 - \frac{3p_{-3}}{R_0^8}\right) z^4$$

$$+ \left(\frac{1}{R_0^2} \left(2p_2 + 2e_2\right) - 8p_4 - \frac{2p_{-2}}{R_0^6}\right) z^3 + \left(\frac{1}{R_0^2} e_1 - 3p_3 - \frac{p_{-1}}{R_0^4}\right) z^2 + (0) z$$
(3.1)

÷

$$+ \left(\frac{1}{R_0^2} \left(2p_{-1} - e_{-1}\right) + 2p_1\right) \\ + \left(\frac{1}{R_0^2} \left(12p_{-3} - 3e_{-3}\right) - 3p_{-1} + 3R_0^4 p_3\right) z^{-2} \\ + \left(\frac{1}{R_0^2} \left(20p_{-4} - 4e_{-4}\right) - 8p_{-2} + 4R_0^6 p_4\right) z^{-3} + \left(5R_0^8 p_5 - 15p_{-3}\right) z^{-4} \\ + \left(6R_0^{10} p_6 - 24p_{-4}\right) z^{-5} + 7p_7 R_0^{12} z^{-6} \\ = -\frac{7}{R_0^2} b_7 z^8 - \frac{6}{R_0^2} b_6 z^7 + \left(-35a_7 - \frac{5}{R_0^2} b_5\right) z^6 + \left(-24a_6 - \frac{4}{R_0^2} b_4\right) z^5 \\ + \left(-15a_5 - \frac{3}{R_0^2} b_3\right) z^4 + \left(-8a_4 - \frac{2}{R_0^2} b_2\right) z^3 + \left(-3a_3 - \frac{1}{R_0^2} b_1\right) z^2 + (0) z \\ + 2a_1 + 2R_0^2 a_2 z^{-1} + 3R_0^4 a_3 z^{-2} + 4R_0^6 a_4 z^{-3} + 5R_0^8 a_5 z^{-4} + 6R_0^{10} a_6 z^{-5} + 7R_0^{12} a_7 z^{-6}.$$

A4.2 Derivation of Equation (3.2)

The displacement jump condition along the Γ_0 (2.12) is

$$(m_0 - n_0)(\kappa_1 + 1)\frac{R_0}{z}\varphi_{11}(z) + (m_0 + n_0)(\kappa_1 + 1)\frac{z}{R_0}\overline{\varphi_{11}}\left(\frac{R_0^2}{z}\right)$$
(2.12)
= $4\mu_1\left(\varphi_0'(z) + \overline{\varphi_0'}\left(\frac{R_0^2}{z}\right) - z\varphi_0''(z) - \frac{z^2}{R_0^2}\psi_0'(z)\right) + (m_0 - n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})\frac{R_0}{z}\varphi_0(z)$
 $+ (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})\left(R_0\overline{\varphi_0'}\left(\frac{R_0^2}{z}\right) + \frac{R_0}{z}\overline{\psi_0}\left(\frac{R_0^2}{z}\right)\right)$
 $+ (m_0 + n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})\frac{z}{R_0}\overline{\varphi_0}(\frac{R_0^2}{z}) + (m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})\left(R_0\varphi_0'(z) + \frac{z}{R_0}\psi_0(z)\right)$
 $+ 4m_0R_0\mu_1\varepsilon_1 + \frac{2\mu_1(m_0 + n_0)(\varepsilon_2 - i\varepsilon_3)}{R_0}z^2 + \frac{2\mu_1(m_0 - n_0)(\varepsilon_2 + i\varepsilon_3)R_0^3}{z^2}.$

Let's derive each item required to be substituted:

.

۰,

$$\frac{R_0}{z}\varphi_{11}(z) = \frac{R_0}{z} \begin{pmatrix} p_7 z^7 + p_6 z^6 + p_5 z^5 + p_4 z^4 + p_3 z^3 + p_2 z^2 + p_1 z^1 \\ + p_0 + p_{-1} z^{-1} + p_{-2} z^{-2} + p_{-3} z^{-3} + p_{-4} z^{-4} \end{pmatrix} (A4.11)$$

$$= R_0 p_7 z^6 + R_0 p_6 z^5 + R_0 p_5 z^4 + R_0 p_4 z^3 + R_0 p_3 z^2 + R_0 p_2 z + R_0 p_1 \\ + R_0 p_0 z^{-1} + R_0 p_{-1} z^{-2} + R_0 p_{-2} z^{-3} + R_0 p_{-3} z^{-4} + R_0 p_{-4} z^{-5},$$

÷

$$\frac{z}{R_{0}}\overline{\varphi_{11}}\left(\frac{R_{0}^{2}}{z}\right) = \frac{z}{R_{0}} \begin{pmatrix} p_{7}\left(R_{0}^{2}z^{-1}\right)^{7} + p_{6}\left(R_{0}^{2}z^{-1}\right)^{6} + p_{5}\left(R_{0}^{2}z^{-1}\right)^{5} \\ + p_{4}\left(R_{0}^{2}z^{-1}\right)^{4} + p_{3}\left(R_{0}^{2}z^{-1}\right)^{3} + p_{2}\left(R_{0}^{2}z^{-1}\right)^{2} \\ + p_{1}\left(R_{0}^{2}z^{-1}\right)^{1} + p_{0} + p_{-1}\left(R_{0}^{2}z^{-1}\right)^{-1} \\ + p_{-2}\left(R_{0}^{2}z^{-1}\right)^{-2} + p_{-3}\left(R_{0}^{2}z^{-1}\right)^{-3} + p_{-4}\left(R_{0}^{2}z^{-1}\right)^{-4} \end{pmatrix} \right)$$

$$= p_{7}R_{0}^{13}z^{-6} + p_{6}R_{0}^{11}z^{-5} + p_{5}R_{0}^{9}z^{-4} + p_{4}R_{0}^{7}z^{-3} + p_{3}R_{0}^{5}z^{-2} + p_{2}R_{0}^{3}z^{-1} \\ + p_{1}R_{0} + \frac{1}{R_{0}}p_{0}z + \frac{1}{R_{0}^{3}}p_{-1}z^{2} + \frac{1}{R_{0}^{5}}p_{-2}z^{3} + \frac{1}{R_{0}^{7}}p_{-3}z^{4} + \frac{1}{R_{0}^{9}}p_{-4}z^{5}, \end{cases}$$

 $\varphi'_0(z), \overline{\varphi'_0}\left(\frac{R_0^2}{z}\right), z\varphi''_0(z), \text{ and } \frac{z^2}{R_0^2}\psi'_0(z) \text{ are derived in the previous condition (A4.5 thru A4.8).}$

$$\frac{R_0}{z}\varphi_0(z) = R_0a_0z^{-1} + R_0a_1 + R_0a_2z + R_0a_3z^2 + R_0a_4z^3 + R_0a_5z^4 + R_0a_6z^5 + R_0a_7z^6,$$
(A4.13)

$$R_{0}\overline{\varphi_{0}'}\left(\frac{R_{0}^{2}}{z}\right) = 7R_{0}^{13}a_{7}z^{-6} + 6R_{0}^{11}a_{6}z^{-5} + 5R_{0}^{9}a_{5}z^{-4} + 4R_{0}^{7}a_{4}z^{-3}$$
(A4.14)
+ $3R_{0}^{5}a_{3}z^{-2} + 2R_{0}^{3}a_{2}z^{-1} + R_{0}a_{1},$

$$\frac{R_0}{z}\overline{\psi}_0\left(\frac{R_0^2}{z}\right) = \frac{R_0}{z} \begin{pmatrix} b_0 + b_1\left(\frac{R_0^2}{z}\right) + b_2\left(\frac{R_0^2}{z}\right)^2 + b_3\left(\frac{R_0^2}{z}\right)^3 \\ + b_4\left(\frac{R_0^2}{z}\right)^4 + b_5\left(\frac{R_0^2}{z}\right)^5 + b_6\left(\frac{R_0^2}{z}\right)^6 + b_7\left(\frac{R_0^2}{z}\right)^7 \end{pmatrix} (A4.15)$$

$$= R_0b_0z^{-1} + R_0^3b_1z^{-2} + R_0^5b_2z^{-3} + R_0^7b_3z^{-4} + R_0^9b_4z^{-5} + R_0^{11}b_5z^{-6} \\ + R_0^{13}b_6z^{-7} + R_0^{15}b_7z^{-8},$$

$$\frac{z}{R_0}\overline{\varphi_0}(\frac{R_0^2}{z}) = \frac{1}{R_0}a_0z + R_0a_1 + R_0^3a_2z^{-1} + R_0^5a_3z^{-2} + R_0^7a_4z^{-3} + R_0^9a_5z^{-4} + R_0^{11}a_6z^{-5} + R_0^{13}a_7z^{-6},$$
(A4.16)

••

$$R_{0}\varphi_{0}'(z) = 7R_{0}a_{7}z^{6} + 6R_{0}a_{6}z^{5} + 5R_{0}a_{5}z^{4}$$

$$+4R_{0}a_{4}z^{3} + 3R_{0}a_{3}z^{2} + 2R_{0}a_{2}z + R_{0}a_{1},$$
(A4.17)

$$\frac{z}{R_0}\psi_0(z) = \frac{1}{R_0}b_0z + \frac{1}{R_0}b_1z^2 + \frac{1}{R_0}b_2z^3 + \frac{1}{R_0}b_3z^4 + \frac{1}{R_0}b_4z^5 + \frac{1}{R_0}b_5z^6 + \frac{1}{R_0}b_6z^7 + \frac{1}{R_0}b_7z^8.$$
(A4.18)

Let's substitute (A4.11) and (A4.12) into L.H.S. of (2.12). Thus, L.H.S. of (2.12) is

$$(m_{0} - n_{0})(\kappa_{1} + 1)\frac{R_{0}}{z}\varphi_{11}(z) + (m_{0} + n_{0})(\kappa_{1} + 1)\frac{z}{R_{0}}\overline{\varphi_{11}}\left(\frac{R_{0}^{2}}{z}\right)$$
(A4.19)

$$= (m_{0} - n_{0})(\kappa_{1} + 1) \begin{pmatrix} R_{0}p_{7}z^{6} + R_{0}p_{6}z^{5} + R_{0}p_{5}z^{4} + R_{0}p_{4}z^{3} \\ + R_{0}p_{3}z^{2} + R_{0}p_{2}z + R_{0}p_{1} + R_{0}p_{0}z^{-1} \\ + R_{0}p_{-1}z^{-2} + R_{0}p_{-2}z^{-3} + R_{0}p_{-3}z^{-4} + R_{0}p_{-4}z^{-5} \end{pmatrix}$$

$$+ (m_{0} + n_{0})(\kappa_{1} + 1) \begin{pmatrix} p_{7}R_{0}^{13}z^{-6} + p_{6}R_{0}^{11}z^{-5} + p_{5}R_{0}^{9}z^{-4} + p_{4}R_{0}^{7}z^{-3} \\ + p_{3}R_{0}^{5}z^{-2} + p_{2}R_{0}^{3}z^{-1} + p_{1}R_{0} + \frac{1}{R_{0}}p_{0}z \\ + \frac{1}{R_{0}^{3}}p_{-1}z^{2} + \frac{1}{R_{0}^{5}}p_{-2}z^{3} + \frac{1}{R_{0}^{7}}p_{-3}z^{4} + \frac{1}{R_{0}^{9}}p_{-4}z^{5} \end{pmatrix}$$

$$= (m_{0} - n_{0})(\kappa_{1} + 1)R_{0}p_{7}z^{6} \\ + \left((m_{0} - n_{0})(\kappa_{1} + 1)R_{0}p_{5} + (m_{0} + n_{0})(\kappa_{1} + 1)\frac{1}{R_{0}^{7}}p_{-3}\right)z^{4} \\ + \left((m_{0} - n_{0})(\kappa_{1} + 1)R_{0}p_{4} + (m_{0} + n_{0})(\kappa_{1} + 1)\frac{1}{R_{0}^{5}}p_{-2}z^{3} \\ + \left((m_{0} - n_{0})(\kappa_{1} + 1)R_{0}p_{4} + (m_{0} + n_{0})(\kappa_{1} + 1)\frac{1}{R_{0}^{3}}p_{-1}z^{2} + \frac{1}{(m_{0} - m_{0})(\kappa_{1} + 1)R_{0}p_{4} + (m_{0} + n_{0})(\kappa_{1} + 1)\frac{1}{R_{0}^{3}}p_{-1}z^{2} + \frac{1}{(m_{0} - m_{0})(\kappa_{1} + 1)R_{0}p_{2} + (m_{0} + n_{0})(\kappa_{1} + 1)\frac{1}{R_{0}^{3}}p_{-1}z^{2} + \frac{1}{(m_{0} - m_{0})(\kappa_{1} + 1)R_{0}p_{2} + (m_{0} + n_{0})(\kappa_{1} + 1)\frac{1}{R_{0}^{3}}p_{-1}z^{2} + \frac{1}{(m_{0} - m_{0})(\kappa_{1} + 1)R_{0}p_{2} + (m_{0} + n_{0})(\kappa_{1} + 1)\frac{1}{R_{0}^{3}}p_{-1}z^{2} + \frac{1}{(m_{0} - m_{0})(\kappa_{1} + 1)R_{0}p_{2} + (m_{0} + m_{0})(\kappa_{1} + 1)\frac{1}{R_{0}^{3}}p_{-1}z^{2} + \frac{1}{(m_{0} - m_{0})(\kappa_{1} + 1)R_{0}p_{2} + (m_{0} + m_{0})(\kappa_{1} + 1)\frac{1}{R_{0}}p_{0}z^{2} + \frac{1}{(m_{0} - m_{0})(\kappa_{1} + 1)R_{0}p_{2} + (m_{0} + m_{0})(\kappa_{1} + 1)\frac{1}{R_{0}}p_{0}z^{2} + \frac{1}{(m_{0} - m_{0})(\kappa_{1} + 1)R_{0}p_{2} + (m_{0} + m_{0})(\kappa_{1} + 1)\frac{1}{R_{0}}p_{0}z^{2} + \frac{1}{(m_{0} - m_{0})(\kappa_{1} + 1)R_{0}p_{2} + (m_{0} + m_{0})(\kappa_{1} + 1)\frac{1}{R_{0}}p_{0}z^{2} + \frac{1}{(m_{0} - m_{0})(\kappa_{1} + 1)R_{0}p_{0}z^{2} + \frac{1}{(m_{0} - m_{0})(\kappa_{1} + 1)R_{0}p_{0}z^{2} + \frac{1}{(m_{0} - m_{0})(\kappa$$

 $+2m_0(\kappa_1+1)R_0p_1 + ((m_0-n_0)(\kappa_1+1)R_0p_0 + (m_0+n_0)(\kappa_1+1)R_0^3p_2)z^{-1}$

••

.

138

•

$$+ \left((m_0 - n_0)(\kappa_1 + 1)R_0p_{-1} + (m_0 + n_0)(\kappa_1 + 1)R_0^5p_3 \right) z^{-2} + \left((m_0 + n_0)(\kappa_1 + 1)R_0^7p_4 + (m_0 - n_0)(\kappa_1 + 1)R_0p_{-2} \right) z^{-3} + \left((m_0 + n_0)(\kappa_1 + 1)R_0^9p_5 + (m_0 - n_0)(\kappa_1 + 1)R_0p_{-3} \right) z^{-4} + \left((m_0 - n_0)(\kappa_1 + 1)R_0p_{-4} + (m_0 + n_0)(\kappa_1 + 1)R_0^{11}p_6 \right) z^{-5} + (m_0 + n_0)(\kappa_1 + 1)R_0^{13}p_7z^{-6}.$$

•

By deploying (A4.13) thru (A4.18) into (2.12), the R.H.S. of (2.12) becomes

$$\begin{split} & 4\mu_1 \left(\varphi_0'(z) + \overline{\varphi_0'} \left(\frac{R_0^2}{z} \right) - z\varphi_0''(z) - \frac{z^2}{R_0^2} \psi_0'(z) \right) + (m_0 - n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0}) \frac{R_0}{z} \varphi_0(z) \\ & + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0}) \left(R_0 \overline{\varphi_0'} \left(\frac{R_0^2}{z} \right) + \frac{R_0}{z} \overline{\psi_0} \left(\frac{R_0^2}{z} \right) \right) \\ & (A4.20) \\ & + (m_0 + n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0}) \frac{z}{R_0} \overline{\varphi_0} \left(\frac{R_0^2}{z} \right) + (m_0 + n_0)(1 - \frac{\mu_1}{\mu_0}) \left(R_0 \varphi_0'(z) + \frac{z}{R_0} \psi_0(z) \right) \\ & + 4m_0 R_0 \mu_1 \varepsilon_1 + \frac{2\mu_1(m_0 + n_0)(\varepsilon_2 - i\varepsilon_3)}{R_0} z^2 + \frac{2\mu_1(m_0 - n_0)(\varepsilon_2 + i\varepsilon_3)R_0^3}{z^2} \\ & = 4\mu_1 \left(-\frac{7}{R_0^2} b_7 z^8 - \frac{6}{R_0^2} b_6 z^7 + \left(-35a_7 - \frac{5}{R_0^2} b_5 \right) z^6 + \left(-24a_6 - \frac{4}{R_0^2} b_4 \right) z^5 \\ & + \left(-15a_5 - \frac{3}{R_0^2} b_3 \right) z^4 + \left(-8a_4 - \frac{2}{R_0^2} b_2 \right) z^3 + \left(-3a_3 - \frac{1}{R_0^2} b_1 \right) z^2 \\ & + (0) z + 2a_1 + 7R_0^{12} a_7 z^{-6} + 6R_0^{10} a_6 z^{-5} + 5R_0^8 a_5 z^{-4} \\ & + 4R_0^6 a_4 z^{-3} + 3R_0^4 a_3 z^{-2} + 2R_0^2 a_2 z^{-1} \\ & + (m_0 - n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0}) \left(\frac{R_0 a_0 z^{-1} + R_0 a_1 + R_0 a_2 z + R_0 a_3 z^2 + R_0 a_4 z^3}{R_0^4 a_2 z^{-1} + R_0^3 b_1 z^{-2} + 2R_0^3 a_2 z^{-1} + R_0 a_1 + R_0 b_0 z^{-1} \\ & + R_0^3 b_1 z^{-2} + R_0^5 b_2 z^{-3} + R_0^7 b_3 z^{-4} + R_0^9 b_4 z^{-5} \\ & + R_0^{11} b_5 z^{-6} + R_0^{13} b_6 z^{-7} + R_0^{15} b_7 z^{-8} \end{array} \right)$$

· .

÷

.

o

$$\begin{split} &+(m_{0}+n_{0})(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}})\left(\frac{\frac{1}{R_{0}}a_{0}z+R_{0}a_{1}+R_{0}^{3}a_{2}z^{-1}+R_{0}^{5}a_{3}z^{-2}+R_{0}^{7}a_{4}z^{-3}}{+R_{0}^{3}a_{5}z^{-4}+R_{0}^{11}a_{6}z^{-5}+R_{0}^{13}a_{7}z^{-6}}\right)\right)\\ &+(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})\left(\frac{7R_{0}a_{7}z^{6}+6R_{0}a_{6}z^{5}+5R_{0}a_{5}z^{4}+4R_{0}a_{4}z^{3}}{+3R_{0}a_{3}z^{2}+2R_{0}a_{2}z+R_{0}a_{1}+\frac{1}{R_{0}}b_{4}z^{5}}{+\frac{1}{R_{0}}b_{5}z^{6}+\frac{1}{R_{0}}b_{5}z^{4}+\frac{1}{R_{0}}b_{4}z^{5}}{+\frac{1}{R_{0}}b_{5}z^{6}+\frac{1}{R_{0}}b_{5}z^{4}+\frac{1}{R_{0}}b_{4}z^{5}}{+\frac{1}{R_{0}}b_{5}z^{6}+\frac{1}{R_{0}}b_{5}z^{6}+\frac{1}{R_{0}}b_{7}z^{8}}\right)\\ &+4m_{0}R_{0}\mu_{1}\varepsilon_{1}+\frac{2\mu_{1}(m_{0}+n_{0})(\varepsilon_{2}-i\varepsilon_{3})}{R_{0}}z^{2}+\frac{2\mu_{1}(m_{0}-n_{0})(\varepsilon_{2}+i\varepsilon_{3})R_{0}^{3}}{z^{2}}\\ &=\left(-4\mu_{1}\frac{7}{R_{0}^{2}}b_{7}+(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})\frac{1}{R_{0}}b_{7}\right)z^{8}\\ &+\left(-4\mu_{1}\frac{6}{R_{0}^{2}}b_{6}+(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})\frac{1}{R_{0}}b_{7}\right)z^{8}\\ &+\left(4\mu_{1}\left(-35a_{7}-\frac{5}{R_{0}^{2}}b_{5}\right)+(m_{0}-n_{0})(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}a_{7}\\ &+\left(4\mu_{1}\left(-24a_{6}-\frac{4}{R_{0}^{2}}b_{4}\right)+(m_{0}-n_{0})(1-\mu_{0})\frac{1}{R_{0}}b_{5}\right)z^{6}\\ &+\left(4\mu_{1}\left(-15a_{5}-\frac{3}{R_{0}^{2}}b_{3}\right)+(m_{0}-n_{0})(1-\mu_{0}\frac{\mu_{1}}{\mu_{0}})\frac{1}{R_{0}}b_{5}\right)z^{4}\\ &+\left(4\mu_{1}\left(-15a_{5}-\frac{3}{R_{0}^{2}}b_{3}\right)+(m_{0}-n_{0})(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}a_{4}\\ &+5(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})R_{0}a_{5}+(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})\frac{1}{R_{0}}b_{5}\right)z^{3}\\ &+\left(4\mu_{1}\left(-8a_{4}-\frac{2}{R_{0}^{2}}b_{2}\right)+(m_{0}-n_{0})(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}a_{3}}\\ &+\left(4\mu_{1}\left(-3a_{3}-\frac{1}{R_{0}^{2}}b_{1}\right)+(m_{0}-n_{0})(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}a_{3}}\\ &+3(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})R_{0}a_{3}+(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})\frac{1}{R_{0}}b_{2}}\right)z^{3}\\ &+\left(\frac{(m_{0}-n_{0})(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}a_{3}+(m_{0}+n_{0})(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}a_{3}}\\ &+\left((m_{0}-n_{0})(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}a_{2}+(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})\frac{1}{R_{0}}b_{2}}\right)z^{4}\\ &+\left(\frac{(m_{0}-n_{0})(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}a_{2}+(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})R_{0}a_{2}}\right)z^{4}\\ &+\left(\frac{(m_{0}-n_{0})(1-\frac{\mu_{1}}{\mu_{0}})R_{0}a_{2}+(m_{0$$

•

.

`

$$+ \begin{pmatrix} 8\mu_1a_1 + (m_0 - n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})R_0a_1 + R_0(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})a_1 \\ + (m_0 + n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})R_0a_1 + 4m_0R_0\mu_1\epsilon_1 + (m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})R_0a_1 \end{pmatrix} \\ + \begin{pmatrix} 8\mu_1R_0^2a_2 + (m_0 - n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})R_0a_0 + 2(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^3a_2 \\ + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0b_0 + (m_0 + n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})R_0^3b_1 \\ + (m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})R_0^5a_3 + 2\mu_1(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^3b_1 \\ + (m_0 + n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})R_0^5a_3 + 2\mu_1(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^5b_2 \\ + (m_0 + n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})R_0^7a_4 + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^5b_2 \\ + (m_0 + n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})R_0^7a_5 + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^7b_3 \\ + \begin{pmatrix} 24\mu_1R_0^{10}a_6 + 6(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^{10}a_6 + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^9b_4 \\ + (m_0 + n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})R_0^{11}a_6 + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^{11}b_5 \\ + (m_0 + n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})R_0^{13}a_7 + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^{11}b_5 \\ + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^{11}b_6z^{-7} + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^{15}b_7z^{-8}. \end{pmatrix} z^{-6}$$

,

. ••

Consequently, (2.12) is expressed in terms of complex coefficients as follows

$$(m_{0} - n_{0})(\kappa_{1} + 1)R_{0}p_{7}z^{6} + \left((m_{0} - n_{0})(\kappa_{1} + 1)R_{0}p_{6} + (m_{0} + n_{0})(\kappa_{1} + 1)\frac{1}{R_{0}^{9}}p_{-4}\right)z^{5} + \left((m_{0} - n_{0})(\kappa_{1} + 1)R_{0}p_{5} + (m_{0} + n_{0})(\kappa_{1} + 1)\frac{1}{R_{0}^{7}}p_{-3}\right)z^{4}$$

$$+ \left((m_{0} - n_{0})(\kappa_{1} + 1)R_{0}p_{4} + (m_{0} + n_{0})(\kappa_{1} + 1)\frac{1}{R_{0}^{5}}p_{-2}\right)z^{3} + \left((m_{0} - n_{0})(\kappa_{1} + 1)R_{0}p_{3} + (m_{0} + n_{0})(\kappa_{1} + 1)\frac{1}{R_{0}^{3}}p_{-1}\right)z^{2}$$

$$(3.2)$$

,

۰

$$+ \left((m_0 - n_0)(\kappa_1 + 1)R_0p_2 + (m_0 + n_0)(\kappa_1 + 1)\frac{1}{R_0}p_0 \right)z + 2m_0(\kappa_1 + 1)R_0p_1 + \left[(m_0 - n_0)(\kappa_1 + 1)R_0p_0 + (m_0 + n_0)(\kappa_1 + 1)R_0^3p_2 \right]z^{-1} + \left[(m_0 - n_0)(\kappa_1 + 1)R_0p_{-1} + (m_0 + n_0)(\kappa_1 + 1)R_0p_{3} \right]z^{-2} + \left[(m_0 + n_0)(\kappa_1 + 1)R_0^7p_4 + (m_0 - n_0)(\kappa_1 + 1)R_0p_{-2} \right]z^{-3} + \left[(m_0 + n_0)(\kappa_1 + 1)R_0^9p_5 + (m_0 - n_0)(\kappa_1 + 1)R_0p_{-3} \right]z^{-4} + \left[(m_0 - n_0)(\kappa_1 + 1)R_0p_{-4} + (m_0 + n_0)(\kappa_1 + 1)R_0^{11}p_6 \right]z^{-5} + (m_0 + n_0)(\kappa_1 + 1)R_0^{13}p_7z^{-6}$$

$$= \left(-4\mu_{1}\frac{7}{R_{0}^{2}}b_{7} + (m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})\frac{1}{R_{0}}b_{7}\right)z^{8} \\ + \left(-4\mu_{1}\frac{6}{R_{0}^{2}}b_{6} + (m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})\frac{1}{R_{0}}b_{6}\right)z^{7} \\ + \left(\frac{4\mu_{1}\left(-35a_{7}-\frac{5}{R_{0}^{2}}b_{5}\right) + (m_{0}-n_{0})(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}a_{7}}{+7(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})R_{0}a_{7} + (m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})\frac{1}{R_{0}}b_{5}}\right)z^{6} \\ + \left(\frac{4\mu_{1}\left(-24a_{6}-\frac{4}{R_{0}^{2}}b_{4}\right) + (m_{0}-n_{0})(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}a_{6}}{+6(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})R_{0}a_{6} + (m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})\frac{1}{R_{0}}b_{4}}\right)z^{5} \\ + \left(\frac{4\mu_{1}\left(-15a_{5}-\frac{43}{R_{0}^{2}}b_{3}\right) + (m_{0}-n_{0})(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}a_{5}}{+5(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})R_{0}a_{5} + (m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})\frac{1}{R_{0}}b_{3}}\right)z^{4} \\ + \left(\frac{4\mu_{1}\left(-8a_{4}-\frac{2}{R_{0}^{2}}b_{2}\right) + (m_{0}-n_{0})(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}a_{4}}{+4(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})R_{0}a_{4} + (m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})\frac{1}{R_{0}}b_{2}}\right)z^{3} \\ + \left(\frac{4\mu_{1}\left(-3a_{3}-\frac{1}{R_{0}^{2}}b_{1}\right) + (m_{0}-n_{0})(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}a_{3} + \frac{2\mu_{1}(m_{0}+n_{0})(s_{2}-i\varepsilon_{3})}{R_{0}}}{+3(m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})R_{0}a_{3} + (m_{0}+n_{0})(1-\frac{\mu_{1}}{\mu_{0}})\frac{1}{R_{0}}b_{1}}\right)z^{2}\right)z^{4}$$

.

$$+ \begin{pmatrix} (m_0 - n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})R_0a_2 + (m_0 + n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})\frac{1}{R_0}a_0 \\ + (m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})\frac{1}{R_0}b_0 + 2(m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})R_0a_2 \end{pmatrix} z \\ + \begin{pmatrix} 8\mu_1a_1 + (m_0 - n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})R_0a_1 + R_0(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})a_0 \\ + (m_0 + n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})R_0a_1 + 4m_0R_0\mu_1\varepsilon_1 + (m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})R_0a_1 \end{pmatrix} \\ + \begin{pmatrix} 8\mu_1R_0^2a_2 + (m_0 - n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})R_0a_0 + 2(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^3a_2 \\ + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0b_0 + (m_0 + n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})R_0^3a_2 \end{pmatrix} z^{-1} \\ + \begin{pmatrix} 12\mu_1R_0^4a_3 + 3(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^5a_3 + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^3b_1 \\ + (m_0 + n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})R_0^5a_3 + 2\mu_1(m_0 - n_0)(\varepsilon_2 + i\varepsilon_3)R_0^3 \end{pmatrix} z^{-2} \\ + \begin{pmatrix} 16\mu_1R_0^6a_4 + 4(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^7a_4 + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^5b_2 \\ + (m_0 + n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})R_0^3a_5 + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^5b_3 \end{pmatrix} z^{-3} \\ + \begin{pmatrix} 24\mu_1R_0^{10}a_6 + 6(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^{11}a_6 + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^{10}b_3 \\ + (m_0 + n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})R_0^{11}a_6 + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^{11}b_5 \\ + (m_0 + n_0)(1 - \kappa_0 \frac{\mu_1}{\mu_0})R_0^{13}a_7 + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^{11}b_5 \\ + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^{13}a_7 + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^{11}b_5 \\ + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^{13}a_7 + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^{11}b_5 \\ + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^{13}a_7 + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^{11}b_5 \\ + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^{13}a_7 + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^{11}b_5 \\ + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^{13}a_7 + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^{11}b_5 \\ + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^{13}a_7 + (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^{11}b_7z^{-8}. \end{pmatrix} z^{-6}$$

A4.3 Derivation of Equation (3.22)

.

••

The traction continuation condition along the Γ_1 (2.13) is

$$\varphi_2'(z) + \overline{\varphi_2'}\left(\frac{R_1^2}{z}\right) - z\varphi_2''(z) - \frac{z^2}{R_1^2}\psi_2'(z) = \varphi_{12}'(z) + \overline{\varphi_{12}'}\left(\frac{R_1^2}{z}\right) - z\varphi_{12}''(z) - \frac{z^2}{R_1^2}\psi_{12}'(z),$$

and the stress potentials associated with (2.13) are defined in Chapter 2 in the following form

.

$$\varphi_{2}(z) = Az + \sum_{k=1}^{\infty} c_{k} z^{-k} = Az + c_{1} z^{-1} + c_{2} z^{-2} + c_{3} z^{-3} + c_{4} z^{-4} + c_{5} z^{-5} + c_{6} z^{-6} + c_{7} z^{-7} + c_{8} z^{-8},$$

$$\vdots$$

$$\psi_{2}(z) = Bz + \sum_{k=1}^{\infty} d_{k} z^{-k} = Bz + d_{1} z^{-1} + d_{2} z^{-2} + d_{3} z^{-3} + d_{4} z^{-4} + d_{5} z^{-5} + d_{6} z^{-6} + d_{7} z^{-7} + d_{8} z^{-8},$$

$$\varphi_{12}(z) = q_{3} z^{3} + q_{2} z^{2} + q_{1} z^{1} + q_{0} + q_{-1} z^{-1} + q_{-2} z^{-2} + q_{-3} z^{-3} + q_{-4} z^{-4} + q_{-5} z^{-5} + q_{-6} z^{-6},$$

$$\psi_{12}(z) = 7q_{-6}z^{-6} + 6q_{-5}z^{-5} + (5q_{-4} - \overline{e}_{-4})z^{-4} + (4q_{-3} - \overline{e}_{-3})z^{-3} + (3q_{-2} - \overline{e}_{-2})z^{-2} + (2q_{-1} - \overline{e}_{-1})z^{-1} + (q_0 - \overline{e}_0) - \overline{e}_1z^1 + (-\overline{e}_2 - q_2)z^2 + (-\overline{e}_3 - 2q_3)z^3 - \overline{e}_4z^4.$$

Let's derive the items on the R.H.S:

4

$$\varphi_{12}'(z) = 3q_3z^2 + 2q_2z^1 + q_1 - q_{-1}z^{-2} - 2q_{-2}z^{-3} - 3q_{-3}z^{-4} - 4q_{-4}z^{-5} - 5q_{-5}z^{-6} - 6q_{-6}z^{-7},$$
(A4.21)

$$\overline{\varphi_{12}'}\left(\frac{R_1^2}{z}\right) = 3q_3 \left(R_1^2 z^{-1}\right)^2 + 2q_2 \left(R_1^2 z^{-1}\right)^1 + q_1 - q_{-1} \left(R_1^2 z^{-1}\right)^{-2} \right)$$

$$-2q_{-2} \left(R_1^2 z^{-1}\right)^{-3} - 3q_{-3} \left(R_1^2 z^{-1}\right)^{-4} - 4q_{-4} \left(R_1^2 z^{-1}\right)^{-5} \right)$$

$$-5q_{-5} \left(R_1^2 z^{-1}\right)^{-6} - 6q_{-6} \left(R_1^2 z^{-1}\right)^{-7}$$

$$= 3q_3 R_1^4 z^{-2} + 2q_2 R_1^2 z^{-1} + q_1 - q_{-1} \frac{1}{R_1^4} z^2 - 2q_{-2} \frac{1}{R_1^6} z^3 - 3q_{-3} \frac{1}{R_1^8} z^4$$

$$-4q_{-4} \frac{1}{R_1^{10}} z^5 - 5q_{-5} \frac{1}{R_1^{12}} z^6 - 6q_{-6} \frac{1}{R_1^{14}} z^7,$$
(A4.22)

$$-z\varphi_{12}''(z) = -z \begin{pmatrix} 3q_3z^2 + 2q_2z^1 + q_1 - q_{-1}z^{-2} - 2q_{-2}z^{-3} \\ -3q_{-3}z^{-4} - 4q_{-4}z^{-5} - 5q_{-5}z^{-6} - 6q_{-6}z^{-7} \end{pmatrix}'$$
(A4.23)
$$= -6q_3z^2 - 2q_2z - 2q_{-1}z^{-2} - 6q_{-2}z^{-3} - 12q_{-3}z^{-4} - 20q_{-4}z^{-5} \\ -30q_{-5}z^{-6} - 42q_{-6}z^{-7},$$

.

$$-\frac{z^{2}}{R_{1}^{2}}\psi_{12}'(z) = -\frac{z^{2}}{R_{1}^{2}} \begin{pmatrix} 7q_{-6}z^{-6} + 6q_{-5}z^{-5} + (5q_{-4} - \overline{e}_{-4})z^{-4} \\ + (4q_{-3} - \overline{e}_{-3})z^{-3} + (3q_{-2} - \overline{e}_{-2})z^{-2} \\ + (2q_{-1} - \overline{e}_{-1})z^{-1} + (q_{0} - \overline{e}_{0}) - \overline{e}_{1}z^{1} \\ + (-\overline{e}_{2} - q_{2})z^{2} + (-\overline{e}_{3} - 2q_{3})z^{3} - \overline{e}_{4}z^{4} \end{pmatrix}'$$

$$= -\frac{z^{2}}{R_{1}^{2}} \begin{pmatrix} -42q_{-6}z^{-7} - 30q_{-5}z^{-6} - 4(5q_{-4} - \overline{e}_{-4})z^{-5} \\ -3(4q_{-3} - \overline{e}_{-3})z^{-4} - 2(3q_{-2} - \overline{e}_{-2})z^{-3} \\ -(2q_{-1} - \overline{e}_{-1})z^{-2} - \overline{e}_{1} \\ +2(-\overline{e}_{2} - q_{2})z + 3(-\overline{e}_{3} - 2q_{3})z^{2} - 4\overline{e}_{4}z^{3} \end{pmatrix}$$

$$= 42q_{-6}\frac{1}{R_{1}^{2}}z^{-5} + 30q_{-5}\frac{1}{R_{1}^{2}}z^{-4} + 4(5q_{-4} - \overline{e}_{-4})\frac{1}{R_{1}^{2}}z^{-3} \\ +3(4q_{-3} - \overline{e}_{-3})\frac{1}{R_{1}^{2}}z^{-2} + 2(3q_{-2} - \overline{e}_{-2})\frac{1}{R_{1}^{2}}z^{-1} + (2q_{-1} - \overline{e}_{-1})\frac{1}{R_{1}^{2}}z^{-1} \\ +\overline{e}_{1}\frac{z^{2}}{R_{1}^{2}} - 2(-\overline{e}_{2} - q_{2})\frac{1}{R_{1}^{2}}z^{3} - 3(-\overline{e}_{3} - 2q_{3})\frac{1}{R_{1}^{2}}z^{4} + 4\overline{e}_{4}\frac{1}{R_{1}^{2}}z^{5}.$$

Also, let's derive the items on the L.H.S:

-5

$$\varphi_{2}'(z) = A - c_{1}z^{-2} - 2c_{2}z^{-3} - 3c_{3}z^{-4} - 4c_{4}z^{-5} - 5c_{5}z^{-6} - 6c_{6}z^{-7} - 7c_{7}z^{-8} - 8c_{8}z^{-9},$$
(A4.25)

•

$$\overline{\varphi_2'}\left(\frac{R_1^2}{z}\right) = A - c_1 \left(R_1^2 z^{-1}\right)^{-2} - 2c_2 \left(R_1^2 z^{-1}\right)^{-3} - 3c_3 \left(R_1^2 z^{-1}\right)^{-4}$$
(A4.26)

$$-4c_4 \left(R_1^2 z^{-1}\right)^{-5} - 5c_5 \left(R_1^2 z^{-1}\right)^{-6} - 6c_6 \left(R_1^2 z^{-1}\right)^{-7}$$

$$-7c_7 \left(R_1^2 z^{-1}\right)^{-8} - 8c_8 \left(R_1^2 z^{-1}\right)^{-9}$$

$$= A - \frac{1}{R_1^4} c_1 z^2 - \frac{2}{R_1^6} c_2 z^3 - \frac{3}{R_1^8} c_3 z^4 - \frac{4}{R_1^{10}} c_4 z^5 - \frac{5}{R_1^{12}} c_5 z^6$$

$$- \frac{6}{R_1^{14}} c_6 z^7 - \frac{7}{R_1^{16}} c_7 z^8 - \frac{8}{R_1^{18}} c_8 z^9,$$

$$z\varphi_{2}''(z) = 2c_{1}z^{-2} + 6c_{2}z^{-3} + 12c_{3}z^{-4} + 20c_{4}z^{-5}$$

$$+ 30c_{5}z^{-6} + 42c_{6}z^{-7} + 56c_{7}z^{-8} + 72c_{8}z^{-9},$$
(A4.27)

.

•

4

.

$$\frac{z^{2}}{R_{1}^{2}}\psi_{2}'(z) = \frac{z^{2}}{R_{1}^{2}} \begin{pmatrix} Bz + d_{1}z^{-1} + d_{2}z^{-2} + d_{3}z^{-3} + d_{4}z^{-4} \\ + d_{5}z^{-5} + d_{6}z^{-6} + d_{7}z^{-7} + d_{8}z^{-8} \end{pmatrix}'$$

$$= \frac{1}{R_{1}^{2}}Bz^{2} - \frac{1}{R_{1}^{2}}d_{1} - \frac{2}{R_{1}^{2}}d_{2}z^{-1} - \frac{3}{R_{1}^{2}}d_{3}z^{-2} - \frac{4}{R_{1}^{2}}d_{4}z^{-3} - \frac{5}{R_{1}^{2}}d_{5}z^{-4} \\ - \frac{6}{R_{1}^{2}}d_{6}z^{-5} - \frac{7}{R_{1}^{2}}d_{7}z^{-6} - \frac{8}{R_{1}^{2}}d_{8}z^{-7}.$$
(A4.28)

Thus, by substituting (A4.25) thru (A4.28) into (2.13), the L.H.S. of (2.13) becomes

$$\begin{aligned} \varphi_{2}'(z) + \overline{\varphi_{2}'}\left(\frac{R_{1}^{2}}{z}\right) &- z\varphi_{2}''(z) - \frac{z^{2}}{R_{1}^{2}}\psi_{2}'(z) \end{aligned} \tag{A4.29} \\ &= \left(A - c_{1}z^{-2} - 2c_{2}z^{-3} - 3c_{3}z^{-4} - 4c_{4}z^{-5} - 5c_{5}z^{-6} - 6c_{6}z^{-7} - 7c_{7}z^{-8} - 8c_{8}z^{-9}\right) \\ &+ \left(\begin{array}{c}A - \frac{1}{R_{1}^{4}}c_{1}z^{2} &- \frac{2}{R_{1}^{6}}c_{2}z^{3} - \frac{3}{R_{1}^{6}}c_{3}z^{4} - \frac{4}{R_{1}^{10}}c_{4}z^{5} - \frac{5}{R_{1}^{12}}c_{5}z^{6} \\ &- \frac{6}{R_{1}^{14}}c_{6}z^{7} - \frac{7}{R_{1}^{16}}c_{7}z^{8} - \frac{8}{R_{1}^{18}}c_{8}z^{9} \end{array}\right) \\ &- \left(\begin{array}{c}2c_{1}z^{-2} + 6c_{2}z^{-3} + 12c_{3}z^{-4} + 20c_{4}z^{-5} + 30c_{5}z^{-6} + 42c_{6}z^{-7} \\ &+ 56c_{7}z^{-8} + 72c_{8}z^{-9} \end{array}\right) \\ &- \left(\begin{array}{c}\frac{1}{R_{1}^{2}}Bz^{2} - \frac{1}{R_{1}^{2}}d_{1} - \frac{2}{R_{1}^{2}}d_{2}z^{-1} - \frac{3}{R_{1}^{2}}d_{3}z^{-2} - \frac{4}{R_{1}^{2}}d_{4}z^{-3} \\ &- \frac{5}{R_{1}^{2}}d_{5}z^{-4} - \frac{6}{R_{1}^{2}}d_{6}z^{-5} - \frac{7}{R_{1}^{2}}d_{7}z^{-6} - \frac{8}{R_{1}^{2}}d_{8}z^{-7} \end{array}\right) \end{aligned}$$

$$= -\frac{8}{R_{1}^{18}}c_{8}z^{9} - \frac{7}{R_{1}^{16}}c_{7}z^{8} - \frac{6}{R_{1}^{14}}c_{6}z^{7} - \frac{5}{R_{1}^{12}}c_{5}z^{6} - \frac{4}{R_{1}^{10}}c_{4}z^{5} - \frac{3}{R_{1}^{8}}c_{3}z^{4} -\frac{2}{R_{1}^{6}}c_{2}z^{3} + \left(-\frac{1}{R_{1}^{4}}c_{1}z^{2} - \frac{1}{R_{1}^{2}}Bz^{2}\right) + \left(2A + \frac{1}{R_{1}^{2}}d_{1}\right) + \frac{2}{R_{1}^{2}}d_{2}z^{-1} + \left(-3c_{1}z^{-2} + \frac{3}{R_{1}^{2}}d_{3}z^{-2}\right) + \left(-8c_{2}z^{-3} + \frac{4}{R_{1}^{2}}d_{4}z^{-3}\right) + \left(-15c_{3}z^{-4} + \frac{5}{R_{1}^{2}}d_{5}z^{-4}\right) + \left(-24c_{4}z^{-5} + \frac{6}{R_{1}^{2}}d_{6}z^{-5}\right) + \left(-35c_{5}z^{-6} + \frac{7}{R_{1}^{2}}d_{7}z^{-6}\right) + \left(-48c_{6}z^{-7} + \frac{8}{R_{1}^{2}}d_{8}z^{-7}\right) - 63c_{7}z^{-8} - 80c_{8}z^{-9}.$$

Also, substituting (A4.21) thru (A4.24) into the R.H.S. of (2.13) yields

:

4

$$\begin{split} \varphi_{12}'(z) + \overline{\varphi_{12}'} \left(\frac{R_1^2}{z}\right) &- z\varphi_{12}''(z) - \frac{z^2}{R_1^2} \psi_{12}'(z) \qquad (A4.30) \\ &= \begin{pmatrix} 3q_3z^2 + 2q_2z^1 + q_1 - q_{-1}z^{-2} - 2q_{-2}z^{-3} - 3q_{-3}z^{-4} \\ -4q_{-4}z^{-5} - 5q_{-5}z^{-6} - 6q_{-6}z^{-7} \end{pmatrix} \\ &+ \begin{pmatrix} 3q_3R_1^4z^{-2} + 2q_2R_1^2z^{-1} + q_1 - q_{-1}\frac{1}{R_1^4}z^2 - 2q_{-2}\frac{1}{R_1^6}z^3 \\ -3q_{-3}\frac{1}{R_1^6}z^4 - 4q_{-4}\frac{1}{R_1^{10}}z^5 - 5q_{-5}\frac{1}{R_1^{12}}z^6 - 6q_{-6}\frac{1}{R_1^{14}}z^7 \end{pmatrix} \\ &- 6q_3z^2 - 2q_2z - 2q_{-1}z^{-2} - 6q_{-2}z^{-3} - 12q_{-3}z^{-4} - 20q_{-4}z^{-5} \\ &- 30q_{-5}z^{-6} - 42q_{-6}z^{-7} + 42q_{-6}\frac{1}{R_1^2}z^{-5} + 30q_{-5}\frac{1}{R_1^2}z^{-4} \\ &+ 4(5q_{-4} - \overline{e}_{-4})\frac{1}{R_1^2}z^{-3} + 3(4q_{-3} - \overline{e}_{-3})\frac{1}{R_1^2}z^{-2} \\ &+ 2(3q_{-2} - \overline{e}_{-2})\frac{1}{R_1^2}z^{-1} + (2q_{-1} - \overline{e}_{-1})\frac{1}{R_1^2}z^{-2} \\ &+ 2(3q_{-2} - \overline{e}_{-2})\frac{1}{R_1^2}z^{-3} - 3(-\overline{e}_3 - 2q_3)\frac{1}{R_1^2}z^{4} + 4\overline{e}_4\frac{1}{R_1^2}z^{5} \\ &= \begin{pmatrix} -6q_{-6}\frac{1}{R_1^{14}}z^7 \end{pmatrix} + \begin{pmatrix} -5q_{-5}\frac{1}{R_1^{12}}z^6 \end{pmatrix} + \begin{pmatrix} 4\overline{e}_4\frac{1}{R_1^2}z^5 - 4q_{-4}\frac{1}{R_1^{10}}z^5 \end{pmatrix} \\ &+ \begin{pmatrix} -3(-\overline{e}_3 - 2q_3)\frac{1}{R_1^2}z^4 - 3q_{-3}\frac{1}{R_1^8}z^4 \end{pmatrix} + \begin{pmatrix} -2(-\overline{e}_2 - q_2)\frac{1}{R_1^2}z^3 - 2q_{-2}\frac{1}{R_1^6}z^3 \end{pmatrix} \\ &+ \begin{pmatrix} \overline{e}_1\frac{z^2}{R_1^2} - q_{-1}\frac{1}{R_1^4}z^2 + 3q_3z^2 - 6q_3z^2 \end{pmatrix} + (2q_2z^1 - 2q_2z) \end{split}$$

$$\begin{split} &+ \left(q_1 + q_1 + (2q_{-1} - \overline{e}_{-1}) \frac{1}{R_1^2} \right) + \left(2q_2 R_1^2 z^{-1} + 2 \left(3q_{-2} - \overline{e}_{-2} \right) \frac{1}{R_1^2} z^{-1} \right) \\ &+ \left(3q_3 R_1^4 z^{-2} - q_{-1} z^{-2} - 2q_{-1} z^{-2} + 3 \left(4q_{-3} - \overline{e}_{-3} \right) \frac{1}{R_1^2} z^{-2} \right) \\ &+ \left(-2q_{-2} z^{-3} + 4 \left(5q_{-4} - \overline{e}_{-4} \right) \frac{1}{R_1^2} z^{-3} - 6q_{-2} z^{-3} \right) \\ &+ \left(-3q_{-3} z^{-4} + 30q_{-5} \frac{1}{R_1^2} z^{-4} - 12q_{-3} z^{-4} \right) \\ &+ \left(-4q_{-4} z^{-5} + 42q_{-6} \frac{1}{R_1^2} z^{-5} - 20q_{-4} z^{-5} \right) \\ &+ \left(-5q_{-5} z^{-6} - 30q_{-5} z^{-6} \right) + \left(-6q_{-6} z^{-7} - 42q_{-6} z^{-7} \right). \end{split}$$

Thus, (2.13) is determined solely in terms of the complex coefficients

.

.

$$\begin{split} &-\frac{8}{R_1^{18}}c_8z^9 - \frac{7}{R_1^{16}}c_7z^8 - \frac{6}{R_1^{14}}c_6z^7 - \frac{5}{R_1^{12}}c_5z^6 - \frac{4}{R_1^{10}}c_4z^5 - \frac{3}{R_1^8}c_3z^4 \qquad (3.22) \\ &-\frac{2}{R_1^6}c_2z^3 + \left(-\frac{1}{R_1^4}c_1 - \frac{1}{R_1^2}B\right)z^2 + \left(2A + \frac{1}{R_1^2}d_1\right) + \frac{2}{R_1^2}d_2z^{-1} \\ &+ \left(-3c_1 + \frac{3}{R_1^2}d_3\right)z^{-2} + \left(-8c_2 + \frac{4}{R_1^2}d_4\right)z^{-3} + \left(-15c_3 + \frac{5}{R_1^2}d_5\right)z^{-4} \\ &+ \left(-24c_4 + \frac{6}{R_1^2}d_6\right)z^{-5} + \left(-35c_5 + \frac{7}{R_1^2}d_7\right)z^{-6} + \left(-48c_6 + \frac{8}{R_1^2}d_8\right)z^{-7} \\ &-63c_7z^{-8} - 80c_8z^{-9} \end{split}$$

$$= \left(-6q_{-6}\frac{1}{R_1^{14}}z^7\right) + \left(-5q_{-5}\frac{1}{R_1^{12}}z^6\right) + \left(4\overline{e}_4\frac{1}{R_1^2} - 4q_{-4}\frac{1}{R_1^{10}}\right)z^5 \\ &+ \left(-3\left(-\overline{e}_3 - 2q_3\right)\frac{1}{R_1^2} - 3q_{-3}\frac{1}{R_1^8}\right)z^4 + \left(-2\left(-\overline{e}_2 - q_2\right)\frac{1}{R_1^2} - 2q_{-2}\frac{1}{R_1^6}\right)z^3 \\ &+ \left(\overline{e}_1\frac{1}{R_1^2} - q_{-1}\frac{1}{R_1^4} + 3q_3 - 6q_3\right)z^2 + \left(2q_2 - 2q_2\right)z \\ &+ \left(q_1 + q_1 + \left(2q_{-1} - \overline{e}_{-1}\right)\frac{1}{R_1^2}\right) + \left(2q_2R_1^2 + 2\left(3q_{-2} - \overline{e}_{-2}\right)\frac{1}{R_1^2}\right)z^{-1} \\ &+ \left(3q_3R_1^4 - q_{-1} - 2q_{-1} + 3\left(4q_{-3} - \overline{e}_{-3}\right)\frac{1}{R_1^2}\right)z^{-3} \\ &+ \left(-2q_{-2} + 4\left(5q_{-4} - \overline{e}_{-4}\right)\frac{1}{R_1^2} - 6q_{-2}\right)z^{-3} \\ \end{array}$$

••

,

$$+ \left(-3q_{-3} + 30q_{-5}\frac{1}{R_1^2} - 12q_{-3}\right)z^{-4} + \left(-4q_{-4} + 42q_{-6}\frac{1}{R_1^2} - 20q_{-4}\right)z^{-5} + \left(-5q_{-5} - 30q_{-5}\right)z^{-6} + \left(-6q_{-6} - 42q_{-6}\right)z^{-7}.$$

A4.4 Derivation of Equation (3.23)

The displacement jump condition along the Γ_1 (2.14) is

$$(m_{1} - n_{1})(k_{1} + 1)\frac{R_{1}}{z}\varphi_{12}(z) + (m_{1} + n_{1})(k_{1} + 1)\frac{z}{R_{1}}\overline{\varphi_{12}}\left(\frac{R_{1}^{2}}{z}\right) \quad (2.14)$$

$$= -4\mu_{1}\left(\varphi_{2}'(z) + \overline{\varphi_{2}'}\left(\frac{R_{1}^{2}}{z}\right) - z\varphi_{2}''(z) - \frac{z^{2}}{R_{1}^{2}}\psi_{2}'(z)\right)$$

$$+ (m_{1} - n_{1})\left(1 - \frac{\mu_{1}}{\mu_{2}}\right)\left(R_{1}\overline{\varphi_{2}'}\left(\frac{R_{1}^{2}}{z}\right) + \frac{R_{1}}{z}\overline{\psi_{2}}\left(\frac{R_{1}^{2}}{z}\right)\right)$$

$$+ (m_{1} - n_{1})\left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2}\right)\frac{R_{1}}{z}\varphi_{2}(z)$$

$$+ (m_{1} + n_{1})\left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2}\right)\frac{z}{R_{1}}\overline{\varphi_{2}}\left(\frac{R_{1}^{2}}{z}\right)$$

$$+ (m_{1} + n_{1})\left(1 - \frac{\mu_{1}}{\mu_{2}}\right)\left(R_{1}\varphi_{2}'(z) + \frac{z}{R_{1}}\psi_{2}(z)\right).$$

Let's derive the each items required to be substituted:

$$\varphi_{12}(z) = q_3 z^3 + q_2 z^2 + q_1 z^1 + q_0 + q_{-1} z^{-1} + q_{-2} z^{-2} + q_{-3} z^{-3} + q_{-4} z^{-4} + q_{-5} z^{-5} + q_{-6} z^{-6},$$
(A4.31)

$$\frac{R_1}{z}\varphi_{12}(z) = R_1q_3z^2 + R_1q_2z + R_1q_1 + R_1q_0z^{-1} + R_1q_{-1}z^{-2}$$
(A4.32)
+ $R_1q_{-2}z^{-3} + R_1q_{-3}z^{-4} + R_1q_{-4}z^{-5} + R_1q_{-5}z^{-6} + R_1q_{-6}z^{-7},$

÷

$$\frac{z}{R_{1}}\overline{\varphi_{12}}\left(\frac{R_{1}^{2}}{z}\right) = \frac{z}{R_{1}}\begin{pmatrix}
q_{3}\left(R_{1}^{2}z^{-1}\right)^{3} + q_{2}\left(R_{1}^{2}z^{-1}\right)^{2} + q_{1}\left(R_{1}^{2}z^{-1}\right)^{1} \\
+q_{0} + q_{-1}\left(R_{1}^{2}z^{-1}\right)^{-1} + q_{-2}\left(R_{1}^{2}z^{-1}\right)^{-2} \\
+q_{-3}\left(R_{1}^{2}z^{-1}\right)^{-3} + q_{-4}\left(R_{1}^{2}z^{-1}\right)^{-4} \\
+q_{-5}\left(R_{1}^{2}z^{-1}\right)^{-5} + q_{-6}\left(R_{1}^{2}z^{-1}\right)^{-6}
\end{pmatrix}$$

$$(A4.33)$$

$$= q_{3}R_{1}^{5}z^{-2} + q_{2}R_{1}^{3}z^{-1} + q_{1}R_{1} + \frac{1}{R_{1}}q_{0}z + q_{-1}\frac{1}{R_{1}^{3}}z^{2} \\
+q_{-2}\frac{1}{R_{1}^{5}}z^{3} + q_{-3}\frac{1}{R_{1}^{7}}z^{4} + q_{-4}\frac{1}{R_{1}^{9}}z^{5} + q_{-5}\frac{1}{R_{1}^{11}}z^{6} + q_{-6}\frac{1}{R_{1}^{13}}z^{7}.$$

The first bracket $\left(\varphi_2'(z) + \overline{\varphi_2'}\left(\frac{R_1^2}{z}\right) - z\varphi_2''(z) - \frac{z^2}{R_1^2}\psi_2'(z)\right)$ in the R.H.S. of (2.14) is same as L.H.S. of (2.13), namely (A4.29).

$$R_{1}\overline{\varphi_{2}'}\left(\frac{R_{1}^{2}}{z}\right) = R_{1}\left(\begin{array}{c}A - \frac{1}{R_{1}^{4}}c_{1}z^{2} - \frac{2}{R_{1}^{6}}c_{2}z^{3} - \frac{3}{R_{1}^{8}}c_{3}z^{4} - \frac{4}{R_{1}^{10}}c_{4}z^{5}\\ -\frac{5}{R_{1}^{12}}c_{5}z^{6} - \frac{6}{R_{1}^{14}}c_{6}z^{7} - \frac{7}{R_{1}^{16}}c_{7}z^{8} - \frac{8}{R_{1}^{18}}c_{8}z^{9}\end{array}\right)$$
(A4.34)
$$= R_{1}A - \frac{1}{R_{1}^{2}}c_{1}z^{2} - \frac{2}{R_{1}^{5}}c_{2}z^{3} - \frac{3}{R_{1}^{7}}c_{3}z^{4} - \frac{4}{R_{1}^{9}}c_{4}z^{5}\\ -\frac{5}{R_{1}^{11}}c_{5}z^{6} - \frac{6}{R_{1}^{13}}c_{6}z^{7} - \frac{7}{R_{1}^{15}}c_{7}z^{8} - \frac{8}{R_{1}^{17}}c_{8}z^{9},$$

$$\frac{R_{1}}{z}\overline{\psi_{2}}\left(\frac{R_{1}^{2}}{z}\right) = \frac{R_{1}}{z} \begin{pmatrix} B\left(\frac{R_{1}^{2}}{z}\right) + d_{1}\left(\frac{R_{1}^{2}}{z}\right)^{-1} + d_{2}\left(\frac{R_{1}^{2}}{z}\right)^{-2} \\ + d_{3}\left(\frac{R_{1}^{2}}{z}\right)^{-3} + d_{4}\left(\frac{R_{1}^{2}}{z}\right)^{-4} + d_{5}\left(\frac{R_{1}^{2}}{z}\right)^{-5} \end{pmatrix} \quad (A4.35)$$

$$= BR_{1}^{3}z^{-2} + d_{1}\frac{1}{R_{1}} + d_{2}\frac{1}{R^{3}}z + d_{3}\frac{1}{R_{1}^{5}}z^{2} + d_{4}\frac{1}{R_{1}^{7}}z^{3} + d_{5}\frac{1}{R_{1}^{9}}z^{4},$$

$$\frac{R_1}{z}\varphi_2(z) = AR_1 + R_1c_1z^{-2} + R_1c_2z^{-3} + R_1c_3z^{-4} + R_1c_4z^{-5},$$
(A4.36)

$$\frac{z}{R_1}\overline{\varphi_2}\left(\frac{R_1^2}{z}\right) = AR_1 + c_1\frac{1}{R_1^3}z^2 + c_2\frac{1}{R_1^5}z^3 + c_3\frac{1}{R_1^7}z^4 + c_4\frac{1}{R_1^9}z^5,$$
(A4.37)

$$R_1\varphi_2'(z) = R_1 \left(A - c_1 z^{-2} - 2c_2 z^{-3} - 3c_3 z^{-4} - 4c_4 z^{-5} \right) \quad (A4.38)$$

..

•

$$= AR_1 - R_1c_1z^{-2} - 2R_1c_2z^{-3} - 3R_1c_3z^{-4} - 4R_1c_4z^{-5},$$

٠

$$\frac{z}{R_{1}}\psi_{2}(z) = B\frac{z}{R_{1}}z + d_{1}\frac{z}{R_{1}}z^{-1} + d_{2}\frac{z}{R_{1}}z^{-2}$$

$$+ d_{3}\frac{z}{R_{1}}z^{-3} + d_{4}\frac{z}{R_{1}}z^{-4} + d_{5}\frac{z}{R_{1}}z^{-5}$$

$$= B\frac{1}{R_{1}}z^{2} + d_{1}\frac{1}{R_{1}} + d_{2}\frac{1}{R_{1}}z^{-1} + d_{3}\frac{1}{R_{1}}z^{-2} + d_{4}\frac{1}{R_{1}}z^{-3} + d_{5}\frac{1}{R_{1}}z^{-4}.$$
(A4.39)

By substituting (A4.31) and (A4.33) into (2.14), the L.H.S. of (2.14) takes the following form

$$(m_{1} - n_{1}) (k_{1} + 1) \frac{R_{1}}{z} \varphi_{1} (z) + (m_{1} + n_{1}) (k_{1} + 1) \frac{z}{R_{1}} \overline{\varphi_{1}} \left(\frac{R_{1}^{2}}{z}\right)$$
(A4.40)

$$= (m_{1} - n_{1}) (k_{1} + 1) \begin{pmatrix} R_{1}q_{3}z^{2} + R_{1}q_{2}z + R_{1}q_{1} + R_{1}q_{0}z^{-1} \\ +R_{1}q_{-1}z^{-2} + R_{1}q_{-2}z^{-3} + R_{1}q_{-3}z^{-4} \\ +R_{1}q_{-4}z^{-5} + R_{1}q_{-5}z^{-6} + R_{1}q_{-6}z^{-7} \end{pmatrix}$$

$$+ (m_{1} + n_{1}) (k_{1} + 1) \begin{pmatrix} q_{3}R_{1}^{5}z^{-2} + q_{2}R_{1}^{3}z^{-1} + q_{1}R_{1} + \frac{1}{R_{1}}q_{0}z \\ +q_{-1}\frac{1}{R_{1}^{3}}z^{2} + q_{-2}\frac{1}{R_{1}^{5}}z^{3} + q_{-3}\frac{1}{R_{1}^{2}}z^{4} \\ +q_{-4}\frac{1}{R_{1}^{2}}z^{5} + q_{-5}\frac{1}{R_{1}^{11}}z^{6} + q_{-6}\frac{1}{R_{1}^{13}}z^{7} \end{pmatrix}$$

$$= \begin{pmatrix} (m_{1} - n_{1}) (k_{1} + 1) R_{1}q_{3}z^{2} + (m_{1} - n_{1}) (k_{1} + 1) R_{1}q_{0}z^{-1} \\ + (m_{1} - n_{1}) (k_{1} + 1) R_{1}q_{-1}z^{-2} \\ + (m_{1} - n_{1}) (k_{1} + 1) R_{1}q_{-2}z^{-3} + (m_{1} - n_{1}) (k_{1} + 1) R_{1}q_{-3}z^{-4} \\ + (m_{1} - n_{1}) (k_{1} + 1) R_{1}q_{-4}z^{-5} + (m_{1} - n_{1}) (k_{1} + 1) R_{1}q_{-5}z^{-6} \\ + (m_{1} - n_{1}) (k_{1} + 1) R_{1}q_{-6}z^{-7} \end{pmatrix}$$

••

:

હ

$$+ \begin{pmatrix} (m_1 + n_1) (k_1 + 1) q_3 R_1^5 z^{-2} + (m_1 + n_1) (k_1 + 1) q_2 R_1^3 z^{-1} \\ + (m_1 + n_1) (k_1 + 1) q_{-1} \frac{1}{R_1^3} z^2 + (m_1 + n_1) (k_1 + 1) \frac{1}{R_1} q_0 z \\ + (m_1 + n_1) (k_1 + 1) q_{-1} \frac{1}{R_1^3} z^2 + (m_1 + n_1) (k_1 + 1) q_{-2} \frac{1}{R_1^3} z^3 \\ + (m_1 + n_1) (k_1 + 1) q_{-3} \frac{1}{R_1^3} z^4 + (m_1 + n_1) (k_1 + 1) q_{-4} \frac{1}{R_1^3} z^7 \\ + (m_1 + n_1) (k_1 + 1) q_{-5} \frac{1}{R_1^{11}} z^6 + (m_1 + n_1) (k_1 + 1) q_{-5} \frac{1}{R_1^{11}} z^6 \\ + (m_1 + n_1) (k_1 + 1) q_{-4} \frac{1}{R_1^3} z^7 + (m_1 + n_1) (k_1 + 1) q_{-5} \frac{1}{R_1^1} z^4 \\ + (m_1 + n_1) (k_1 + 1) q_{-4} \frac{1}{R_1^5} z^5 + (m_1 + n_1) (k_1 + 1) q_{-3} \frac{1}{R_1^7} z^4 \\ + (m_1 + n_1) (k_1 + 1) q_{-2} \frac{1}{R_1^5} z^3 \\ + \left((m_1 - n_1) (k_1 + 1) R_1 q_3 + (m_1 + n_1) (k_1 + 1) \frac{1}{R_1} q_0 \right) z \\ + 2m_1 (k_1 + 1) R_1 q_1 \\ + ((m_1 - n_1) (k_1 + 1) R_1 q_{-1} + (m_1 + n_1) (k_1 + 1) R_1^5 q_2) z^{-1} \\ + \left[(m_1 - n_1) (k_1 + 1) R_1 q_{-2} z^{-3} + (m_1 - n_1) (k_1 + 1) R_1 q_{-3} z^{-4} \\ + (m_1 - n_1) (k_1 + 1) R_1 q_{-4} z^{-5} + (m_1 - n_1) (k_1 + 1) R_1 q_{-5} z^{-6} \\ + (m_1 - n_1) (k_1 + 1) R_1 q_{-6} z^{-7}. \end{cases}$$

Also, the R.H.S. of (2.14) is re-written as

٠

$$-4\mu_{1}\left(\varphi_{2}'(z) + \overline{\varphi_{2}'}\left(\frac{R_{1}^{2}}{z}\right) - z\varphi_{2}''(z) - \frac{z^{2}}{R_{1}^{2}}\psi_{2}'(z)\right)$$
(A4.41)
+ $(m_{1} - n_{1})\left(1 - \frac{\mu_{1}}{\mu_{2}}\right)\left(R_{1}\overline{\varphi_{2}'}\left(\frac{R_{1}^{2}}{z}\right) + \frac{R_{1}}{z}\overline{\psi_{2}}\left(\frac{R_{1}^{2}}{z}\right)\right)$.
+ $(m_{1} - n_{1})\left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2}\right)\frac{R_{1}}{z}\varphi_{2}(z) + (m_{1} + n_{1})\left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2}\right)\frac{z}{R_{1}}\overline{\varphi_{2}}\left(\frac{R_{1}^{2}}{z}\right)$

. •

$$\begin{split} &+ \left(m_{1} + n_{1}\right)\left(1 - \frac{\mu_{1}}{\mu_{2}}\right)\left(R_{1}\varphi_{2}'\left(z\right) + \frac{z}{R_{1}}\psi_{2}\left(z\right)\right)\\ &= \left.-4\mu_{1}\left(\begin{array}{c} -\frac{8}{R_{1}^{3}}c_{8}z^{9} - \frac{7}{R_{1}^{3}}c_{7}z^{8} - \frac{6}{R_{1}^{3}}c_{8}z^{7} - \frac{5}{R_{1}^{2}}c_{5}z^{6} - \frac{4}{R_{1}^{3}}c_{4}z^{5} \\ -\frac{3}{R_{1}^{3}}c_{3}z^{4} - \frac{2}{R_{1}^{2}}c_{2}z^{3} + \left(-\frac{1}{R_{1}^{4}}c_{1}z^{2} - \frac{1}{R_{1}^{2}}Bz^{2}\right) \\ + \left(2A + \frac{1}{R_{1}^{4}}d_{1}\right) + \frac{2}{R_{1}^{2}}d_{2}z^{-1} + \left(-3c_{1}z^{-2} + \frac{3}{R_{1}^{2}}d_{3}z^{-2}\right) \\ + \left(-8c_{2}z^{-3} + \frac{4}{R_{1}^{4}}d_{4}z^{-3}\right) + \left(-15c_{3}z^{-4} + \frac{5}{R_{1}^{2}}d_{5}z^{-4}\right) \\ + \left(-24c_{4}z^{-5} + \frac{6}{R_{1}^{4}}d_{6}z^{-5}\right) + \left(-35c_{5}z^{-6} + \frac{7}{R_{1}^{2}}d_{7}z^{-6}\right) \\ + \left(-48c_{6}z^{-7} + \frac{8}{R_{1}^{2}}d_{8}z^{-7}\right) - 63c_{7}z^{-8} - 80c_{8}z^{-9} \\ \right) \\ + \left(m_{1} - n_{1}\right)\left(1 - \frac{\mu_{1}}{\mu_{2}}\right)\left(\left(\begin{array}{c}R_{1}A - \frac{1}{R_{1}^{2}}c_{1}z^{2} - \frac{2}{R_{1}^{2}}c_{2}z^{3} \\ -\frac{3}{R_{1}^{2}}c_{3}z^{4} - \frac{4}{R_{1}^{2}}d_{4}z^{5} \\ + d_{3}\frac{1}{R_{1}^{2}}z^{2} + d_{4}\frac{1}{R_{1}^{2}}z^{3} + d_{5}\frac{1}{R_{1}^{2}}z^{4} \\ + d_{3}\frac{1}{R_{1}^{2}}z^{2} + d_{4}\frac{1}{R_{1}^{2}}z^{3} + d_{5}\frac{1}{R_{1}^{2}}z^{4} \\ + (m_{1} - n_{1})\left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2}\right)\left(AR_{1} + c_{1}\frac{1}{R_{1}^{2}}z^{2} + c_{2}\frac{1}{R_{1}^{2}}z^{3} + c_{3}\frac{1}{R_{1}^{2}}z^{4} + c_{4}\frac{1}{R_{1}^{2}}z^{5}\right) \\ + \left(m_{1} + n_{1})\left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2}\right)\left(AR_{1} + c_{1}\frac{1}{R_{1}^{2}}z^{2} + c_{2}\frac{1}{R_{1}^{2}}z^{3} + c_{3}\frac{1}{R_{1}^{2}}z^{4} + c_{4}\frac{1}{R_{1}^{2}}z^{5}\right) \\ + \left(m_{1} + n_{1})\left(1 - \frac{\mu_{1}}{\mu_{2}}\right)\left(AR_{1} + c_{1}\frac{1}{R_{1}^{2}}z^{2} + c_{2}\frac{1}{R_{1}^{2}}z^{3} + c_{3}\frac{1}{R_{1}^{2}}z^{-4} + c_{4}\frac{1}{R_{1}^{2}}z^{5}\right) \\ + \left(\frac{R_{1} - R_{1}c_{2}z^{-2}}{R_{1}c_{3}z^{-4}} - 4R_{1}c_{4}z^{-5}}\right) \\ + \left(\frac{R_{1} - R_{1}c_{2}z^{-2}}{R_{1}c_{3}z^{-4}} - 4R_{1}c_{4}z^{-5}}\right) \\ + \left(\frac{R_{1} - R_{1}c_{2}z^{-2}}{R_{1}c_{3}z^{-4}} - 4R_{1}c_{4}z^{-5}}\right) \\ + \left(\frac{R_{1} - R_{1}c_{3}z^{-4}}{R_{1}^{2}} + 2R_{1}\frac{1}{R_{1}^{2}}}z^{-2} + 2R_{1}\frac{1}{R_{1}^{2}}}z^{-4}\right)\right)\right)$$

÷

.

.

$$= \begin{pmatrix} 4\mu_{1}\frac{8}{R_{1}^{15}}c_{8}z^{9} + 4\mu_{1}\frac{7}{R_{1}^{16}}c_{7}z^{8} + 4\mu_{1}\frac{6}{R_{1}^{17}}c_{6}z^{7} + 4\mu_{1}\frac{5}{R_{1}^{12}}c_{5}z^{6} + 4\mu_{1}\frac{4}{R_{1}^{10}}c_{4}z^{5} \\ + 4\mu_{1}\frac{8}{R_{1}^{16}}c_{3}z^{4} + 4\mu_{1}\frac{2}{R_{1}^{1}}c_{2}z^{3} + 4\mu_{1}\left(\frac{1}{R_{1}^{1}}c_{1}z^{2} + \frac{1}{R_{1}^{1}}Bz^{2}\right) \\ -4\mu_{1}\left(2A + \frac{1}{R_{1}^{1}}d_{1}\right) - 4\mu_{1}\frac{2}{R_{1}^{2}}d_{2}z^{-1} - 4\mu_{1}\left(-3c_{1}z^{-2} + \frac{3}{R_{1}^{2}}d_{5}z^{-4}\right) \\ -4\mu_{1}\left(-24c_{4}z^{-5} + \frac{6}{R_{1}^{2}}d_{6}z^{-5}\right) - 4\mu_{1}\left(-15c_{5}z^{-6} + \frac{7}{R_{1}^{2}}d_{7}z^{-6}\right) \\ -4\mu_{1}\left(-48c_{6}z^{-7} + \frac{8}{R_{1}^{2}}d_{8}z^{-7}\right) + 4\mu_{1}63c_{7}z^{-8} + 4\mu_{1}80c_{8}z^{-9} \end{pmatrix} \\ + \begin{pmatrix} (m_{1} - n_{1})\left(1 - \frac{\mu_{1}}{\mu_{2}}\right)BR_{1}^{3}z^{-2} + (m_{1} - n_{1})\left(1 - \frac{\mu_{1}}{\mu_{2}}\right)\left(d_{3}\frac{1}{R_{1}^{2}}z^{2} - \frac{1}{R_{1}^{2}}c_{1}z^{2}\right) \\ + (m_{1} - n_{1})\left(1 - \frac{\mu_{1}}{\mu_{2}}\right)d_{2}\frac{1}{R_{3}}z^{4} + (m_{1} - n_{1})\left(1 - \frac{\mu_{1}}{\mu_{2}}\right)\left(d_{3}\frac{1}{R_{1}^{2}}z^{2} - \frac{1}{R_{1}^{2}}c_{1}z^{2}\right) \\ + (m_{1} - n_{1})\left(1 - \frac{\mu_{1}}{\mu_{2}}\right)d_{2}\frac{1}{R_{1}^{3}}z^{4} - \frac{3}{R_{1}^{2}}c_{3}z^{4}\right) - (m_{1} - n_{1})\left(1 - \frac{\mu_{1}}{\mu_{2}}\right)\frac{4}{R_{1}^{2}}c_{4}z^{5} \end{pmatrix} \\ + \begin{pmatrix} (m_{1} - n_{1})\left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2}\right)R_{1}c_{2}z^{-3} + (m_{1} - n_{1})\left(1 - \frac{\mu_{1}}{\mu_{2}}\right)\frac{4}{R_{1}^{2}}c_{4}z^{5} \end{pmatrix} \\ + \left((m_{1} - n_{1})\left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2}\right)R_{1}c_{2}z^{-3} + (m_{1} - n_{1})\left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2}\right)R_{1}c_{3}z^{-4} \\ + (m_{1} - n_{1})\left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2}\right)R_{1}c_{2}z^{-3} + (m_{1} - n_{1})\left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2}\right)R_{1}c_{3}z^{-4} \\ + (m_{1} + n_{1})\left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2}\right)C_{1}\frac{R_{1}^{2}}z^{5} \end{pmatrix} \\ + \begin{pmatrix} (m_{1} + n_{1})\left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2}\right)C_{1}\frac{R_{1}^{2}}z^{5} \\ (m_{1} + n_{1})\left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2}\right)C_{1}\frac{R_{1}^{2}}z^{5} \\ + (m_{1} + n_{1})\left(1 - \frac{\mu_{1}}{\mu_{2}}\right)\left(d_{3}\frac{R_{1}}{R_{1}}z^{-2} - R_{1}c_{1}z^{-2}\right) \\ + (m_{1} + n_{1})\left(1 - \frac{\mu_{1}}{\mu_{2}}\right)\left(d_{3}\frac{R_{1}}{R_{1}}z^{-2} - R_{1}c_{1}z^{-2}\right) \\ + \left((m_{1} + n_{1})\left(1 - \frac{\mu_{1}}{\mu_{2}}\right)\left(d_{3}\frac{R_{1}}{R_{1}}z^{-2} - R_{1}c_{1}z^{-2}\right) \\ + \left((m_{1} + n_{1}$$

 $= 4\mu_1 \frac{4}{R_1^{10}} c_4 z^5 - (m_1 - n_1) \left(1 - \frac{\mu_1}{\mu_2}\right) \frac{4}{R_1^9} c_4 z^5$

· · · · · · · · ·

·

$$\begin{split} -(w^{i}+u^{i})\left(1-\frac{h^{2}}{h^{2}}\right)q^{i}H^{i}c^{i}s^{2}_{2}\right)+(w^{i}-u^{i})\left(1+\frac{h^{2}}{h^{2}}k^{2}\right)c^{i}H^{i}c^{i}s^{2}_{2}\right)\\ -qh^{i}\left(-5qc^{4}c_{+}\frac{B_{4}}{2}q^{2}c_{-}q\right)+(w^{i}-u^{i})\left(1+\frac{h^{2}}{h^{2}}k^{2}\right)H^{i}c^{i}s^{2}_{-}q\right)\\ +(w^{i}+u^{i})\left(1-\frac{h^{2}}{h^{2}}\right)\left(q^{2}\frac{B_{4}}{2}s^{2}-q\right)+(w^{i}-u^{i})\left(1+\frac{h^{2}}{h^{2}}k^{2}\right)H^{i}c^{3}s^{2}_{-}q\right)\\ +(w^{i}+u^{i})\left(1-\frac{h^{2}}{h^{2}}\right)\left(q^{2}\frac{B_{4}}{2}s^{2}-q\right)+(w^{i}-u^{i})\left(1+\frac{h^{2}}{h^{2}}k^{2}\right)H^{i}c^{3}s^{2}_{-}q\right)\\ +(w^{i}-u^{i})\left(1-\frac{h^{2}}{h^{2}}\right)\left(q^{2}\frac{B_{4}}{2}s^{2}-q\right)+(w^{i}-u^{i})\left(1+\frac{h^{2}}{h^{2}}k^{2}\right)H^{i}c^{3}s^{2}_{-}q\right)\\ +(w^{i}-u^{i})\left(1+\frac{h^{2}}{h^{2}}k^{2}\right)d^{2}t^{2}s^{2}-q\right)+(w^{i}-u^{i})\left(1+\frac{h^{2}}{h^{2}}k^{2}\right)H^{i}c^{3}s^{2}_{-}q\right)\\ +(w^{i}-u^{i})\left(1+\frac{h^{2}}{h^{2}}k^{2}\right)d^{2}t^{2}s^{2}-q^{2}h^{i}\left(\frac{1}{h^{2}}t^{2}s^{2}-\frac{1}{h^{2}}h^{2}\right)H^{i}c^{3}s^{2}_{-}g\right)\\ +(w^{i}-u^{i})\left(1+\frac{h^{2}}{h^{2}}k^{2}\right)d^{2}t^{2}s^{2}-q^{2}h^{i}\left(\frac{1}{h^{2}}t^{2}s^{2}-\frac{1}{h^{2}}h^{2}\right)H^{i}c^{3}s^{2}_{-}g\right)\\ +(w^{i}-u^{i})\left(1-\frac{h^{2}}{h^{2}}\right)q^{2}t^{2}s^{2}-q^{2}h^{i}\left(\frac{1}{h^{2}}t^{2}s^{2}-\frac{1}{h^{2}}h^{2}\right)H^{i}c^{2}s^{2}_{-}g\right)\\ +(w^{i}-u^{i})\left(1-\frac{h^{2}}{h^{2}}\right)q^{2}t^{2}s^{2}s^{2}-q^{2}h^{i}\left(\frac{1}{h^{2}}t^{2}s^{2}-\frac{1}{h^{2}}h^{2}s^{2}s^{2}_{-}g\right)H^{i}c^{2}s^{2}_{-}g\right)\\ +(w^{i}-u^{i})\left(1-\frac{h^{2}}{h^{2}}\right)q^{2}t^{2}s^{2}s^{2}-q^{2}h^{i}\left(\frac{1}{h^{2}}t^{2}s^{2}-\frac{1}{h^{2}}h^{2}s^{2}s^{2}_{-}g\right)H^{i}s^{2}s^{2}_{-}g\right)\\ +(w^{i}-u^{i})\left(1-\frac{h^{2}}{h^{2}}\right)q^{2}t^{2}s^{2}s^{2}-q^{2}h^{i}\left(\frac{1}{h^{2}}t^{2}s^{2}-\frac{1}{h^{2}}h^{2}s^{2}s^{2}_{-}g\right)H^{i}s^{2}s^{2}_{-}g\right)g^{2}H^{i}s^{2}s^{2}s^{2}_{-}g\right)H^{i}s^{2}s^{2}s^{2}_{-}g\right)H^{i}s^{2}s^{2}s^{2}_{-}g\right)H^{i}s^{2}s^{2}s^{2}_{-}g\right)H^{i}s^{2}s^{2}s^{2}_{-}g\right)g^{2}H^{i}s^{2}s^{2}s^{2}_{-}g\right)g^{2}H^{i}s^{2}s^{2}s^{2}_{-}g\right)g^{2}H^{i}s^{2}s^{2}s^{2}_{-}g\right)g^{2}H^{i}s^{2}s^{2}s^{2}_{-}g\right)g^{2}H^{i}s^{2}s^{2}s^{2}_{-}g\right)g^{2}H^{i}s^{2}s^{2}s^{2}_{-}g\right)g^{2}H^{i}s^{2}s^{2}s^{2}_{-}g\right)g^{2}H^{i}s^{2}s^{2}s^{2}_{-}g\right)g^{2}g^{2}H^{i}s^{2}s^{2}s^{2}_{-}g\right)g^{2}g^{2}H^{i}s^{2}s^{2}s^{2}_{-}g\right)g^$$

..

122

Thus, (2.14) is solely expressed in terms of the coefficients in the following form

,

÷

$$\begin{array}{l} & (m_{1}+n_{1})\left(k_{1}+1\right)q_{-6}\frac{1}{R_{1}^{13}}z^{7}+(m_{1}+n_{1})\left(k_{1}+1\right)q_{-5}\frac{1}{R_{1}^{11}}z^{6} \\ & (3.23) \\ & +\left(m_{1}+n_{1}\right)\left(k_{1}+1\right)q_{-4}\frac{1}{R_{1}^{6}}z^{5}+\left(m_{1}+n_{1}\right)\left(k_{1}+1\right)q_{-3}\frac{1}{R_{1}^{2}}z^{4} \\ & +\left(m_{1}+n_{1}\right)\left(k_{1}+1\right)q_{-2}\frac{1}{R_{1}^{5}}z^{3} \\ & +\left(\left(m_{1}-n_{1}\right)\left(k_{1}+1\right)R_{1}q_{2}+\left(m_{1}+n_{1}\right)\left(k_{1}+1\right)\frac{1}{R_{1}}q_{0}\right)z+2m_{1}\left(k_{1}+1\right)R_{1}q_{1} \\ & +\left(\left(m_{1}-n_{1}\right)\left(k_{1}+1\right)R_{1}q_{2}+\left(m_{1}+n_{1}\right)\left(k_{1}+1\right)R_{1}^{2}q_{2}\right)z^{-1} \\ & +\left(\left(m_{1}-n_{1}\right)\left(k_{1}+1\right)R_{1}q_{0}+\left(m_{1}+n_{1}\right)\left(k_{1}+1\right)R_{1}^{2}q_{3}\right)z^{-2} \\ & +\left(\left(m_{1}-n_{1}\right)\left(k_{1}+1\right)R_{1}q_{-2}z^{-3}+\left(m_{1}-n_{1}\right)\left(k_{1}+1\right)R_{1}q_{-3}z^{-4} \\ & +\left(\left(m_{1}-n_{1}\right)\left(k_{1}+1\right)R_{1}q_{-2}z^{-5}+\left(m_{1}-n_{1}\right)\left(k_{1}+1\right)R_{1}q_{-3}z^{-4} \\ & +\left(m_{1}-n_{1}\right)\left(k_{1}+1\right)R_{1}q_{-6}z^{-7} \\ & = \left(4\mu_{1}\frac{4}{R_{1}^{10}}c_{4}-\left(m_{1}-n_{1}\right)\left(1-\frac{\mu_{1}}{\mu_{2}}\right)\frac{4}{R_{1}^{8}}c_{4}+\left(m_{1}+n_{1}\right)\left(1+\frac{\mu_{1}}{\mu_{2}}k_{2}\right)c_{4}\frac{1}{R_{1}^{2}}\right)z^{5} \\ & +\left(\frac{4\mu_{1}\frac{3}{R_{1}^{2}}c_{3}+\left(m_{1}-n_{1}\right)\left(1-\frac{\mu_{1}}{\mu_{2}}\right)\left(d_{5}\frac{1}{R_{1}^{2}}-\frac{3}{R_{1}^{2}}c_{3}\right)}{\left(d_{1}\frac{1}{R_{1}^{2}}-\frac{2}{R_{1}^{2}}c_{2}\right)}\right)z^{4} \\ & +\left(\frac{4\mu_{1}\frac{1}{R_{1}^{2}}c_{2}+\left(m_{1}-n_{1}\right)\left(1-\frac{\mu_{1}}{\mu_{2}}\right)\left(d_{3}\frac{1}{R_{1}^{2}}-\frac{1}{R_{1}^{2}}c_{1}\right)}{\left(d_{1}\frac{1}{R_{1}^{2}}-\frac{1}{R_{1}^{2}}c_{1}\right)}\right)z^{2} \\ & +\left(m_{1}+n_{1}\right)\left(1+\frac{\mu_{1}}{\mu_{2}}k_{2}\right)c_{1}\frac{1}{R_{1}^{3}}+\left(m_{1}+n_{1}\right)\left(1-\frac{\mu_{1}}{\mu_{2}}\right)R_{1}^{2}\right)z^{2} \\ & +\left(m_{1}-n_{1}\right)\left(1-\frac{\mu_{1}}{\mu_{2}}\right)d_{2}\frac{1}{R_{1}^{3}}z^{2} \\ & -4\mu_{1}\left(2A+\frac{1}{R_{1}^{2}}d_{1}\right)+\left(m_{1}+n_{1}\right)\left(1+\frac{\mu_{1}}{\mu_{2}}k_{2}\right)A_{1}\right)d_{1} \\ \end{array}$$

$$+ (m_{1} + n_{1}) \left(1 - \frac{\mu_{1}}{\mu_{2}}\right) \left(AR_{1} + d_{1}\frac{1}{R_{1}}\right) + (m_{1} - n_{1}) \left(1 - \frac{\mu_{1}}{\mu_{2}}\right) \left(AR_{1} + d_{1}\frac{1}{R_{1}}\right) + (m_{1} - n_{1}) \left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2}\right) AR_{1} + \left(-4\mu_{1}\frac{2}{R_{1}^{2}}d_{2} + (m_{1} + n_{1}) \left(1 - \frac{\mu_{1}}{\mu_{2}}\right) d_{2}\frac{1}{R_{1}}\right) z^{-1} + \left(\frac{-4\mu_{1} \left(-3c_{1} + \frac{3}{R_{1}^{2}}d_{3}\right) + (m_{1} - n_{1}) \left(1 - \frac{\mu_{1}}{\mu_{2}}\right) BR_{1}^{3} \\+ (m_{1} - n_{1}) \left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2}\right) R_{1}c_{1} \\+ (m_{1} + n_{1}) \left(1 - \frac{\mu_{1}}{\mu_{2}}\right) \left(d_{3}\frac{1}{R_{1}} - R_{1}c_{1}\right) \right) z^{-2} + \left(\frac{-4\mu_{1} \left(-8c_{2} + \frac{4}{R_{1}^{2}}d_{4}\right) + (m_{1} - n_{1}) \left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2}\right) R_{1}c_{2} \\+ (m_{1} + n_{1}) \left(1 - \frac{\mu_{1}}{\mu_{2}}\right) \left(d_{4}\frac{1}{R_{1}} - 2R_{1}c_{2}\right) \right) z^{-3} + \left(\frac{-4\mu_{1} \left(-15c_{3} + \frac{5}{R_{1}^{2}}d_{5}\right) + (m_{1} - n_{1}) \left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2}\right) R_{1}c_{3} \\+ (m_{1} + n_{1}) \left(1 - \frac{\mu_{1}}{\mu_{2}}\right) \left(d_{5}\frac{1}{R_{1}} - 3R_{1}c_{3}\right) \right) z^{-4} + \left(\frac{-4\mu_{1} \left(-24c_{4} + \frac{6}{R_{1}^{2}}d_{6}\right) + (m_{1} - n_{1}) \left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2}\right) R_{1}c_{4} \\- (m_{1} + n_{1}) \left(1 - \frac{\mu_{1}}{\mu_{2}}\right) 4R_{1}c_{4} \right) z^{-5}.$$

4 '

•

.

.

.

۰

.

÷

APPENDIX 5 38 Algebraic Equations

(3.3) - (3.21) and (3.24) - (3.42) can be re-written as follows: Traction continuation along the Γ_0

1) z^{5}

$$24a_6 + \frac{4}{R_0^2}b_4 + \frac{4}{R_0^2}e_4 - 24p_6 + \frac{12}{R_0^2}p_4 - \frac{4}{R_0^{10}}p_{-4} = 0,$$

2) z^4

$$15a_5 + \frac{3}{R_0^2}b_3 + \frac{3}{R_0^2}e_3 - 15p_5 + \frac{6}{R_0^2}p_3 - \frac{3}{R_0^8}p_{-3} = 0,$$

3) z^{3}

$$8a_4 + \frac{2}{R_0^2}b_2 + \frac{2}{R_0^2}e_2 - 8p_4 + \frac{2}{R_0^2}p_2 - \frac{2}{R_0^6}p_{-2} = 0,$$

4) z^2

$$3a_3 + \frac{1}{R_0^2}b_1 + \frac{1}{R_0^2}e_1 - 3p_3 - \frac{1}{R_0^4}p_{-1} = 0,$$

5) z^{0}

.

••

$$-2a_1 - \frac{1}{R_0^2}e_{-1} + 2p_1 + \frac{2}{R_0^2}p_{-1} = 0,$$

6) z^{-1}

$$-2R_0^2a_2 - \frac{2}{R_0^2}e_{-2} + 2R_0^2p_2 + \frac{6}{R_0^2}p_{-2} = 0,$$

τ

7)
$$z^{-2}$$

 $-3R_0^4a_3 - \frac{3}{R_0^2}e_{-3} + 3R_0^4p_3 - 3p_{-1} + \frac{12}{R_0^2}p_{-3} = 0,$
8) z^{-3}
 $-4R_0^6a_4 - \frac{4}{R_0^2}e_{-4} + 4R_0^6p_4 - 8p_{-2} + \frac{20}{R_0^2}p_{-4} = 0,$
9) z^{-4}

$$-5R_0^8a_5 + 5R_0^8p_5 - 15p_{-3} = 0,$$

Displacement jump along the Γ_0 10) z^5

.

$$0 = (m_0 - n_0)(\kappa_1 + 1)R_0p_6 + (m_0 + n_0)(\kappa_1 + 1)\frac{1}{R_0^9}p_{-4} + \left(96\mu_1 - (m_0 - n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})R_0 - 6(m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})R_0\right)a_6 + \left(-(m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})\frac{1}{R_0} + \mu_1\frac{16}{R_0^2}\right)b_4,$$

۰

11) z^4

.

••

4

$$0 = (m_0 - n_0)(\kappa_1 + 1)R_0p_5 + (m_0 + n_0)(\kappa_1 + 1)\frac{1}{R_0^7}p_{-3} + \left(60\mu_1 - (m_0 - n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})R_0 - 5(m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})R_0\right)a_5 + \left(-(m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})\frac{1}{R_0} + \mu_1\frac{12}{R_0^2}\right)b_3,$$

.

•

12) z^3

$$0 = (m_0 - n_0)(\kappa_1 + 1)R_0p_4 + (m_0 + n_0)(\kappa_1 + 1)\frac{1}{R_0^5}p_{-2} + \left(32\mu_1 - (m_0 - n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})R_0 - 4(m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})R_0\right)a_4 + \left(-(m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})\frac{1}{R_0} + \mu_1\frac{8}{R_0^2}\right)b_2,$$

.

.

×

•

 $(13)^{'}z^{2}$

$$\frac{2\mu_1(m_0+n_0)(\varepsilon_2-i\varepsilon_3)}{R_0} = (m_0-n_0)(\kappa_1+1)R_0p_3 + (m_0+n_0)(\kappa_1+1)\frac{1}{R_0^3}p_{-1} \\
+ \left(12\mu_1-(m_0-n_0)(1+\kappa_0\frac{\mu_1}{\mu_0})R_0 - 3(m_0+n_0)(1-\frac{\mu_1}{\mu_0})R_0\right)a_3 \\
+ \left(-(m_0+n_0)(1-\frac{\mu_1}{\mu_0})\frac{1}{R_0} + \mu_1\frac{4}{R_0^2}\right)b_1,$$

14) z

$$0 = (m_0 - n_0)(\kappa_1 + 1)R_0p_2 + (m_0 + n_0)(\kappa_1 + 1)\frac{1}{R_0}p_0 + \left(-(m_0 - n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})R_0 - 2(m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})R_0\right)a_2 - (m_0 + n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})\frac{1}{R_0}a_0 - (m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})\frac{1}{R_0}b_0,$$

15) z^0

.

.

4

 $4m_0R_0\mu_1arepsilon_1$

$$= 2m_{0}(\kappa_{1}+1)R_{0}p_{1} + \begin{pmatrix} -8\mu_{1} - (m_{0} - n_{0})(1 + \kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0} - (m_{0} - n_{0})(1 - \frac{\mu_{1}}{\mu_{0}})R_{0} \\ -(m_{0} + n_{0})(1 + \kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0} - (m_{0} + n_{0})(1 - \frac{\mu_{1}}{\mu_{0}})R_{0} \end{pmatrix} a_{1},$$

16) z^{-1}

$$0 = (m_0 + n_0)(\kappa_1 + 1)R_0^3 p_2 + (m_0 - n_0)(\kappa_1 + 1)R_0 p_0 + \left(-8\mu_1 R_0^2 - 2(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^3 - (m_0 + n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})R_0^3\right) a_2 - (m_0 - n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})R_0 a_0 - (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0 b_0,$$

17) z^{-2}

•

$$2\mu_{1}(m_{0} - n_{0})(\varepsilon_{2} + i\varepsilon_{3})R_{0}^{3}$$

$$= (m_{0} + n_{0})(\kappa_{1} + 1)R_{0}^{5}p_{3} + (m_{0} - n_{0})(\kappa_{1} + 1)R_{0}p_{-1}$$

$$+ \left(-12\mu_{1}R_{0}^{4} - 3(m_{0} - n_{0})(1 - \frac{\mu_{1}}{\mu_{0}})R_{0}^{5} - (m_{0} + n_{0})(1 + \kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}^{5}\right)a_{3}$$

$$- (m_{0} - n_{0})(1 - \frac{\mu_{1}}{\mu_{0}})R_{0}^{3}b_{1},$$

18) z^{-3}

$$0 = (m_0 + n_0)(\kappa_1 + 1)R_0^7 p_4 + (m_0 - n_0)(\kappa_1 + 1)R_0 p_{-2} + \left(-16\mu_1 R_0^6 - 4(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^7 - (m_0 + n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})R_0^7\right) a_4 - (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^5 b_2,$$

. 、

19) z⁻⁴

4

.

.

$$0 = (m_0 + n_0)(\kappa_1 + 1)R_0^9 p_5 + (m_0 - n_0)(\kappa_1 + 1)R_0 p_{-3} + \left(-20\mu_1 R_0^8 - 5(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^9 - (m_0 + n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})R_0^9\right) a_5 - (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^7 b_3,$$

.

Traction continuation along the Γ_1

20) z^5

$$0 = \frac{4}{R_1^{10}}c_4 + 4\frac{1}{R_1^2}e_4 - 4\frac{1}{R_1^{10}}q_{-4},$$

21) z^4
$$0 = \frac{3}{R_1^8}c_3 + 3\frac{1}{R_1^2}e_3 + 6\frac{1}{R_1^2}q_3 - 3\frac{1}{R_1^8}q_{-3},$$

22) z^3

•

4

.

$$0 = \frac{2}{R_1^6} c_2 + 2\frac{1}{R_1^2} e_2 + 2\frac{1}{R_1^2} q_2 - 2\frac{1}{R_1^6} q_{-2},$$

23) z^2

$$-\frac{1}{R_1^2}B = \frac{1}{R_1^4}c_1 + \frac{1}{R_1^2}e_1 - 3q_3 - \frac{1}{R_1^4}q_{-1}$$

24) z^0

$$2A = -\frac{1}{R_1^2}d_1 - \frac{1}{R_1^2}e_{-1} + 2q_1 + 2\frac{1}{R_1^2}q_{-1},$$

25) z^{-1}

;

$$0 = -\frac{2}{R_1^2}d_2 - 2\frac{1}{R_1^2}e_{-2} + 2R_1^2q_2 + 6\frac{1}{R_1^2}q_{-2},$$

26) z^{-2}

$$0 = 3c_1 - \frac{3}{R_1^2}d_3 - 3\frac{1}{R_1^2}e_{-3} + 3R_1^4q_3 - 3q_{-1} + 12\frac{1}{R_1^2}q_{-3},$$

,

27)
$$z^{-3}$$

$$0 = 8c_2 - \frac{4}{R_1^2}d_4 - 4\frac{1}{R_1^2}e_{-4} - 8q_{-2} + 20\frac{1}{R_1^2}q_{-4},$$

28) z^{-4}

$$0 = 15c_3 - \frac{5}{R_1^2}d_5 - 15q_{-3} + 30\frac{1}{R_1^2}q_{-5},$$

•

Displacement jump along the Γ_1 29) z^5

$$0 = \left(\frac{16}{R_1^{10}}\mu_1 - (m_1 - n_1)\left(1 - \frac{\mu_1}{\mu_2}\right)\frac{4}{R_1^9} + (m_1 + n_1)\left(1 + \frac{\mu_1}{\mu_2}k_2\right)\frac{1}{R_1^9}\right) - (m_1 + n_1)(k_1 + 1)\frac{1}{R_1^9}q_{-4},$$

.

 c_4

.

.

30) z^4

$$0 = \left(\frac{12}{R_1^8}\mu_1 - (m_1 - n_1)\left(1 - \frac{\mu_1}{\mu_2}\right)\frac{3}{R_1^7} + (m_1 + n_1)\left(1 + \frac{\mu_1}{\mu_2}k_2\right)\frac{1}{R_1^7}\right)c_3 + (m_1 - n_1)\left(1 - \frac{\mu_1}{\mu_2}\right)\frac{1}{R_1^9}d_5 - (m_1 + n_1)\left(k_1 + 1\right)\frac{1}{R_1^7}q_{-3},$$

31) z^{3}

$$0 = \left(\frac{8}{R_1^6}\mu_1 - (m_1 - n_1)\left(1 - \frac{\mu_1}{\mu_2}\right)\frac{2}{R_1^5} + (m_1 + n_1)\left(1 + \frac{\mu_1}{\mu_2}k_2\right)\frac{1}{R_1^5}\right)c_2 + (m_1 - n_1)\left(1 - \frac{\mu_1}{\mu_2}\right)\frac{1}{R_1^7}d_4 - (m_1 + n_1)\left(k_1 + 1\right)\frac{1}{R_1^5}q_{-2},$$

32) z^2

.

$$\left(-\left(m_{1}+n_{1}\right)\left(1-\frac{\mu_{1}}{\mu_{2}}\right)\frac{1}{R_{1}}-4\mu_{1}\frac{1}{R_{1}^{2}}\right)B$$

•

$$= \left(\frac{4}{R_1^4}\mu_1 - (m_1 - n_1)\left(1 - \frac{\mu_1}{\mu_2}\right)\frac{1}{R_1^3} + (m_1 + n_1)\left(1 + \frac{\mu_1}{\mu_2}k_2\right)\frac{1}{R_1^3}\right)c_1 + (m_1 - n_1)\left(1 - \frac{\mu_1}{\mu_2}\right)\frac{1}{R_1^5}d_3 - (m_1 - n_1)\left(k_1 + 1\right)R_1q_3 - (m_1 + n_1)\left(k_1 + 1\right)q_{-1}\frac{1}{R_1^3},$$

33) z^1

•

$$0 = (m_1 - n_1) \left(1 - \frac{\mu_1}{\mu_2} \right) \frac{1}{R^3} d_2 - (m_1 - n_1) (k_1 + 1) R_1 q_2 - (m_1 + n_1) (k_1 + 1) \frac{1}{R_1} q_0,$$

•

.

34) z^0

$$- \left(-8\mu_1 A + 2m_1 \left(1 + \frac{\mu_1}{\mu_2} k_2\right) A R_1 + 2m_1 \left(1 - \frac{\mu_1}{\mu_2}\right) A R_1\right)$$

$$= \left(-\frac{4}{R_1^2}\mu_1 + (m_1 + n_1) \left(1 - \frac{\mu_1}{\mu_2}\right) \frac{1}{R_1} + (m_1 - n_1) \left(1 - \frac{\mu_1}{\mu_2}\right) \frac{1}{R_1}\right) d_1$$

$$- 2m_1 \left(k_1 + 1\right) R_1 q_1,$$

35) z^{-1}

$$0 = \left(-\frac{8}{R_1^2}\mu_1 + (m_1 + n_1)\left(1 - \frac{\mu_1}{\mu_2}\right)\frac{1}{R_1}\right)d_2 - (m_1 + n_1)(k_1 + 1)R_1^3q_2 - (m_1 - n_1)(k_1 + 1)R_1q_0,$$

36) z^{-2}

•

$$- (m_1 - n_1) \left(1 - \frac{\mu_1}{\mu_2} \right) BR_1^3$$

$$= \left(12\mu_1 + (m_1 - n_1) \left(1 + \frac{\mu_1}{\mu_2} k_2 \right) R_1 - (m_1 + n_1) \left(1 - \frac{\mu_1}{\mu_2} \right) R_1 \right) c_1$$

$$+ \left(-\frac{12}{R_1^2} \mu_1 + (m_1 + n_1) \left(1 - \frac{\mu_1}{\mu_2} \right) \frac{1}{R_1} \right) d_3$$

$$- (m_1 + n_1) (k_1 + 1) R_1^5 q_3 - (m_1 - n_1) (k_1 + 1) R_1 q_{-1},$$

.

.

÷

37)
$$z^{-3}$$

$$0 = \left(+32\mu_1 + (m_1 - n_1)\left(1 + \frac{\mu_1}{\mu_2}k_2\right)R_1 - 2R_1\left(m_1 + n_1\right)\left(1 - \frac{\mu_1}{\mu_2}\right)\right)c_2 + \left(-\frac{16}{R_1^2}\mu_1 + (m_1 + n_1)\left(1 - \frac{\mu_1}{\mu_2}\right)\frac{1}{R_1}\right)d_4 - (m_1 - n_1)\left(k_1 + 1\right)R_1q_{-2},$$

38) z^{-4}

$$0 = \left(+60\mu_1 + (m_1 - n_1)\left(1 + \frac{\mu_1}{\mu_2}k_2\right)R_1 - 3R_1(m_1 + n_1)\left(1 - \frac{\mu_1}{\mu_2}\right)\right)c_3 \\ + \left(-\frac{20}{R_1^2}\mu_1 + (m_1 + n_1)\left(1 - \frac{\mu_1}{\mu_2}\right)\frac{1}{R_1}\right)d_5 - (m_1 - n_1)(k_1 + 1)R_1q_{-3}.$$

Now p_k and q_k are substituted into the above equations.

.

 $p_{-4..6}$ are :

٠

,

.

•

$$\begin{split} p_6 &= \left(g_7 f_{-1} + g_6 f_0 + g_5 f_1 + g_4 f_2 + g_3 f_3 + g_2 f_4\right), \\ p_5 &= \left(g_7 f_{-2} + g_6 f_{-1} + g_5 f_0 + g_4 f_1 + g_3 f_2 + g_2 f_3 + g_1 f_4\right), \\ p_4 &= \left(g_7 f_{-3} + g_6 f_{-2} + g_5 f_{-1} + g_4 f_0 + g_3 f_1 + g_2 f_2 + g_1 f_3 + g_0 f_4 + \frac{1}{2} e_4\right), \\ p_3 &= \left(g_6 f_{-3} + g_5 f_{-2} + g_4 f_{-1} + g_3 f_0 + g_2 f_1 + g_1 f_2 + g_0 f_3 + \frac{1}{2} e_3\right), \\ p_2 &= \left(g_5 f_{-3} + g_4 f_{-2} + g_3 f_{-1} + g_2 f_0 + g_1 f_1 + g_0 f_2 + \frac{1}{2} e_2\right), \\ p_1 &= \left(g_4 f_{-3} + g_3 f_{-2} + g_2 f_{-1} + g_1 f_0 + g_0 f_1 + \frac{1}{2} e_1\right), \\ p_0 &= \left(g_3 f_{-3} + g_2 f_{-2} + g_1 f_{-1} + g_0 f_0 + \frac{1}{2} e_0\right), \\ p_{-1} &= \left(g_2 f_{-3} + g_1 f_{-2} + g_0 f_{-1} + \frac{1}{2} e_{-1}\right), \\ p_{-2} &= \left(g_1 f_{-3} + g_0 f_{-2} + \frac{1}{2} e_{-2}\right), \\ p_{-3} &= \left(g_0 f_{-3} + \frac{1}{2} e_{-3}\right), \end{split}$$

.

••

$$p_{-4} = \left(\frac{1}{2}e_{-4}\right).$$

 $q_{-6..3}$ are :

$$\begin{split} q_{3} &= \left(h_{-1}f_{4} + \frac{1}{2}e_{3}\right), \\ q_{2} &= \left(h_{-2}f_{4} + h_{-1}f_{3} + \frac{1}{2}e_{2}\right), \\ q_{1} &= \left(h_{-3}f_{4} + h_{-2}f_{3} + h_{-1}f_{2} + \frac{1}{2}e_{1}\right), \\ q_{0} &= \left(h_{-4}f_{4} + h_{-3}f_{3} + h_{-2}f_{2} + h_{-1}f_{1} + \frac{1}{2}e_{0}\right), \\ q_{-1} &= \left(h_{-5}f_{4} + h_{-4}f_{3} + h_{-3}f_{2} + h_{-2}f_{1} + h_{-1}f_{0} + \frac{1}{2}e_{-1}\right), \\ q_{-2} &= \left(h_{-6}f_{4} + h_{-5}f_{3} + h_{-4}f_{2} + h_{-3}f_{1} + h_{-2}f_{0} + h_{-1}f_{-1} + \frac{1}{2}e_{-2}\right), \\ q_{-3} &= \left(h_{-7}f_{4} + h_{-6}f_{3} + h_{-5}f_{2} + h_{-4}f_{1} + h_{-3}f_{0} + h_{-2}f_{-1} + h_{-1}f_{-2} + \frac{1}{2}e_{-3}\right), \\ q_{-4} &= \left(\begin{array}{c}h_{-8}f_{4} + h_{-7}f_{3} + h_{-6}f_{2} + h_{-5}f_{1} + h_{-4}f_{0} + h_{-3}f_{-1} + h_{-2}f_{-2} \\ &+ h_{-1}f_{-3} + \frac{1}{2}e_{-4}\end{array}\right), \\ q_{-5} &= \left(\begin{array}{c}h_{-8}f_{3} + h_{-7}f_{2} + h_{-6}f_{1} + h_{-5}f_{0} + h_{-4}f_{-1} + h_{-3}f_{-2} + h_{-2}f_{-3} \\ &+ h_{-1}f_{-4} + \frac{1}{2}e_{-5}\end{array}\right), \\ q_{-6} &= \left(h_{-8}f_{2} + h_{-7}f_{1} + h_{-6}f_{0} + h_{-5}f_{-1} + h_{-4}f_{-2} + h_{-3}f_{-3} + h_{-2}f_{-4} + \frac{1}{2}e_{-6}\right). \end{split}$$

Consequently, the first 9 equations from the traction continuation condition along Γ_0 , namely, (3.43) thru (3.51) are; 1) z^5

$$0 = 24a_6 + \frac{4}{R_0^2}b_4 - \frac{4}{R_0^{10}}\frac{1}{2}e_{-4} + \frac{10}{R_0^2}e_4 + \frac{12}{R_0^2}g_7f_{-3} + \frac{12}{R_0^2}g_6f_{-2}$$

:

$$+ \left(\frac{12}{R_0^2}g_5 - 24g_7\right)f_{-1} + \left(\frac{12}{R_0^2}g_4 - 24g_6\right)f_0 + \left(\frac{12}{R_0^2}g_3 - 24g_5\right)f_1 \\ + \left(\frac{12}{R_0^2}g_2 - 24g_4\right)f_2 + \left(\frac{12}{R_0^2}g_1 - 24g_3\right)f_3 + \left(\frac{12}{R_0^2}g_0 - 24g_2\right)f_4,$$

2) z^4

$$0 = 15a_5 + \frac{3}{R_0^2}b_3 - \frac{3}{R_0^8}\frac{1}{2}e_{-3} + \frac{6}{R_0^2}e_3 + \left(\frac{6}{R_0^2}g_6 - \frac{3}{R_0^8}g_0\right)f_{-3} + \left(\frac{6}{R_0^2}g_5 - 15g_7\right)f_{-2} + \left(\frac{6}{R_0^2}g_4 - 15g_6\right)f_{-1} + \left(\frac{6}{R_0^2}g_3 - 15g_5\right)f_0 + \left(\frac{6}{R_0^2}g_2 - 15g_4\right)f_1 + \left(\frac{6}{R_0^2}g_1 - 15g_3\right)f_2 + \left(\frac{6}{R_0^2}g_0 - 15g_2\right)f_3 - 15g_1f_4,$$

3) z^{3}

$$0 = 8a_4 + \frac{2}{R_0^2}b_2 - \frac{1}{R_0^6}e_{-2} + \frac{3}{R_0^2}e_2 - 4e_4 + \left(\frac{2}{R_0^2}g_5 - 8g_7 - \frac{2}{R_0^6}g_1\right)f_{-3} \\ + \left(\frac{2}{R_0^2}g_4 - 8g_6 - \frac{2}{R_0^6}g_0\right)f_{-2} + \left(\frac{2}{R_0^2}g_3 - 8g_5\right)f_{-1} + \left(\frac{2}{R_0^2}g_2 - 8g_4\right)f_0 \\ + \left(\frac{2}{R_0^2}g_1 - 8g_3\right)f_1 + \left(\frac{2}{R_0^2}g_0 - 8g_2\right)f_2 - 8g_1f_3 - 8g_0f_4,$$

4) z^{2}

$$0 = 3a_3 + \frac{1}{R_0^2}b_1 - \frac{1}{2}\frac{1}{R_0^4}e_{-1} + \frac{1}{R_0^2}e_1 - 3\frac{1}{2}e_3 + \left(-3g_6 - \frac{1}{R_0^4}g_2\right)f_{-3} + \left(-3g_5 - \frac{1}{R_0^4}g_1\right)f_{-2} + \left(-3g_4 - \frac{1}{R_0^4}g_0\right)f_{-1} - 3g_3f_0 - 3g_2f_1 - 3g_1f_2 - 3g_0f_3,$$

÷

5) z^{0}

۰

$$0 = -2a_1 + e_1 + \left(2g_4 + \frac{2}{R_0^2}g_2\right)f_{-3}$$

:

4

$$+\left(2g_3 + \frac{2}{R_0^2}g_1\right)f_{-2} + \left(2g_2 + \frac{2}{R_0^2}g_0\right)f_{-1} + 2g_1f_0 + 2g_0f_1,$$

6) z^{-1}

$$0 = -2R_0^2 a_2 + \frac{1}{R_0^2} e_{-2} + R_0^2 e_2 + \left(2R_0^2 g_5 + \frac{6}{R_0^2}g_1\right) f_{-3} + \left(2R_0^2 g_4 + \frac{6}{R_0^2}g_0\right) f_{-2} + 2R_0^2 g_3 f_{-1} + 2R_0^2 g_2 f_0 + 2R_0^2 g_1 f_1 + 2R_0^2 g_0 f_2,$$

7) z^{-2}

$$0 = -3R_0^4 a_3 + \frac{3}{R_0^2} e_{-3} - \frac{3}{2} e_{-1} + 3R_0^4 \frac{1}{2} e_3 + \left(3R_0^4 g_6 - 3g_2 + \frac{12}{R_0^2} g_0\right) f_{-3} + \left(3R_0^4 g_5 - 3g_1\right) f_{-2} + \left(3R_0^4 g_4 - 3g_0\right) f_{-1} + 3R_0^4 g_3 f_0 + 3R_0^4 g_2 f_1 + 3R_0^4 g_1 f_2 + 3R_0^4 g_0 f_3,$$

 $8)z^{-3}$

$$0 = -4R_0^6 a_4 + \frac{6}{R_0^2} e_{-4} - 4e_{-2} + 2R_0^6 e_4 + (4R_0^6 g_7 - 8g_1) f_{-3} + (4R_0^6 g_6 - 8g_0) f_{-2} + 4R_0^6 g_5 f_{-1} + 4R_0^6 g_4 f_0 + 4R_0^6 g_3 f_1 + 4R_0^6 g_2 f_2 + 4R_0^6 g_1 f_3 + 4R_0^6 g_0 f_4,$$

9) z^{-4}

$$0 = -5R_0^8 a_5 - 15\frac{1}{2}e_{-3} - 15g_0 f_{-3} + 5R_0^8 g_7 f_{-2} + 5R_0^8 g_6 f_{-1} + 5R_0^8 g_5 f_0 + 5R_0^8 g_4 f_1 + 5R_0^8 g_3 f_2 + 5R_0^8 g_2 f_3 + 5R_0^8 g_1 f_4 .$$

In addition, non-dimensional parameter $M_0 = \frac{m_0+n_0}{2\mu_1}R_0$ is deployed into displacement jump equations along the Γ_0 . Thus, another 10 equations, such as (3.52) thru

÷

168

.

4

(3.61) are; 10) z⁵

$$0 = \left(96\mu_1 - (m_0 - n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})R_0 - 6(m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})R_0\right)a_6$$

+ $\left(-(m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})\frac{1}{R_0} + \mu_1 \frac{16}{R_0^2}\right)b_4$
+ $(m_0 + n_0)(\kappa_1 + 1)\frac{1}{2R_0^9}e_{-4}$
+ $(m_0 - n_0)(\kappa_1 + 1)R_0g_7f_{-1} + (m_0 - n_0)(\kappa_1 + 1)R_0g_6f_0$
+ $(m_0 - n_0)(\kappa_1 + 1)R_0g_5f_1 + (m_0 - n_0)(\kappa_1 + 1)R_0g_4f_2$
+ $(m_0 - n_0)(\kappa_1 + 1)R_0g_3f_3 + (m_0 - n_0)(\kappa_1 + 1)R_0g_2f_4,$

$$0 = \left(48\frac{1}{M_0} - \frac{(m_0 - n_0)}{(m_0 + n_0)}(1 + \kappa_0\frac{\mu_1}{\mu_0}) - 6(1 - \frac{\mu_1}{\mu_0})\right)a_6$$

+ $\left(-(1 - \frac{\mu_1}{\mu_0})\frac{1}{R_0^2} + \frac{8}{M_0}\frac{1}{R_0^2}\right)b_4 + (\kappa_1 + 1)\frac{1}{2R_0^{10}}e_{-4}$
+ $\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_7f_{-1} + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_6f_0$
+ $\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_5f_1 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_4f_2$
+ $\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_3f_3 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_2f_4,$

11) z^4

$$0 = \left(60\mu_1 - (m_0 - n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})R_0 - 5(m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})R_0 \right) a_5 + \left(-(m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})\frac{1}{R_0} + \mu_1 \frac{12}{R_0^2} \right) b_3 + (m_0 + n_0)(\kappa_1 + 1)\frac{1}{2R_0^7}e_{-3} + (m_0 + n_0)(\kappa_1 + 1)\frac{1}{R_0^7}g_0f_{-3} + (m_0 - n_0)(\kappa_1 + 1)R_0g_7f_{-2} + (m_0 - n_0)(\kappa_1 + 1)R_0g_6f_{-1}$$

÷

.

લ

$$+(m_0 - n_0)(\kappa_1 + 1)R_0g_5f_0 + (m_0 - n_0)(\kappa_1 + 1)R_0g_4f_1 +(m_0 - n_0)(\kappa_1 + 1)R_0g_3f_2 + (m_0 - n_0)(\kappa_1 + 1)R_0g_2f_3 + (m_0 - n_0)(\kappa_1 + 1)R_0g_1f_4,$$

$$\begin{array}{lll} 0 & = & \left(30\frac{1}{M_0} - \frac{(m_0 - n_0)}{(m_0 + n_0)}(1 + \kappa_0\frac{\mu_1}{\mu_0}) - 5(1 - \frac{\mu_1}{\mu_0})\right)a_5 \\ & \quad + \left(-(1 - \frac{\mu_1}{\mu_0})\frac{1}{R_0^2} + \frac{1}{M_0}\frac{6}{R_0^2}\right)b_3 + (\kappa_1 + 1)\frac{1}{2R_0^8}e_{-3} \\ & \quad + (\kappa_1 + 1)\frac{1}{R_0^8}g_0f_{-3} + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_7f_{-2} + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_6f_{-1} \\ & \quad + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_5f_0 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_4f_1 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_3f_2 \\ & \quad + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_2f_3 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_1f_4, \end{array}$$

12) z^3

1

$$0 = \left(32\mu_1 - (m_0 - n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})R_0 - 4(m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})R_0\right)a_4 \\ + \left(-(m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})\frac{1}{R_0} + \mu_1 \frac{8}{R_0^2}\right)b_2 \\ + (m_0 + n_0)(\kappa_1 + 1)\frac{1}{R_0^5}\frac{1}{2}e_{-2} + (m_0 - n_0)(\kappa_1 + 1)R_0\frac{1}{2}e_4 \\ + \left((m_0 - n_0)(\kappa_1 + 1)R_0g_7f_{-3} + (m_0 + n_0)(\kappa_1 + 1)\frac{1}{R_0^5}g_1f_{-3}\right) \\ + \left((m_0 - n_0)(\kappa_1 + 1)R_0g_6f_{-2} + (m_0 + n_0)(\kappa_1 + 1)\frac{1}{R_0^5}g_0f_{-2}\right) \\ + (m_0 - n_0)(\kappa_1 + 1)R_0g_5f_{-1} + (m_0 - n_0)(\kappa_1 + 1)R_0g_4f_0 \\ + (m_0 - n_0)(\kappa_1 + 1)R_0g_3f_1 + (m_0 - n_0)(\kappa_1 + 1)R_0g_2f_2 \\ + (m_0 - n_0)(\kappa_1 + 1)R_0g_1f_3 + (m_0 - n_0)(\kappa_1 + 1)R_0g_0f_4,$$

•

$$0 = \left(16\frac{1}{M_0} - \frac{(m_0 - n_0)}{(m_0 + n_0)}(1 + \kappa_0 \frac{\mu_1}{\mu_0}) - 4(1 - \frac{\mu_1}{\mu_0})\right) a_4$$

÷
$$+ \left(-(1 - \frac{\mu_1}{\mu_0}) \frac{1}{R_0^2} + \frac{1}{M_0} \frac{4}{R_0^2} \right) b_2 + (\kappa_1 + 1) \frac{1}{R_0^6} \frac{1}{2} e_{-2} + \frac{(m_0 - n_0)}{(m_0 + n_0)} (\kappa_1 + 1) \frac{1}{2} e_4 \\ + \left(\frac{(m_0 - n_0)}{(m_0 + n_0)} (\kappa_1 + 1) g_7 + (\kappa_1 + 1) \frac{1}{R_0^6} g_1 \right) f_{-3} \\ + \left(\frac{(m_0 - n_0)}{(m_0 + n_0)} (\kappa_1 + 1) g_6 + (\kappa_1 + 1) \frac{1}{R_0^6} g_0 \right) f_{-2} + \frac{(m_0 - n_0)}{(m_0 + n_0)} (\kappa_1 + 1) g_5 f_{-1} \\ + \frac{(m_0 - n_0)}{(m_0 + n_0)} (\kappa_1 + 1) g_4 f_0 + \frac{(m_0 - n_0)}{(m_0 + n_0)} (\kappa_1 + 1) g_3 f_1 \\ + \frac{(m_0 - n_0)}{(m_0 + n_0)} (\kappa_1 + 1) g_2 f_2 + \frac{(m_0 - n_0)}{(m_0 + n_0)} (\kappa_1 + 1) g_1 f_3 + \frac{(m_0 - n_0)}{(m_0 + n_0)} (\kappa_1 + 1) g_0 f_4,$$

13)
$$z^2 \quad \left[\frac{2\mu_1(m_0+n_0)(\varepsilon_2-i\varepsilon_3)}{R_0}=0\right]$$

..

,

$$0 = \left(12\mu_{1} - (m_{0} - n_{0})(1 + \kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0} - 3(m_{0} + n_{0})(1 - \frac{\mu_{1}}{\mu_{0}})R_{0}\right)a_{3} \\ + \left(-(m_{0} + n_{0})(1 - \frac{\mu_{1}}{\mu_{0}})\frac{1}{R_{0}} + \mu_{1}\frac{4}{R_{0}^{2}}\right)b_{1} + (m_{0} + n_{0})(\kappa_{1} + 1)\frac{1}{R_{0}^{3}}\frac{1}{2}e_{-1} \\ + (m_{0} - n_{0})(\kappa_{1} + 1)R_{0}\frac{1}{2}e_{3} \\ + \left((m_{0} - n_{0})(\kappa_{1} + 1)R_{0}g_{6}f_{-3} + (m_{0} + n_{0})(\kappa_{1} + 1)\frac{1}{R_{0}^{3}}g_{2}f_{-3}\right) \\ + \left((m_{0} - n_{0})(\kappa_{1} + 1)R_{0}g_{5}f_{-2} + (m_{0} + n_{0})(\kappa_{1} + 1)\frac{1}{R_{0}^{3}}g_{1}f_{-2}\right) \\ + \left((m_{0} - n_{0})(\kappa_{1} + 1)R_{0}g_{4}f_{-1} + (m_{0} + n_{0})(\kappa_{1} + 1)\frac{1}{R_{0}^{3}}g_{0}f_{-1}\right) \\ + (m_{0} - n_{0})(\kappa_{1} + 1)R_{0}g_{3}f_{0} + (m_{0} - n_{0})(\kappa_{1} + 1)R_{0}g_{2}f_{1} \\ + (m_{0} - n_{0})(\kappa_{1} + 1)R_{0}g_{1}f_{2} + (m_{0} - n_{0})(\kappa_{1} + 1)R_{0}g_{0}f_{3},$$

$$0 = \left(6\frac{1}{M_0} - \frac{(m_0 - n_0)}{(m_0 + n_0)}(1 + \kappa_0\frac{\mu_1}{\mu_0}) - 3(1 - \frac{\mu_1}{\mu_0})\right)a_3 + \left(-(1 - \frac{\mu_1}{\mu_0})\frac{1}{R_0^2} + \frac{1}{M_0}\frac{2}{R_0^2}\right)b_1 + (\kappa_1 + 1)\frac{1}{R_0^4}\frac{1}{2}e_{-1} + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)\frac{1}{2}e_3 + \left(\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_6 + (\kappa_1 + 1)\frac{1}{R_0^4}g_2\right)f_{-3}$$

.

.

•

$$+ \left(\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_5 + (\kappa_1 + 1)\frac{1}{R_0^4}g_1\right)f_{-2} \\ + \left(\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_4 + (\kappa_1 + 1)\frac{1}{R_0^4}g_0\right)f_{-1} \\ + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_3f_0 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_2f_1 \\ + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_1f_2 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_0f_3$$

ŕ

14) z

••

$$0 = + \left(-(m_0 - n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})R_0 - 2(m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})R_0 \right) a_2$$

-(m_0 + n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})\frac{1}{R_0}a_0 - (m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})\frac{1}{R_0}b_0
+(m_0 + n_0)(\kappa_1 + 1)\frac{1}{R_0}\frac{1}{2}e_0 + (m_0 - n_0)(\kappa_1 + 1)R_0\frac{1}{2}e_2
+ $\left((m_0 - n_0)(\kappa_1 + 1)R_0g_5f_{-3} + (m_0 + n_0)(\kappa_1 + 1)\frac{1}{R_0}g_3f_{-3} \right)$
+ $\left((m_0 - n_0)(\kappa_1 + 1)R_0g_4f_{-2} + (m_0 + n_0)(\kappa_1 + 1)\frac{1}{R_0}g_2f_{-2} \right)$
+ $\left((m_0 - n_0)(\kappa_1 + 1)R_0g_3f_{-1} + (m_0 + n_0)(\kappa_1 + 1)\frac{1}{R_0}g_1f_{-1} \right)$
+ $\left((m_0 - n_0)(\kappa_1 + 1)R_0g_2f_0 + (m_0 + n_0)(\kappa_1 + 1)\frac{1}{R_0}g_0f_0 \right)$
+ $(m_0 - n_0)(\kappa_1 + 1)R_0g_1f_1 + (m_0 - n_0)(\kappa_1 + 1)R_0g_0f_2,$

$$0 = \left(-\frac{(m_0 - n_0)}{(m_0 + n_0)}(1 + \kappa_0 \frac{\mu_1}{\mu_0})R_0 - 2(1 - \frac{\mu_1}{\mu_0})R_0\right)a_2 - (1 + \kappa_0 \frac{\mu_1}{\mu_0})\frac{1}{R_0}a_0$$

- $(1 - \frac{\mu_1}{\mu_0})\frac{1}{R_0}b_0 + (\kappa_1 + 1)\frac{1}{R_0}\frac{1}{2}e_0 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)R_0\frac{1}{2}e_2$
+ $\left(\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)R_0g_5 + (\kappa_1 + 1)\frac{1}{R_0}g_3\right)f_{-3}$
+ $\left(\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)R_0g_4 + (\kappa_1 + 1)\frac{1}{R_0}g_2\right)f_{-2}$

đ

$$+ \left(\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)R_0g_3 + (\kappa_1 + 1)\frac{1}{R_0}g_1\right)f_{-1} \\ + \left(\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)R_0g_2 + (\kappa_1 + 1)\frac{1}{R_0}g_0\right)f_0 \\ + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)R_0g_1f_1 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)R_0g_0f_2$$

15)
$$z^0$$
 $(4m_0R_0\mu_1\varepsilon_1=0)$

$$0 = \begin{pmatrix} -8\mu_1 - (m_0 - n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})R_0 - (m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0 \\ -(m_0 + n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})R_0 - (m_0 + n_0)(1 - \frac{\mu_1}{\mu_0})R_0 \end{pmatrix} a_1 \\ +2m_0(\kappa_1 + 1)R_0 \frac{1}{2}e_1 + 2m_0(\kappa_1 + 1)R_0g_4f_{-3} + 2m_0(\kappa_1 + 1)R_0g_3f_{-2} \\ +2m_0(\kappa_1 + 1)R_0g_2f_{-1} + 2m_0(\kappa_1 + 1)R_0g_1f_0 + 2m_0(\kappa_1 + 1)R_0g_0f_1, \end{pmatrix}$$

$$0 = \begin{pmatrix} -4\frac{1}{M_0} - \frac{(m_0 - n_0)}{(m_0 + n_0)}(1 + \kappa_0\frac{\mu_1}{\mu_0}) - \frac{(m_0 - n_0)}{(m_0 + n_0)}(1 - \frac{\mu_1}{\mu_0}) \\ -(1 + \kappa_0\frac{\mu_1}{\mu_0}) - (1 - \frac{\mu_1}{\mu_0}) \end{pmatrix} a_1 + \frac{m_0(\kappa_1 + 1)}{(m_0 + n_0)}e_1 + \frac{2m_0(\kappa_1 + 1)}{(m_0 + n_0)}g_4f_{-3} + \frac{2m_0(\kappa_1 + 1)}{(m_0 + n_0)}g_3f_{-2} + \frac{2m_0(\kappa_1 + 1)}{(m_0 + n_0)}g_2f_{-1} + \frac{2m_0(\kappa_1 + 1)}{(m_0 + n_0)}g_1f_0 + \frac{2m_0(\kappa_1 + 1)}{(m_0 + n_0)}g_0f_1,$$

16) z^{-1}

$$0 = -(m_0 - n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})R_0 a_0 + \begin{pmatrix} -8\mu_1 R_0^2 - 2(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^3 \\ -(m_0 + n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})R_0^3 \end{pmatrix} a_2$$

$$-(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0 b_0 + (m_0 - n_0)(\kappa_1 + 1)R_0 \frac{1}{2}e_0$$

$$+(m_0 + n_0)(\kappa_1 + 1)R_0^3 \frac{1}{2}e_2 + \begin{pmatrix} (m_0 - n_0)(\kappa_1 + 1)R_0 g_3 f_{-3} \\ +(m_0 + n_0)(\kappa_1 + 1)R_0^3 g_5 f_{-3} \end{pmatrix}$$

$$+((m_0 - n_0)(\kappa_1 + 1)R_0 g_2 f_{-2} + (m_0 + n_0)(\kappa_1 + 1)R_0^3 g_4 f_{-2})$$

173

+
$$((m_0 - n_0)(\kappa_1 + 1)R_0g_1f_{-1} + (m_0 + n_0)(\kappa_1 + 1)R_0^3g_3f_{-1})$$

+ $((m_0 - n_0)(\kappa_1 + 1)R_0g_0f_0 + (m_0 + n_0)(\kappa_1 + 1)R_0^3g_2f_0)$
+ $(m_0 + n_0)(\kappa_1 + 1)R_0^3g_1f_1 + (m_0 + n_0)(\kappa_1 + 1)R_0^3g_0f_2,$

$$\begin{array}{lll} 0 & = & -\frac{\left(m_{0}-n_{0}\right)}{\left(m_{0}+n_{0}\right)}\left(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}}\right)a_{0}+\left(\begin{array}{c} -4\frac{1}{M_{0}}R_{0}^{2}-\frac{\left(m_{0}-n_{0}\right)}{\left(m_{0}+n_{0}\right)}2\left(1-\frac{\mu_{1}}{\mu_{0}}\right)R_{0}^{2}\right)}{-\left(1+\kappa_{0}\frac{\mu_{1}}{\mu_{0}}\right)R_{0}^{2}}\right)a_{2}\\ & -\frac{\left(m_{0}-n_{0}\right)}{\left(m_{0}+n_{0}\right)}\left(1-\frac{\mu_{1}}{\mu_{0}}\right)b_{0}+\frac{\left(m_{0}-n_{0}\right)}{\left(m_{0}+n_{0}\right)}\left(\kappa_{1}+1\right)\frac{1}{2}e_{0}+\left(\kappa_{1}+1\right)R_{0}^{2}\frac{1}{2}e_{2}\\ & +\left(\frac{\left(m_{0}-n_{0}\right)}{\left(m_{0}+n_{0}\right)}\left(\kappa_{1}+1\right)g_{3}+\left(\kappa_{1}+1\right)R_{0}^{2}g_{5}\right)f_{-3}\\ & +\left(\frac{\left(m_{0}-n_{0}\right)}{\left(m_{0}+n_{0}\right)}\left(\kappa_{1}+1\right)g_{2}+\left(\kappa_{1}+1\right)R_{0}^{2}g_{4}\right)f_{-2}\\ & +\left(\frac{\left(m_{0}-n_{0}\right)}{\left(m_{0}+n_{0}\right)}\left(\kappa_{1}+1\right)g_{1}+\left(\kappa_{1}+1\right)R_{0}^{2}g_{3}\right)f_{-1}\\ & +\left(\frac{\left(m_{0}-n_{0}\right)}{\left(m_{0}+n_{0}\right)}\left(\kappa_{1}+1\right)g_{0}+\left(\kappa_{1}+1\right)R_{0}^{2}g_{2}\right)f_{0}+\left(\kappa_{1}+1\right)R_{0}^{2}g_{1}f_{1}+\left(\kappa_{1}+1\right)R_{0}^{2}g_{0}f_{2},\end{array}$$

17)
$$z^{-2}$$
 $(2\mu_1(m_0 - n_0)(\varepsilon_2 + i\varepsilon_3)R_0^3 = 0)$

..

$$0 = \left(-12\mu_{1}R_{0}^{4} - 3(m_{0} - n_{0})(1 - \frac{\mu_{1}}{\mu_{0}})R_{0}^{5} - (m_{0} + n_{0})(1 + \kappa_{0}\frac{\mu_{1}}{\mu_{0}})R_{0}^{5}\right)a_{3}$$

- $(m_{0} - n_{0})(1 - \frac{\mu_{1}}{\mu_{0}})R_{0}^{3}b_{1} + (m_{0} - n_{0})(\kappa_{1} + 1)R_{0}\frac{1}{2}e_{-1}$
+ $(m_{0} + n_{0})(\kappa_{1} + 1)R_{0}^{5}\frac{1}{2}e_{3} + \left(\begin{array}{c}(m_{0} - n_{0})(\kappa_{1} + 1)R_{0}g_{2}f_{-3}\\ +(m_{0} + n_{0})(\kappa_{1} + 1)R_{0}g_{1}f_{-2} + (m_{0} + n_{0})(\kappa_{1} + 1)R_{0}^{5}g_{6}f_{-3}\end{array}\right)$
+ $((m_{0} - n_{0})(\kappa_{1} + 1)R_{0}g_{1}f_{-2} + (m_{0} + n_{0})(\kappa_{1} + 1)R_{0}^{5}g_{4}f_{-1})$
+ $((m_{0} - n_{0})(\kappa_{1} + 1)R_{0}g_{0}f_{-1} + (m_{0} + n_{0})(\kappa_{1} + 1)R_{0}^{5}g_{4}f_{-1})$
+ $(m_{0} + n_{0})(\kappa_{1} + 1)R_{0}^{5}g_{3}f_{0} + (m_{0} + n_{0})(\kappa_{1} + 1)R_{0}^{5}g_{2}f_{1}$
+ $(m_{0} + n_{0})(\kappa_{1} + 1)R_{0}^{5}g_{1}f_{2} + (m_{0} + n_{0})(\kappa_{1} + 1)R_{0}^{5}g_{0}f_{3},$

$$0 = \left(-\frac{6}{M_0}R_0^4 - \frac{(m_0 - n_0)}{(m_0 + n_0)}3(1 - \frac{\mu_1}{\mu_0})R_0^4 - (1 + \kappa_0\frac{\mu_1}{\mu_0})R_0^4\right)a_3$$

$$-\frac{(m_0 - n_0)}{(m_0 + n_0)}(1 - \frac{\mu_1}{\mu_0})R_0^2b_1 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)\frac{1}{2}e_{-1}$$

$$+(\kappa_1 + 1)R_0^4\frac{1}{2}e_3 + \left(\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_2 + (\kappa_1 + 1)R_0^4g_6\right)f_{-3}$$

$$+ \left(\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_1 + (\kappa_1 + 1)R_0^4g_5\right)f_{-2}$$

$$+ \left(\frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_0 + (\kappa_1 + 1)R_0^4g_4\right)f_{-1} + (\kappa_1 + 1)R_0^4g_3f_0$$

$$+(\kappa_1 + 1)R_0^4g_2f_1 + (\kappa_1 + 1)R_0^4g_1f_2 + (\kappa_1 + 1)R_0^4g_0f_3,$$

18) z^{-3}

÷

.

$$0 = \left(-16\mu_1 R_0^6 - 4(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^7 - (m_0 + n_0)(1 + \kappa_0 \frac{\mu_1}{\mu_0})R_0^7\right) a_4 -(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^5 b_2 + (m_0 - n_0)(\kappa_1 + 1)R_0 \frac{1}{2}e_{-2} +(m_0 + n_0)(\kappa_1 + 1)R_0^7 \frac{1}{2}e_4 + \left(\begin{array}{c}(m_0 + n_0)(\kappa_1 + 1)R_0^7 g_7 f_{-3} \\+(m_0 - n_0)(\kappa_1 + 1)R_0^7 g_6 f_{-2} + (m_0 - n_0)(\kappa_1 + 1)R_0 g_1 f_{-3}\end{array}\right) + \left((m_0 + n_0)(\kappa_1 + 1)R_0^7 g_6 f_{-2} + (m_0 - n_0)(\kappa_1 + 1)R_0 g_0 f_{-2}\right) +(m_0 + n_0)(\kappa_1 + 1)R_0^7 g_5 f_{-1} +(m_0 + n_0)(\kappa_1 + 1)R_0^7 g_2 f_2 + (m_0 + n_0)(\kappa_1 + 1)R_0^7 g_3 f_1 +(m_0 + n_0)(\kappa_1 + 1)R_0^7 g_2 f_2 + (m_0 + n_0)(\kappa_1 + 1)R_0^7 g_1 f_3 . +(m_0 + n_0)(\kappa_1 + 1)R_0^7 g_0 f_4,$$

$$0 = \left(-8\frac{1}{M_0}R_0^6 - \frac{(m_0 - n_0)}{(m_0 + n_0)}4(1 - \frac{\mu_1}{\mu_0})R_0^6 - (1 + \kappa_0\frac{\mu_1}{\mu_0})R_0^6\right)a_4$$

$$-\frac{(m_0 - n_0)}{(m_0 + n_0)}(1 - \frac{\mu_1}{\mu_0})R_0^4b_2 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)\frac{1}{2}e_{-2}$$

.

$$+ (\kappa_1 + 1)R_0^6 \frac{1}{2}e_4 + \left((\kappa_1 + 1)R_0^6 g_7 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_1\right)f_{-3} \\ + \left((\kappa_1 + 1)R_0^6 g_6 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_0\right)f_{-2} + (\kappa_1 + 1)R_0^6 g_5 f_{-1} \\ + (\kappa_1 + 1)R_0^6 g_4 f_0 + (\kappa_1 + 1)R_0^6 g_3 f_1 + (\kappa_1 + 1)R_0^6 g_2 f_2 \\ + (\kappa_1 + 1)R_0^6 g_1 f_3 + (\kappa_1 + 1)R_0^6 g_0 f_4,$$

19) z^{-4} .

,

.

$$0 = \left(-20\mu_1 R_0^8 - 5(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^9 - (m_0 + n_0)(1 + \kappa_0\frac{\mu_1}{\mu_0})R_0^9\right)a_5$$

- $(m_0 - n_0)(1 - \frac{\mu_1}{\mu_0})R_0^7b_3 + (m_0 - n_0)(\kappa_1 + 1)R_0\frac{1}{2}e_{-3}$
+ $(m_0 - n_0)(\kappa_1 + 1)R_0g_0f_{-3} + (m_0 + n_0)(\kappa_1 + 1)R_0^9g_7f_{-2}$
+ $(m_0 + n_0)(\kappa_1 + 1)R_0^9g_6f_{-1} + (m_0 + n_0)(\kappa_1 + 1)R_0^9g_5f_0$
+ $(m_0 + n_0)(\kappa_1 + 1)R_0^9g_4f_1 + (m_0 + n_0)(\kappa_1 + 1)R_0^9g_3f_2$
+ $(m_0 + n_0)(\kappa_1 + 1)R_0^9g_2f_3 + (m_0 + n_0)(\kappa_1 + 1)R_0^9g_1f_4,$

$$0 = \left(-10\frac{1}{M_0}R_0^8 - \frac{(m_0 - n_0)}{(m_0 + n_0)}5(1 - \frac{\mu_1}{\mu_0})R_0^8 - (1 + \kappa_0\frac{\mu_1}{\mu_0})R_0^8\right)a_5 - \frac{(m_0 - n_0)}{(m_0 + n_0)}(1 - \frac{\mu_1}{\mu_0})R_0^6b_3 + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)\frac{1}{2}e_{-3} + \frac{(m_0 - n_0)}{(m_0 + n_0)}(\kappa_1 + 1)g_0f_{-3} + (\kappa_1 + 1)R_0^8g_7f_{-2} + (\kappa_1 + 1)R_0^8g_6f_{-1} + (\kappa_1 + 1)R_0^8g_5f_0 + (\kappa_1 + 1)R_0^8g_4f_1 + (\kappa_1 + 1)R_0^8g_3f_2 + (\kappa_1 + 1)R_0^8g_2f_3 + (\kappa_1 + 1)R_0^8g_1f_4.$$

Other 9 equations, (3.62) thru (3.70) from the traction continuation condition along Γ_1 are;

20) z^5

$$0 = +\frac{4}{R_1^{10}}c_4 + 4\frac{1}{R_1^2}e_4$$

-4
$$\frac{1}{R_1^{10}}\begin{pmatrix} h_{-8}f_4 + h_{-7}f_3 + h_{-6}f_2 + h_{-5}f_1 + h_{-4}f_0 \\ +h_{-3}f_{-1} + h_{-2}f_{-2} + h_{-1}f_{-3} + \frac{1}{2}e_{-4} \end{pmatrix},$$

$$0 = +\frac{4}{R_{1}^{10}}c_{4} + 4\frac{1}{R_{1}^{2}}e_{4} - \frac{2}{R_{1}^{10}}e_{-4} - \frac{4}{R_{1}^{10}}h_{-1}f_{-3} - \frac{4}{R_{1}^{10}}h_{-2}f_{-2}$$

$$-\frac{4}{R_{1}^{10}}h_{-3}f_{-1} - \frac{4}{R_{1}^{10}}h_{-4}f_{0} - \frac{4}{R_{1}^{10}}h_{-5}f_{1} - \frac{4}{R_{1}^{10}}h_{-6}f_{2}$$

$$-\frac{4}{R_{1}^{10}}h_{-7}f_{3} - \frac{4}{R_{1}^{10}}h_{-8}f_{4},$$

21) z^4

;

$$0 = +\frac{3}{R_1^8}c_3 + 3\frac{1}{R_1^2}e_3 + 6\frac{1}{R_1^2}\left(h_{-1}f_4 + \frac{1}{2}e_3\right) \\ -3\frac{1}{R_1^8}\left(\begin{array}{c}h_{-7}f_4 + h_{-6}f_3 + h_{-5}f_2 + h_{-4}f_1 + h_{-3}f_0 \\ +h_{-2}f_{-1} + h_{-1}f_{-2} + \frac{1}{2}e_{-3}\end{array}\right),$$

$$0 = +\frac{3}{R_1^8}c_3 + \frac{6}{R_1^2}e_3 + \left(\frac{6}{R_1^2}h_{-1} - \frac{3}{R_1^8}h_{-7}\right)f_4 -\frac{3}{R_1^8}h_{-6}f_3 - \frac{3}{R_1^8}h_{-5}f_2 - \frac{3}{R_1^8}h_{-4}f_1 - \frac{3}{R_1^8}h_{-3}f_0 -\frac{3}{R_1^8}h_{-2}f_{-1} - \frac{3}{R_1^8}h_{-1}f_{-2} - \frac{1}{R_1^8}\frac{3}{2}e_{-3},$$

22) z^3

4

$$0 = +\frac{2}{R_1^6}c_2 + 2\frac{1}{R_1^2}e_2 + 2\frac{1}{R_1^2}\left(h_{-2}f_4 + h_{-1}f_3 + \frac{1}{2}e_2\right)$$

•

$$-2\frac{1}{R_1^6} \left(\begin{array}{c} h_{-6}f_4 + h_{-5}f_3 + h_{-4}f_2 + h_{-3}f_1 \\ + h_{-2}f_0 + h_{-1}f_{-1} + \frac{1}{2}e_{-2} \end{array} \right),$$

$$0 = \frac{2}{R_1^6} c_2 + \frac{3}{R_1^2} e_2 - \frac{1}{R_1^6} e_{-2} + \left(\frac{2}{R_1^2} h_{-2} - \frac{2}{R_1^6} h_{-6}\right) f_4 + \left(\frac{2}{R_1^2} h_{-1} - \frac{2}{R_1^6} h_{-5}\right) f_3 - \frac{2}{R_1^6} h_{-4} f_2 - \frac{2}{R_1^6} h_{-3} f_1 - \frac{2}{R_1^6} h_{-2} f_0 - \frac{2}{R_1^6} h_{-1} f_{-1},$$

23) z^2

:

4

,

.

,

$$-\frac{1}{R_1^2}B = +\frac{1}{R_1^4}c_1 + \frac{1}{R_1^2}e_1 - 3\left(h_{-1}f_4 + \frac{1}{2}e_3\right) \\ -\frac{1}{R_1^4}\left(h_{-5}f_4 + h_{-4}f_3 + h_{-3}f_2 + h_{-2}f_1 + h_{-1}f_0 + \frac{1}{2}e_{-1}\right)$$

$$\begin{aligned} -\frac{1}{R_1^2}B &= +\frac{1}{R_1^4}c_1 - \frac{3}{2}e_3 + \frac{1}{R_1^2}e_1 - \frac{1}{R_1^4}\frac{1}{2}e_{-1} + \left(-3h_{-1} - \frac{1}{R_1^4}h_{-5}\right)f_4 \\ &- \frac{1}{R_1^4}h_{-4}f_3 - \frac{1}{R_1^4}h_{-3}f_2 - \frac{1}{R_1^4}h_{-2}f_1 - \frac{1}{R_1^4}h_{-1}f_0, \end{aligned}$$

24) z^0

$$2A = -\frac{1}{R_1^2} d_1 - \frac{1}{R_1^2} e_{-1} + 2\left(h_{-3}f_4 + h_{-2}f_3 + h_{-1}f_2 + \frac{1}{2}e_1\right) + 2\frac{1}{R_1^2}\left(h_{-5}f_4 + h_{-4}f_3 + h_{-3}f_2 + h_{-2}f_1 + h_{-1}f_0 + \frac{1}{2}e_{-1}\right),$$

$$2A = -\frac{1}{R_1^2}d_1 + e_1 + \left(2h_{-3} + \frac{2}{R_1^2}h_{-5}\right)f_4 + \left(2h_{-2} + \frac{2}{R_1^2}h_{-4}\right)f_3 \\ + \left(2h_{-1} + \frac{2}{R_1^2}h_{-3}\right)f_2 + \frac{2}{R_1^2}h_{-2}f_1 + \frac{2}{R_1^2}h_{-1}f_0,$$

,

25)
$$z^{-1}$$

$$0 = -\frac{2}{R_1^2} d_2 - 2\frac{1}{R_1^2} e_{-2} + 2R_1^2 \left(h_{-2}f_4 + h_{-1}f_3 + \frac{1}{2}e_2 \right) + 6\frac{1}{R_1^2} \left(h_{-6}f_4 + h_{-5}f_3 + h_{-4}f_2 + h_{-3}f_1 + h_{-2}f_0 + h_{-1}f_{-1} + \frac{1}{2}e_{-2} \right),$$

$$0 = -\frac{2}{R_1^2} d_2 + R_1^2 e_2 + \frac{1}{R_1^2} e_{-2} + \left(2R_1^2 h_{-2} + \frac{6}{R_1^2} h_{-6}\right) f_4 + \left(2R_1^2 h_{-1} + \frac{6}{R_1^2} h_{-5}\right) f_3 + \frac{6}{R_1^2} h_{-4} f_2 + \frac{6}{R_1^2} h_{-3} f_1 + \frac{6}{R_1^2} h_{-2} f_0 + \frac{6}{R_1^2} h_{-1} f_{-1},$$

26) z^{-2}

•

4

$$0 = +3c_1 - \frac{3}{R_1^2}d_3 - 3\frac{1}{R_1^2}e_{-3} + 3R_1^4\left(h_{-1}f_4 + \frac{1}{2}e_3\right) -3\left(h_{-5}f_4 + h_{-4}f_3 + h_{-3}f_2 + h_{-2}f_1 + h_{-1}f_0 + \frac{1}{2}e_{-1}\right) +12\frac{1}{R_1^2}\left(h_{-7}f_4 + h_{-6}f_3 + h_{-5}f_2 + h_{-4}f_1 + h_{-3}f_0 + h_{-2}f_{-1} + h_{-1}f_{-2} + \frac{1}{2}e_{-3}\right),$$

$$0 = +3c_1 - \frac{3}{R_1^2}d_3 + R_1^4 \frac{3}{2}e_3 - \frac{3}{2}e_{-1} + \frac{3}{R_1^2}e_{-3} + \left(\frac{12}{R_1^2}h_{-7} + 3R_1^4h_{-1} - 3h_{-5}\right)f_4 + \left(\frac{12}{R_1^2}h_{-6} - 3h_{-4}\right)f_3 + \left(\frac{12}{R_1^2}h_{-5} - 3h_{-3}\right)f_2 + \left(\frac{12}{R_1^2}h_{-4} - 3h_{-2}\right)f_1 + \left(\frac{12}{R_1^2}h_{-3} - 3h_{-1}\right)f_0 + \frac{12}{R_1^2}h_{-2}f_{-1} + \frac{12}{R_1^2}h_{-1}f_{-2},$$

27) z^{-3}

$$0 = +8c_2 - \frac{4}{R_1^2}d_4 - 4\frac{1}{R_1^2}e_{-4} -8\left(h_{-6}f_4 + h_{-5}f_3 + h_{-4}f_2 + h_{-3}f_1 + h_{-2}f_0 + h_{-1}f_{-1} + \frac{1}{2}e_{-2}\right)$$

.

÷

,

$$+20\frac{1}{R_{1}^{2}}\left(\begin{array}{c}h_{-8}f_{4}+h_{-7}f_{3}+h_{-6}f_{2}+h_{-5}f_{1}+h_{-4}f_{0}+h_{-3}f_{-1}\\+h_{-2}f_{-2}+h_{-1}f_{-3}+\frac{1}{2}e_{-4}\end{array}\right),$$

$$0 = +8c_2 - \frac{4}{R_1^2}d_4 - 4e_{-2} + \frac{6}{R_1^2}e_{-4} + \left(\frac{20}{R_1^2}h_{-8} - 8h_{-6}\right)f_4 + \left(\frac{20}{R_1^2}h_{-7} - 8h_{-5}\right)f_3 + \left(\frac{20}{R_1^2}h_{-6} - 8h_{-4}\right)f_2 + \left(\frac{20}{R_1^2}h_{-5} - 8h_{-3}\right)f_1 + \left(\frac{20}{R_1^2}h_{-4} - 8h_{-2}\right)f_0 + \left(\frac{20}{R_1^2}h_{-3} - 8h_{-1}\right)f_{-1} + \frac{20}{R_1^2}h_{-2}f_{-2} + \frac{20}{R_1^2}h_{-1}f_{-3},$$

28) z^{-4}

$$0 = +15c_3 - \frac{5}{R_1^2}d_5 - 15\begin{pmatrix} h_{-7}f_4 + h_{-6}f_3 + h_{-5}f_2 + h_{-4}f_1 \\ +h_{-3}f_0 + h_{-2}f_{-1} + h_{-1}f_{-2} + \frac{1}{2}e_{-3} \end{pmatrix} \\ + 30\frac{1}{R_1^2}\begin{pmatrix} h_{-8}f_3 + h_{-7}f_2 + h_{-6}f_1 + h_{-5}f_0 + h_{-4}f_{-1} \\ +h_{-3}f_{-2} + h_{-2}f_{-3} + h_{-1}f_{-4} + \frac{1}{2}e_{-5} \end{pmatrix},$$

$$0 = 15c_3 - \frac{5}{R_1^2}d_5 - \frac{15}{2}e_{-3} + \frac{15}{R_1^2}e_{-5} - 15h_{-7}f_4 + \left(\frac{30}{R_1^2}h_{-8} - 15h_{-6}\right)f_3 + \left(\frac{30}{R_1^2}h_{-7} - 15h_{-5}\right)f_2 + \left(\frac{30}{R_1^2}h_{-6} - 15h_{-4}\right)f_1 + \left(\frac{30}{R_1^2}h_{-5} - 15h_{-3}\right)f_0 + \left(\frac{30}{R_1^2}h_{-4} - 15h_{-2}\right)f_{-1} + \left(\frac{30}{R_1^2}h_{-3} - 15h_{-1}\right)f_{-2} + \frac{30}{R_1^2}h_{-2}f_{-3} + \frac{30}{R_1^2}h_{-1}f_{-4}.$$

Again, non-dimensional parameter $M_1 = \frac{m_1+n_1}{2\mu_1}R_1$ is deployed into displacement jump equations along the Γ_1 . Finally, the other 10 equations, (3.71) thru (3.80) are;

180

÷

$$0 = \left(+\frac{16}{R_1^{10}}\mu_1 - (m_1 - n_1)\left(1 - \frac{\mu_1}{\mu_2}\right)\frac{4}{R_1^9} + (m_1 + n_1)\left(1 + \frac{\mu_1}{\mu_2}k_2\right)\frac{1}{R_1^9}\right)c_4 - (m_1 + n_1)(k_1 + 1)\frac{1}{R_1^9} \left(\begin{array}{c} h_{-8}f_4 + h_{-7}f_3 + h_{-6}f_2 + h_{-5}f_1 \\ + h_{-4}f_0 + h_{-3}f_{-1} + h_{-2}f_{-2} + h_{-1}f_{-3} + \frac{1}{2}e_{-4} \end{array} \right),$$

.

$$0 = \left(\frac{1}{M_1}\frac{8}{R_1^9} - \frac{(m_1 - n_1)}{(m_1 + n_1)}\left(1 - \frac{\mu_1}{\mu_2}\right)\frac{4}{R_1^9} + \left(1 + \frac{\mu_1}{\mu_2}k_2\right)\frac{1}{R_1^9}\right)c_4 - (k_1 + 1)\frac{1}{R_1^9}h_{-8}f_4 - (k_1 + 1)\frac{1}{R_1^9}h_{-7}f_3 - (k_1 + 1)\frac{1}{R_1^9}h_{-6}f_2 - (k_1 + 1)\frac{1}{R_1^9}h_{-5}f_1 - (k_1 + 1)\frac{1}{R_1^9}h_{-4}f_0 - (k_1 + 1)\frac{1}{R_1^9}h_{-3}f_{-1} - (k_1 + 1)\frac{1}{R_1^9}h_{-2}f_{-2} - (k_1 + 1)\frac{1}{R_1^9}h_{-1}f_{-3} - (k_1 + 1)\frac{1}{R_1^9}\frac{1}{2}e_{-4},$$

30) z^4

.

29) z⁵

$$0 = \left(+\frac{12}{R_1^8} \mu_1 - (m_1 - n_1) \left(1 - \frac{\mu_1}{\mu_2} \right) \frac{3}{R_1^7} + (m_1 + n_1) \left(1 + \frac{\mu_1}{\mu_2} k_2 \right) \frac{1}{R_1^7} \right) c_3 + (m_1 - n_1) \left(1 - \frac{\mu_1}{\mu_2} \right) \frac{1}{R_1^9} d_5 - \frac{(m_1 + n_1) (k_1 + 1)}{R_1^7} \left(\begin{array}{c} h_{-7} f_4 + h_{-6} f_3 + h_{-5} f_2 + h_{-4} f_1 + h_{-3} f_0 \\ + h_{-2} f_{-1} + h_{-1} f_{-2} + \frac{1}{2} e_{-3} \end{array} \right),$$

$$0 = \left(\frac{1}{M_1}\frac{6}{R_1^7} - \frac{(m_1 - n_1)}{(m_1 + n_1)}\left(1 - \frac{\mu_1}{\mu_2}\right)\frac{3}{R_1^7} + \left(1 + \frac{\mu_1}{\mu_2}k_2\right)\frac{1}{R_1^7}\right)c_3 + \frac{(m_1 - n_1)}{(m_1 + n_1)}\left(1 - \frac{\mu_1}{\mu_2}\right)\frac{1}{R_1^9}d_5 - \frac{(k_1 + 1)}{R_1^7}\frac{1}{2}e_{-3} - \frac{(k_1 + 1)}{R_1^7}h_{-7}f_4 - \frac{(k_1 + 1)}{R_1^7}h_{-6}f_3 - \frac{(k_1 + 1)}{R_1^7}h_{-5}f_2 - \frac{(k_1 + 1)}{R_1^7}h_{-4}f_1 - \frac{(k_1 + 1)}{R_1^7}h_{-3}f_0 - \frac{(k_1 + 1)}{R_1^7}h_{-2}f_{-1} - \frac{(k_1 + 1)}{R_1^7}h_{-1}f_{-2},$$

÷

31)
$$z^3$$

$$0 = \left(+\frac{8}{R_1^6} \mu_1 - (m_1 - n_1) \left(1 - \frac{\mu_1}{\mu_2} \right) \frac{2}{R_1^5} + (m_1 + n_1) \left(1 + \frac{\mu_1}{\mu_2} k_2 \right) \frac{1}{R_1^5} \right) c_2 + (m_1 - n_1) \left(1 - \frac{\mu_1}{\mu_2} \right) \frac{1}{R_1^7} d_4 - (m_1 + n_1) (k_1 + 1) \frac{1}{R_1^5} \left(\begin{array}{c} h_{-6}f_4 + h_{-5}f_3 + h_{-4}f_2 + h_{-3}f_1 \\ + h_{-2}f_0 + h_{-1}f_{-1} + \frac{1}{2}e_{-2} \end{array} \right),$$

$$0 = \left(\frac{1}{M_{1}}\frac{4}{R_{1}^{5}} - \frac{(m_{1} - n_{1})}{(m_{1} + n_{1})}\left(1 - \frac{\mu_{1}}{\mu_{2}}\right)\frac{2}{R_{1}^{5}} + \left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2}\right)\frac{1}{R_{1}^{5}}\right)c_{2} \\ + \frac{(m_{1} - n_{1})}{(m_{1} + n_{1})}\left(1 - \frac{\mu_{1}}{\mu_{2}}\right)\frac{1}{R_{1}^{7}}d_{4} - \frac{(k_{1} + 1)}{R_{1}^{5}}\frac{1}{2}e_{-2} - \frac{(k_{1} + 1)}{R_{1}^{5}}h_{-6}f_{4} - \frac{(k_{1} + 1)}{R_{1}^{5}}h_{-5}f_{3} \\ - \frac{(k_{1} + 1)}{R_{1}^{5}}h_{-4}f_{2} - \frac{(k_{1} + 1)}{R_{1}^{5}}h_{-3}f_{1} - \frac{(k_{1} + 1)}{R_{1}^{5}}h_{-2}f_{0} - \frac{(k_{1} + 1)}{R_{1}^{5}}h_{-1}f_{-1},$$

32) z^2

.

$$\begin{pmatrix} -(m_1+n_1)\left(1-\frac{\mu_1}{\mu_2}\right)\frac{1}{R_1}-4\mu_1\frac{1}{R_1^2}\right)B \\ = \left(+\frac{4}{R_1^4}\mu_1-(m_1-n_1)\left(1-\frac{\mu_1}{\mu_2}\right)\frac{1}{R_1^3}+(m_1+n_1)\left(1+\frac{\mu_1}{\mu_2}k_2\right)\frac{1}{R_1^3}\right)c_1 \\ +(m_1-n_1)\left(1-\frac{\mu_1}{\mu_2}\right)\frac{1}{R_1^5}d_3-(m_1-n_1)\left(k_1+1\right)R_1\left(h_{-1}f_4+\frac{1}{2}e_3\right) \\ -(m_1+n_1)\left(k_1+1\right)\frac{1}{R_1^3}\left(h_{-5}f_4+h_{-4}f_3+h_{-3}f_2+h_{-2}f_1+h_{-1}f_0+\frac{1}{2}e_{-1}\right),$$

$$\begin{pmatrix} -\left(1-\frac{\mu_1}{\mu_2}\right)\frac{1}{R_1}-\frac{1}{M_1}\frac{2}{R_1} \end{pmatrix} B \\ = \left(\frac{1}{M_1}\frac{2}{R_1^3}-\frac{(m_1-n_1)}{(m_1+n_1)}\left(1-\frac{\mu_1}{\mu_2}\right)\frac{1}{R_1^3}+\left(1+\frac{\mu_1}{\mu_2}k_2\right)\frac{1}{R_1^3}\right)c_1 \\ +\frac{(m_1-n_1)}{(m_1+n_1)}\left(1-\frac{\mu_1}{\mu_2}\right)\frac{1}{R_1^5}d_3-\frac{(m_1-n_1)}{(m_1+n_1)}\left(k_1+1\right)R_1\frac{1}{2}e_3$$

$$-\frac{(k_{1}+1)}{R_{1}^{3}}\frac{1}{2}e_{-1} + \left(-\frac{(m_{1}-n_{1})}{(m_{1}+n_{1})}(k_{1}+1)R_{1}h_{-1} - \frac{(k_{1}+1)}{R_{1}^{3}}h_{-5}\right)f_{4}$$
$$-\frac{(k_{1}+1)}{R_{1}^{3}}h_{-4}f_{3} - \frac{(k_{1}+1)}{R_{1}^{3}}h_{-3}f_{2} - \frac{(k_{1}+1)}{R_{1}^{3}}h_{-2}f_{1} - \frac{(k_{1}+1)}{R_{1}^{3}}h_{-1}f_{0},$$

33) z^1

$$0 = + (m_1 - n_1) \left(1 - \frac{\mu_1}{\mu_2} \right) \frac{1}{R^3} d_2 - (m_1 - n_1) (k_1 + 1) R_1 \left(h_{-2} f_4 + h_{-1} f_3 + \frac{1}{2} e_2 \right) - (m_1 + n_1) (k_1 + 1) \frac{1}{R_1} \left(h_{-4} f_4 + h_{-3} f_3 + h_{-2} f_2 + h_{-1} f_1 + \frac{1}{2} e_0 \right),$$

$$0 = \frac{(m_1 - n_1)}{(m_1 + n_1)} \left(1 - \frac{\mu_1}{\mu_2} \right) \frac{1}{R^3} d_2 - \frac{(m_1 - n_1)}{2(m_1 + n_1)} (k_1 + 1) R_1 e_2$$

- $(k_1 + 1) \frac{1}{R_1} \frac{1}{2} e_0 + \left(-\frac{(m_1 - n_1)}{(m_1 + n_1)} (k_1 + 1) R_1 h_{-2} - (k_1 + 1) \frac{1}{R_1} h_{-4} \right) f_4$
+ $\left(-\frac{(m_1 - n_1)}{(m_1 + n_1)} (k_1 + 1) R_1 h_{-1} - (k_1 + 1) \frac{1}{R_1} h_{-3} \right) f_3$
- $(k_1 + 1) \frac{1}{R_1} h_{-2} f_2 - (k_1 + 1) \frac{1}{R_1} h_{-1} f_1,$

34) z^0

•

$$- \left(-8\mu_1 A + 2m_1 \left(1 + \frac{\mu_1}{\mu_2}k_2\right)AR_1 + 2m_1 \left(1 - \frac{\mu_1}{\mu_2}\right)AR_1\right) \\ = \left(-\frac{4}{R_1^2}\mu_1 + (m_1 + n_1)\left(1 - \frac{\mu_1}{\mu_2}\right)\frac{1}{R_1} + (m_1 - n_1)\left(1 - \frac{\mu_1}{\mu_2}\right)\frac{1}{R_1}\right)d_1 \\ - 2m_1 \left(k_1 + 1\right)R_1 \left(h_{-3}f_4 + h_{-2}f_3 + h_{-1}f_2 + \frac{1}{2}e_1\right),$$

$$\frac{4A}{M_1} - \frac{2m_1A}{(m_1 + n_1)} \left(1 + \frac{\mu_1}{\mu_2} k_2 \right) - \frac{2m_1A}{(m_1 + n_1)} \left(1 - \frac{\mu_1}{\mu_2} \right)$$
$$= \left(-\frac{2}{R_1^2} \frac{1}{M_1} + \left(1 - \frac{\mu_1}{\mu_2} \right) \frac{1}{R_1^2} + \frac{(m_1 - n_1)}{(m_1 + n_1)} \left(1 - \frac{\mu_1}{\mu_2} \right) \frac{1}{R_1^2} \right) d_1$$

. .

4

.

÷

$$-\frac{m_1(k_1+1)}{(m_1+n_1)}e_1 - \frac{2m_1(k_1+1)}{(m_1+n_1)}h_{-3}f_4 - \frac{2m_1(k_1+1)}{(m_1+n_1)}h_{-2}f_3 - \frac{2m_1(k_1+1)}{(m_1+n_1)}h_{-1}f_2,$$

35) z^{-1}

.

$$0 = \left(-\frac{8}{R_1^2} \mu_1 + (m_1 + n_1) \left(1 - \frac{\mu_1}{\mu_2} \right) \frac{1}{R_1} \right) d_2$$

- $(m_1 + n_1) (k_1 + 1) R_1^3 \left(h_{-2}f_4 + h_{-1}f_3 + \frac{1}{2}e_2 \right)$
- $(m_1 - n_1) (k_1 + 1) R_1 \left(h_{-4}f_4 + h_{-3}f_3 + h_{-2}f_2 + h_{-1}f_1 + \frac{1}{2}e_0 \right),$

$$0 = \left(-\frac{4}{R_{1}}\frac{1}{M_{1}} + \left(1 - \frac{\mu_{1}}{\mu_{2}}\right)\frac{1}{R_{1}}\right)d_{2} - (k_{1}+1)R_{1}^{3}\frac{1}{2}e_{2} \\ -\frac{(m_{1}-n_{1})}{(m_{1}+n_{1})}(k_{1}+1)R_{1}\frac{1}{2}e_{0} + \left(-(k_{1}+1)R_{1}^{3}h_{-2} - \frac{(m_{1}-n_{1})}{(m_{1}+n_{1})}(k_{1}+1)R_{1}h_{-4}\right)f_{4} \\ + \left(-(k_{1}+1)R_{1}^{3}h_{-1} - \frac{(m_{1}-n_{1})}{(m_{1}+n_{1})}(k_{1}+1)R_{1}h_{-3}\right)f_{3} \\ -\frac{(m_{1}-n_{1})}{(m_{1}+n_{1})}(k_{1}+1)R_{1}h_{-2}f_{2} - \frac{(m_{1}-n_{1})}{(m_{1}+n_{1})}(k_{1}+1)R_{1}h_{-1}f_{1},$$

.

36) z^{-2}

,

$$- (m_1 - n_1) \left(1 - \frac{\mu_1}{\mu_2} \right) BR_1^3$$

$$= + \left[12\mu_1 + (m_1 - n_1) \left(1 + \frac{\mu_1}{\mu_2} k_2 \right) R_1 - (m_1 + n_1) \left(1 - \frac{\mu_1}{\mu_2} \right) R_1 \right] c_1$$

$$+ \left[-\frac{12}{R_1^2} \mu_1 + (m_1 + n_1) \left(1 - \frac{\mu_1}{\mu_2} \right) \frac{1}{R_1} \right] d_3 - (m_1 + n_1) (k_1 + 1) R_1^5 \left(h_{-1} f_4 + \frac{1}{2} e_3 \right)$$

$$- (m_1 - n_1) (k_1 + 1) R_1 \left(h_{-5} f_4 + h_{-4} f_3 + h_{-3} f_2 + h_{-2} f_1 + h_{-1} f_0 + \frac{1}{2} e_{-1} \right),$$

÷

.

$$-\frac{(m_1-n_1)}{(m_1+n_1)}\left(1-\frac{\mu_1}{\mu_2}\right)BR_1^2$$

.

:

4

$$= \left(\frac{6}{M_{1}} + \frac{(m_{1} - n_{1})}{(m_{1} + n_{1})}\left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2}\right) - \left(1 - \frac{\mu_{1}}{\mu_{2}}\right)\right)c_{1} \\ + \left(-\frac{6}{R_{1}^{2}}\frac{1}{M_{1}} + \left(1 - \frac{\mu_{1}}{\mu_{2}}\right)\frac{1}{R_{1}^{2}}\right)d_{3} - (k_{1} + 1)R_{1}^{4}\frac{1}{2}e_{3} - \frac{(m_{1} - n_{1})}{(m_{1} + n_{1})}(k_{1} + 1)\frac{1}{2}e_{-1} \\ + \left(-(k_{1} + 1)R_{1}^{4}h_{-1} - \frac{(m_{1} - n_{1})}{(m_{1} + n_{1})}(k_{1} + 1)h_{-5}\right)f_{4} - \frac{(m_{1} - n_{1})}{(m_{1} + n_{1})}(k_{1} + 1)h_{-4}f_{3} \\ - \frac{(m_{1} - n_{1})}{(m_{1} + n_{1})}(k_{1} + 1)h_{-3}f_{2} - \frac{(m_{1} - n_{1})}{(m_{1} + n_{1})}(k_{1} + 1)h_{-2}f_{1} - \frac{(m_{1} - n_{1})}{(m_{1} + n_{1})}(k_{1} + 1)h_{-1}f_{0},$$

$$(7) z^{-3}$$

37) z^{-3}

$$0 = \left[+32\mu_{1} + (m_{1} - n_{1}) \left(1 + \frac{\mu_{1}}{\mu_{2}} k_{2} \right) R_{1} - 2R_{1} (m_{1} + n_{1}) \left(1 - \frac{\mu_{1}}{\mu_{2}} \right) \right] c_{2} + \left[-\frac{16}{R_{1}^{2}} \mu_{1} + (m_{1} + n_{1}) \left(1 - \frac{\mu_{1}}{\mu_{2}} \right) \frac{1}{R_{1}} \right] d_{4} - (m_{1} - n_{1}) (k_{1} + 1) R_{1} \left(\begin{array}{c} h_{-6}f_{4} + h_{-5}f_{3} + h_{-4}f_{2} + h_{-3}f_{1} \\ +h_{-2}f_{0} + h_{-1}f_{-1} + \frac{1}{2}e_{-2} \end{array} \right),$$

$$\begin{array}{ll} 0 & = & \left(\frac{16}{M_1} + \frac{(m_1 - n_1)}{(m_1 + n_1)} \left(1 + \frac{\mu_1}{\mu_2} k_2\right) - 2 \left(1 - \frac{\mu_1}{\mu_2}\right)\right) c_2 \\ & \quad + \left(-\frac{1}{R_1^2} \frac{8}{M_1} + \left(1 - \frac{\mu_1}{\mu_2}\right) \frac{1}{R_1^2}\right) d_4 - \frac{(m_1 - n_1) \left(k_1 + 1\right)}{2 \left(m_1 + n_1\right)} e_{-2} \\ & \quad - \frac{(m_1 - n_1)}{(m_1 + n_1)} \left(k_1 + 1\right) h_{-6} f_4 - \frac{(m_1 - n_1)}{(m_1 + n_1)} \left(k_1 + 1\right) h_{-5} f_3 \\ & \quad - \frac{(m_1 - n_1)}{(m_1 + n_1)} \left(k_1 + 1\right) h_{-4} f_2 - \frac{(m_1 - n_1)}{(m_1 + n_1)} \left(k_1 + 1\right) h_{-3} f_1 \\ & \quad - \frac{(m_1 - n_1)}{(m_1 + n_1)} \left(k_1 + 1\right) h_{-2} f_0 - \frac{(m_1 - n_1)}{(m_1 + n_1)} \left(k_1 + 1\right) h_{-1} f_{-1}, \end{array}$$

38) z^{-4}

$$0 = \left(+60\mu_1 + (m_1 - n_1)\left(1 + \frac{\mu_1}{\mu_2}k_2\right)R_1 - 3R_1(m_1 + n_1)\left(1 - \frac{\mu_1}{\mu_2}\right)\right)c_3 + \left(-\frac{20}{R_1^2}\mu_1 + (m_1 + n_1)\left(1 - \frac{\mu_1}{\mu_2}\right)\frac{1}{R_1}\right)d_5 - (m_1 - n_1)(k_1 + 1)R_1q_{-3},$$

÷

•

•

4

$$\begin{array}{lll} 0 & = & \left(\frac{30}{M_{1}} + \frac{(m_{1} - n_{1})}{(m_{1} + n_{1})} \left(1 + \frac{\mu_{1}}{\mu_{2}}k_{2}\right) - 3 \left(1 - \frac{\mu_{1}}{\mu_{2}}\right)\right) c_{3} \\ & & + \left(-\frac{1}{R_{1}^{2}}\frac{10}{M_{1}} + \left(1 - \frac{\mu_{1}}{\mu_{2}}\right)\frac{1}{R_{1}^{2}}\right) d_{5} - \frac{(m_{1} - n_{1}) \left(k_{1} + 1\right)}{(m_{1} + n_{1})} h_{-7}f_{4} \\ & & - \frac{(m_{1} - n_{1}) \left(k_{1} + 1\right)}{(m_{1} + n_{1})} h_{-6}f_{3} - \frac{(m_{1} - n_{1}) \left(k_{1} + 1\right)}{(m_{1} + n_{1})} h_{-5}f_{2} - \frac{(m_{1} - n_{1}) \left(k_{1} + 1\right)}{(m_{1} + n_{1})} h_{-4}f_{1} \\ & - \frac{(m_{1} - n_{1}) \left(k_{1} + 1\right)}{(m_{1} + n_{1})} h_{-3}f_{0} - \frac{(m_{1} - n_{1}) \left(k_{1} + 1\right)}{(m_{1} + n_{1})} h_{-2}f_{-1} \\ & - \frac{(m_{1} - n_{1}) \left(k_{1} + 1\right)}{(m_{1} + n_{1})} h_{-1}f_{-2} - \frac{(m_{1} - n_{1}) \left(k_{1} + 1\right)}{2 \left(m_{1} + n_{1}\right)} e_{-3}. \end{array}$$

,

.,

.

APPENDIX 6 Stress Fields Around Crack Tips

A6.1 Stress Field

Stress fields are expressed in terms of stress potentials as follows

$$\sigma_{xx} = \operatorname{Re}\left[\varphi_{1}'(z) - \varphi_{1}'(\bar{z}) + 2\overline{\varphi_{1}'(z)} + (\bar{z} - z)\overline{\varphi_{1}''(z)} + \dot{X}'(\bar{z})\right], \qquad (4.7)$$

$$\sigma_{yy} = \operatorname{Re}\left[\varphi_{1}'(z) + \varphi_{1}'(\bar{z}) + (z - \bar{z})\overline{\varphi_{1}''(z)} - X'(\bar{z})\right], \qquad (4.7)$$

$$\sigma_{xy} = \operatorname{Im}\left[\varphi_{1}'(z) - \varphi_{1}'(\bar{z}) + 2\overline{\varphi_{1}'(z)} + (\bar{z} - z)\overline{\varphi_{1}''(z)} + X'(\bar{z})\right].$$

Let's derive $\varphi_1'(z)$ first. From (2.32)

.

$$\frac{d}{dz}\varphi_{1}(z) = \frac{d}{dz} \left[\frac{1}{2}(z-a)^{-\frac{1}{2}}(z-b)^{-\frac{1}{2}}Y(z) + \frac{1}{2}X(z) \right]$$

$$= -\frac{1}{4}(z-a)^{-\frac{3}{2}}(z-b)^{-\frac{1}{2}}Y(z) - \frac{1}{4}(z-a)^{-\frac{1}{2}}(z-b)^{-\frac{3}{2}}Y(z) + \frac{1}{2}(z-a)^{-\frac{1}{2}}(z-b)^{-\frac{3}{2}}Y(z) + \frac{1}{2}(z-a)^{-\frac{1}{2}}(z-b)^{-\frac{1}{2}}Y'(z) + \frac{1}{2}X'(z) \\
= -\frac{1}{4}\frac{(z-b)+(z-a)}{(z-a)^{\frac{3}{2}}(z-b)^{\frac{3}{2}}}Y(z) + \frac{1}{2}(z-a)^{-\frac{1}{2}}(z-b)^{-\frac{1}{2}}Y'(z) + \frac{1}{2}X'(z) \\
= \frac{1}{4}\frac{a+b-2z}{(z-a)^{\frac{3}{2}}(z-b)^{\frac{3}{2}}}Y(z) + \frac{1}{2}(z-a)^{\frac{1}{2}}(z-b)^{\frac{1}{2}}Y'(z) + \frac{1}{2}X'(z).$$
(A6.1)

Thus,

$$\begin{aligned}
\varphi_1''(z) & \cdot & (A6.2) \\
&= \frac{d}{dz} \begin{bmatrix} \frac{1}{4} (a+b-2z) (z-a)^{-\frac{3}{2}} (z-b)^{-\frac{3}{2}} Y(z) \\ &+ \frac{1}{2} (z-a)^{-\frac{1}{2}} (z-b)^{-\frac{1}{2}} Y'(z) + \frac{1}{2} X'(z) \end{bmatrix} \\
&= \frac{1}{4} (-2) (z-a)^{-\frac{3}{2}} (z-b)^{-\frac{3}{2}} Y(z) - \frac{3}{8} (a+b-2z) (z-a)^{-\frac{5}{2}} (z-b)^{-\frac{3}{2}} Y(z) \\ &- \frac{3}{8} (a+b-2z) (z-a)^{-\frac{3}{2}} (z-b)^{-\frac{5}{2}} Y(z) + \frac{1}{4} (a+b-2z) (z-a)^{-\frac{3}{2}} (z-b)^{-\frac{3}{2}} Y'(z) \end{aligned}$$

$$-\frac{1}{4}(z-a)^{-\frac{3}{2}}(z-b)^{-\frac{1}{2}}Y'(z) - \frac{1}{4}(z-a)^{-\frac{1}{2}}(z-b)^{-\frac{3}{2}}Y'(z) +\frac{1}{2}(z-a)^{-\frac{1}{2}}(z-b)^{-\frac{1}{2}}Y''(z) + \frac{1}{2}X''(z).$$

The 2nd and 3rd terms in (A6.2) turn out

$$-\frac{3}{8}(a+b-2z)\left[(z-a)^{-\frac{5}{2}}(z-b)^{-\frac{3}{2}}+(z-a)^{-\frac{3}{2}}(z-b)^{-\frac{5}{2}}\right]Y(z) \quad (A6.3)$$

$$= -\frac{3}{8}(a+b-2z)\frac{(z-b)+(z-a)}{(z-a)^{\frac{5}{2}}(z-b)^{\frac{5}{2}}}Y(z)$$

$$= \frac{3}{8}\frac{(a+b-2z)^{2}}{(z-a)^{\frac{5}{2}}(z-b)^{\frac{5}{2}}}Y(z).$$

The 4th, 5th and 6th terms are also re-written as

÷

$$\frac{(a+b-2z)-(z-b)-(z-a)}{4(z-a)^{\frac{3}{2}}(z-b)^{\frac{3}{2}}}Y'(z) = \frac{(a+b-2z)}{2(z-a)^{\frac{3}{2}}(z-b)^{\frac{3}{2}}}Y'(z).$$
 (A6.4)

Thus,

.

$$\varphi_{1}''(z) = -\frac{1}{2(z-a)^{\frac{3}{2}}(z-b)^{\frac{3}{2}}}Y(z) + \frac{3(a+b-2z)^{2}}{8(z-a)^{\frac{5}{2}}(z-b)^{\frac{5}{2}}}Y(z)$$

$$+\frac{(a+b-2z)}{2(z-a)^{\frac{3}{2}}(z-b)^{\frac{3}{2}}}Y'(z) + \frac{1}{2(z-a)^{\frac{1}{2}}(z-b)^{\frac{1}{2}}}Y''(z) + \frac{1}{2}X''(z).$$
(A6.5)

Since our aim is to determine the values of stress potentials near crack tips, Y(z), Y'(z) and Y''(z) need to be determined near crack tip a and b. First let's determine the stress potential Y(z) in the neighbourhood of crack tip a as follows

$$Y(z) = Y(a) + Y'(a) (z - a) + \frac{1}{2!} Y''(a) (z - a)^2 + \dots$$

$$+ \frac{1}{n!} Y^{(n)}(a) (z - a)^n + \dots$$

$$= Y(a) + Y'(a) r_1 e^{i\theta_1} + \frac{1}{2!} Y''(a) r_1^2 e^{i2\theta_1} + \dots + \frac{1}{n!} Y^{(n)}(a) r_1^n e^{in\theta_1} + \dots ,$$
(A6.6)

where $(z - a) = r_1 e^{i\theta_1}$ $(0 \le \theta_1 \le 2\pi)$. Thus,

$$Y(z) = Y(a) + Y'(a)r_1e^{i\theta_1} + \dots,$$
 (A6.7)

$$Y'(z) = Y'(a) + Y''(a)r_1e^{i\theta_1}...,$$
(A6.8)

$$Y''(z) = Y''(a) + Y'''(a)r_1e^{i\theta_1}\dots$$
(A6.9)

Substituting (A6.7) and (A6.8) into (A6.1) yields

$$\begin{split} \varphi_1'(z) &= \left[\frac{1}{4} \frac{(a+b-2z)}{(z-a)^{\frac{3}{2}}(z-b)^{\frac{3}{2}}} Y(z) + \frac{1}{2(z-a)^{\frac{1}{2}}(z-b)^{\frac{1}{2}}} Y'(z) + \frac{1}{2} X'(z) \right]_{around \ z=a} \\ &= \frac{1}{4} \frac{-(a-b) \left[Y(a) + Y'(a)r_1e^{i\theta_1} \right]}{(r_1e^{i\theta_1})^{\frac{3}{2}}(a-b)^{\frac{3}{2}}} + \frac{1}{2} \frac{\left[Y'(a) + Y''(a)r_1e^{i\theta_1} \right]}{(r_1e^{i\theta_1})^{\frac{1}{2}}(a-b)^{\frac{1}{2}}} + \frac{1}{2} X'(a) \\ &= \frac{1}{4} \frac{-Y(a)}{(r_1e^{i\theta_1})^{\frac{3}{2}}(a-b)^{\frac{1}{2}}} + \frac{1}{4} \frac{-Y'(a)}{(r_1e^{i\theta_1})^{\frac{1}{2}}(a-b)^{\frac{1}{2}}} + \frac{1}{2} \frac{Y'(a)}{(r_1e^{i\theta_1})^{\frac{1}{2}}(a-b)^{\frac{1}{2}}} \\ &+ \frac{1}{2} \frac{Y''(a)(r_1e^{i\theta_1})^{\frac{1}{2}}}{(a-b)^{\frac{1}{2}}} + \frac{1}{2} X'(a). \end{split}$$

Taking only leading-order items yields

$$\varphi_1'(z) = \frac{1}{4} \frac{-Y'(a)}{(r_1 e^{i\theta_1})^{\frac{1}{2}} (a-b)^{\frac{1}{2}}} + \frac{1}{2} \frac{Y'(a)}{(r_1 e^{i\theta_1})^{\frac{1}{2}} (a-b)^{\frac{1}{2}}}$$

$$= \frac{1}{4} \frac{Y'(a)}{\sqrt{r_1 e^{i\frac{\theta_1}{2}} \sqrt{a-b}}}.$$
(A6.11)

Note that b > a so that $\sqrt{a-b} = i\sqrt{b-a}$ and $i = \cos \frac{\pi}{2} + i \sin \frac{\pi}{2} = e^{i\frac{\pi}{2}}$. Therefore,

$$\varphi_1'(z) = \frac{Y'(a)}{4\sqrt{r_1}e^{i\frac{\theta_1}{2}}e^{i\frac{\pi}{2}}\sqrt{b-a}} = \frac{Y'(a)}{4\sqrt{r_1}\sqrt{b-a}}e^{-i\left(\frac{\theta_1+\pi}{2}\right)}.$$
(A6.12)

In addition,

••

,

$$\varphi_1'(\bar{z}) = \frac{Y'(a)}{4\sqrt{r_1}e^{-i\frac{\theta_1}{2}}e^{i\frac{\pi}{2}}\sqrt{b-a}} = \frac{Y'(a)}{4\sqrt{r_1}\sqrt{b-a}}e^{i\left(\frac{\theta_1-\pi}{2}\right)}.$$
(A6.13)

The exponential terms can be expressed with trigonometric functions in the following form

$$e^{-i\left(\frac{\theta_1+\pi}{2}\right)} = \cos\left(\frac{\theta_1+\pi}{2}\right) - i\sin\left(\frac{\theta_1+\pi}{2}\right) = -\sin\frac{\theta_1}{2} - i\cos\frac{\theta_1}{2},$$
$$e^{i\left(\frac{\theta_1-\pi}{2}\right)} = \cos\left(\frac{\theta_1-\pi}{2}\right) + i\sin\left(\frac{\theta_1-\pi}{2}\right) = \sin\frac{\theta_1}{2} - i\cos\frac{\theta_1}{2}.$$

Substituting the above equations into (A6.12) and (A6.13) leads to

$$\varphi_1'(z) = \frac{Y'(a)}{4\sqrt{r_1}\sqrt{b-a}} \left(-\sin\frac{\theta_1}{2} - i\cos\frac{\theta_1}{2} \right),$$

$$\varphi_1'(\bar{z}) = \frac{Y'(a)}{4\sqrt{r_1}\sqrt{b-a}} \left(\sin\frac{\theta_1}{2} - i\cos\frac{\theta_1}{2} \right).$$
(A6.14)

Note that $\operatorname{Re}(\varphi_1'(z)) = -\operatorname{Re}(\varphi_1'(\overline{z}))$, which means $\varphi_1'(z)$ has multi-values at the crack face. Therefore, we need to define a new $\varphi_1'(\overline{z})$ value leading single valueness at the crack face. Now, let's consider a branch cut which gives us single valueness and makes $\varphi_1'(z)$ single-valued function at the crack face. The branches of $e^{-i\frac{\theta_1}{2}}$ are $e^{-i\frac{1}{2}(\theta_1+2n\pi)}$ and $e^{-i\frac{1}{2}(\theta_1-2n\pi)}$. Let's take $e^{-i\frac{1}{2}(\theta_1-2\pi)}$ for $\varphi_1'(\overline{z})$. From (6.13), $\varphi_1'(\overline{z})$ can be expressed as follows

$$\begin{aligned}
\varphi_1'(\bar{z}) &= \frac{Y'(a)}{4\sqrt{r_1}\sqrt{b-a}e^{-i\frac{1}{3}(\theta_1-2\pi)}e^{i\frac{\pi}{2}}} = \frac{Y'(a)}{4\sqrt{r_1}\sqrt{b-a}}e^{i\left(\frac{\theta_1}{2}-\frac{3\pi}{2}\right)} \\
&= \frac{Y'(a)}{4\sqrt{r_1}\sqrt{b-a}}\left[\cos\left(\frac{\theta_1}{2}-\frac{3\pi}{2}\right)+i\sin\left(\frac{\theta_1}{2}-\frac{3\pi}{2}\right)\right] \\
&= \frac{Y'(a)}{4\sqrt{r_1}\sqrt{b-a}}\left[-\sin\frac{\theta_1}{2}+i\cos\frac{\theta_1}{2}\right].
\end{aligned}$$
(A6.15)

Thus, at the branch, $\varphi_1'(z)$ has a single value at the upper/lower (+/-) crack face

÷

such as

$$\varphi_1'(z) = \frac{Y'(a)}{4\sqrt{r_1}\sqrt{b-a}} \left(-\sin\frac{\theta_1}{2} - i\cos\frac{\theta_1}{2} \right),$$
(A6.16)
$$\varphi_1'(\bar{z}) = \frac{Y'(a)}{4\sqrt{r_1}\sqrt{b-a}} \left[-\sin\frac{\theta_1}{2} + i\cos\frac{\theta_1}{2} \right].$$

Next, $\varphi_1''(z)$ is derived. In view of (4.7), (A6.5) is rewritten as

$$(z-\bar{z})\varphi_{1}''(z) = (z-\bar{z}) \begin{bmatrix} \frac{(a+b-2z)}{2(z-a)^{\frac{3}{2}}(z-b)^{\frac{3}{2}}}Y'(z) - \frac{1}{2(z-a)^{\frac{3}{2}}(z-b)^{\frac{3}{2}}}Y(z) \\ + \frac{3(a+b-2z)^{2}}{8(z-a)^{\frac{5}{2}}(z-b)^{\frac{5}{2}}}Y(z) + \frac{1}{2(z-a)^{\frac{1}{2}}(z-b)^{\frac{1}{2}}}Y''(z) + \frac{1}{2}X''(z) \end{bmatrix}_{around \ z=a}$$
(A6.17)

At $z \approx a$, from (A6.7), (A6.8) and (A6.9) are

$$(z-a) = r_1 e^{i\theta_1},$$

 $Y(z) = Y(a) + Y'(a)r_1 e^{i\theta_1},$
 $Y'(z) = Y'(a) \text{ and } Y''(z) = Y''(a)$

In addition,

÷

$$(z-\bar{z}) = r_1 \left(\cos^{i}\theta_1 + i\sin\theta_1\right) - r_1 \left(\cos\theta_1 - i\sin\theta_1\right) = 2r_1 i\sin\theta_1.$$

Substituting above equations into (A6.17) yields

$$(z - \bar{z}) \varphi_{1}''(z) = 2r_{1}i \sin \theta_{1} \begin{bmatrix} \frac{(-a+b)Y'(a)}{2(r_{1}e^{i\theta_{1}})^{\frac{3}{2}}(a-b)^{\frac{3}{2}}} - \frac{Y(a)+Y'(a)r_{1}e^{i\theta_{1}}}{2(r_{1}e^{i\theta_{1}})^{\frac{3}{2}}(a-b)^{\frac{3}{2}}} \\ + \frac{3(-a+b)^{2}[Y(a)+Y'(a)r_{1}e^{i\theta_{1}}]}{8(r_{1}e^{i\theta_{1}})^{\frac{5}{2}}(a-b)^{\frac{5}{2}}} \end{bmatrix}$$
(A6.18)
+ $2r_{1}i \sin \theta_{1} \frac{Y''(a)}{2(r_{1}e^{i\theta_{1}})^{\frac{1}{2}}(a-b)^{\frac{1}{2}}} + 2r_{1}i \sin \theta_{1} \frac{1}{2}X''(z).$

Note that Y(a) = 0 by definition. By taking leading order singularity only, we get

$$(z-\bar{z})\varphi_1''(z) = -\frac{Y'(a)\sin\theta_1}{4\sqrt{r_1}\sqrt{b-a}e^{i\frac{3}{2}\theta_1}} = -\frac{Y'(a)\sin\theta_1}{4\sqrt{r_1}\sqrt{b-a}}e^{-i\frac{3}{2}\theta_1}.$$
 (A6.19)

Now, let's check single valueness of $(z - \bar{z}) \varphi_1''(z)$ at a branch cut. We choose $\theta_1 + 2\pi$ first.

٠

.

$$e^{-i\left(3\pi + \frac{3\theta_1}{2}\right)} = \cos\left(3\pi + \frac{3\theta_1}{2}\right) - i\sin\left(3\pi + \frac{3\theta_1}{2}\right) = -\cos\frac{3\theta_1}{2} + i\sin\frac{3\theta_1}{2}, \\ e^{i\left(3\pi + \frac{3\theta_1}{2}\right)} = \cos\left(3\pi + \frac{3\theta_1}{2}\right) + i\sin\left(3\pi + \frac{3\theta_1}{2}\right) = -\cos\frac{3\theta_1}{2} - i\sin\frac{3\theta_1}{2}.$$

Note that $\operatorname{Re}(\varphi_1''(z)) = \operatorname{Re}(\varphi_1''(\overline{z}))$ and $\operatorname{Im}(\varphi_1''(z)) = -\operatorname{Im}(\varphi_1''(\overline{z}))$ and it has single value at the crack face. Thus,

$$(z - \bar{z})\varphi_1''(z) = -\frac{Y'(a)\sin\theta_1}{4\sqrt{r_1}\sqrt{b-a}} \left(-\cos\frac{3\theta_1}{2} + i\sin\frac{3\theta_1}{2}\right).$$
 (A6.20)

Let's substitute (A6.16) and (A6.20) into $(4.7)_2$ to define stress fields.

$$\sigma_{yy} = \operatorname{Re} \left[\varphi_1'(z) + \varphi_1'(\bar{z}) + (z - \bar{z}) \overline{\varphi_1''(z)} - X'(\bar{z}) \right]$$

$$= -\frac{Y'(a)}{2\sqrt{r_1}\sqrt{b-a}} \sin \frac{\theta_1}{2} + \frac{Y'(a)}{4\sqrt{r_1}\sqrt{b-a}} \sin \theta_1 \cos \frac{3\theta_1}{2} - \operatorname{Re} \left[X'(\bar{z}) \right]$$

$$= \frac{Y'(a)}{\sqrt{r_1}\sqrt{b-a}} \left[-\frac{1}{2} \sin \frac{\theta_1}{2} + \frac{1}{4} \sin \theta_1 \cos \frac{3\theta_1}{2} \right] + O'(r_1).$$
(A6.21)

Note that

.

÷

$$\begin{aligned} & -\frac{1}{2}\sin\frac{\theta_{1}}{2} + \frac{1}{4}\cos\frac{3\theta_{1}}{2}\sin\theta_{1} \\ &= -\frac{1}{2}\sin\frac{\theta_{1}}{2} + \frac{1}{8}\left[\sin\left(\frac{3\theta_{1}}{2} + \theta_{1}\right) - \sin\left(\frac{3\theta_{1}}{2} - \theta_{1}\right)\right] \\ &= -\frac{1}{2}\sin\frac{\theta_{1}}{2} + \frac{1}{8}\left(\sin\frac{5\theta_{1}}{2} - \sin\frac{\theta_{1}}{2}\right) \end{aligned}$$

$$= \frac{1}{8}\sin\frac{5\theta_1}{2} - \frac{5}{8}\sin\frac{\theta_1}{2}.$$

Thus,

$$\sigma_{yy} = \frac{Y'(a)}{\sqrt{r_1}\sqrt{b-a}} \left(-\frac{5}{8} \sin\frac{\theta_1}{2} + \frac{1}{8} \sin\frac{5\theta_1}{2} \right) + O(r_1).$$
(A6.22)

In view of (4.7), adding σ_{yy} to σ_{xx} leads to the expression

$$\sigma_{yy} + \sigma_{xx} = 2 \operatorname{Re}\left[\varphi_1'(z) + \overline{\varphi_1'(z)}\right]. \tag{A6.23}$$

Thus,

.

.

$$\sigma_{xx} = -\frac{Y'(a)}{\sqrt{r_1}\sqrt{b-a}} \sin\frac{\theta_1}{2} - \frac{Y'(a)}{\sqrt{r_1}\sqrt{b-a}} \left(-\frac{5}{8}\sin\frac{\theta_1}{2} + \frac{1}{8}\sin\frac{5\theta_1}{2}\right) + O(r_1)$$

$$= \frac{Y'(a)}{\sqrt{r_1}\sqrt{b-a}} \left[-\frac{3}{8}\sin\frac{\theta_1}{2} - \frac{1}{8}\sin\frac{5\theta_1}{2}\right] + O(r_1).$$
(A6.24)

Also, σ_{xy} is given by

$$\sigma_{xy} = \operatorname{Im} \left[\varphi_1'(z) - \varphi_1'(\bar{z}) + 2\overline{\varphi_1'(z)} + (\bar{z} - z)\overline{\varphi_1''(z)} + X'(\bar{z}) \right]$$
(A6.25)
$$= \frac{Y'(a)}{4\sqrt{r_1}\sqrt{b-a}} \left[-\cos\frac{\theta_1}{2} - \cos\frac{\theta_1}{2} + 2\cos\frac{\theta_1}{2} - \sin\theta_1\sin\frac{3\theta_1}{2} \right] + O(r_1)$$

$$= \frac{Y'(a)}{\sqrt{r_1}\sqrt{b-a}} \left[-\frac{1}{8}\cos\frac{\theta_1}{2} + \frac{1}{8}\cos\frac{5\theta_1}{2} \right] + O(r_1).$$

Consequently, the stress fields σ_{yy} , σ_{xx} and σ_{xy} near crack tip *a* have been determined as follows

$$\sigma_{yy} = \frac{Y'(a)}{\sqrt{r_1}\sqrt{b-a}} \left(-\frac{5}{8} \sin \frac{\theta_1}{2} + \frac{1}{8} \sin \frac{5\theta_1}{2} \right) + O(r_1), \quad (A6.26)$$

$$\sigma_{xx} = \frac{Y'(a)}{\sqrt{r_1}\sqrt{b-a}} \left(-\frac{3}{8} \sin \frac{\theta_1}{2} - \frac{1}{8} \sin \frac{5\theta_1}{2} \right) + O(r_1),$$

•

$$\sigma_{xy} = \frac{Y'(a)}{\sqrt{r_1}\sqrt{b-a}} \left(-\frac{1}{8}\cos\frac{\theta_1}{2} + \frac{1}{8}\cos\frac{5\theta_1}{2} \right) + O(r_1).$$

In the similar way of (A6.6), the stress potential Y(z) near crack tip b can also be determined in the following form

$$Y(z) = Y(b) + Y'(b) (z - b) + \frac{1}{2!} Y''(b) (z - b)^2 + \dots$$

$$+ \frac{1}{n!} Y^{(n)}(b) (z - b)^n + \dots$$

$$= Y(b) + Y'(b) r_2 e^{i\theta_2} + \frac{1}{2!} Y''(b) r_2^2 e^{i2\theta_2} + \dots + \frac{1}{n!} Y^{(n)}(b) r_2^n e^{in\theta_2} + \dots ,$$
(A6.27)

where $(z-b) = r_2 e^{i\theta_2}$ $(-\pi \le \theta_2 \le \pi)$. And

$$Y(z) = Y(b) + Y'(b)r_2e^{i\theta_2},$$
(A6.28)

$$Y'(z) = Y'(b) + Y''(b)r_2e^{i\theta_2},$$
(A6.29)

$$Y''(z) = Y''(b) + Y'''(b)r_2e^{i\theta_2}.$$
(A6.30)

Thus,

લ

.

:

$$\begin{split} \varphi_1'(z) &= \left[\frac{1}{4} \frac{(a+b-2z)}{(z-a)^{\frac{3}{2}}(z-b)^{\frac{3}{2}}} Y(z) + \frac{1}{2(z-a)^{\frac{1}{2}}(z-b)^{\frac{1}{2}}} Y'(z) + \frac{1}{2} X'(z) \right]_{around \ z=b} \\ &= \frac{1}{4} \frac{-(b-a)}{(b-a)^{\frac{3}{2}}(r_2 e^{i\theta_2})^{\frac{3}{2}}} \left[Y(b) + Y'(b) r_2 e^{i\theta_2} \right] \\ &+ \frac{1}{2(b-a)^{\frac{1}{2}}(r_2 e^{i\theta_2})^{\frac{1}{2}}} \left[Y'(b) + Y''(b) r_2 e^{i\theta_2} \right] + \frac{1}{2} X'(z). \end{split}$$

Taking only leading-order items yields

.

$$\varphi_1'(z) = \frac{1}{4} \frac{Y'(b)}{\sqrt{r_2}e^{i\frac{\theta_2}{2}}\sqrt{b-a}} = \frac{1}{4} \frac{Y'(b)}{\sqrt{r_2}\sqrt{b-a}} \left(\cos\frac{\theta_2}{2} - i\sin\frac{\theta_2}{2}\right).$$
(A6.32)

In addition,

:

4

$$\varphi_1'(\bar{z}) = \frac{1}{4} \frac{Y'(b)}{\sqrt{r_2}e^{-i\frac{\theta_2}{2}}\sqrt{b-a}} = \frac{1}{4} \frac{Y'(b)}{\sqrt{r_2}\sqrt{b-a}} \left(\cos\frac{\theta_2}{2} + i\sin\frac{\theta_2}{2}\right).$$
(A6.33)

Thus, the $\varphi'_1(z)$ has single value at the crack face since $\operatorname{Re}[\varphi'_1(z)] = \operatorname{Re}[\varphi'_1(\bar{z})]$ and $\operatorname{Im}[\varphi'_1(z)] = -\operatorname{Im}[\varphi'_1(\bar{z})]$.

Similar to (A6.17) and (A6.18),

$$(z-\bar{z})\varphi_{1}''(z) = 2r_{2}i\sin\theta_{2} \begin{bmatrix} \frac{(a+b-2z)}{2(z-a)^{\frac{3}{2}}(z-b)^{\frac{3}{2}}} [Y'(b) + Y''(b)r_{2}e^{i\theta_{2}}] \\ -\frac{1}{2(z-a)^{\frac{3}{2}}(z-b)^{\frac{3}{2}}} [Y(b) + Y'(b)r_{2}e^{i\theta_{2}}] \\ +\frac{3(a+b-2z)^{2}}{8(z-a)^{\frac{5}{2}}(z-b)^{\frac{5}{2}}} [Y(b) + Y'(b)r_{2}e^{i\theta_{2}}] \\ +\frac{1}{2(z-a)^{\frac{1}{2}}(z-b)^{\frac{1}{2}}} [Y''(b)] + \frac{1}{2}X''(z) \end{bmatrix}_{around z=b}$$

$$= \frac{-2i\sin\theta_{2}}{2(b-a)^{\frac{1}{2}}(r_{2})^{\frac{1}{2}}(e^{i\theta_{2}})^{\frac{3}{2}}} Y'(b) + \frac{-2(r_{2})^{\frac{1}{2}}i\sin\theta_{2}}{2(b-a)^{\frac{1}{2}}(e^{i\theta_{2}})^{\frac{1}{2}}} Y''(b) + \frac{3*2i\sin\theta_{2}}{8(b-a)^{\frac{1}{2}}(r_{2})^{\frac{1}{2}}(e^{i\theta_{2}})^{\frac{3}{2}}} Y'(b) \\ +\frac{2(r_{2})^{\frac{1}{2}}i\sin\theta_{2}}{2(b-a)^{\frac{1}{2}}(e^{i\theta_{2}})^{\frac{1}{2}}} Y''(b) + \frac{1}{2}X''(z)2r_{2}i\sin\theta_{2}.$$

Note that Y(b) = 0 by definition. By taking leading order singularity only, we get

$$(z-\bar{z})\,\varphi_1''(z) = -\frac{Y'(b)e^{i\frac{\pi}{2}}\sin\theta_2}{4\sqrt{r_2}\sqrt{b-a}e^{i\frac{3}{2}\theta_2}} = -\frac{Y'(b)\sin\theta_2}{4\sqrt{r_2}\sqrt{b-a}}e^{i\left(\frac{\pi}{2}-\frac{3}{2}\theta_2\right)}.$$
(A6.35)

By taking $\theta_2 + 2\pi$, the single valueness of $(z - \bar{z}) \varphi_1''(z)$ at a branch cut is obtained in the following expression

$$\frac{\pi}{2} - \frac{3}{2}\theta_2 = \frac{\pi}{2} - \frac{3}{2}(\theta_2 + 2\pi) = -\frac{\pi}{2} - \frac{3}{2}\theta_2,$$

$$e^{-i(-\frac{\pi}{2} - \frac{3}{2}\theta_2)} = \cos\left(-\frac{\pi}{2} - \frac{3}{2}\theta_2\right) - i\sin\left(-\frac{\pi}{2} - \frac{3}{2}\theta_2\right) = -\sin\frac{3\theta_2}{2} + i\cos\frac{3\theta_2}{2},$$

$$e^{i\left(-\frac{\pi}{2}-\frac{3}{2}\theta_{2}\right)} = \cos\left(-\frac{\pi}{2}-\frac{3}{2}\theta_{2}\right) + i\sin\left(-\frac{\pi}{2}-\frac{3}{2}\theta_{2}\right) = -\sin\frac{3\theta_{2}}{2} - i\cos\frac{3\theta_{2}}{2}$$

Thus,

$$(z - \bar{z}) \varphi_1''(z) = -\frac{Y'(b)\sin\theta_2}{4\sqrt{r_2}\sqrt{b-a}} \left(-\sin\frac{3\theta_2}{2} - i\cos\frac{3\theta_2}{2}\right).$$
(A6.36)

By substituting (A6.32), (A6.33) and (A6.36) into (4.7), we get stress fields around crack tip b as follows

$$\sigma_{yy} = \frac{Y'(b)}{\sqrt{r_2}\sqrt{b-a}} \left(\frac{5}{8}\cos\frac{\theta_2}{2} - \frac{1}{8}\cos\frac{5\theta_2}{2}\right) + O(r_2), \qquad (A6.37)$$

$$\sigma_{xx} = \frac{Y'(b)}{\sqrt{r_2}\sqrt{b-a}} \left(\frac{3}{8}\cos\frac{\theta_2}{2} + \frac{1}{8}\cos\frac{5\theta_2}{2}\right) + O(r_2), \qquad (A6.37)$$

$$\sigma_{xy} = \frac{Y'(b)}{\sqrt{r_2}\sqrt{b-a}} \left(-\frac{1}{8}\sin\frac{\theta_2}{2} + \frac{1}{8}\sin\frac{5\theta_2}{2}\right) + O(r_2).$$

The Y(z), (2.31) determined in Chapter 2 yields

$$Y(z) = \sum_{k=-\infty}^{\infty} f_k z^k = \dots f_{-2} z^{-2} + f_{-1} z^{-1} + f_0 + f_1 z^1 + f_2 z^2 \dots , \qquad (A6.38)$$

$$Y'(z) = \sum_{k=-\infty}^{\infty} k f_k z^{k-1} = \dots - 2 f_{-2} z^{-3} - f_{-1} z^{-2} + 0 + f_1 + 2 f_2 z \dots .$$

Near the crack tip a and b,

$$Y'(a) = \sum_{k=-\infty}^{\infty} k f_k a^{k-1}, Y'(b) = \sum_{k=-\infty}^{\infty} k f_k b^{k-1}.$$
 (A6.39)

Finally, with (A6.39) and b - a = 2l substituted into (6.26), we have stress fields at

...

crack tips a in the following forms

$$\sigma_{yy} = \frac{1}{\sqrt{r_1}\sqrt{2l}} \left(-\frac{5}{8} \sin\frac{\theta_1}{2} + \frac{1}{8} \sin\frac{5\theta_1}{2} \right) \left[\sum_{k=-\infty}^{\infty} kf_k a^{k-1} \right] + O(r_1), \quad (4.8)$$

$$\sigma_{xx} = \frac{1}{\sqrt{r_1}\sqrt{2l}} \left(-\frac{3}{8} \sin\frac{\theta_1}{2} - \frac{1}{8} \sin\frac{5\theta_1}{2} \right) \left[\sum_{k=-\infty}^{\infty} kf_k a^{k-1} \right] + O(r_1), \quad (4.8)$$

$$\sigma_{xy} = \frac{1}{\sqrt{r_1}\sqrt{2l}} \left(-\frac{1}{8} \cos\frac{\theta_1}{2} + \frac{1}{8} \cos\frac{5\theta_1}{2} \right) \left[\sum_{k=-\infty}^{\infty} kf_k a^{k-1} \right] + O(r_1),$$

where $(z - a) = r_1 e^{i\theta_1} \ (0 \le \theta_1 \le 2\pi).$

Also, the stresses, (A6.37) near crack tip b are determined as

$$\sigma_{yy} = \frac{1}{\sqrt{r_2}\sqrt{2l}} \left(\frac{5}{8} \cos \frac{\theta_2}{2} - \frac{1}{8} \cos \frac{5\theta_2}{2} \right) \left[\sum_{k=-\infty}^{\infty} kf_k b^{k-1} \right] + O(r_2), \quad (4.9)$$

$$\sigma_{xx} = \frac{1}{\sqrt{r_2}\sqrt{2l}} \left(\frac{3}{8} \cos \frac{\theta_2}{2} + \frac{1}{8} \cos \frac{5\theta_2}{2} \right) \left[\sum_{k=-\infty}^{\infty} kf_k b^{k-1} \right] + O(r_2), \quad (4.9)$$

$$\sigma_{xy} = \frac{1}{\sqrt{r_2}\sqrt{2l}} \left(-\frac{1}{8} \sin \frac{\theta_2}{2} + \frac{1}{8} \sin \frac{5\theta_2}{2} \right) \left[\sum_{k=-\infty}^{\infty} kf_k b^{k-1} \right] + O(r_2),$$

,

.

÷

where $(z-b) = r_2 e^{i\theta_2}$ $(-\pi \le \theta_2 \le \pi)$.

١